De-Sheng Pei Naima Hamid Marriya Sultan Suman Thodhal Yoganandham

Reservoir Ecotoxicology



Reservoir Ecotoxicology



Pictorial Illustration of the reservoir ecotoxicology

De-Sheng Pei • Naima Hamid • Marriya Sultan • Suman Thodhal Yoganandham

Reservoir Ecotoxicology



De-Sheng Pei (**b**) School of Public Health Chongqing Medical University Chongqing, China

Marriya Sultan School of Public Health Chongqing Medical University Chongqing, China Naima Hamid School of Public Health Chongqing Medical University Chongqing, China

Suman Thodhal Yoganandham School of Public Health Chongqing Medical University Chongqing, China

ISBN 978-3-031-26343-9 ISBN 978-3-031-26344-6 (eBook) https://doi.org/10.1007/978-3-031-26344-6

© The Editor(s) (if applicable) and The Author(s), under exclusive license to Springer Nature Switzerland AG 2023

This work is subject to copyright. All rights are solely and exclusively licensed by the Publisher, whether the whole or part of the material is concerned, specifically the rights of translation, reprinting, reuse of illustrations, recitation, broadcasting, reproduction on microfilms or in any other physical way, and transmission or information storage and retrieval, electronic adaptation, computer software, or by similar or dissimilar methodology now known or hereafter developed.

The use of general descriptive names, registered names, trademarks, service marks, etc. in this publication does not imply, even in the absence of a specific statement, that such names are exempt from the relevant protective laws and regulations and therefore free for general use.

The publisher, the authors, and the editors are safe to assume that the advice and information in this book are believed to be true and accurate at the date of publication. Neither the publisher nor the authors or the editors give a warranty, expressed or implied, with respect to the material contained herein or for any errors or omissions that may have been made. The publisher remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

This Springer imprint is published by the registered company Springer Nature Switzerland AG The registered company address is: Gewerbestrasse 11, 6330 Cham, Switzerland

This book is in memory of my parents (Mr. Chao Pei and Ms. Jinhua Wang), who passed away on October 22, 2020 and February 14, 2014, respectively. -De-Sheng Pei I write this book in loving memory of my beloved parents, Mr. Hamid Javeed Awan, Ms. Shasita Hamid. and husband. Mohsin Khubaib Ahmed. -Naima Hamid I would like to dedicate this book to my beloved parents, Mr. Muhammad Sultan, Ms. Tahira Sultan, and my siblings, Igra Sultan, Tayyab Sultan, and Abdul Basit Sultan. -Marriya Sultan I would like to dedicate this book to my child S. Krishiv and the support of my family, which helped me for contributing to the book.

-Suman Thodhal Yoganandham

Foreword

This book, written by De-Sheng Pei, Naima Hamid, Marriya Sultan, and Suman Thodhal Yoganandham, introduces a modern scientific concept of reservoir ecotoxicology, which plays a vital role in water resource management. The reservoir as a specific geographic area is important in irrigation, hydroelectric power, flood control, and shipping. Reservoir ecotoxicology has noticeable and different features, compared to other types of ecotoxicology. The chapters on the reservoir's environmental characteristics, study methods for identifying pollutants, and its ecotoxicological effects on aquatic species demonstrated that national and international collaborative effort is required to resolve the problems. Furthermore, it is of great concern that, in many countries, the majority of reservoir water is also used for drinking purposes. Moreover, reservoirs protect native flora and fauna at the population, community, and ecosystem levels, ensuring the ecosystem's integrity and resilience.

Therefore, I believe this book will be a valuable resource for students, scientists, policymakers, and regulatory authorities working to protect precious freshwater resources.

Jun Xia

Academician of Chinese Academy of Sciences (CAS), The Research Institute for Water Security (RIWS), Wuhan University, Wuhan, Hubei, China

vii

Preface

Reservoir construction is the key strategy for water resource management. In recent years, the migration of pollutants and their transformation into the reservoir ecosystem has become a global problem. Intensified anthropogenic activities and unsustainable practices have amplified the pollutant levels in the reservoir. Therefore, this book intends to highlight the environmental characteristics of the reservoir, study methods of ecotoxicology, and toxicological mechanism pathways of pollutants in the reservoir.

Introductory Chap. 1 proposes the term reservoir ecotoxicology and highlights the important role of reservoirs in irrigation, hydroelectric power, flood control, and shipping. Moreover, the reservoirs protect the native flora and fauna at the population, community, and ecosystem levels, and support ecosystem integrity and resilience. Chapter 2 describes the ecosystem characterization of the reservoir and understands the effect of various environmental stressors on the species diversity of various communities. Hydrologic changes, sedimentation flux, nutrient loading, and erosion are all major environmental stressors that affect water quality and disturb the species composition of flora and fauna in the reservoir.

Chapter 3 describes the major sources of pollution and the trend of pollution distribution in the Three Gorges Reservoir Area (TGRA), which serves as an example for understanding pollutant dynamics in the reservoir ecosystem. Heavy metals (HMs) were the most prevalent class of pollutants found in the TGRA from 2008 to 2016. Phthalate esters (PAEs) mainly DEHP showed higher health risks to the aquatic species. Moreover, intensified pyrogenic and petrogenic combustion are considered as the major sources of polycyclic aromatic hydrocarbons (PAHs). Similarly, agricultural activities, hospital waste, and industrial effluent increase the fingerprints of pharmaceutical and personal care products (PPCPs) in the TGRA.

Chapter 4 highlights the important factors involved in the migration and transformation of the pollutants. HMs were prevalent in the TGRA but found within the safety limits prescribed by the US EPA. The hydro-fluctuation (HFB) belt greatly affects the HMs migration from the inner layers of soil to the upper layers. Seasonal trends show a lower pollution load in summer, autumn, and spring than in winter due to the varied impoundment levels. Overall, the TGRA's ecological health has been relatively stable. Chapter 5 summarizes that excessive nutrients and algal growth degrade the reservoir water quality. The abundant release of phosphates and nitrates in the reservoir from natural and anthropogenic sources causes eutrophication and accelerates the process of algal blooms.

Chapter 6 highlights the increasing sources and abundance of microplastics (MPs) contamination in the reservoir ecosystem. Moreover, this chapter introduces different strategies and techniques used for MPs sampling, pre-treatment, and characterization. Chapter 7 describes the microorganism distribution in the TGRA. The authors review the microorganism species contribution before the dam construction. The ecological risk was assessed using the bacteria-based index of biotic integrity (Ba-IBI), which revealed that 25% of all sampling sites were Excellent, while 50% and 25% were Good and Fair, respectively.

Chapter 8 highlights the primary producers that play a vital role in the transfer of pollutants through food chain. Major primary producers including *Vibrio fischeri* (bacterium), *Raphidocelis subcapitata* (microalgae), *Spirodela polyrhiza* (floating macrophytes), and *Myriophyllum* (submerged macrophytes) are widely used as the ecotoxicity assessment species. Chapter 9 introduces various ecotoxicological methods using invertebrates for the assessment of the reservoir's water quality. The most commonly used standard species in ecotoxicity testing are Daphnids (*Daphnia magna, Daphnia pulex*, and *Ceriodaphnia dubia*). Previous studies also reported the use of crustaceans, such as amphipods, branchiopods, insect species, and rotifers. Furthermore, numerous acute and chronic toxicity tests that are widely used for toxicity screening in water bodies have been approved by international organizations.

Chapter 10 reviews the model fish species that are involved in ecotoxicological studies. Water quality and fish health are interrelated and can be used as the pollution indicator for monitoring reservoir pollution. Based on the contaminant type and toxicity endpoints, zebrafish, rainbow trout, Japanese medaka, and fathead minnow are the most recommended model fishes. Chapter 11 covers the topic of the ecotox-icology methodology of sediment toxicity in the reservoir. The reservoir sediment serves as a sink as well as the contaminant source. When the pollutant is released into the water column, it can be dangerous for biotic species. Moreover, aquatic species including algae, amphipods, and bivalves are commonly used for measuring aquatic toxicity.

Chapter 12 discusses the significance of mesocosm in the reservoir ecosystem. Moreover, mesocosm have broadly assessed the toxic effects of water-borne contaminants on various biological levels of organizations, as well as their impact on the overall ecosystem. Several studies have been conducted to assess the impact of contaminants, such as pesticides, microplastics, and persistent organic pollutants on macroinvertebrates and fish species with endpoints including species richness, diversity, and morphological & physicochemical parameters. Moreover, these systems mimic the natural environment to generate real-world exposure scenarios based on the study objectives, available cost, and time. Preface

Chapter 13 presents information on the molecular toxicity mechanism of HMs in the reservoir. Heavy metal bioaccumulation may have toxic effects on various tissues and organs. Heavy metal exposure inhibits growth, proliferation, differentiation, and damage repair. The molecular pathways of heavy metal toxicity in the reservoir include reactive oxygen species (ROS) formation, enzyme inactivation, DNA damage, cell death, etc. Chap. 14 explains the intricate mechanistic toxicity pathway caused by persistent organic pollutants (POPs) exposure. POPs mainly disrupt the aryl hydrocarbon receptor (AhR) signaling pathway. Furthermore, when polychlorinated biphenyls (PCBs), polycyclic aromatic hydrocarbons (PAHs), and organochlorinated pesticides (OCPs) were exposed to mice and zebrafish, they may induce neurodevelopmental disorders (NDD), oxidative stress, DNA damage, endocrine disruption, etc.

Chapter 15 discusses the molecular toxicity pathway caused by MPs. MPs in the reservoir can adsorb environmental pollutants, including heavy metals and organic pollutants, implying combined toxicity effects on organisms. MPs exposure may generate reactive oxygen species (ROS) and cause developmental and reproductive failure. Chapter 16 describes that plasticizers (PAEs) and bisphenol A (BPA) exposure can affect growth development by altering the thyroid and estrogen axis, leading to infertility. It also decreases egg production *via* reducing steroidogenesis, activating peroxisome proliferator-activated receptors (PPARs), and enhancing oxidative stress levels. Chapter 17 highlights toxicity effects caused by PPCPs exposure in the reservoir. These emerging chemicals may disturb the reproductive cycle and decrease sperm production. Particularly, sulfonamides (SAs) and tetracycline (TC) can affect the detoxification metabolism pathway. Even exposure to a low level of PPCP exhibits cytotoxicity and genetic toxicity with increased levels of apoptosis and DNA damage.

Chapter 18 describes the importance of the adverse outcome pathway (AOP) of the pollutants in the reservoir. The AOP framework aids to decipher molecular or biochemical data from field-sampled organisms exposed to a mixture of contaminants. The resulting endpoints are then used to infer the potential ecological risk. A more advanced AOP, known as quantitative adverse outcome pathway (qAOP), includes biology-based computational models delineating kev event (KE) correlations and combining a molecular initiating event (MIE) with an adverse outcome (AO). Chapter 19 highlights the problem of invasive alien species caused in the reservoir. Many exotic plant and fish species made their way to the reservoir region after the reservoir construction. The majority of the invasive species were the result of factors such as impoundments, international trade via water, road access, and recreational activities, such as aquaculture. Chapter 20 provides final thoughts and concluding remarks of this book. It highlights reservoir ecotoxicology as an emerging academic discipline is a multidisciplinary study, including ecology, ecotoxicology, environmental science, environmental hygiene, reservoir management, water resources engineering, etc.

This book presents reservoir ecotoxicology as a modern scientific concept, and there are numerous serious ecotoxicological issues in the reservoir that must be addressed thoroughly. It also provides the recent advances in the theoretical background of legacy and emerging pollutants in the reservoir. Furthermore, detailed mechanistic toxicity trends via model species will help academic scientists and government officials develop background knowledge and formulate mitigation plans based on multidisciplinary research.

Chongqing, China Lahore, Pakistan De-Sheng Pei Naima Hamid

Acknowledgments

The authors are grateful for the support from the High-level Talents Project of Chongqing Medical University (No. R4014) and the Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Contents

Part I Introduction

1	An	Introduction to Reservoir Ecotoxicology	3
	De	-Sheng Pei	
	1	Definition of Reservoir Ecotoxicology	3
	2	Types of Reservoirs	4
	3	Distribution of Reservoirs in the World	4
	4	Environmental Impacts of Reservoirs	6
		4.1 Water Temperature Change	6
		4.2 Reservoir Sedimentation Risk	7
		4.3 Greenhouse Gas Pollution and Climate Change	7
		4.4 Water Pollution	8
		4.5 Effects of Water-Level Fluctuation Zone	8
	5	Conclusion	9
	Re	ferences	9
Par	t II	Environmental Characteristics in Reservoir Ecosystems	
Par	t II	Environmental Characteristics in Reservoir Ecosystems	15
Par 2	t II Ch	Environmental Characteristics in Reservoir Ecosystems	15
Par 2	t II Ch Ma	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem	15
Par 2	t II Ch Ma 1	Environmental Characteristics in Reservoir Ecosystems aracterization of Reservoir Ecosystem arriya Sultan and De-Sheng Pei Introduction Site Spacific Charactere	15 15
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems aracterization of Reservoir Ecosystem	15 15 16
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems arriva Sultan and De-Sheng Pei Introduction Site-Specific Characters 2.1 Biophysical Setting 2.2 Climate	15 15 16 16
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems arriva Sultan and De-Sheng Pei Introduction Site-Specific Characters 2.1 Biophysical Setting 2.2 Climate	15 15 16 16
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem arriya Sultan and De-Sheng Pei Introduction	15 15 16 16 16 18
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem arriya Sultan and De-Sheng Pei Introduction	15 16 16 16 18 18
Par 2	t II Ch Ma 1 2	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem arriya Sultan and De-Sheng Pei Introduction	15 16 16 18 18
Par 2	t II Ch Ma 1 2 3	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem arriya Sultan and De-Sheng Pei Introduction Site-Specific Characters 2.1 Biophysical Setting 2.2 Climate 2.3 Geology Physical Transport Mechanisms 3.1 Hydrology 3.2 Erosion	15 16 16 18 18 18
Par 2	t II Ch Ma 1 2 3 4	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem marriya Sultan and De-Sheng Pei Introduction Site-Specific Characters 2.1 Biophysical Setting 2.2 Climate 2.3 Geology Physical Transport Mechanisms 3.1 Hydrology 3.2 Erosion Natural Communities	15 16 16 18 18 18 19 19
Par 2	t II Ch Ma 1 2 3 4	Environmental Characteristics in Reservoir Ecosystems maracterization of Reservoir Ecosystem marriya Sultan and De-Sheng Pei Introduction Site-Specific Characters 2.1 Biophysical Setting 2.2 Climate 2.3 Geology Physical Transport Mechanisms 3.1 Hydrology 3.2 Erosion Natural Communities 4.1	15 16 16 18 18 18 19 19 20

		4.3 Macroinvertebrate Community Composition	22	
		4.4 Plant Communities and Distribution Patterns	23	
	5	Conclusion	24	
	Re	terences	24	
3	M	ain Sources and Distribution of Pollutants in the Reservoir	29	
	N٤	ima Hamid and De-Sheng Pei		
	1	Introduction	29	
	2	Sources of Pollutants	30	
	3	Source Apportionment Techniques	31	
	4	Pollution in the TGRA	31	
	5	Spatial Distribution of Pollutants in the TGRA	32	
		5.1 Heavy Metals	32	
		5.2 Polycyclic Aromatic Hydrocarbons (PAHs)	34	
		5.3 Pharmaceuticals and Personal Care Products (PPCPs)	35	
		5.4 Phthalates (PAEs)	35	
		5.5 Microplastics (MPs)	35	
		5.6 Nutrient or Algal Pollution	36	
		5.7 Microbial Diversity and Pollution	37	
	6	Conclusion	38	
	Re	ferences	38	
4	Migration and Transformation of Pollutants in the Reservoir 4			
	Na	aima Hamid and De-Sheng Pei		
	1	Introduction	41	
	2	Fluctuating Impoundment Level Effects	42	
	3	Status and Migration of Pollutants in the TGRA	44	
	4	Temporal and Seasonal Variations of Pollutants	44	
	5	Ecological Health of the TGRA	45	
	6	Conclusion	45	
	Re	ferences	48	
5	н	armful Algal Bloom in the Reservoir	51	
5	Su	man Thodhal Yoganandham and De-Sheng Pei	51	
	1	Introduction	51	
	2	The Formation of Algal Bloom in the Water Bodies	53	
	3	Conditions that Stimulate Harmful Algal Blooms in the		
		Reservoir	54	
	4	Reservoir Algal Bloom in Different Countries	55	
	5	Methods Used to Control Harmful Algal Bloom in the		
		Reservoirs	56	
		5.1 Chemicals	57	
		5.2 Aeration	57	
		5.3 Mixing	58	
		5.4 Ultrasound	58	
	6	Conclusion	59	
	Re	eferences	59	

6	Mi	Microplastics Pollution in the Reservoir: Occurrence, Extraction,		
	an	d Characterization		
	Ma	arriya Sultan, Suman Thodhal Yoganandham, and De-Sheng Pei		
	1	Introduction		
	2	Sources of MPs in Reservoirs		
	3	Occurrence and Distribution of MPs in Reservoirs		
	4	MPs Sampling and Analysis Techniques		
		4.1 Sample Collection		
		4.2 Sample Pre-treatment		
		4.3 Characterization of MPs		
	5	Conclusion		
	Re	ferences		
7	Di	stribution of Microorganisms in the Reservoir		
	Na	ima Hamid and De-Sheng Pei		
	1	Introduction		
	2	Factors Affecting the Microbial Diversity in the Aquatic		
		Environment		
	3	Microbial Diversity and Phylogenetic Distribution Pattern in the		
		TGRA		
	4	Effect of Environmental Variations on Microbial Diversity in the		
		TGR		
	5	The TGRA Ecological Health via Bacteria-Based Index		
	6	Conclusion		
	Re	ferences		
Par	t III	I Study Methods of Ecotoxicology in the Reservoir		
8	Ec	otoxicity Test Methods of Primary Producers in the Reservoir .		
	Ma	arriya Sultan and De-Sheng Pei		
	1	Introduction		
	2	Standardized Ecotoxicity Tests Using Primary Producers		
		2.1 Luminescence Inhibition Test of <i>Vibrio fischeri</i>		
		2.2 Growth Inhibition Test of <i>Raphidocelis subcapitata</i>		
		2.3 Growth Inhibition Test of <i>Spirodela polyrhiza</i>		
		2.4 Vegetative Shoot Inhibition Test of <i>Myriophyllum spicatum</i> .		
	3	Ecotoxicity Assessment of Freshwater Reservoirs Using Primary		
		Producers		
	4	Concerns in Standardized Primary Producers-Based Ecotoxicity		
		Tests		
	5	Conclusions		
	Re	ferences		

9	Ecotoxicology Methods of Reservoir Water Using Invertebrates	99
	Marriya Sultan and De-Sheng Pei	
	1 Introduction	99
	2 Invertebrates for Ecotoxicological Testing	100
	3 Invertebrate Test Species	100
	4 Ecotoxicity Tests	101
	4.1 Acute Toxicity Tests	103
	4.2 Chronic Toxicity Tests	105
	5 Bioaccumulation	106
	6 Conclusion	107
	References	108
10		110
10	Marriya Sultan and De-Sheng Pei	113
	1 Introduction	113
	2 Ecotoxicological Importance of Fish and Its Use in Ecotoxicity	
	Testing	114
	3 Fish Models	114
	4 Fish Reservoir Ecotoxicity Tests	117
	4.1 Acute Toxicity Tests	118
	4.2 Chronic Toxicity Tests	120
	4.3 Bioconcentration and Bioaccumulation Tests	121
	5 Limitations of Fish Ecotoxicity Testing	122
	6 Conclusions	122
	References	122
11	Ecotoxicology Methodology of Sediment Toxicity in the	
	Reservoir	127
	Suman Thodhal Yoganandham and De-Sheng Pei	
	1 Introduction	127
	2 Sediments Collection	128
	3 Manipulations of Sediments Before Testing	129
	4 Pathways for Pollutant Exposure	129
	5 Test Organisms for Sediment Toxicity	130
	5.1 Algae	131
	5.2 Amphipods	131
	5.3 Bivalves	131
	6 Toxicity Endpoints	132
	7 Conclusion	132
	References	133
12	Mesocosm Study in the Reservoir Ecosystem	137
	Marriya Sultan and De-Sheng Pei	
	1 Introduction	137
	2 Use of Freshwater Mesocosms in Lakes and Reservoir	
	Ecosystems	138
	3 Significance of Mesocosms in Ecotoxicity Testing	139

	4 Size and Types of Mesocosms	139
	5 General Design of Aquatic Mesocosms	143
	5.1 Preliminary Aspects	143
	5.2 Mesocosm Assembly	143
	5.3 Protocol of Mesocosm Preparation	144
	6 Limitations	145
	7 Conclusion	145
	References	146
Par	t IV Ecotoxicological Effects and Mechanism of Pollutants in the	
12	Reservoir Malaulan Tariaita Machanism of Haam Matala in the Basemain	152
15	Suman Thodhal Yoganandham and De-Sheng Pei	155
	1 Introduction	153
	2 Heavy Metal Pollution in the Reservoir	154
	3 Bioaccumulation of Heavy Metals in the Reservoir's Organisms	155
	4 Molecular Toxicity Mechanism of Heavy Metals	156
	5 Mechanism of Heavy Metal's Detoxification	158
	6 Conclusion	158
	References	159
14	Molecular Toxicity Mechanism of Persistent Organic Pollutants	
	(POPs) in the Reservoir	163
	Naima Hamid and De-Sheng Pei	
	1 Introduction	163
	2 POPs Mechanistic Toxicity in the Three Gorges Reservoir (TGR)	165
	3 AhR Metabolism Mechanism	165
	4 POPs Toxicogenetic Endpoints <i>In vitro</i> and <i>In vivo</i>	165
	5 Conclusion	168
	References	169
15	Molecular Toxicity Mechanism of Microplastics in the Reservoir De-Sheng Pei and Suman Thodhal Yoganandham	173
	1 Introduction	173
	2 MPs Distribution in the Reservoir	174
	3 Toxicity Mechanisms of MPs	175
	4 Combined Toxicity of MPs with Other Environmental Pollutants	177
	5 AOPs Relevance to MPs Toxicity	178
	6 Conclusion	179
	References	179
16	Molecular Toxicity Mechanism of Plasticizers in the Reservoir	183
	Naima Hamid and De-Sheng Pei	
	1 Introduction	183
	2 DEHP Mechanistic Toxicity in the Three Gorges Reservoir Area	
	(TGRA)	184

	3	Plasticizers' Toxic Effect on Peroxisome Proliferator-Activated	105
	4	The Impairment of Thyroid and Estrogen Axes by Dathalates and	185
	4	RPA	185
		4.1 Phthalates and BPA Effects on Thyroid Axis	186
		4.7 Phthalates and BPA Effects on Estrogen Axis	187
	5	Combined Reproductive Effects of Plasticizers	187
	6	Conclusion	188
	Re	oferences	192
17	м	alecular Tavicity Mechanism of PPCPs in the Reservair	107
1/	Na	aima Hamid and De-Sheng Pei	177
	1	Introduction	197
	2	PPCPs Mechanistic Toxicity in the Three Gorges Reservoir Area	
		(TGRA)	198
	3	Mechanism of Detoxification Metabolism Pathway	199
	4	PPCPs Toxicogenetic Endpoints In vitro or In vivo	201
	5	Computational Approaches for Toxicological Screening	201
	6	Conclusion	204
	Re	ferences	205
18	Ac	lverse Outcome Pathways (AOPs) of Pollutants in the	
	Re	eservoir	209
	Μ	arriya Sultan and De-Sheng Pei	
	1	Introduction	209
	2	Development of AOPs for Assessment of Water Pollutants	211
		2.1 Bioassays Prioritization for AOPs	212
		2.2 Design of Mechanistic Test Approaches	213
	3	Endpoints in the AOP Framework for Risk Assessment	214
	4	Use of the AOP for Risk Assessment of Pollutants	214
	5	Outlook and Prospects	215
	6	Conclusion	216
	Re	ferences	216
19	In	vasive Alien Species Problem in the Reservoir	219
	Μ	arriya Sultan and De-Sheng Pei	
	1	Introduction	219
	2	Impact of Local and Landscape Variables on Exotic Species	
		Invasion	220
	3	Exotic Plant Species in the TGR Region	222
	4	Exotic Fish Species Along the TGR Region	223
	5	Effect of Exotic Species on the Reservoir Ecosystem	224
	6	Management of Alien Species and Implications for the Reservoir	
		Conservation	225
	7	Conclusion	226
	Re	ferences	226

Part V Conclusions

20	Final Thoughts and Concluding Remarks		
	De-Sheng Pei and Naima Hamid		
	1 Summarized Viewpoint and Conclusion	231	
	References	233	

About the Authors



De-Sheng Pei is a professor at Chongqing Medical University, China. Dr. Pei got his Ph.D. degree from the Institute of Hydrobiology (IHB), Chinese Academy of Sciences (CAS) in 2006. From June 2006 to February 2009, Dr. Pei worked at IHB (CAS) as an assistant professor. From February 2009 to February 2011, as a senior postdoctoral research associate, Dr. Pei concentrated his study on the mechanisms of DNA damage and repair at Northeastern University (USA). Thereafter, Dr. Pei worked as a Research Fellow at Dana-Farber Cancer Institute/Harvard Medical School (USA) to study the mechanism of carcinogenesis of neuroblastoma from February 2011 to February 2013. In August 2012, Dr. Pei got the professor position at the Chongging Institute of Green and Intelligent Technology (CIGIT), Chinese Academy of Sciences (CAS). From February 2012 to September 2021, he led the Research Center for Environment and Health (RCEH) at CIGIT. From September 2021 till now, he was introduced to Chongqing Medical University as a distinguished professor of excellent high-level talent. His discoveries have been published in more than 120 journals, including Cancer Cell, Nucleic Acids Research, Brain, PLOS Genetics, Biomaterials, Journal of Hazardous Materials, Environmental Pollution, etc.



Naima Hamid joined Lahore College for Women University as an assistant professor in November 2021. She completed doctoral studies at the University of Chinese Academy of Sciences (UCAS), China, which ranked 84th in the World University rankings for 2020–2021. She was awarded the prestigious Chinese Academy of Sciences (CAS)-The World Academy of Sciences (TWAS) presidential scholarship for three years (2017-2020). Her research work was focused on the combined toxicity of phthalates or pharmaceuticals and personal care products (PPCPs) with endocrinedisrupting perspectives. She published more than 25 international SCI articles, including Journal of Hazardous Materials, Science of the Total Environment, Environmental Pollution. Aquatic toxicology, Chemosphere, etc. She remains associated with the Ministry of Climate Change, Pakistan, and worked on the National Climate Change Policy. As a research scholar, she joined Pakistan Meteorological Department with the climate change and modeling research group. Her research interests are focused on ecotoxicology and ecological risk assessment of hazardous chemicals in environmental matrices. Presently, she is involved in the development of integrated experimental and computational monitoring methods of environmental pollutants with special reference to their toxicological mechanisms.



Marriya Sultan is currently enrolled in a Ph.D. program in environmental sciences at the Chongqing Institute of Green and Intelligent Technology, Chinese Academy of Sciences under a CAS-TWAS fellowship 2019. Previously, she obtained her M.Phil. degree in environmental sciences from Quaid-I-Azam University, Islamabad, Pakistan. Currently, she is working on a Ph. D. project, entitled "Co-existing characteristics of microplastics and DEHP in the Three Gorges Reservoir Area and mechanism of compound reproductive toxicity on zebrafish model," under the supervision of Prof. & Dr. De-Sheng Pei. Her expertise includes the characterization of organic and inorganic environmental pollutants (exposure routes, dose-response, and mode of action), rigorous evaluations of contamination using novel techniques, and analysis of environmental pollutants data.



Suman Thodhal Yoganandham got his Ph.D. degree from Sathyabama University in 2017. From June 2014 to December 2018, Dr. S. Thodhal Yoganandham worked at the Center of Ocean Research in Sathyabama University as a senior scientific assistant. From December 2018 to March 2021, as a postdoctoral research fellow, Dr. S. Thodhal Yoganandham concentrated his study on the characterization of marine plastic debris and the effects of microplastics on aquatic organisms at Henan Normal University & Chongqing Institute of Green and Intelligent Technology, Chinese Academy of Sciences (China). He currently works as a postdoctoral research fellow for environment engineering at Changwon National University, South Korea. His research mainly focuses on micropollutants and their environmental impacts.

Abbreviations

ABS	Absorbance Spectrum
AFT	Acute Fish Toxicity
AgNPs	Silver Nanoparticles
AHRR	Aryl hydrocarbon receptor repressor
AhRs	Aryl hydrocarbon receptors
Al	Aluminum
AO	Adverse outcome
AOP	Adverse Outcome Pathway
AOX	Alternative oxidase
APEs	Alkylphenol ethoxylates
APHA	American Public Health Association
APs	Alkylphenols
As	Arsenic
ASD	Autism Spectrum Disorder
ASTM	American Society for Testing and Materials
ATR	Attenuated Total Reflectance
Ba-IBI	Bacteria-based Index of Biotic Integrity
BBP	Benzyl-butyl phthalate
BCF	Bioconcentration Factor
BCFK	Bioconcentration Factor Kinetics
BF	Bezafibrate
BPA	Bisphenol A
CAT	Catalase
CBZ	Carbamazepine
CCA	Canonical Correspondence Analysis
Cd	Cadmium
CFB	Cytophaga Flavobacterium Bacteroides
CI	Chemical index
CI	Combination index
Со	Cobalt
COD	Chemical Oxygen Demand

СР	Chlorella pyrenoidosa
Cr	Chromium
CYP1A, CYP11A1	Cytochrome P450 family
DBP	Di-butyl phthalate
DDT	Dichlorodiphenyltrichloroethane
DEGs	Differentially Expressed Genes
DEHP	Di-(2-ethylhexyl) phthalate
DEP	Di-ethyl phthalate
DIC	Diclofenac
DMP	Dimethyl phthalate
DNA	Deoxyribonucleic acid
DNOP	Di-n-octyl phthalate
DO	Dissolved Oxygen
EC ₅₀	Half maximal effective concentration
ECCC	Environment and Climate Change Canada
EDCs	Endocrine-Disrupting Chemicals
EDS	Energy Dispersive Spectroscopy
EE	Ethinyl estradiol
EE2	Ethynylestradiol
ELISA	Enzyme-linked immunoassay
EPA	Environmental Protection Agency
ER	Estrogen Receptor
ERCs	Environment Relevant Concentrations
ERα	Estrogen Receptor alpha
Erβ	Estrogen Receptor beta
ESEM	Environmental Scanning Electron Microscopy
EU	European Union
F1	First generation
F2	Second generation
FA	Factor Analysis
FAO	Food and Agriculture Organization
Fe	Iron
FET	Fish Embryo Toxicity
FTIR	Fourier transform infrared
GI	Germination Index
GMO	Glycol monostearate
GMS	Glycerin monostearate
GO	Gene Ontology
GPx	Glutathione peroxidase
GST	Glutathione S-transferase
HA	Humic acid
HAB	Harmful algal bloom
HFB	Hydro-fluctuation Belt
Hg	Mercury

HM	Heavy Metals
HPF	Hours post fertilization
HPF	Hours per fertilization
HPG	Hypothalamus pituitary gonad
HPT	Hypothalamus pituitary thyroid
IA	Independent addition
Igeo	Geoaccumulation Index
IM	Indomethacin
KE	Key Events
KERs	Key Event Relationships
LC ₅₀	Lethal Concentration that kills 50% population
LC-MS	Liquid chromatography mass spectroscopic analysis
LH	Luteinizing hormone
LOEC	Low Observed Effect Concentration
LPO	Lipid peroxidation
MAPK	Mitogen-activated protein kinase
MDA	Malondialdehyde
MDR	Molecular Diagnostic Ratios
MET	Methionine
MIE	Molecular Initiating Event
MLR	Multiple Linear Regression
Mn	Manganese
MO	Microorganisms
MPs	Microplastics
NDD	Neurodevelopmental Disorders
Ni	Nickel
NMR	Nuclear magnetic resonance spectroscopy
NOEC	No Observed Effect Concentration
NOEL	No Observed Effect Limit
NPs	Nanoparticles
NSAIDs	Non-steroidal anti-inflammatory drugs
OCPs	Organochlorine Pesticides
OECD	Organization for Economic Co-operation and Development
OM	Organic Matter
PA	Polyamide
PAEs	Phthalate Esters
PAHs	Polycyclic Aromatic Hydrocarbons
Pb	Lead
PBBs	Polybrominated Biphenyls
PBDEs	Polybrominated Diphenyl Ethers
PCA	Principle Component Analysis
PCBs	Polychlorinated Biphenyls
PCDFs	Polychlorinated Dibenzofurans
PCNs	Polychlorinated Naphthalenes

PCPs	Personal care products
PE	Polyethylene
PECs	Predicted Effect Concentrations
PET	Polyethylene Terephthalate
PFOS	Perfluorooctane Sulfonates
PKs	Pharmaco-kinetics
POPs	Persistent Organic Pollutants
PPARs	Peroxisome proliferator receptors
PPCPs	Pharmaceutical and Personal care products
PS	Polystyrene
PVA	Polyvinyl alcohol
PVC	Polyvinyl chloride
qAOP	Quantitative Adverse Outcome Pathway
RA	Redundancy Analysis
RNA	Ribonucleic acid
ROS	Reactive Oxygen Species
RXR	Retinoid X receptor
SAs	Sulfonamides
SDZ	Sulfadiazine
SEM	Scanning Electron Microscopy
SEPA	State Environmental Policy Act
SMR	Sulfamerazine
SMX	Sulfamethoxazole
SO	S. obliquus
SOD	Superoxide dismutase
SPY	Sulfapyridine
ST	Sulfameter
SULTs	Sulfotransferases
T ₂	Testosterone
T ₃	Triiodothyronine
T_4	Thyroxine
TBT	Tributyltin
TC	Tetracycline
TCDD	Tetrachlorodibenzo-p-dioxin
TGD	Three Gorges Dam
TGDR	Three Gorges Dam Region
TGR	Three Gorges Reservoir
TGRA	Three Gorges Reservoir Area
TH	Tetracycline hydrochloride
TH	Thyroid axis
TMP	Trimethoprim
TOC	Total Organic Carbon
TOX ₂₁	Toxicity testing in the 21st century
TPs	Transmembrane proteins
	-

Thyrotropin-releasing hormone
Toxic Substances Control Act
Thyroid-stimulating hormone
Glucuronosyltransferases
United States Environmental Protection Agency
Vitellogenin
Vitellin
Water-Level Fluctuation Zone
Xenobiotic Response Element
Zinc

Part I Introduction

Chapter 1 An Introduction to Reservoir Ecotoxicology



De-Sheng Pei 💿

Abstract Reservoir ecotoxicology is a new term proposed by us for the description of ecology and toxicology in the reservoir, focusing on the effects of toxic substances and especially pollutants in the reservoir environment. As we know, reservoir construction is the key strategy for water resource production and management, such as irrigation, hydroelectric power, flood control, and shipping. Different types of reservoirs possess distinct pollution characteristics, and the toxicological impacts of contaminants on wildlife in the reservoirs deserve in-depth studies. Reservoir ecotoxicology aims to protect reservoir native flora and fauna at the population, community, and ecosystem levels and support their ecosystem integrity and resilience. This chapter provides an overview of reservoir ecotoxicology with an emphasis on topics and perspectives that are different from other fields of ecotoxicology.

Keywords Reservoir ecotoxicology \cdot Dams \cdot Ecosystem \cdot Water resource \cdot Environmental impacts

1 Definition of Reservoir Ecotoxicology

The term reservoir ecotoxicology is derived from ecotoxicology focusing on ecology and toxicology in the reservoir, which is a new term proposed as a modern scientific subdiscipline of environmental toxicology. Reservoirs as open-air storage areas of many water supply systems around the world possess their own environmental characteristics [1]. Thus, reservoir ecotoxicology has noticeable and different features, compared to other ecotoxicology. To well address reservoir ecotoxicology, we should clarify the environmental characteristics of reservoirs, including the source, distribution, and migration of pollutants in the reservoir. Moreover, we need to figure out the diversity of species composition (such as benthos, plankton, and nekton), food chain, and reservoir ecosystem in the hydro-fluctuation belt, backwater zone, reservoir bank, riparian zone, open water belt, and deep-water belt. Thus, reservoir ecotoxicology can be broadly defined as the scientific study of interactions among organisms including humans and their reservoir environment, typically focusing on protecting reservoir native flora and fauna at the population, community, and ecosystem levels. Besides, the ecotoxicological effect and molecular mechanisms of toxicity in the reservoir should be different because of the specific surrounding environment, compared to other water bodies [2].

2 Types of Reservoirs

Reservoirs are large open storage structures for collecting and storing water, and water bodies contained by embankments or a dam. Reservoirs play important roles in irrigation, flood control, hydroelectric power, water supply, and leisure activities [3, 4]. Reservoirs can be classified into different types according to construction purposes and dam location [5, 6]. Based on the terrain trait, reservoirs can be divided into four types: plain reservoirs [7], valley reservoirs [8], underground reservoirs [9], and coastal reservoirs [10]. According to use features, reservoirs are also classified as follows: storage reservoirs [11], flood control reservoirs [12], distribution reservoirs [13], and multipurpose reservoirs [14]. Based on their function, reservoirs are also classified into three types: bank-side reservoirs [15], valley-dammed reservoirs [16], and service reservoirs [17]. Thus, types of reservoirs may be named differently in terms of multiple classification criteria. Here, we summarized the major types of reservoirs based on different classifications (Table 1.1).

3 Distribution of Reservoirs in the World

Although there are many types of reservoirs, the first and usually largest type of reservoir is valley-dammed reservoirs. Indeed, around 30-40% of irrigation water was supplied by dammed reservoirs globally [18]. Hydropower generated 16.6% of the world's electricity by 2015 and 71% of all renewable electricity in 2016 [18, 19]. The Jawa Dam in Jordan is the world's oldest dam, which was built around 3000 BCE to store irrigation water [20]. Globally, 36,222 dams were identified and they are spatially concentrated along major river basins in Asia, North America, South America, and Europe (Fig. 1.1) [21]. As shown in Fig. 1.1, dammed reservoirs are mainly used for irrigation (34%) and hydroelectricity (25%). Although many developed countries in North America, Europe, and Oceania reduce the construction number of dammed reservoirs since the 1970s, developing countries in Africa, Asia, and South America accelerate the progress of the construction of dammed reservoirs. To date, Asia has the highest number (10,138) of dammed reservoirs completed and possesses 28% of worldwide dam construction. India's first dam is the Kallanai built on the Cauvery River. India alone has over 5200 dams for various purposes, among which Indira Sagar Dam Reservoir is the largest reservoir in India followed by Nagarjuna Sagar Reservoir.

Classification				
criteria	Types	Features	Subcategories	Examples
Terrain	Plain reservoirs	Shallow, low dams, and widespread water surfaces	River-type and dam-type	Qianmu Dang, Haiyan, Zhe- jiang, China
	Valley- dammed reservoirs	Located in narrow valley areas where tremendous amounts of water can be held in by the valley's sides and a dam	Large-scale, medium-sized, and small	Owen Falls Dam, Jinja, Uganda
	Underground reservoirs	Stored water in the gap between soil and rock or cave	Dam-type, non-dam- type, and funnel fill- ing type	Mihuaishun, Beijing, China; Montsouris Reservoir, Paris, France
	Coastal reservoirs	A water body is enclosed by a barrier or barriers inside a large waterbody for specific purposes	Inner river, mouth, beyond river mouth, and beside river mouth	Punggol Res- ervoir, Singapore
Construction purpose	Storage reservoirs	To maintain mini- mum supplies of water for irrigation, hydroelectric gener- ation, etc.	Large-scale, medium-sized, and small	Owen Falls Dam, Jinja, Uganda; Kariba's reser- voir, Zambezi
	Flood control reservoirs	Temporarily store the flood water and release it slowly at a safe rate after the floods	Detention basins and retarding basins	Three Gorges Reservoir, China
	Distribution reservoirs	Small storage reser- voir to distribute and manage water in a city	Surface reservoirs, elevated reservoirs, and standpipes	Location in water supply systems
	Multipurpose reservoirs	Protect the down- stream areas from floods and provide irrigation, water sup- ply, hydroelectric purpose, etc.	Irrigation, power, flood control, municipal and industrial, recrea- tion, and fish and wildlife benefits	Bhakra Dam, Nangal, India; Nagarjuna Sagar Dam, Andhra Pradesh, India
Function	Bank-side reservoirs	Made by diverting water from rivers and streams to an existing reservoir	By excavation, encircling bund, and embankment	JhongJhuang Bank-Side Reservoir, Tai- wan, China
	Valley- dammed reservoirs	Water is contained by the walls of a valley in mountain ranges	Large-scale, medium-sized, and small	Lake Mead Reservoir, Nevada, USA

Table 1.1 Classification of reservoirs in the world

(continued)

Classification criteria	Types	Features	Subcategories	Examples
	Service reservoirs	Reservoirs located above the ground or below the ground	Water towers and other elevated structures	Honor Oak Service Reser- voir, London, UK



Fig. 1.1 The spatial location of world dams based on their functions. Data were retrieved from AQUASTAT, GRanD, WRI, and GDAT and modified by Zhang TB et al. [21]. Irrigation (Green), Hydroelectricity (Violet blue), Water Supply (Blue), Flood Control (Red), Livestock (Yellow), Recreation (Magenta), and Other (Gray)

4 Environmental Impacts of Reservoirs

Reservoirs were established for generating hydroelectric power, controlling floods, improving water quality for irrigation and drinking, and offering recreational opportunities. However, the construction of reservoirs potentially may impede the flow of essential nutrients, affect aquatic life, and release greenhouse gas [18, 22]. Due to anti-seasonal hydrological regulation for most flood control reservoirs, dynamic change in a water-level fluctuation zone (WLFZ) also raises critical environmental concerns in the riparian zone [23].

4.1 Water Temperature Change

As we know, water released from deep outlets below the surface of large dams can significantly disturb water temperature in the downstream river, consequently affecting aquatic biota and river health. A previous study showed that cold-water pollution occurred in the downstream reaches of 11 major Murray–Darling Basin (Australia) dams, where water temperature varied up to 11.1–16.7 °C [24, 25]. Although the adverse consequences of water temperature pollution on river health were widely recognized, little was done to correct this problem in Australia. Thus, species of native warm water fish may be extinct in cold-water pollution-affected rivers. The ecosystem downstream can be adversely affected, and reduced temperatures can be observed in the 250–350 km downstream [25]. Thermal suppression is considered by the operators of Burrendong Dam on the Macquarie River (Eastern Australia) via hanging a geotextile curtain on a new outlet turret to selectively release the surface water [26].

4.2 Reservoir Sedimentation Risk

Sedimentation is one of the most important threats to reservoir ecosystems around the world. Reservoir sedimentation is the filling of the reservoir behind a dam with sediment carried into the reservoir by streams. The water-storing capacity of the reservoir is automatically reduced after the deposition of sediment. When the process of sediment deposition lasts longer, the whole reservoir will be silted up and lose function [27, 28]. The accumulation of sediment limits the storage and lifespan of the reservoir [29], and the reservoir's life can be divided into three stages: continuous and rapidly occurring sediment accumulation, partial sediment balance, and full sediment balance. Aquatic organisms are sensitive to changes in sediment supply and flow regimes [30]. Moreover, reservoir sediment is a primary carrier of suspended pollutants including nitrogen, phosphorous, and heavy metals [31], which greatly affects the reservoir environment.

4.3 Greenhouse Gas Pollution and Climate Change

Dammed reservoirs are thought to be an important source of greenhouse gases (GHGs) in the atmosphere [32]. In the reservoir, carbon dioxide is emitted into the atmosphere through the deforestation process, contributing to 20% of the GHGs emissions [33]. GHGs emissions from reservoir water surfaces were estimated to account for 0.8 (0.5–1.2) Pg CO₂ equivalents per year, with the majority of this forcing due to CH₄ [34]. The terrestrial organic matter stocked in the reservoir may fuel microbial decomposition and convert organic matter to CO₂, CH₄, and N₂O [35]. The climate impact of CH₄ was up to 25 times greater than CO₂ on a 100-year scale [36]. Currently, atmospheric CH₄ concentrations in the last 650 ky are the highest. Moreover, the GHG footprint of reservoirs deserves to quantify based on whether they have generated new fluxes visible to the atmosphere [37]. Of note, China firstly announced "dual carbon goals: carbon peaking and carbon neutrality" at the 75th session of the United Nations General Assembly on September 22, 2020

[38]. China will aim to achieve peak CO_2 emissions before 2030 and carbon neutrality before 2060 [39], demonstrating China's determination in pursuing green and low-carbon development and its responsibility as a major and responsible country to actively tackle climate change and safeguard a bright future for humanity. Thus, reducing greenhouse gas emissions from reservoirs should be an important tache, contributing to realizing this goal.

4.4 Water Pollution

Water quality in reservoirs is deteriorating due to population growth, industrialization, and urbanization. Reservoirs have a more vulnerable and complex ecosystem than rivers because they do not have a weak self-purifying ability. Reservoirs as environmental sinks collect not only sediments but also most of the pollutants that are washed into them. Nutrients (nitrogen and phosphorus) are extracted from flooded plants and soil. Reservoir eutrophication may occur due to large influxes of organic loads and/or nutrients [18, 40]. Moreover, heavy metals have recently been identified as major pollutants in sediments of the reservoir, such as As, Cd, Cr, Cu, Ni, Pb, and Zn [41].

4.5 Effects of Water-Level Fluctuation Zone

Reservoir water-level fluctuation zone (WLFZ) is a new and fragile ecosystem and attracts high attention, which refers to the drawdown area formed between the highest and the lowest water level due to seasonal water-level drawdown and periodic water storage. In the Three Gorges Reservoir (China), the height of WLFZ is 30 m because of the discharge-storage cycle between 145 m and 175 m at the Three Gorges Dam. The WLFZ usually contains three belts: the littoral zone (wooded wetland, wet meadow, marsh, and aquatic vegetation), the riparian zone, and the riparian ecotone zone [42]. Due to the annually cyclic variation of the water level in the Three Gorges Reservoir, frequent alternation of erosion and deposition profoundly disturbs the geochemical and biological processes and affect the water quality and ecosystem of the reservoir. Consequently, nutrients (typically P and N) and heavy and trace metals may be gradually deposited in the WLFZ at the low-water level, but released into the water at the high-water level, potentially triggering eutrophication incidents or water pollution [43]. Besides, many original vegetation species failed to survive under prolonged submergence conditions during the inundation stage [44], resulting in the reduction of vegetation diversity in the reservoir. Therefore, to well understand the complex geomorphological, geochemical, and biological processes of reservoir WLFZ needs more in-depth studies.

5 Conclusion

The main aim of reservoir ecotoxicology is to protect the reservoir's native flora and fauna at the population, community, and ecosystem levels. There are different types of reservoirs in the world. Thus, we should figure out the environmental characteristics of reservoirs, including creature types, the composition of reservoir ecological systems, major pollutants in the reservoir, and their migration & conversion. The toxicological impact of contaminants on reservoir organisms should be considered at the population and community levels based on adverse outcome pathways (AOPs). Contaminants exposure is complicated and should reflect bioavailability and other factors. Acute and chronic exposure to combined pollutants, especially because of bioconcentration, bioaccumulation, and biomagnification, will be helpful to elucidate the mechanistic effects at the biochemical level and the biological pathways. Advanced research approaches should be introduced to assess reservoir environmental impacts from both prospective and retrospective views.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The author declares no conflict of interest.

References

- 1. H. Keppler, Earth's deep water reservoir. Nature 507, 174–175 (2014)
- M. Szara, A. Baran, A. Klimkowicz-Pawlas, M. Tarnawski, Ecotoxicological characteristics and ecological risk assessment of trace elements in the bottom sediments of the Rożnów reservoir (Poland). Ecotoxicology 29, 45–57 (2020)
- 3. R.W. Herschy, Dams and reservoirs, role, in *Encyclopedia of Lakes and Reservoirs*, ed. by L. Bengtsson, R.W. Herschy, R.W. Fairbridge, (Springer, Dordrecht, 2012), pp. 191–199
- H. Ma, J.A. Nittrouer, X. Fu, G. Parker, Y. Zhang, Y. Wang, Y. Wang, M.P. Lamb, J. Cisneros, J. Best, D.R. Parsons, B. Wu, Amplification of downstream flood stage due to damming of finegrained rivers. Nat. Commun. 13, 3054 (2022)
- D. Nikolić, S. Skorić, M. Lenhardt, A. Hegediš, J. Krpo-Ćetković, Risk assessment of using fish from different types of reservoirs as human food—a study on European perch (Perca fluviatilis). Environ. Pollut. 257, 113586 (2020)
- 6. J. Galizia Tundisi, Reservoirs: New challenges for ecosystem studies and environmental management. Water Security **4–5**, 1–7 (2018)
- Q. Guo, J. Yu, X. Li, T. Chen, C. Wang, Z. Li, W. Ma, C. Ding, M. Yang, A systematic study on the odorants characterization and evaluation in a plain reservoir with wetlands ecosystem. J. Hazard. Mater. **393**, 122404 (2020)
- Q. Li, Z. Li, Y. Zhang, X. Wu, L. Yang, L. Lu, Y. Chen, Sedimentation supports life-cycle CH4 production and accumulation in a river valley reservoir: A hierarchical Bayesian modeling approach. Water Res. 222, 118861 (2022)
- J. Menéndez, J.M. Fernández-Oro, M. Galdo, L. Álvarez, A. Bernardo-Sánchez, Numerical investigation of underground reservoirs in compressed air energy storage systems considering

different operating conditions: Influence of thermodynamic performance on the energy balance and round-trip efficiency. J. Energy Storage **46**, 103816 (2022)

- S.-Q. Yang, Chapter 2—coastal reservoirs' design and applications, in *Coastal Reservoir Technology and Applications*, ed. by S.-Q. Yang, (Elsevier, 2022), pp. 39–84
- H. Hong, J. Qiu, Y. Liang, Environmental factors influencing the distribution of total and fecal coliform bacteria in six water storage reservoirs in the Pearl River Delta Region, China. J. Environ. Sci. 22, 663–668 (2010)
- K. Huang, L. Ye, L. Chen, Q. Wang, L. Dai, J. Zhou, V.P. Singh, M. Huang, J. Zhang, Risk analysis of flood control reservoir operation considering multiple uncertainties. J. Hydrol. 565, 672–684 (2018)
- 13. G.H. Abplanalp, Multiple use of distribution reservoirs. J. AWWA 68, 574-578 (1976)
- G. Di Baldassarre, N. Wanders, A. AghaKouchak, L. Kuil, S. Rangecroft, T.I.E. Veldkamp, M. Garcia, P.R. van Oel, K. Breinl, A.F. Van Loon, Water shortages worsened by reservoir effects. Nat. Sustain. 1, 617–622 (2018)
- J.A. Steel, A. Duncan, Modelling the ecological aspects of bankside reservoirs and implications for management. Hydrobiologia 395, 133–147 (1999)
- O. Bábek, O. Kielar, Z. Lenďáková, K. Mandlíková, J. Sedláček, J. Tolaszová, Reservoir deltas and their role in pollutant distribution in valley-type dam reservoirs: Les Království Dam, Elbe River, Czech Republic. CATENA 184, 104251 (2020)
- A. Sathasivan, K.C. Bal Krishna, I. Fisher, Development and application of a method for quantifying factors affecting chloramine decay in service reservoirs. Water Res. 44, 4463–4472 (2010)
- T. Maavara, Q. Chen, K. Van Meter, L.E. Brown, J. Zhang, J. Ni, C. Zarfl, River dam impacts on biogeochemical cycling. Nat. Rev. Earth Environ. 1, 103–116 (2020)
- E.F. Moran, M.C. Lopez, N. Moore, N. Müller, D.W. Hyndman, Sustainable hydropower in the 21st century. Proc. Natl. Acad. Sci. 115, 11891–11898 (2018)
- 20. G.S.R. Murthy, K.G. Murty, G. Raghupathy, Designing earth dams optimally, in *Case Studies in Operations Research: Applications of Optimal Decision Making*, ed. by K.G. Murty, (Springer, New York, 2015), pp. 129–165
- T.A. Zhang, J. Urpelainen, W. Schlenker, Power of the river: Introducing the global dam tracker (GDAT). Res. Projects. Energypolicy.columbia.edu 565, 1–22 (2018)
- 22. Z. Guo, W.J. Boeing, E. Borgomeo, Y. Xu, Y. Weng, Linking reservoir ecosystems research to the sustainable development goals. Sci. Total Environ. **781**, 146769 (2021)
- 23. Y. Li, B. Gao, D. Xu, J. Lu, H. Zhou, L. Gao, Heavy metals in the water-level-fluctuation zone soil of the Three Gorges Reservoir, China: Remobilization and catchment-wide transportation. J. Hydrol. 612, 128108 (2022)
- 24. R. Gray, H.A. Jones, J.N. Hitchcock, L. Hardwick, D. Pepper, A. Lugg, J.R. Seymour, S.M. Mitrovic, Mitigation of cold-water thermal pollution downstream of a large dam with the use of a novel thermal curtain. River Res. Appl. 35, 855–866 (2019)
- A. Lugg, C. Copeland, Review of cold water pollution in the Murray–Darling Basin and the impacts on fish communities. Ecol. Manage. Restor. 15, 71–79 (2014)
- 26. R. Gray, Effectiveness of Cold Water Pollution Mitigation at Burrendong Dam Using an Innovative Thermal Curtain, (Master dissertation) (University of Technology, Sydney, 2016)
- C.-H. Wu, C.-N. Chen, C.-H. Tsai, C.-T. Tsai, Estimating sediment deposition volume in a reservoir using the physiographic soil erosion-deposition model. Int. J. Sediment Res. 27, 362–377 (2012)
- 28. L. Khaba, J.A. Griffiths, Calculation of reservoir capacity loss due to sediment deposition in the Muela reservoir, Northern Lesotho. Int. Soil Water Conserv. Res. **5**, 130–140 (2017)
- J. Sedláček, O. Bábek, T.M. Grygar, Z. Lenďáková, J. Pacina, J. Štojdl, M. Hošek, J. Elznicová, A closer look at sedimentation processes in two dam reservoirs. J. Hydrol. 605, 127397 (2022)
- 30. F. Cattanéo, J. Guillard, S. Diouf, J. O'Rourke, D. Grimardias, Mitigation of ecological impacts on fish of large reservoir sediment management through controlled flushing—the case of the Verbois dam (Rhône River, Switzerland). Sci. Total Environ. **756**, 144053 (2021)
- R. Ammar, V. Kazpard, M. Wazne, A.G. El Samrani, N. Amacha, Z. Saad, L. Chou, Reservoir sediments: A sink or source of chemicals at the surface water-groundwater interface. Environ. Monit. Assess. 187, 579 (2015)
- 32. P.S. Keller, R. Marcé, B. Obrador, M. Koschorreck, Global carbon budget of reservoirs is overturned by the quantification of drawdown areas. Nat. Geosci. 14, 402–408 (2021)
- 33. P.M. Fearnside, S. Pueyo, Greenhouse-gas emissions from tropical dams. Nature Climate Change 2, 382–384 (2012)
- 34. B.R. Deemer, J.A. Harrison, S. Li, J.J. Beaulieu, T. DelSontro, N. Barros, J.F. Bezerra-Neto, S.M. Powers, M.A. dos Santos, J.A. Vonk, Greenhouse gas emissions from reservoir water surfaces: A new global synthesis. Bioscience 66, 949–964 (2016)
- 35. J.T. Huttunen, T.S. Väisänen, S.K. Hellsten, M. Heikkinen, H. Nykänen, H. Jungner, A. Niskanen, M.O. Virtanen, O.V. Lindqvist, O.S. Nenonen, P.J. Martikainen, Fluxes of CH4, CO2, and N2O in hydroelectric reservoirs Lokka and Porttipahta in the northern boreal zone in Finland. Glob. Biogeochem. Cycles 16, 3-1-3-17 (2002)
- 36. G. Yvon-Durocher, A.P. Allen, D. Bastviken, R. Conrad, C. Gudasz, A. St-Pierre, N. Thanh-Duc, P.A. del Giorgio, Methane fluxes show consistent temperature dependence across microbial to ecosystem scales. Nature 507, 488–491 (2014)
- 37. Y.T. Prairie, J. Alm, J. Beaulieu, N. Barros, T. Battin, J. Cole, P. del Giorgio, T. DelSontro, F. Guérin, A. Harby, J. Harrison, S. Mercier-Blais, D. Serça, S. Sobek, D. Vachon, Greenhouse gas emissions from freshwater reservoirs: What does the atmosphere see? Ecosystems 21, 1058–1071 (2018)
- Y.-M. Wei, K. Chen, J.-N. Kang, W. Chen, X.-Y. Wang, X. Zhang, Policy and management of carbon peaking and carbon neutrality: A literature review. Engineering (2022)
- C. Zhao, S. Ju, Y. Xue, T. Ren, Y. Ji, X. Chen, China's energy transitions for carbon neutrality: Challenges and opportunities. Carbon Neutrality 1, 7 (2022)
- J. Yu, Y. Zeng, J. Chen, P. Liao, H. Yang, C. Yin, Organic phosphorus regeneration enhanced since eutrophication occurred in the sub-deep reservoir. Environ. Pollut. 306, 119350 (2022)
- X. Wang, L. Zhang, Z. Zhao, Y. Cai, Heavy metal pollution in reservoirs in the hilly area of southern China: Distribution, source apportionment and health risk assessment. Sci. Total Environ. 634, 158–169 (2018)
- Y. Bao, P. Gao, X. He, The water-level fluctuation zone of three gorges reservoir—a unique geomorphological unit. Earth Sci. Rev. 150, 14–24 (2015)
- 43. W. Shi, M. Du, C. Ye, Q. Zhang, Divergent effects of hydrological alteration and nutrient addition on greenhouse gas emissions in the water level fluctuation zone of the three gorges reservoir, China. Water Res. **201**, 117308 (2021)
- 44. C. Peng, L. Zhang, H. Qin, D. Li, Revegetation in the water level fluctuation zone of a reservoir: An ideal measure to reduce the input of nutrients and sediment. Ecol. Eng. 71, 574–577 (2014)

Part II Environmental Characteristics in Reservoir Ecosystems

Chapter 2 Characterization of Reservoir Ecosystem



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter intends to describe the ecosystem characterization of the reservoir and understands the effect of various environmental stressors on the species diversity of various communities. The ecological trait of reservoirs is subjective to their association with the watershed and functional dynamics utilities of the impounded water. Various temporal and spatial studies related to fish assemblage, microbial diversity, and the composition of macroinvertebrates & vegetation revealed that the ecosystem of the reservoir has been changed in terms of species richness and composition due to the stressors like hydrologic alterations, nutrient loading, sedimentation flux, and erosion, resulting in the modification of ecological characteristics. Moreover, anthropogenic activities have affected the water quality, which is also the major cause of changes in species composition of flora and fauna in the reservoir.

Keywords Ecosystem \cdot Ecological characterization \cdot Diversity \cdot Species composition \cdot Water quality

1 Introduction

Reservoir characterization is the quantitative description of reservoir information gathered from different sources. According to the report by Kelkar [1], reservoir characterization is the elucidation of reservoir properties that reliably incorporates data of different qualities and capacities. In the process of reservoir characterization, information composed at several levels needs to be combined into a single, ample, and coherent depiction of the reservoir [2].

Characterization of the reservoir ecosystem provides a precise description of the present ecological tenets of the site and assists in the identification of ecological resources of interest, such as critical habitats, wetlands, and endangered species that are affected by remedial actions. Moreover, it also aids to ascertain areas that require further sampling or monitoring and provides ecological information for exposure and toxicity assessments [3].

The ecological subtleties of the reservoirs are influenced by their association with the watershed, functional dynamics, and manifold utilities of the impounded water.

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_2

Reservoirs pose various challenges for ecosystem management. Since different associations exist between different operational processes in the reservoir that impact the ecosystem directly or indirectly, the integration of ecologically changing aspects of the reservoirs with management is necessary. For this reason, integrated basin-scale analysis and a proper understanding of reservoir engineering and operational techniques are required [4].

Figure 2.1a explains the different components on the basis of which the ecology of the reservoir is characterized. It includes specific characteristics (location, existing resources, and environmental conditions geology of an area) and physical transport mechanisms, including hydrology, erosion, sedimentation, and the description of prevalent natural communities in habitats around the reservoir area. The services that the reservoir provides become diversified by the passage of time. Moreover, the water quality parameter is also vital and if it is well-upheld, then the overall degradation could be prevented [5]. Figure 2.1b represents the influence of various drivers, such as climate, geology, and hydrological alterations on ecosystem components, such as changes in the diversity of various natural communities.

2 Site-Specific Characters

2.1 Biophysical Setting

For example, the Three Gorges Reservoir Area ranges 663 km from the west of Chongqing to the east of Yichang City (Hubei province) alongside the upper end of the Yangtze River (Changjiang) and lies about 44 km upstream from the Yichang gauging station [6]. The overall area is around 58,000 km², together with the 19 administrative units on both edges of the river. The average depth of the reservoir is about 70 m whereas the maximum depth in the anterior is around 170 m [7]. The overall storage capacity of the reservoir is 3.93×10^{10} m³ with an installed capacity of 18,200 MW and a rated power of 4990 MW, whereas the lowest and highest water discharges are 1580 and 98,800 m³/s, respectively. The reservoir executes services of flood regulation, power generation, irrigation, and course-plotting [8].

2.2 Climate

The region has a southeast subtropical monsoon with a mean yearly temperature of 16.5 °C and annual precipitation of almost 1100 mm [9]. During summer subtropical high pressure from the western pacific drifts over this region and forms a drought climate and increases the temperature. It also brings humid air from the sea in the south, which meets the airstream of the southwest (Guizhou and Hubei), which offers abundant humidity for precipitation. Rainfall frequently occurs between May



Fig. 2.1 General ecological characterization of reservoir systems. (a) Components of ecological characterization; (b) Influence of drivers and stressors on ecosystem components

and September in the form of the rainstorm and accounts for 60–80% of annual rainfall [10].

2.3 Geology

The topography of the TGR region is uneven and intricate. 74% of the landscape is composed of mountainous areas and 21.7% comprises hills [11], whereas the plains along the river valleys cover 4.3% of the remaining area in the reservoir region. The geology of the region entails a translucent basement composed of metamorphic and magmatic rocks and Jurassic sedimentary cover comprised of inter-embedded limestone, sandstone, and shale [11, 12]. Based on the age and rock type, the TGR area could be categorized into three geological sections. The initial one extends from Fengjie to Badong, interbedded with dolomite, carbonates, and shale; the next is Jurassic extending from Badong to Zigui, and the third is a granitic area that ranges from Zigui to Yichang [13].

Karst landform is widespread all over the TGR region as it is popular in a temperate and subtropical climate in southwest China [14]. Cave mines development is also swift and many caves are being quarried. Water from reservoir seeping into the regions's limestone caves, mines and underground rivers causes the caves to become unstable and eventually collapse. Reservoir water could discharge into underground rivers, mines, and limestone caves, which are distributed in the region due to caves become unstable and ultimately collapse [15].

3 Physical Transport Mechanisms

3.1 Hydrology

The hydrological conditions of water systems are disturbed by anthropogenic activities and climate. Since 2003, after the impoundment of the TGR, this area's ecosystem has been transformed, altering the water and sediment discharges downstream [16]. Dam operations can lead to substantial alterations in the water flow regime, particularly concerning monthly flow patterns and the magnitude of the flood [17, 18]. Dam operations usually disrupt the sequence of sediment pathways and modify the river hydrology, morphology, and ecology [19]. The TGD has strongly affected the downstream hydrological regime of the Yangtze River.

The first dam impoundment was commenced in 2003 with defiance of gravity to 139 m. Following another impoundment in 2006, which raised the water level to 156 m. The third elevation in the year raised the level to 172.8 m. Later in 2010, a remarkable water level of 175 m was experienced [20]. Human activities substantially influence the total runoff, which could result in rainfall erosion in agricultural land [21]. Moreover, according to the description by Mei et al. [22], the minimum

and maximum water levels downstream of the TGD decreased significantly in retort to erosion, despite the fact the yearly maximum water level was raised because of other anthropological activities in the pre-dam era. Modifications in hydrological processes may considerably distress spatial and time-based circulation, availability, and quality of water, resulting in reshaping the biota of the region [23].

3.2 Erosion

Soil permeability, resistance to erosion, thickness, anti-eroding ability, etc., have strong effects on the erosive process. In the TGR area, soils vary from yellow earth to yellow-brown earth and brown earth, whereas non-regionalized soil varies from purple, chalky, and paddy soil. Yellow, yellow-brown, and chalky soils create surface runoff because of their sticky nature and poor infiltration. The purple soil is the derivative of mudstone, which is thinner in texture and has good infiltration during a rainstorm. In riverine valleys, the dominant soil types are yellow, purple, and rendzina alluvial soil [11]. Because of the presence of these soil types, soil erosion may occur and cause the loss of nutrients [24]. During the mid-1990s, the problem of soil erosion comprised 82.9% of the overall reservoir area. Due to the clearing of vegetation for rebuilding road networks and urban development, a considerable quantity of soil organic material was lost, causing the degradation of soil [11].

The soil erosion emerges in the central portion of the TGR region, Chongqing Municipality, which accounted for 48.6% of the entire area [26]. According to previous studies [27], the risk of soil erosion will increase in the future. Moreover, sedimentation and erosion in the Yangtze River is a probable risk to the protection of the TGR [27]. The soil erosion in the TGR area has been promoted since 2000, due to the resettlement of urban and rural people near the reservoir region. Occurrence of geological hazards prompted by increasing the storage of the reservoir [28]. These factors might also aggravate the problem of soil erosion, whereas the existing ecological projects could help to reduce the extent and intensity of soil erosion in the reservoir region [29]. Other factors impelling soil erosion encompass urban expansion and reconstruction [11]. According to a previous report [26], the discrepancy in soil erosion in 2001 and 2006 illustrates the possible correlation between soil erosion and the precipitation intensity in the TGRA.

4 Natural Communities

Reservoirs are dynamic bodies with vertical zonation on the horizontal axis. Along the reservoir region, spatial heterogeneity of various regions and transference of nutrients to a downstream region takes place. The concentrations of various nutrients, trophic states, and operational processes in the upstream area affect metabolic

Community	Dominant species	Reference
Fish	Aristichthys nobilis, Coreius heterodon, Coreius guichenoti, Hypophthalmichthys molitrix, Pelteobagrus vachelli, Rhinogobio cylindricus, Hemiculter bleekeri, Cyprinus carpio in abundance. Protosalanx hyalocranius, Ictalurus punctatus, Megalobrama amblycephala	[7]
Microbiota	Alpha-, Beta-, Gamma Proteobacteria, Verrucomicrobia, and Planctomycetes, Spirochaetes, Nitrospirae, Chloroflexi, and Acidobacteria	[31]
	Firmicutes, Proteobacteria, and Actinobacteria	[32]
	Chironomidae, Heptageniidae, and Baetidae	[33]
	Diatoma vulgare Melosira varians, Cocconeis placentula, Gyrosigma scalproides, and Oscillatoria tenuis, M. varians, Cymbella affinis, D. vulgare, Eucapsis alpina, and M. granulata, M. varians, C. affinis, and C. placentula,	[34]
Macroinvertebrates	Nais–Polypedilum, Limnodrilus	[35]
	Branchiura sowerbyi, Bothrioneurum vejdovskyanum, Nematoda spp, Polypedilum scalaenum, Limnodrilus hoffmeisteri, Stictochironomus sp., Teneridrilus mastix, Nais variabilis, Paranais frici, Procladius sp., and Polypedilum scalaenum	[36]
Plants	Pinus massoniana Lamb., Cupressus funebris Endl.	[37]
	Bidens tripartita, Cynodon dactylon, Cyperus rotundus, Digitaria sp., Echinochloa crus-galli, Setaria viridis, Polygo- num lapathifolium, and Xanthium sibiricum	[38]

Table 2.1 Summary of various groups of natural communities in the TGR region

progressions of a downstream reservoir and modify the overall water quality [4]. The altered watercourses, the segregating effect of the dam, and the transformed sediment composition have modified the habitations and associated vegetation, vertebrates, and invertebrate species in the riparian and reservoir ecosystems. Besides, variation in water level due to changes in hydrological patterns has resulted in the loss of land-dwelling and riparian flora fauna [30]. A summary of different natural communities found in the TGR region is shown in Table 2.1.

4.1 Fish Communities

Reservoirs being important artificial ecosystems alter the ecological and hydrological physiognomies of a river [39]. The ecological and hydrological regimes of the river are modified by artificial ecosystems like dams due to changes in physical, biological, and chemical variables [40]. For physical gradients, reservoir region is distributed into riverine, transitional, and lacustrine zones in the longitudinal direction [41]. In the riverine zone, surface and deep-water sediments are well mixed and the environment is lotic. The ecotone between the upstream and the lacustrine zone is called a transitional zone, whereas the lacustrine zone is the stratified lake-like area [40]. The flow rate decreases from upstream to downstream in the TGRA, due to different niches for organisms were created and gradients of different nutrients were found [42]. The zonation configuration is very useful to study the time- and space-based pattern of fish accumulations. To manage fish diversity, it is important to understand the assemblage of fish along the gradient [7] Information related to the characteristics of fish congregations in dams is also very vital for the development of operative conservation policies [43].

In a study by Lin et al. [7], fish assemblage patterns in river dam gradient were assessed in the TGR region. The reservoir area was defined as the riverine zone, transitional zone, and lacustrine zone. Three fish zones were reported in the TGR. The riverine zones were subjugated by rheophilic species, such as *Coreius guichenoti* and *Pelteobagrus vachelli*. The transitional zones were found the fish species, such as *Coreius heterodon* and *Rhinogobio cylindricus*. The lacustrine zone was dominated by eurytopic species including *Aristichthys nobilis*, *Cyprinus carpio*, *Hypophthalmichthys molitrix*, and *Hemiculter bleekeri*. Moreover, the lacustrine zone also contained 18 alien species like *Protosalanx hyalocranius*, *Ictalurus punctatus*, *Megalobrama amblycephala*, and *Tilapia*.

4.2 Microbial Diversity

The microbial populations are mostly dominated by the homogeneous selection, ecological drift, and limitation of dispersal in the sediments of deep-water reservoirs [42]. In the previous studies [44, 45], Proteobacteria dominated in the bacterial communities; whereas among Protistan communities, Ochrophyta, Fungi, *Ciliophora*, and *Chlorophyta* were dominant species in major hotspot areas of the TGR region [46]. In a similar study, it was observed that in bacterial communities, Proteobacteria, such as Alphaproteobacteria and Betaproteobacteria, were the most diverse among all samples, followed by Bacteroidetes (Bacteroidia and Flavobacteriia). Chloroflexi (Anaerolineae), and other Phvla including Planctomycetes, Acidobacteria, Spirochaetes, Verrucomicrobia, Nitrospirae, Actinobacteria, Firmicutes, Chlorobi, and Euryarchaeota. In a previous study [46] utilizing the metabarcoding approach, 6217 non-singleton protistan and 26 protistan phyla were identified, and Ochrophyta (Bacillariophyta), Fungi (Ascomycota), Ciliophora (Litostomatea and Spirotrichea), Chlorophyta (Chlorophyceae) were dominant, compared to others.

Among algae communities, almost 103 species from 45 genera and 4 families were identified in River Daxi, Caotang, and Meixi in the Fengjie district of Chongqing, in which *Diatoma vulgare*, *Melosira varians*, *Cocconeis placentula*, *Gyrosigma scalproides*, *Oscillatoria tenuis*, *Cymbella affinis*, *D. vulgare*, *Eucapsis alpina*, and *M. granulata* were common and their composition in the region was influenced by environmental factors, like pH, temperature, total nitrogen, and total phosphorus [34]. Another study investigated the taxonomic groups of bacteria in the TGR region and reported that *Firmicutes* instead of *Proteobacteria* and *Actinobacteria* are the predominant group at the phyla level. However, *Proteobacteria* and *Actinobacteria* can contribute to more than 50% of total bacteria in surface water [32], which are indicators of freshwater bacteria. The high levels of *Firmicutes* are usually detected in wastewater [47] where it is involved in solid waste biodegradation [47, 48], implying that surface water of the TGR might be contaminated with it.

In freshwater bacterial clusters like alpha-, beta-, gamma-, and deltaproteobacteria, Actinobacteria, Cyanobacteria, Cytophaga-Flavobacterium-Bacteroides (CFB), Firmicutes, Planctomycetes, and Verrucomicrobia are usually present. Alpha-. Beta-. Gammaproteobacteria, Verrucomicrobia, and Planctomycetes were found in the TGR according to the previous study [31]. Besides, Spirochaetes, Nitrospirae, Chloroflexi, and Acidobacteria were also detected.

4.3 Macroinvertebrate Community Composition

Macroinvertebrates are ubiquitous and diverse and are utilized as biological indicators in ecological assessments. Their populations are affected by anthropological stress and natural factors like temperature, light penetration, water chemistry, food resources, and habitat diversity [33]. Time-based variations in macroinvertebrate composition could be associated with life-history patterns in the community, in response to food availability and sporadic alterations in physicochemical properties. Seasonal variation leads to food abundance and impacts the life cycles of the aquatic community [49].

In a previous study [33], 27 main tributaries of the TGRC were evaluated for species richness during the four seasons. 87 taxa were recorded and the overall richness of the taxa varied in different seasons (61 families in spring, 40 families in summer, 47 families in autumn, and 52 families in winter). The most abundant taxa in all seasons were *Chironomidae*, *Heptageniidae*, and *Baetidae*.

Previous data acquired before the damming revealed that the benthic community has drastically altered in the reservoir. Xie reported that before the construction of the reservoir [50], mayflies and caddisflies were common macroinvertebrates among benthic fauna. However, the occasional presence of caddisflies and no occurrence of mayflies have been detected. Later, *Oligochaetes* and *Chironomids* with morphological adaptations dominated the taxon composition and are still presently dominating [51, 52]. *Nais–Polypedilum* community type was more common, which occurred in the low-water discharge inflow during winter and spring after the second impounding year. *Limnodrilus* community occurred in the high discharge inflow during autumn and summer [35]. Many studies showed that the community of macroinvertebrates in the TGR is regulated by the subtropical monsoon climate with seasonal cycles of destruction followed by reestablishment [36].

In a 10-year survey study [36], 49 taxa were collected, which are classified into three groups, *Chironomidae*, *Tubificidae*, and *Naididae*. During the three impoundment stages, 10 taxa were found, which include *Polypedilum scalaenum*, *Branchiura sowerbyi*, *Limnodrilus hoffmeisteri*, *Stictochironomus* sp., *Teneridrilus mastix*, *Nais variabilis*, *Paranais frici*, *Procladius* sp., *Bothrioneurum vejdovskyanum*, and *Nematoda* spp. *Polypedilum scalaenum* was the most frequent taxon, occurring in 70% of all 40 surveys. Other frequent taxa included *B. sowerbyi*, *Procladius* sp., *N. inflata*, *Nematoda* spp., *T. mastix*, *Stictochironomus* sp., and *L. hoffmeisteri*.

4.4 Plant Communities and Distribution Patterns

The TGR region is known to be one of the 25 biodiversity hotspots around the globe [53] and one of the three richest biodiversity centers in China [54]. Factors, such as high temperature in summer and extensive floods in winter, alter the water level and affect aquatic and terrestrial species of the TGR. Because of the distinctive geography and landscape, some of the species present in the TGR region have survived the late Tertiary and Quaternary periods, and this area is well known for the presence of a high level of rare, endemic, and ancient species [30]. Previously, the region was dominated by original subtropical vegetation *Castanopsis* spp. and *Phoebe* spp. [55], but it has been substituted by secondary forests composed of *Pinus massoniana Lamb*. and *Cupressus funebris Endl.*), shrublands, grasslands, and croplands [37]. Various studies have been carried out to assess the impact of the TGD on local biodiversity [31, 37, 56–58]. A total of 22 plant communities (four woody, nine shrubs, and nine kinds of grasses) were vulnerable to the submergence [37].

Before the closing of the TGR, the vascular plants of the downward areas were studied thoroughly by different research scholars. Wang et al. [59] collected 377 vascular plants in the downstream area of the TGR, whereas in another study [58], 392 different plant species were reported around the reservoir region. These studies suggest that the TGR impoundment has reduced the floral diversity of the downstream region. *Bidens tripartita, Cynodon dactylon, Cyperus rotundus, Digitaria* sp., *Echinochloa crus-galli, Setaria viridis, Polygonum lapathifolium*, and *Xanthium sibiricum* were most widely distributed in the whole TGR [38].

A recent study using quantitative methods found 150 vascular plant species belonging to 130 genera of 56 families in the TGR region [60]. The majority of species were annual herbs, which confirms that the region is ecologically distressed and is at an initial phase of succession. The medium and steep slopes of the riparian zone have comparatively fewer species, whereas the upper zone had more diverse species and community types. At the site scale, the major influence that affects the structuring of communities was the landscape. While at community scale habitat, soil, nitrogen, organic matter, and flooding have severely impacted the structure.

5 Conclusion

The overall aim of this chapter is to understand the characterization of the reservoir ecosystem and to identify the core drivers and stressors that affect the composition and distribution of the various natural communities in the region. Previous studies have revealed that species composition is affected by various stressors, and drivers that are part of ecology and disturbance in ecological entities affects the species richness and type. Over the years, the ecology of the reservoir region has changed gradually because of anthropogenic activities and natural disturbances. The reservoir has also affected the downstream ecology of the Yangtze River and has also reduced the number of some important species especially fish biodiversity. These results of various studies highlight the need of conserving native fish in the TGR region.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of Interest The authors declare no conflict of interest.

References

- 1. M. Kelkar, Application of geostatistics for reservoir characterization-accomplishments and challenges. J. Can. Pet. Technol. **39** (2000)
- 2. J.R. Fanchi, Shared Earth Modeling: Methodologies for Integrated Reservoir Simulations (Gulf Professional Publishing, 2002)
- R.H. Kennedy, K.W. Thornton, D.E. Ford, Characterization of the reservoir ecosystem, in Microbial Processes in Reservoirs, (Springer, 1985), pp. 27–38
- J.G. Tundisi, Reservoirs: New challenges for ecosystem studies and environmental management. Water Security 4, 1–7 (2018)
- J. Tundisi, T. Matsumura-Tundisi, J. Tundisi, Reservoirs and human well being: New challenges for evaluating impacts and benefits in the neotropics. Braz. J. Biol. 68, 1133–1135 (2008)
- B.-J. Fu, B.-F. Wu, Y.-H. Lü, Z.-H. Xu, J.-H. Cao, D. Niu, G.-S. Yang, Y.-M. Zhou, Three Gorges Project: Efforts and challenges for the environment. Prog. Phys. Geogr. 34, 741–754 (2010)
- P. Lin, X. Gao, F. Liu, M. Li, H. Liu, Long-term monitoring revealed fish assemblage zonation in the Three Gorges Reservoir. J. Oceanol. Limnol. 37, 1258–1267 (2019)
- W. Guo, H. Wang, J. Xu, Z. Xia, Ecological operation for Three Gorges Reservoir. Water Sci. Eng. 4, 143–156 (2011)
- C. Ye, S. Li, Y. Yang, X. Shu, J. Zhang, Q. Zhang, Advancing analysis of spatio-temporal variations of soil nutrients in the water level fluctuation zone of China's Three Gorges Reservoir using self-organizing map. PLoS One 10 (2015)
- H. Binghui, Soil erosion regularity in Three Gorges Reservoir Area and its controlling measures, in *12th ISCO Conference Department of Soil and Water Conservation*, (Southwest Agriculture University, Beibei, 2002), pp. 252–258

- 11. X. Xu, Y. Tan, G. Yang, H. Li, W. Su, Soil erosion in the Three Gorges Reservoir area. Soil Res. **49**, 212–222 (2011)
- S. Wu, L. Shi, R. Wang, C. Tan, D. Hu, Y. Mei, R. Xu, Zonation of the landslide hazards in the forereservoir region of the Three Gorges Project on the Yangtze River. Eng. Geol. 59, 51–58 (2001)
- W. Jietao, Geomorphological evolution of the Hengshixi anticline of the Three Gorges Area through isobases: A model of Yangtze River capture. Int. J. Simul. Syst. Sci. Technol. 17 (2016)
- K. Li, C. Zhu, L. Wu, L. Huang, Problems caused by the Three Gorges Dam construction in the Yangtze River basin: A review. Environ. Rev. 21, 127–135 (2013)
- J. Wang, The Three Gorges Area and the linking of the upper and middle reaches of the Yangtze River. J. Geogr. Inf. Syst. 10, 301–322 (2018)
- X. Cai, L. Feng, Y. Wang, X. Chen, Influence of the Three Gorges Project on the water resource components of Poyang Lake watershed: Observations from TRMM and GRACE. Adv. Meteorol. 2015 (2015)
- R.J. Batalla, C.M. Gomez, G.M. Kondolf, Reservoir-induced hydrological changes in the Ebro River basin (NE Spain). J. Hydrol. 290, 117–136 (2004)
- F.J. Magilligan, K.H. Nislow, Changes in hydrologic regime by dams. Geomorphology 71, 61–78 (2005)
- 19. R.J. Batalla, Sediment deficit in rivers caused by dams and instream gravel mining. Are view with examples from NE Spain. Cuaternario y Geomorfología **17**(2), 79–91 (2003)
- X. Zhang, Z. Dong, H. Gupta, G. Wu, D. Li, Impact of the Three Gorges Dam on the hydrology and ecology of the Yangtze River. Water 8, 590 (2016)
- X. Chen, G. Xu, W. Zhang, H. Peng, H. Xia, X. Zhang, Q. Ke, J. Wan, Spatial variation pattern analysis of hydrologic processes and water quality in Three Gorges Reservoir Area. Water 11, 2608 (2019)
- X. Mei, Z. Dai, P. Van Gelder, J. Gao, Linking Three Gorges Dam and downstream hydrological regimes along the Yangtze River, China. Earth Space Sci. 2, 94–106 (2015)
- N.C. Kleinstreuer, K. Sullivan, D. Allen, S. Edwards, D.L. Mendrick, M. Embry, J. Matheson, J.C. Rowlands, S. Munn, E. Maull, Adverse outcome pathways: From research to regulation scientific workshop report. Regul. Toxicol. Pharmacol. **76**, 39–50 (2016)
- 24. Y. Li, C. Liu, C. Zhao, J. Huang, Spatiotemporal features of soil and water loss in the Three Gorges Reservoir Area of Chongqing. Acta Geograph. Sin. 63, 502–513 (2008)
- 25. R. Du, M. Shi, J. Yuan, Impacts of Soil Losses on the Ecology and Environment in Three Gorges Reservoir Area, Yangtze River (Science Press, Beijing. in Chinese, 1994)
- 26. Y. Li, C. Liu, X. Yuan, Spatiotemporal features of soil and water loss in Three Gorges Reservoir Area of Chongqing. J. Geogr. Sci. 19, 81–94 (2009)
- X.-Z. Yuan, Y.-W. Zhang, H. Liu, S. Xiong, B. Li, W. Deng, The littoral zone in the Three Gorges Reservoir, China: Challenges and opportunities. Environ. Sci. Pollut. Res. 20, 7092–7102 (2013)
- X. Xu, Y. Tan, G. Yang, Environmental impact assessments of the Three Gorges Project in China: Issues and interventions. Earth Sci. Rev. 124, 115–125 (2013)
- G. Zhao, X. Mu, Z. Wen, F. Wang, P. Gao, Soil erosion, conservation, and eco-environment changes in the Loess Plateau of China. Land Degrad. Dev. 24, 499–510 (2013)
- 30. T. New, Z. Xie, Impacts of large dams on riparian vegetation: Applying global experience to the case of China's Three Gorges Dam. Biodivers. Conserv. **17**, 3149–3163 (2008)
- S. Wang, R.M. Dong, C.Z. Dong, L. Huang, H. Jiang, Y. Wei, L. Feng, D. Liu, G. Yang, C. Zhang, Diversity of microbial plankton across the Three Gorges Dam of the Yangtze River, China. Geosci. Front. 3, 335–349 (2012)
- 32. Z. Yu, J. Yang, S. Amalfitano, X. Yu, L. Liu, Effects of water stratification and mixing on microbial community structure in a subtropical deep reservoir. Sci. Rep. 4, 1–7 (2014)
- 33. S. Chi, S. Li, S. Chen, M. Chen, J. Zheng, J. Hu, Temporal variations in macroinvertebrate communities from the tributaries in the Three Gorges Reservoir Catchment, China. Rev. Chil. Hist. Nat. 90 (2017)

- 34. J. Fu, L. Liu, X. He, H. Zhang, C. Dong, Y. Yang, J. Shi, Z. Wu, Dynamics of epilithic algae communities and their relationship with environmental factors during storage and non-storage periods in the Three Gorges Reservoir. Huan Jing Ke Xue 40, 3099 (2019)
- M.L. Shao, Z.C. Xie, X.Q. Han, M. Cao, Q.H. Cai, Macroinvertebrate community structure in Three-Gorges Reservoir, China. Int. Rev. Hydrobiol. 93, 175–187 (2008)
- B. Li, Q.-h. Cai, M. Zhang, M.-I. Shao, Macroinvertebrate community succession in the Three-Gorges Reservoir ten years after impoundment, Quat. Int., 380 (2015) 247–255
- Z. Tian, W. Chen, C. Zhao, Y. Chen, B. Zheng, Plant biodiversity and its conservation strategy in the in-undation and resettlement districts of the Yangtze Three Gorges, China. Acta Ecol. Sin. 27, 3110–3118 (2007)
- 38. Q. Wang, X. Yuan, H. Liu, Y. Zhang, Z. Cheng, B. Li, Effect of long-term winter flooding on the vascular flora in the drawdown area of the Three Georges Reservoir, China. Pol. J. Ecol. 60, 95–106 (2012)
- 39. P.F. Verdonschot, *Ecological Characterization of Surface waters in the Province of Overijssel* (Verdonschot, Amsterdam, 1990)
- 40. E.D. Oliveira, E. Goulart, C. Minte-Vera, Fish diversity along spatial gradients in the Itaipu Reservoir, Paraná, Brazil. Braz. J. Biol. **64**, 447–458 (2004)
- Y. Xu, Q. Cai, M. Shao, X. Han, Patterns of asynchrony for phytoplankton fluctuations from reservoir mainstream to a tributary bay in a giant dendritic reservoir (Three Gorges Reservoir, China). Aquat. Sci. 74, 287–300 (2012)
- 42. Y. Li, Y. Gao, W. Zhang, C. Wang, P. Wang, L. Niu, H. Wu, Homogeneous selection dominates the microbial community assembly in the sediment of the Three Gorges Reservoir. Sci. Total Environ. 690, 50–60 (2019)
- A.A. Agostinho, L.C. Gomes, N.C. Santos, J.C. Ortega, F.M. Pelicice, Fish assemblages in neotropical reservoirs: Colonization patterns, impacts and management. Fish. Res. 173, 26–36 (2016)
- 44. Y. Xie, J. Wang, Y. Wu, C. Ren, C. Song, J. Yang, H. Yu, J.P. Giesy, X. Zhang, Using in situ bacterial communities to monitor contaminants in river sediments. Environ. Pollut. 212, 348–357 (2016)
- 45. G. Saxena, E.M. Marzinelli, N.N. Naing, Z. He, Y. Liang, L. Tom, S. Mitra, H. Ping, U.M. Joshi, S. Reuben, Ecogenomics reveals metals and land-use pressures on microbial communities in the waterways of a megacity. Environ. Sci. Technol. 49, 1462–1471 (2015)
- 46. Y. Xie, T. Floehr, X. Zhang, H. Xiao, J. Yang, P. Xia, G.A. Burton Jr., H. Hollert, In situ microbiota distinguished primary anthropogenic stressor in freshwater sediments. Environ. Pollut. 239, 189–197 (2018)
- 47. A. Hu, X. Yang, N. Chen, L. Hou, Y. Ma, C.-P. Yu, Response of bacterial communities to environmental changes in a mesoscale subtropical watershed, Southeast China. Sci. Total Environ. **472**, 746–756 (2014)
- M.A.M. Mondol, H.J. Shin, M.T. Islam, Diversity of secondary metabolites from marine bacillus species: Chemistry and biological activity. Mar. Drugs 11, 2846–2872 (2013)
- C. Clenaghan, P. Giller, J. O'halloran, R. Hernan, Stream macroinvertebrate communities in a conifer-afforested catchment in Ireland: Relationships to physico-chemical and biotic factors. Freshwater Biol. 40, 175–193 (1998)
- 50. C. Xie, Prediction of effects of the Three-Gorge Project on macroinvertebrates in the reservoir area, in *Collection of Papers on Impact and Countermeasure of the Three-Gorge Project on Ecology and Environment*, ed. by The Project Leading Group of Ecology and Environment of the Three-Gorge Project, Chinese Academy of Sciences, (Science Press, Beijing. in Chinese, 1987)
- M. Callisto, M. Goulart, F. Barbosa, O. Rocha, Biodiversity assessment of benthic macroinvertebrates along a reservoir cascade in the lower São Francisco river (northeastern Brazil). Braz. J. Biol. 65, 229–240 (2005)

- 52. M. Shao, Z. Xie, L. Ye, Q. Cai, Monthly change of community structure of zoobenthos in Xiangxi Bay after impoundment of the Three Gorges Reservoir, China. Front. Biol. China 2, 229–234 (2007)
- J. Lopez-Pujol, M.-X. Ren, Biodiversity and the Three Gorges Reservoir: A troubled marriage. J. Nat. Hist. 43, 2765–2786 (2009)
- 54. J. López-Pujol, H.-F. Wang, Z.-Y. Zhang, Conservation of Chinese Plant Diversity: An Overview, Research in Biodiversity–Models and Applications (Rijeka, InTech, 2011), pp. 163–202
- 55. Z.-Y. Wu, Vegetation of China (中国植被)., agris.fao.org, (1980)
- 56. J. Wu, J. Huang, X. Han, Z. Xie, X. Gao, *Three-Gorges Dam—Experiment in Habitat Fragmentation?* (American Association for the Advancement of Science, 2003)
- Z. Xie, Characteristics and conservation priority of threatened plants in the Yangtze valley. Biodivers. Conserv. 12, 65–72 (2003)
- Z. Zhong, D. Qi, *The Illustrated Species Catalog and Biodiversity in the Hydro-Fluctuation Belt* of *Three Gorges Reservoir*, vol 10 (Southwest China Normal University Press, Chongqing, 2008)
- 59. Y. Wang, E. Li, J. Wu, A preliminary study on the vascular plant flora of the water-levelfluctuating zone in the Three-Gorge Reservoir Area. Wuhan Botanical Res. **20**, 265–274 (2002)
- 60. C. Chen, M. Ma, S. Wu, J. Jia, Y. Wang, Complex effects of landscape, habitat and reservoir operation on riparian vegetation across multiple scales in a human-dominated landscape. Ecol. Indic. 94, 482–490 (2018)

Chapter 3 Main Sources and Distribution of Pollutants in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract This chapter particularly aims to identify the sources and the pollutants distribution trend observed in the Three Gorges Reservoir Area (TGRA). It was found that, among various classes of pollutants, heavy metals (HM) were the most widely reported from 2008 to 2016. The general trend showed that the HM levels were relatively high in this area. Moreover, phthalate esters (PAEs) including DEHP showed higher risks to the aquatic species with concentrations double the prescribed limit by WHO and US EPA. Polycyclic aromatic hydrocarbons (PAHs) enter the TGRA from the pyrogenic and petrogenic combustion sources. Besides, the PPCPs sources could be attributed to agricultural activities, hospital waste, and industrial waste. Higher microplastic pollution loads were particularly found in the upper stream, which also increases the ecological risks. In summary, detailed temporal studies are imperative to foresee the clear pollution trend. Also, regulative authorities should properly implement the rules and restrict the use of hazardous chemicals that enter the TGRA.

Keywords Pollutants · Three Gorges Reservoir Area · Sources · Distribution

1 Introduction

Over the past decades, increased urbanization, industrialization, and exponential population growth have led to the increasing exploitation of natural resources [1]. The continuous release of pollutants into the environment had posed detrimental effects on the environment, particularly on the reservoir ecosystem [2, 3]. Numerous pollutants enter the reservoir through natural processes (runoff, flooding, and transport) and a variety of anthropogenic interventions (industrial activities, agriculture activities, and tourism) [4]. The prominent pollutants that are released in the reservoirs include different classes of organic and inorganic pollutants possessing persistent, toxic, and bioaccumulative properties. Different classes of pollutants are frequently found in the reservoir ecosystem (Fig. 3.1) [5].

Globally, the reservoir ecosystem is manmade that is used for irrigation, water supply, industrial, domestic, flood mitigation, and hydropower generation purpose. Through direct and indirect sources, chemical contaminants pollute the reservoir

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_3



Fig. 3.1 General classification of the pollutants found in the reservoir ecosystem

water and cause human health risks. Previously, a 50-year sedimentary record of heavy metal pollution was reported in the Lot rivers, France [6]. From the geoaccumulation index (Igeo), it was found that Lot river sediments were considered the most polluted reservoirs, particularly with cadmium (Cd) and zinc (Zn) [6]. Similarly, higher concentrations of eight heavy metals including iron (Fe), Zn, chromium (Cr), manganese (Mn), cobalt (Co), lead (Pb), cadmium (Cd), and nickel (Ni) were detected in the Al-Najaf sea depression reservoir, Iraq [7]. The majority of the published studies were focused on heavy metal pollution; therefore, more studies are needed to elucidate more classes of pollutants trends in reservoirs.

2 Sources of Pollutants

Pollution sources can be classified as point and non-point sources. Point source refers to any single identifiable point where pollutants enter the water body. Whereas, non-point source involves various contaminants enter in the water from different sources. Generally, non-point sources are easy to identify but difficult to manage [8, 9]. Previously, various studies have demonstrated the sources of water pollution. Every class of pollutants possesses different sources. For example, PAHs

mainly originated from pyrogenic or petrogenic combustion [10, 11]. Phthalates (PAEs) ubiquitous presence in the environment is from industrial or wastewater treatment plant waste [12, 13]. PCBs sources are attributed mainly to agricultural waste [14]. Similarly, heavy metals came from industrial activities and wastewater treatment plants [15]. The major sources of PPCPs are hospital waste, industrial waste, and domestic waste [16].

3 Source Apportionment Techniques

Numerous source apportionment techniques have been used to identify the possible sources and their quantitative contribution to the Three Gorges Reservoir Area (TGRA). The most widely used techniques include the principal component analysis (PCA), molecular diagnostic ratios (MDRs), Monte Carlo source apportionment, regression analysis, geoaccumulation index, enrichment index, and isotopes analysis [2, 17]. Nevertheless, each technique has its own advantages and disadvantages, implying that the use of multiple methods may reduce bias instead of an individual method.

4 Pollution in the TGRA

In 1994, the TGRA was implemented to promote sustainable water resources in China. The TGRA is also recognized as China's Golden Waterway, because it plays a key role in the national economy [18]. It is the longest river in Asia, and flows throughout China [19]. In 2009, the TGRA project was completed and comprises the upper, middle, and lower reaches. However, with the construction and its functional use, the untreated wastewater discharge imposed threats to ecological and public health effects [3, 18]. Similarly, various agricultural and industrial pollutants were reported to be discharged into the river and are responsible for ecological and health risks to freshwater ecosystems [15, 20, 21].

Generally, in the TGRA, pollutants enter mainly from the industrial wastewater, domestic wastewater, and agricultural discharges, which include different organic and inorganic pollutants, such as heavy metals (HM), polycyclic aromatic hydrocarbons (PAHs), polychlorinated biphenyls, phthalates (PCBs), microplastics (MPs), microbial and algal pollutants, organic matter (OM), pharmaceutical and personal care products (PPCPs) [3, 20, 22, 23]. Each pollutant category comprises different points and non-point sources. According to the previous studies from 2003 to 2014, the TGRA holds many pollutants, of which 37.2% and 62.7% accounted for industrial and domestic wastewater, respectively [10, 24]. Moreover, this could be attributed to rapid industrialization, urbanization, and an increase in population. Besides, chemical oxygen demand (COD) is considered the major indicator of water quality and was reported to increase from 2003 to 2014, suggesting 68.2% of the domestic wastewater sources [14, 25, 26].

5 Spatial Distribution of Pollutants in the TGRA

5.1 Heavy Metals

In the TGRA, heavy metals are the most widely studied pollutants. Previously a study quantified Hg, Cr, As, Cu, Cd, Fe, Mn, and Zn in soil and sediments samples from 2008 to 2009 and showed that Cd, As, Cu, Pb, and Zn were higher in the upper and lower reaches of the TRGA. Source apportionment techniques, such as factor analysis (FA) and multiple linear regression (MLR) suggested that As and Cd are the main chemical pollutants with 45% contribution from domestic sewage and 59% from industrial waste [27] (Fig. 3.2). Gao et al. evaluated the HMs status in surface water for the years 2008–2013 in the TGRA [17]. It was found that the highest metal levels were observed in the upper reaches with the concentrations of Cu (10.3 μ g/L), Hg (0.05 μ g/L), Cd (1.47 μ g/L), As (3.06 μ g/L), and Pb (15.02 μ g/L). Similarly, in 2015–2016, Cu, Zn, Cd, Pb, and Cr were found in the surface water and sediments, and the PCA results implied that the HM levels were lesser in the drinking water and reached the surface water quality standards of China. The HMs were found to be non-polluted in 2015–2016. Cr levels in the sediments were also non-polluted, except for Cu, Zn, Pb, and Cd, which were slightly polluted in the TGRA [3].

Bioaccumulation and ecological risk of main HMs (Cu, Fe, Hg, Zn, Cd, and Pb) were observed in aquatic invertebrates and fish from the TGRA [25]. It was found that, in the impoundment years from 2003 to 2010, all the HM levels were under the criteria of safety guidelines [25]. Moreover, Cu, Zn, Hg, and Fe levels in fish and aquatic invertebrates were relatively higher, compared to the levels before impoundment. Also, the pollution trends of HMs were the same in the upper, middle, and lower reaches [25] (Table 3.1).



Fig. 3.2 A general illustration of the sources of pollutants in the TGRA

Chemical	Sampling			D.C
class	time	Matrix	Concentrations/levels	Reference
Heavy metals	2008	Surface water	Cu: 10.3 µg/L, Hg: 0.05 µg/L, Cd: 1.47 µg/L, As: 3.06 µg/L, Pb: 15.02 µg/L	[17]
Heavy metals	2013	Surface water	Cu: 3.01 μg/L, Hg: 0.016 μg/L, Cd: 0.771 μg/L, As: 1.53 μg/L, Zn: 10.43 μg/ L, Pb: 7.89 μg/L	[17]
Heavy metals	2008–2013	Surface water	Cu: 8.94 µg/L, Hg: 0.03 µg/L, Cd: 1.02 µg/L, As: 2.33 µg/L, Zn: 13.03 µg/ L, Pb: 11.20 µg/L	[17]
Heavy metals	July 2015	Sediments	As: 0.05–50.90 μg/L, Mo: 0.30–1.63 μg/ L, W: 0.01–0.42 μg/L	[28]
Heavy metals	2011–2012	Fish	Hg: 17.8 ng/g, Cd: 18.8 ng/g, Pb: 27.3 ng/g, Cu: 515.0 ng/g, Fe: 6969.7 ng/g, Zn: 6163 ng/g	[25]
Heavy metals	December 2015	Surface water	Cu: 1.21 µg/L, Zn: 12.98 µg/L, Pb: 0.04 µg/L, Cr: 0.47 µg/L, Cd: 0.02 µg/L	[3]
Heavy metals	December 2015	Sediments	Cu: 58.9 mg/kg, Zn: 165.9 mg/kg, Pb: 56.7 mg/kg, Cr: 96.5 mg/kg, Cd: 1.14 mg/kg	[3]
Heavy metals	2014	Sediments	Cd: 0.99 mg/kg, Cr: 94.2 mg/kg, Cu: 69.0 mg/kg, Ni: 40.8 mg/kg, Pb: 56.7 mg/kg, Zn: 161.0 mg/kg	[29]
Heavy metals	2014	Sediments	Cd: 1.18 mg/kg, Cr: 115.8 mg/kg, Cu: 79.14 mg/kg, Ni: 48.70 mg/kg, Pb: 71.17 mg/kg, Zn: 167.47 mg/kg,	[30]
Heavy metals	2016	Sediments	Cd: 1.01 mg/kg, Cr: 86.4 mg/kg, Cu: 49.5 mg/kg, Ni: 38.6 mg/kg, Pb: 54.5 mg/kg, Zn: 185.1 mg/kg	[29]
Heavy metals	2008	Soil	Cd: 0.77 mg/kg, Cr: 44.2 mg/kg, Cu: 32.0 mg/kg	[27]
Heavy metals	September 2008	Soil	Hg: 1.73 mg/kg, As: 3.98 mg/kg, Cr: 0.56 mg/kg, Cd: 3.18 mg/kg, Pb: 1.58 mg/kg, Cu: 1.29 mg/kg, Zn: 1.17 mg/kg, Mn: 1.23 mg/kg	[23]
Heavy metals	June 2009	Soil	Hg: 3.20 mg/kg, As: 1.30 mg/kg, Cr: 0.67 mg/kg, Cd: 4.27 mg/kg, Pb: 2.11 mg/kg, Cu: 1.64 mg/kg, Zn: 1.49 mg/kg, Mn: 1.26 mg/kg	[23]
РАН	2015	Surface water	ΣPAHs mean: 23–1630 ng/L	[10]
PAEs	2015	Surface water	DEHP: upper reaches 6.21 µg/L, middle reaches 1.48 µg/L, lower reaches 0.38 µg/L	[22]
Tetracycline	August 2015	Surface water	OTC: 10.62 ng/L, TC: 22.91 ng/L, CTC: 30.06 ng/L, TCs: 6.6 ng/L	[31]
Sulfonamide	August 2015	Surface water		[31]

Table 3.1 Summary of pollutants levels in the TGRA environment and biological matrices

(continued)

Chemical	Sampling			
class	time	Matrix	Concentrations/levels	Reference
			SDZ: 95.39 ng/L, TMP: 119.0 ng/ L, STZ: 30.12 ng/L, ST: 14.37 ng/ L, SMZ: 78.55 ng/L, SAs: 337.5 ng/L	
Quinolones	August 2015	Surface water	OFL: 31.12 ng/L, ENR: 19.32 ng/ L, FQs: 50.4 ng/L	[31]
Tetracycline	August 2015	Sediments	OTC: 4.16 ng/g, TC: 31.76 ng/ g, CTC:14.68 ng/g, TCs: 51.4 ng/g	[31]
Sulfonamide	August 2015	Sediments	SDZ: 7.02 ng/g, TMP: 11.71 ng/g, STZ: 3.01 ng/g, ST: 35.93 ng/g, SMZ: 24.91 ng/g, SAs: 82.6 ng/g	[31]
Sulfonamide	2011	Surface water	SMX: 7.9 ng/L, SDZ: 1.45 ng/L	[32]
Sulfonamide	2015	Surface water	SMX: 289.3 ng/L, SMR: 6.57 ng/ L, SPY: 76.57 ng/L, ST: 6.68 ng/ L, SDZ: 1.93 ng/L	[20]
Quinolones	August 2015	Sediments	OFL: 22.46 ng/g, ENR: 42.59 ng/g, FQs: 68.1 ng/g	[31]
Microplastics	July 2017	Sediments	$ \begin{array}{c} 1-5 \text{ mm particles } (33.24-10^3 \text{ particles/} \\ m^{-2}) \\ 0.5-1 \text{ mm particles } (23.78-10^3 \text{ particles/} \\ m^{-2}) \\ 0.1-0.5 \text{ mm particles } (34.35-10^3 \text{ particles/} \\ cles/m^{-2}) \end{array} $	[33]
Microplastics	August 2016	Surface water	0.5–5 mm (1597–12,611 n/m ³)	[34]
Microplastics	August 2016	Sediments	0.5–5 mm (25–300 n/kg)	[34]

 Table 3.1 (continued)

5.2 Polycyclic Aromatic Hydrocarbons (PAHs)

PAHs fingerprints in the TGRA and their toxicological impacts were evaluated at the highest impoundment level of 175 m in 2015. It was found that 16 PAHs congeners ranged in the upper reaches (83–1631 ng/L), middle reaches (354–1159 ng/L), and lower reaches (23–747 ng/L) [10]. Among PAHs structural composition, low molecular weight (LMW) compounds were >85% of high molecular weight compounds (HMW). Diagnostic molecular ratios (DMRs) were used for source apportionment, implying that wood and coal combustion, heavy traffic, industrial emissions, gasoline combustion, and agriculture were the major sources of PAHs contamination in the TGRA. Besides, ecological risks were ranked as the upper reaches > middle reaches > lower reaches [10]. Transgenic Tg(cyp1a:gfp) zebrafish evaluated the toxicological aspects in the form of genetic expression, which revealed the contrary scenario may be due to the higher levels of HMW found in the middle and lower reaches, which is also confirmed by the prediction of Cox hazard proportional model [10].

5.3 Pharmaceuticals and Personal Care Products (PPCPs)

PPCPs involve various human and veterinary antibiotics, cosmetics, and daily-use products when released into the environment, which is also known as emerging pollutants [35]. However, in the TGRA, limited literature is available on PPCPs. A study reported by Yan et al. determine the distribution of antibiotics in the TGRA [31]. Among the detected pharmaceuticals, sulfonamides (SAs) levels were higher than the total concentration of tetracyclines (TCs) and quinolones (FQs). The concentrations of SAs, TCs, and FQs ranged from 21.55–536.86, 3.69–438.76, and 15.78–213.84 ng/L, respectively [31]. Furthermore, the class 1 integron gene (Int11) showed more proliferation and propagation among bacterial-resistant genes [31]. Similarly, a recent study reported the individual and combined toxicogenetic effects of sulfonamides in the TGRA [20]. It was found that, among individual SAs, sulfamethoxazole (SMX: 289.3 ng/L) exhibited the highest levels followed by sulfamerazine (SMR: 6.57 ng/L), sulfapyridine (SPY: 76.57 ng/L), sulfameter (ST: 6.68 ng/L), and sulfadiazine (SDZ: 1.93 ng/L) [20]. The major sources were the emissions from the pharmaceutical industries and hospitals abundantly found in the upper part of the TGRA. However, the lower levels in the middle reach of the TGRA were due to lower anthropogenic pressure and less human intervention/ urbanization [20].

5.4 Phthalates (PAEs)

PAEs as plasticizers are ubiquitously released in the TGRA. Among six priority (PAEs), Di-(2-ethylhexyl) phthalate (DEHP) is considered the most toxic compound. The DEHP levels were previously identified in the TGRA and evaluated in vivo or in vitro exposure at environmentally relevant concentrations (ERCs) (Table 3.2) [22]. It was found that DEHP levels were ranked as follows: upper reaches ($6.21 \mu g/L$) > middle reaches ($1.48 \mu g/L$) > lower reaches ($0.38 \mu g/L$) in the TGRA surface water, and also double that of the United States Environmental Protection Agency (US EPA) guidelines, indicating high-level ecological risk [22]. The sources and primary reason for these high concentrations could be related to greater production and consumption rate of PAEs in this area [22]. Moreover, toxicological bioassays revealed that DEHP at ERCs significantly impaired the normal functions of aquatic species in the TGRA.

5.5 Microplastics (MPs)

MPs refer to the small plastic items that can break down into smaller particles size usually <5 mm. Previous studies indicated that various sizes of microplastics existed

		Abundance/levels in					
River	Abundance/levels in water	Sediment	References				
Microplastics							
Yangtze River	$\begin{array}{c} 34.1\times10^5 \text{ to } 136.1\times10^5 \\ \text{particles/km}^2 \end{array}$	-	[36]				
Estuaries of Yangtze River tributaries	1.92×10^5 to 118.9×10^5 particles/km ²	-	[36]				
Yulin River	2.00 × 10 ⁻² to 7.00 × 10 ⁻¹ items/L,	-	[37]				
Yangtze River	1597–12,611 particles m^{-3}	25–300 particles kg ⁻¹ w	[38]				
Xiangxi Bay TGR	0.55×10^{5} -342 × 10 ⁵ par- ticles km ⁻²	80-864 particles m ⁻²	[39]				
Xiangxi River (Three Gorges Reservoir)		550-14,580 particles m ⁻²	[33]				
Yangtze Estuary	500 n/m ³ to 10,200 n/m ³		[40]				
DEHP							
Three Gorges Reservoir	0.38–6.21 µg/L	-	[22]				
Yangtze River tributaries	1.7–394.4 ng/L	10.9-1107.1 ng/g	[41]				

Table 3.2 Summary of the distribution of microplastics and DEHP in the TGRA surface water and sediments

in the TGRA, which can produce a high ecological risk to aquatic species [33]. Limited studies have been reported regarding the microplastic distribution in the TGRA. A recent study reported by Zhang et al. [33] indicated that microplastic abundance in surface water ranged from 1597 to 12,611 n/m³. Whereas, in sediments, it was 25–300 n/kg wet weight and urban areas were the most contaminated, compared to the countryside. Among different types of MPs, fibers (polystyrene) with small-sized particles were the most dominant microplastics. Similarly, Mathias et al. [34] reported that, in surface water of the TGRA, microplastic size from <0.5 to 5 mm occupied 1597–12,611 n/m³ fractions, whereas, in the sediments, it ranged from 25 to 300 n/kg (Table 3.2). The authors mentioned that wastewater from the nearby plant, domestic waste, and the runoff from the river might be the major sources of MPs pollution in the TGRA. Besides, the flourished tourism might contribute to the elevated MPs levels.

5.6 Nutrient or Algal Pollution

Organic matter (OM) is a vital parameter in the terrestrial environment. However, it enters the surface water via leaching, runoff, and flooding, and ultimately affects biogeochemical cycling and disturbs homeostasis. Moreover, many pollutants, such as trace HM, POPs, and PAHs, may enter the surface water. However, very few studies are available regarding the effect of dissolved organic matter (DOM) in the TGRA. A study reported by Jiang et al. [18] evaluated the DOM extracted from the soils in the TGRA area, especially the water fluctuation zone via various analytical methods. It was found that samples from the TGRA were comprised of OM, poly-saccharides, and lignin. Besides, the compositional complexity of soil OM is very crucial, because OM plays an important role in the fate of contaminants in the TGRA [18].

A study reported that phosphate waste was discarded in the Xiangxi river bay [11]. Phosphorous leaching was determined under both neutral and acidic conditions. It was found that phosphorous release was totally dependent on solubility, which increased as the pH decreased [11]. The authors further recommended that phosphate rock's weathering or leaching should be monitored timely, because it could be the point source of pollution for the aquatic species and harmful to public health [11].

5.7 Microbial Diversity and Pollution

In 2015, microbial abundance and water quality were detected along the upper, middle, and lower reaches of the TGRA by Niu et al. [42]. Results indicated that, at the highest chemical oxygen demand (COD) level, the highest number/abundance of *Firmicutes* and *Bacillus* were found in the upper, middle, and down streams [42]. Besides, the results of redundancy analysis revealed that COD and phosphates were the primary environmental parameters that significantly affected the bacterial community structure [42]. Furthermore, the general trend of bacterial richness decreased in the order: downstream > middle stream > upstream [42].

Moreover, mercury is a carcinogenic heavy metal in the environment. The organic form of mercury, methyl mercury (MeHg), is more hazardous to human health [23]. Microbial communities are involved in controlling in MeHg toxicity and movement in the environment [17]. Therefore, the identification of microbial diversity involved in MeHg degradation in the TGRA is important for the ecosystem and public health [23]. A study reported by Xiang et al. [19] indicated that MeHg contamination in the TGRA soils was relatively higher in both summer and winter areas with the impoundment of 175–155 m. It was found that *Deltaproteobacteria* and *Methanomicrobia* showed higher abundance as Hg methylators seasonally [19]. In short, seasonal variations may enhance microbial community abundance as well as MeHg methylation [19]. Moreover, more detailed seasonal level studies are recommended to clearly comprehend the pollution situation in the TGRA.

6 Conclusion

In summary, this chapter aims to identify the sources and the distribution of the pollutants ubiquitously present in the TGRA. Various studies from 2008 to 2016 revealed elevated HM levels. Furthermore, PAHs fingerprints in the TGRA showed pyrogenic and petrogenic combustion activities. DEHP levels were found to be exceeded the WHO and US EPA guidelines, indicating high ecological risks. PPCPs, such as sulfonamides, are mainly derived from industrial/pharmaceutical and hospital waste. Moreover, microplastic pollution was also high mainly in the upper stream of the TGRA, and runoff from the river transport might be considered as the potential source of microplastic pollution. Microbe and DOM that indirectly aggravated pollution were also affected by the seasonal impoundment rate. In conclusion, more detailed temporal studies are recommended to depict a clear picture of the pollution load in the reservoir.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- N. Hamid, J.H. Syed, M. Junaid, A. Mahmood, J. Li, G. Zhang, R.N. Malik, Elucidating the urban levels, sources and health risks of polycyclic aromatic hydrocarbons (PAHs) in Pakistan: Implications for changing energy demand. Sci. Total Environ. 619–620, 165–175 (2018)
- N. Hamid, J.H. Syed, A. Kamal, F. Aziz, S. Tanveer, U. Ali, A. Cincinelli, A. Katsoyiannis, I.C. Yadav, J. Li, R.N. Malik, G. Zhang, A review on the abundance, distribution and eco-biological risks of PAHs in the key environmental matrices of South Asia, in *Reviews of Environmental Contamination and Toxicology*, ed. by P. de Voogt, vol. 240, (Springer, Cham, 2017), pp. 1–30
- L. Lin, C. Li, W. Yang, L. Zhao, M. Liu, Q. Li, J.C. Crittenden, Spatial variations and periodic changes in heavy metals in surface water and sediments of the Three Gorges Reservoir, China. Chemosphere 240, 124837 (2020)
- 4. L. Lu, J. Liu, Z. Li, X. Zou, J. Guo, Z. Liu, J. Yang, Y. Zhou, Antibiotic resistance gene abundances associated with heavy metals and antibiotics in the sediments of Changshou Lake in the Three Gorges Reservoir Area, China. Ecol. Indic. 113, 106275 (2020)
- F.M. Windsor, M.G. Pereira, C.R. Tyler, S.J. Ormerod, River organisms as indicators of the distribution and sources of persistent organic pollutants in contrasting catchments. Environ. Pollut. 255, 113144 (2019)
- S. Audry, J. Schäfer, G. Blanc, J.-M. Jouanneau, Fifty-year sedimentary record of heavy metal pollution (Cd, Zn, Cu, Pb) in the Lot River reservoirs (France). Environ. Pollut. 132, 413–426 (2004)
- A.M. Hussein, D. Neama Jabbar, A.R. Ali, Spatial distribution and evaluation of heavy metals in surface sediments of the Al-Najaf Sea depression reservoir, Iraq. Alex. Eng. J. 59, 5197–5206 (2020)

- 3 Main Sources and Distribution of Pollutants in the Reservoir
- H. Jiang, M. Qiang, Q. Fan, M. Zhang, Scientific research driven by large-scale infrastructure projects: A case study of the Three Gorges Project in China. Technol. Forecast. Soc. Chang. 134, 61–71 (2018)
- B. Li, X. Yuan, H. Xiao, Z. Chen, Design of the dike-pond system in the littoral zone of a tributary in the Three Gorges Reservoir, China. Ecol. Eng. 37, 1718–1725 (2011)
- Y.-M. Tang, M. Junaid, A. Niu, S. Deng, D.-S. Pei, Diverse toxicological risks of PAHs in surface water with an impounding level of 175m in the Three Gorges Reservoir Area, China. Sci. Total Environ. 580, 1085–1096 (2017)
- 11. L.-G. Jiang, B. Liang, Q. Xue, C.-W. Yin, Characterization of phosphorus leaching from phosphate waste rock in the Xiangxi River watershed, Three Gorges Reservoir, China. Chemosphere 150, 130–138 (2016)
- J. Zhao, G. Fu, K. Lei, Y. Li, Multivariate analysis of surface water quality in the Three Gorges Area of China and implications for water management. J. Environ. Sci. 23, 1460–1471 (2011)
- Z. Shi, A. Wen, X. Zhang, D. Yan, Comparison of the soil losses from 7Be measurements and the monitoring data by erosion pins and runoff plots in the Three Gorges Reservoir region, China. Appl. Radiat. Isot. 69, 1343–1348 (2011)
- Z. Cao, X. Zhang, N. Ai, Effect of sediment on concentration of dissolved phosphorus in the Three Gorges Reservoir. Int. J. Sediment Res. 26, 87–95 (2011)
- Y. Feng, Q. Bao, X. Xiao, M. Lin, Geo-accumulation vector model for evaluating the heavy metal pollution in the sediments of Western Dongting Lake. J. Hydrol. 573, 40–48 (2019)
- 16. Y. Bao, P. Gao, X. He, The water-level fluctuation zone of Three Gorges Reservoir—a unique geomorphological unit. Earth Sci. Rev. **150**, 14–24 (2015)
- 17. Q. Gao, Y. Li, Q. Cheng, M. Yu, B. Hu, Z. Wang, Z. Yu, Analysis and assessment of the nutrients, biochemical indexes and heavy metals in the Three Gorges Reservoir, China. Water Res. 92(2016), 262–274 (2008–2013)
- T. Jiang, J. Kaal, J. Liang, Y. Zhang, S. Wei, D. Wang, N.W. Green, Composition of dissolved organic matter (DOM) from periodically submerged soils in the Three Gorges Reservoir areas as determined by elemental and optical analysis, infrared spectroscopy, pyrolysis-GC–MS and thermally assisted hydrolysis and methylation. Sci. Total Environ. 603-604, 461–471 (2017)
- Y. Xiang, Y. Wang, C. Zhang, H. Shen, D. Wang, Water level fluctuations influence microbial communities and mercury methylation in soils in the Three Gorges Reservoir, China. J. Environ. Sci. 68, 206–217 (2018)
- N. Hamid, M. Junaid, D.-S. Pei, Individual and combined mechanistic toxicity of sulfonamides and their implications for ecological risk assessment in the Three Gorges Reservoir Area (TGRA), China. J. Hazardous Mater. 382, 121106 (2020)
- J. Zhang, S. Li, R. Dong, C. Jiang, M. Ni, Influences of land use metrics at multi-spatial scales on seasonal water quality: A case study of river systems in the Three Gorges Reservoir Area, China. J. Clean. Prod. 206, 76–85 (2019)
- M. Junaid, P.-P. Jia, Y.-M. Tang, W.-X. Xiong, H.-Y. Huang, P.R. Strauss, W.-G. Li, D.-S. Pei, Mechanistic toxicity of DEHP at environmentally relevant concentrations (ERCs) and ecological risk assessment in the Three Gorges Reservoir Area, China. Environ. Pollut. 242, 1939–1949 (2018)
- C. Ye, S. Li, Y. Zhang, Q. Zhang, Assessing soil heavy metal pollution in the water-levelfluctuation zone of the Three Gorges Reservoir, China. J. Hazard. Mater. 191, 366–372 (2011)
- 24. J. Liu, T. Jiang, R. Huang, D. Wang, J. Zhang, S. Qian, D. Yin, H. Chen, A simulation study of inorganic sulfur cycling in the water level fluctuation zone of the Three Gorges Reservoir, China and the implications for mercury methylation. Chemosphere 166, 31–40 (2017)
- 25. C. Sang, Y. Zheng, Q. Zhou, D. Li, G. Liang, Y. Gao, Effects of water impoundment and waterlevel manipulation on the bioaccumulation pattern, trophic transfer and health risk of heavy metals in the food web of Three Gorges Reservoir (China). Chemosphere 232, 403–414 (2019)
- B. Zhang, J.-S. Guo, F. Fang, Z. Li, C. Fu, Concentration of nutrients in the soil in water-levelfluctuating zone of Three Gorges Reservoir. Ecohydrol. Hydrobiol. 12, 105–114 (2012)

- 27. Y. Huang, C. Fu, Z. Li, F. Fang, W. Ouyang, J. Guo, Effect of dissolved organic matters on adsorption and desorption behavior of heavy metals in a water-level-fluctuation zone of the Three Gorges Reservoir, China. Ecotoxicol. Environ. Saf. 185, 109695 (2019)
- L. Gao, B. Gao, W. Peng, D. Xu, S. Yin, Assessing potential release tendency of As, Mo and W in the tributary sediments of the Three Gorges Reservoir, China. Ecotoxicol. Environ. Saf. 147, 342–348 (2018)
- 29. H. Bing, Y. Wu, J. Zhou, H. Sun, X. Wang, H. Zhu, Spatial variation of heavy metal contamination in the riparian sediments after two-year flow regulation in the Three Gorges Reservoir, China. Sci. Total Environ. 649, 1004–1016 (2019)
- 30. H. Zhu, H. Bing, Y. Wu, J. Zhou, H. Sun, J. Wang, X. Wang, The spatial and vertical distribution of heavy metal contamination in sediments of the Three Gorges Reservoir determined by anti-seasonal flow regulation. Sci. Total Environ. 664, 79–88 (2019)
- M. Yan, C. Xu, Y. Huang, H. Nie, J. Wang, Tetracyclines, sulfonamides and quinolones and their corresponding resistance genes in the Three Gorges Reservoir, China. Sci. Total Environ. 631–632, 840–848 (2018)
- 32. C. Wu, X. Huang, J.D. Witter, A.L. Spongberg, K. Wang, D. Wang, J. Liu, Occurrence of pharmaceuticals and personal care products and associated environmental risks in the central and lower Yangtze River, China. Ecotoxicol. Environ. Saf. 106, 19–26 (2014)
- 33. K. Zhang, X. Chen, X. Xiong, Y. Ruan, H. Zhou, C. Wu, P.K.S. Lam, The hydro-fluctuation belt of the Three Gorges Reservoir: Source or sink of microplastics in the water? Environ. Pollut. 248, 279–285 (2019)
- 34. F.T. Mathias, D.H. Fockink, G.R. Disner, V. Prodocimo, J.L.C. Ribas, L.P. Ramos, M.M. Cestari, H.C. Silva de Assis, Effects of low concentrations of ibuprofen on freshwater fish Rhamdia quelen. Environ. Toxicol. Pharmacol. 59, 105–113 (2018)
- 35. M. Di, X. Liu, W. Wang, J. Wang, Manuscript prepared for submission to environmental toxicology and pharmacology pollution in drinking water source areas: Microplastics in the Danjiangkou reservoir, China. Environ. Toxicol. Pharmacol. 65, 82–89 (2019)
- 36. K. Zhang, W. Gong, J. Lv, X. Xiong, C. Wu, Accumulation of floating microplastics behind the Three Gorges Dam. Environ. Pollut. 204, 117–123 (2015)
- 37. Y. Mao, H. Li, W. Gu, G. Yang, Y. Liu, Q. He, Distribution and characteristics of microplastics in the Yulin River, China: Role of environmental and spatial factors. Environ. Pollut. 265, 115033 (2020)
- M. Di, J. Wang, Microplastics in surface waters and sediments of the Three Gorges Reservoir, China. Sci. Total Environ. 616, 1620–1627 (2018)
- 39. K. Zhang, X. Xiong, H. Hu, C. Wu, Y. Bi, Y. Wu, B. Zhou, P.K. Lam, J. Liu, Occurrence and characteristics of microplastic pollution in Xiangxi Bay of Three gorges Reservoir, China. Environ. Sci. Technol. 51, 3794–3801 (2017)
- S. Zhao, L. Zhu, T. Wang, D. Li, Suspended microplastics in the surface water of the Yangtze Estuary system, China: First observations on occurrence, distribution. Mar. Pollut. Bull. 86, 562–568 (2014)
- 41. L. Lin, L. Dong, X. Meng, Q. Li, Z. Huang, C. Li, R. Li, W. Yang, J. Crittenden, Distribution and sources of polycyclic aromatic hydrocarbons and phthalic acid esters in water and surface sediment from the Three Gorges Reservoir. J. Environ. Sci. 69, 271–280 (2018)
- 42. A. Niu, L.-Y. Song, Y.-H. Xiong, C.-J. Lu, M. Junaid, D.-S. Pei, Impact of water quality on the microbial diversity in the surface water along the Three Gorge Reservoir (TGR), China. Ecotoxicol. Environ. Saf. 181, 412–418 (2019)

Chapter 4 Migration and Transformation of Pollutants in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract This chapter particularly aims to elucidate the factors involved in the migration and transformation of pollutants in the Three Gorges Reservoir Area (TGRA). However, limited seasonal studies available make it difficult to comprehend the real situation of different pollutants. More literature related to heavy metals (HMs) are available, which showed that, from 2003 to 2010, the HMs levels were found within the limits of safety guidelines as prescribed by the US EPA. Moreover, the hydro-fluctuation belt (HFB) greatly affected the migration of HMs from the inner layers of soil to the upper layers. In the upper reaches of the TGRA (2004–2015), the ecological health was found to be lower in summer, autumn, and spring, compared to winter. From 2003 to 2014, the chemical oxygen demand (COD) in the TGRA has increased, out of which domestic wastewater contributed about 68.2%. Overall, the ecological health of the TGRA for the past 12 years was relatively stable, except in winters when the pollution load was relatively high due to the varied impoundment levels.

Keywords Pollutants · Seasonal trend · Migration · Ecological risks · Reservoir

1 Introduction

In recent years, the migration of pollutants and their transformation into the freshwater ecosystem has become a global problem [1, 2]. Large-scale reservoirs have many diversified positive and negative impacts. Economic benefits, such as flood control, power generation, and water storage are some positive aspects [3, 4]. However, intensified anthropogenic activities and unsustainable practices become some negative aspects that amplified the pollutant levels in the reservoir [5, 6]. Generally, toxic pollutants are classified into two major categories, organic and inorganic pollutants [7]. Mostly, they are persistent in nature as well as possess long-distance transmission properties. The majority of the pollutants have high bioaccumulation and biomagnification properties and are difficult to degrade [8]. Besides, the biological effect of organic pollutants is more complex, because they can cause serious health problems, such as endocrine disruption abilities [9].



Fig. 4.1 Major factors that are involved in the migration and transformation of pollutants in the reservoir

Previously, it was reported that the majority of the persistent organic pollutants (POPs) are hydrophobic and tend to accumulate in aquatic species, whereas some are hydrophilic, which can easily migrate in water [10]. Polycyclic aromatic hydrocarbons (PAHs), polychlorinated biphenyls (PCBs), microplastics (MPs), microbial and algal pollutants, organic matter (OM), and pharmaceutical and personal care products (PPCPs) are the common pollutants ubiquitously found in freshwater, sediments, and fish bodies [11].

Bioavailability and transformation of the pollutants in the reservoir also depend on environmental/abiotic factors [12, 13], such as temperature, dissolved organic carbon, chemical oxygen demand, pH, precipitation, and humidity (Fig. 4.1). Moreover, biotic factors include living communities' feeding habits, exchange surfaces, and metabolic activity. Reservoir impoundment also has a significant impact on the migration and transformation of pollutants [14]. Sang et al. [10] reported that with the increasing water impoundment level and seasonal variations, pollutant load has also been enhanced [10].

2 Fluctuating Impoundment Level Effects

Water fluctuation level further regulates the pollutants transfer and migration along with the biogeochemistry of the sediments and the response of the microbial community, which helps in the sorption and desorption of the pollutants mainly heavy metals. The Three Gorges Reservoir Area (TGRA) is considered the largest



Fig. 4.2 Water fluctuation level affects the pollution load and elevates the health risk among aquatic species



Fig. 4.3 Hydro-fluctuation belt observed across the TGRA

hydropower plant in the world, also attracts attention due to its variation in the water impoundment level for many years. Furthermore, the continuous contamination of persistent toxic pollutants in the TGRA poses high ecological risks to the aquatic species (Fig. 4.2).

From the previous studies, it was found that the water impoundment level may fluctuate from 145 to 175 m, which also varied from the hydrodynamic conditions [15, 16]. In the TGRA, a special zone has been created due to the seasonal impoundment, which is called the hydro-fluctuation belt (HFB) (Fig. 4.3) [17]. This belt comprises a vertical height of 30 m with a length of 662 km² and covered a total area of about 349 km² along the whole Yangtze River [18]. The HFB zone has gained much attraction from all over the world due to its unique characteristics [19]. Moreover, during wet and dry seasons, the rise and fall of the water levels in the TGRA is different from the natural water flow, resulting in higher chances of pollutants may enter in the reservoir, especially at lower water levels in summers with high precipitation [16, 18].

3 Status and Migration of Pollutants in the TGRA

A study reported by Zhang et al. [16] investigated the microplastic sources and sinks along with the influence of HFB in the TGRA. From the sediment samples, it was found that more microplastics were accumulated in the HFB area due to the influence of runoff or water impoundment, implying that the HFB acts as both the source and sink for the microplastics. However, additional studies are required to determine the real-time status and the transport and fate of the microplastics. Similarly, the heavy metals (HMs) quantified in the upper reaches of the TGRA were higher in concentration compared to the lower reaches, indicating that these chemicals were migrated from the domestic sewage and industrial waste [20]. On contrary, the HMs levels were decreased in sediments during the impoundment period, due to faster flow velocity, resulting in the low deposition of particles and high release of HMs from sediments [21].

In 2015–2016, heavy metals (As, Cr, Cd, Cu, Pb, and Zn) levels were quantified in 46 river sections of the mainstream and tributaries of the TGRA under high and lower impoundment levels and evaluated their health risks [22]. It was found that the levels of HMs were observed ranking as Zn > As > Cu > Cr > Pb > Cd, which were lower than the permissible limit prescribed by the Chinese standard except for Zn [22]. Moreover, the results showed that As, Cd, Cu, and Cr exhibited a significant decrease in the levels from 2015 to 2016. Health risk assessment revealed that heavy metals caused higher risks in the mainstream, compared to the tributaries [22].

Similarly, the migration of HMs pollutants and their trophic transfer are found in aquatic species of the TGRA [10]. Results showed that, during the impoundment years (2003–2010), HMs levels were found within the limits of safety guidelines as prescribed by the US EPA [10]. Moreover, Cu, Zn, Hg, and Fe levels in fish were increased, compared to the concentrations reported before impoundment [10]. A study reported by Huang et al. [20] investigated the migration of HMs (Cd, Cr, and Cu) within the adjacent soil of the TGRA and tracked their adsorption and desorption in the reservoir. Results indicated that water fluctuation/impoundment level greatly affected the migration of HMs from the inner layers of soil to the upper layers. Moreover, hydraulic power further increased the release of Cd and Cu from the lower layer [20], while Cr was mostly released in the environment with continuous recession and inundation [21].

4 Temporal and Seasonal Variations of Pollutants

The published literature regarding the seasonal trend of pollutants is limited. Therefore, it is difficult to elucidate the seasonal status of pollutants [23]. Moreover, the water fluctuation level affected the water biochemistry of the aquatic ecosystem, which might increase the release of inorganic and organic pollutants [18]. Consequently, more detailed seasonal studies and strict implementation of the industrial discharge rules should protect the ecological integrity of the TGRA.

Long-term temporal studies from 1992 to 2016 reported the levels of chemical oxygen demand (COD) through the potassium permanganate index [22]. The result demonstrated that total phosphorous levels decreased from $40.9\% \pm 9.9\%$ in 2003 and from $22.2\% \pm 9.7\%$ in 2016. However, the total nitrogen $(1.3 \pm 2.4\%)$ and ammonia $(8.2\% \pm 2.6\%)$ levels were increased, respectively [24]. The significant reductions might be due to the decrease in the water flow, which leads to an increase in the sediment settlement/sink. Similarly, another study reported by the bulletin on the ecological and environmental monitoring (SEPA) from the years 2003 to 2014 found that COD in the TGRA has reached up to 1.98 million tons, out of which domestic wastewater contributed around 68.2% of the total COD [25] (Fig. 4.4).

Moreover, anti-seasonal operation levels with the water volume further elevated the pollutant dilution capacity [20]. Besides, the algal blooms and the growth rate of phytoplankton have been widely increased with a factor of 2.7% in the mainstream since 2004, which may deteriorate the water quality and intensify the pollution in the TGRA [22]. Moreover, phytoplankton growth in the TGRA is mainly supported by phosphorus during all seasons. Therefore, by limiting the growth level of phosphorous, short- or long-term eutrophication might be avoided.

5 Ecological Health of the TGRA

A study reported by Zhao et al. [22] found that ecological health was lower in summer, autumn, and spring, compared to winter. The ecological health status of the TGRA from the year 2004 to 2015 was determined using the chemical index method. It was found that the overall ecological health of the TGRA for the past 12 years was relatively stable. However, in winter, the pollution load was relatively high due to the varied impoundment levels (Fig. 4.5).

Various parameters play an important role in determining the ecological health of the TGRA, such as intensified anthropogenic activities (domestic, industrial, and agricultural waste) [26, 27] and natural processes (eutrophication, volatilization, photodegradation, and microbial degradation) [28, 29]. However, we cannot completely control natural activities, but it is in our hands to limit man-made pollution by strictly following the guidelines and complying with our safety limits [30], which ensures ecological integrity.

6 Conclusion

In summary, this chapter summarizes the factors involved in the migration and transformation of pollutants in the TGRA. However, it was found that HMs are available, which showed that from 2003 to 2010. HMs levels were found within the



Fig. 4.4 Wastewater discharge and its contribution of COD and ammonia in the TGRA. (a) Annual temporal comparison of the domestic and industrial wastewater discharge. (b) its percentage contribution of COD and ammonia from 2003 to 2014 observed in the TGRA. (Source: SEPA, 2004–2015)

limits of safety guidelines as prescribed by the US EPA. Furthermore, the HFB greatly influences the migration of HMs from inner the layers of soil to the upper layers. The COD in the TGRA has increased up to 1.98 million tons from 2003 to 2014 and domestic wastewater contributed the maximum with 68.2%. In short, the overall ecological health of the TGRA in the upper reaches is relatively stable but it may vary due to fluctuation in the impoundment levels.



Fig. 4.5 Annual seasonal ecological risk index in the upper reach of the TGRA for the years 2004–2015 (Source: SEPA, 2004–2015)

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- R.L. Howard, K. Kertell, J.C. Truett, Chapter 15—freshwater invertebrates: Their regulation and importance to vertebrates, in *The Natural History of an Arctic Oil Field*, ed. by J.C. Truett, S.R. Johnson, (Academic Press, San Diego, 2000), p. 307-XVIII
- S. Birk, D. Chapman, L. Carvalho, B.M. Spears, H.E. Andersen, C. Argillier, S. Auer, A. Baattrup-Pedersen, L. Banin, M. Beklioğlu, E. Bondar-Kunze, A. Borja, P. Branco, T. Bucak, A.D. Buijse, A.C. Cardoso, R.-M. Couture, F. Cremona, D. de Zwart, C.K. Feld, M.T. Ferreira, H. Feuchtmayr, M.O. Gessner, A. Gieswein, L. Globevnik, D. Graeber, W. Graf, C. Gutiérrez-Cánovas, J. Hanganu, U. Işkın, M. Järvinen, E. Jeppesen, N. Kotamäki, M. Kuijper, J.U. Lemm, S. Lu, A.L. Solheim, U. Mischke, S.J. Moe, P. Nõges, T. Nõges, S.J. Ormerod, Y. Panagopoulos, G. Phillips, L. Posthuma, S. Pouso, C. Prudhomme, K. Rankinen, J.J. Rasmussen, J. Richardson, A. Sagouis, J.M. Santos, R.B. Schäfer, R. Schinegger, S. Schmutz, S.C. Schneider, L. Schülting, P. Segurado, K. Stefanidis, B. Sures, S.J. Thackeray, J. Turunen, M.C. Uyarra, M. Venohr, P.C. von der Ohe, N. Willby, D. Hering, Impacts of multiple stressors on freshwater biota across spatial scales and ecosystems. Nat. Ecol. Evol. 4, 1060–1068 (2020)
- 3. J. Chen, K. Ma, X. Pang, H. Yang, Secondary migration of hydrocarbons in Ordovician carbonate reservoirs in the Lunnan area, Tarim Basin. J. Pet. Sci. Eng. **188**, 106962 (2020)
- M. Basheer, V. Nechifor, A. Calzadilla, K. Siddig, M. Etichia, D. Whittington, D. Hulme, J.J. Harou, Collaborative management of the Grand Ethiopian Renaissance Dam increases economic benefits and resilience. Nat. Commun. 12, 5622 (2021)
- M. Varol, M.R. Sünbül, Macroelements and toxic trace elements in muscle and liver of fish species from the largest three reservoirs in Turkey and human risk assessment based on the worst-case scenarios. Environ. Res. 184, 109298 (2020)
- 6. Y. Mao, Y. Liu, H. Li, Q. He, H. Ai, W. Gu, G. Yang, Distinct responses of planktonic and sedimentary bacterial communities to anthropogenic activities: Case study of a tributary of the Three Gorges Reservoir, China. Sci. Total Environ. 682, 324–332 (2019)
- S.M. Richardson, J.M. Hanson, A. Locke, Effects of impoundment and water-level fluctuations on macrophyte and macroinvertebrate communities of a dammed tidal river. Aquat. Ecol. 36, 493–510 (2002)
- N. Hamid, J.H. Syed, A. Kamal, F. Aziz, S. Tanveer, U. Ali, A. Cincinelli, A. Katsoyiannis, I.C. Yadav, J. Li, A review on the abundance, distribution and eco-biological risks of PAHs in the key environmental matrices of South Asia. Rev. Environ. Contam. Toxicol. 240, 1–30 (2016)
- L.L. Wei, Q. Zhou, C.X. Xie, J. Wang, J. Li, Bioaccumulation and biomagnification of heavy metals in Three Gorges Reservoir and effect of biological factors. Huan Jing Ke Xue 37, 325–334 (2016)
- C. Sang, Y. Zheng, Q. Zhou, D. Li, G. Liang, Y. Gao, Effects of water impoundment and waterlevel manipulation on the bioaccumulation pattern, trophic transfer and health risk of heavy metals in the food web of Three Gorges Reservoir (China). Chemosphere 232, 403–414 (2019)
- N. Hamid, M. Junaid, D.-S. Pei, Combined toxicity of endocrine-disrupting chemicals: A review. Ecotoxicol. Environ. Saf. 215, 112136 (2021)
- J. Yang, M. Zhou, K. Yu, K.Y.-H. Gin, M. Hassan, Y. He, Heavy metals in a typical city-riverreservoir system of East China: Multi-phase distribution, microbial response and ecological risk. J. Environ. Sci. 112, 343–354 (2022)

- Y. Cheng, F. Zhao, J. Wu, P. Gao, Y. Wang, J. Wang, Migration characteristics of arsenic in sediments under the influence of cascade reservoirs in Lancang River basin. J. Hydrol. 606, 127424 (2022)
- 14. Y. Yu, Y.C. Wang, H.D. Zhou, B. Gao, G.F. Zhao, Biomagnification of heavy metals in the aquatic food chain in Daning River of the Three Gorges Reservoir during initial impoundment. Huan Jing Ke Xue 34, 3847–3853 (2013)
- Y.-M. Tang, M. Junaid, A. Niu, S. Deng, D.-S. Pei, Diverse toxicological risks of PAHs in surface water with an impounding level of 175m in the Three Gorges Reservoir Area, China. Sci. Total Environ. 580, 1085–1096 (2017)
- K. Zhang, X. Chen, X. Xiong, Y. Ruan, H. Zhou, C. Wu, P.K.S. Lam, The hydro-fluctuation belt of the Three Gorges Reservoir: Source or sink of microplastics in the water? Environ. Pollut. 248, 279–285 (2019)
- 17. X. Xu, Y. Tan, G. Yang, Environmental impact assessments of the Three Gorges Project in China: Issues and interventions. Earth Sci. Rev. **124**, 115–125 (2013)
- C. Ye, S. Li, Y. Zhang, Q. Zhang, Assessing soil heavy metal pollution in the water-levelfluctuation zone of the Three Gorges Reservoir, China. J. Hazard. Mater. 191, 366–372 (2011)
- K. Zhang, X. Xiong, H. Hu, C. Wu, Y. Bi, Y. Wu, B. Zhou, P.K.S. Lam, J. Liu, Occurrence and characteristics of microplastic pollution in Xiangxi Bay of Three Gorges Reservoir, China. Environ. Sci. Technol. 51, 3794–3801 (2017)
- 20. Y. Huang, C. Fu, Z. Li, F. Fang, W. Ouyang, J. Guo, Effect of dissolved organic matters on adsorption and desorption behavior of heavy metals in a water-level-fluctuation zone of the Three Gorges Reservoir, China. Ecotoxicol. Environ. Saf. 185, 109695 (2019)
- L. Lin, C. Li, W. Yang, L. Zhao, M. Liu, Q. Li, J.C. Crittenden, Spatial variations and periodic changes in heavy metals in surface water and sediments of the Three Gorges Reservoir, China. Chemosphere 240, 124837 (2020)
- 22. L. Zhao, D. Gong, W. Zhao, L. Lin, W. Yang, W. Guo, X. Tang, Q. Li, Spatial-temporal distribution characteristics and health risk assessment of heavy metals in surface water of the Three Gorges Reservoir, China. Sci. Total Environ., 134883 (2019)
- K.-W. Zhu, Y.-C. Chen, S. Zhang, B. Lei, Z.-M. Yang, L. Huang, Vegetation of the water-level fluctuation zone in the Three Gorges Reservoir at the initial impoundment stage. Global Ecol. Conserv. 21, e00866 (2020)
- 24. Z. Jia, S. Li, Assessment of soil heavy metals for eco-environment and human health in a rapidly urbanization area of the upper Yangtze Basin. Sci. Rep. 8, 3256 (2018)
- 25. SEPA, Bulletin on the Ecological and Environmental Monitoring Results of the Three Gorges Project (State Environmental Protection Administration, Beijing, 2004–2015)
- 26. Y. Yang, T. Chen, X. Liu, S. Wang, K. Wang, R. Xiao, X. Chen, T. Zhang, Ecological risk assessment and environment carrying capacity of soil pesticide residues in vegetable ecosystem in the Three Gorges Reservoir Area. J. Hazard. Mater. 435, 128987 (2022)
- L. Li, M. Gou, N. Wang, L. La, C. Liu, Do ecological restoration programs reduce forest fragmentation? Case study of the Three Gorges Reservoir Area, China. Ecological Engineering 172, 106410 (2021)
- 28. B. Li, N. Chen, W. Wang, C. Wang, R.J.P. Schmitt, A. Lin, G.C. Daily, Eco-environmental impacts of dams in the Yangtze River Basin, China. Sci. Total Environ. 774, 145743 (2021)
- J. Zhang, S. Li, R. Dong, C. Jiang, M. Ni, Influences of land use metrics at multi-spatial scales on seasonal water quality: A case study of river systems in the Three Gorges Reservoir Area, China. J. Clean. Prod. 206, 76–85 (2019)
- 30. J. Sedláček, O. Bábek, T. Nováková, Sedimentary record and anthropogenic pollution of a complex, multiple source fed dam reservoirs: An example from the Nové Mlýny reservoir, Czech Republic. Sci. Total Environ. 574, 1456–1471 (2017)
Chapter 5 Harmful Algal Bloom in the Reservoir



Suman Thodhal Yoganandham and De-Sheng Pei 💿

Abstract The formation of harmful algal blooms is governed by a complex interaction of chemical, biological, physical, and geological processes along with reservoir-associated factors, such as water discharge, sediment deposition, and turbulence intensity. The abundant release of phosphates and nitrates in reservoirs from anthropogenic and natural sources leads to eutrophication, which accelerates the excessive growth of algae and induces algal blooms. Globally, algal blooms occurred in many reservoirs and become a core reservoir environmental problem. This chapter summarizes the types of algal blooms, elucidates their formation, and puts forward the methods for controlling harmful algal blooms, which may improve the management efficiency of reservoirs and lay the groundwork for future studies. Although there are various methods for treating harmful algal blooms in the reservoir, we can not only rely on any specific method to solve all problems. Therefore, an approach involving the combined methods is encouraged to apply to different types of algal blooms. Further, in-depth studies are needed to monitor and block invasive algal flora.

Keywords Harmful algal bloom · Reservoir · Formation of algal bloom · Toxin

1 Introduction

In the environments worldwide, the harmful algal bloom is a significant problem [1]. Algal blooms are not new and naturally occur in fertile regions for at least two centuries in history [2]. Nevertheless, with the growing environmental pollution over the last century, the frequency and geographical range of algal blooms have significantly increased in small and large lakes, rivers, reservoirs, wetlands, and other surface waters. As a result, a severe and persistent toxic algal bloom is currently spreading to large water bodies in the world, such as the Erie Lakes, Ontario, Okeechobee, Winnipeg (North America, Fig. 5.1), Kasumigaura (Japan), Taihu (China), the Caspian Sea (Europe), the Kinneret Lake (Israel), and Victoria (Africa) [3–5]. Some of the blooms are due to planktonic algae floating in the water, but often, the term may also apply to the aggregation of microscopic benthic algae or macro alga attached to the surfaces. *Planktothrix* spp., *Microcystis* spp.,



Fig. 5.1 Toxic blue-green algal bloom in the Copco Reservoir of California, USA (*Photo from David McLain/Alamy*)

Cylindrospermopsis spp., *Anabaena* spp., *Oscillatoria* spp., and *Aphanizomenon* spp. produce cyanobacterial metabolites and toxins in freshwater, which may damage the human health and ruin the environment [6, 7].

Blooms are dense accumulations in marine, brackish, and freshwater bodies of microscopic algae or cyanobacterial cells, frequently contributing to noticeable discoloration of the water [8]. The effects of algal toxins on humans after acute or chronic exposure may cause kidney and liver toxicity, and the causal relations must still be identified in the long run [2, 9]. Besides, many algal blooms are often associated with the deaths of animals, pets, and birds because of environmental pollution [8]. Generally, toxins compounds produced by cyanobacteria are becoming increasingly problematic in drinking water reservoirs, and off-flavor blends are essential in aquaculture operations [10]. Cyanotoxins are the toxins produced by cyanobacteria, which are photosynthetic prokaryotes and are known to produce a wide range of secondary metabolites [11]. Depending on their toxic activity, cyanotoxins are divided into hepatotoxins (microcystins, nodularin), cytotoxins (cylindrospermopsin), neurotoxins (anatoxin-a, saxitoxin), and dermatotoxins (lyngbyatoxin A, aplysiatoxins, and endotoxin-LPS) [12]. Most reported incidents of poisoning by cyanotoxins were associated with hepatotoxic and neurotoxic blooms of cyanobacteria. However, exposure to cyanotoxins, such as microcystins (MCs) and anatoxin-a (ATX-a), leads to the death of terrestrial and aquatic animals, as well as humans [13].

Harmful algal blooms indicate environmental instability and are often triggered by multiple ecological changes. These changes fall into three general categories: (a) watershed development; (b) climate-related changes; and (c) biological changes affecting the consumption, integrity, and viability of cyanobacterial cells. In eutrophic and meso-eutrophic waters, harmful algal flora is generally more frequent and more extreme. It occasionally appears in underproductive systems, especially with human impacts, such as acidification and organic loading [14].

2 The Formation of Algal Bloom in the Water Bodies

While several kinds of algae can trigger harmful algal blooms in freshwater species, cyanobacteria usually contribute to the most frequent and severe blooms [15]. Blooms are created when the algae are high within a given region, originally due to sustained growth of algal populations, typically accompanied by the physical mechanism that further concentrates cells [16]. A common characteristic of several harmful algal species is their way of generating cysts or stages of rest [17]. The majority of bloom-forming algal are photoautotrophic, and light availability is essential. However, besides carbon, nutrients including nitrogen and phosphorus are often required to create cellular material. Silicon or iron may sometimes be restricted by algal growth. The availability of light and nutrients and cell distribution within the water body is influenced by physical factors. Similar principles can be used to control the production of toxic algal blooms in freshwater and marine environments. Marine algal-blooming species use proliferation strategies to accomplish high densities [16]. For example, large harmful algal bloom species form large, gelatinous colonies, extremely dense blooms, or allelopathic toxins that are toxic to naturally grazing organisms like zooplankton and fish [18].

Nutrients such as nitrogen and phosphorous are a natural part of aquatic ecosystems and essential for plant growth (Fig. 5.2). Yet, when they spill out of urban and rural surfaces and flow into a river, lake, pond, or reservoir, they serve as fertilizer and encourage algae and bacteria growth [18], whereas N_2 could be directly fixed



Fig. 5.2 Illustration of nutrient sources of harmful algal blooms

from the atmosphere for cell growth [19]. Warm water provides a competitive advantage for cyanobacteria. These bacteria grow more quickly at higher temperatures than benign algae, which can trigger a feedback loop after a bloom. The algae sludge absorbs more sunlight as the blooms grow thicker, resulting in algal growth. Eutrophic cyanobacteria (rich in nutrients) often substitute for other forms of algal biomass [7].

Increased anthropogenic fertilizer emissions affect lakes and reservoirs' ecology, chemistry, and physical properties. The cyano-harmful algal bloom often increases as an enrichment response [20]. Cyanobacteria survive well in the reservoir in high retention periods, because they have slow growth rates, compared to other algal types [21]. Thus, while most harmful algal blooms occur in standing waters, they can grow in very slow-flowing rivers. In 1991, after dry weather in Australia, the largest recorded cyanobacterial bloom of *Anabaena circinalis* formed 1000 km from the Barwon Darling River [22]. The stability of the water column is critical because cyanobacteria can control their growth by generating gas vacuoles to maintain an optimum position for light harvesting [23]. This mechanism is ineffective in gusty and turbulent situations, as cells are distributed over all mixed layers, but thick cyanobacterial scums can accrue on the water surface and concentrate more. When the wind blown onto the lake shores of the lake, they significantly increase the risk of toxic exposure [24].

3 Conditions that Stimulate Harmful Algal Blooms in the Reservoir

The dynamics of harmful algal blooms depend upon the synergistic stimulus of different environmental aspects, including nutrients inflow, temperature, light, pH, N₂ fixation, and hydrodynamics of the reservoir [21, 25, 26]. The construction of large-scale reservoirs is an efficient way of using water supplies, generating electricity from hydropower, and reducing flood risk. The hydrodynamic state with less water velocity and a longer residence duration causes water quality problems [27]. In man-made reservoirs, eutrophication has gained significant attention because of its harmful impact on the aquatic ecosystem and animal health [28]. There is a greater probability that algal blooms will occur, mainly while these blooms are related to (toxic) cyanobacteria. Under natural conditions, phytoplankton and cyanobacteria usually occur in aquatic environments. Unique features may allow cyanobacteria to prevail. These features include cellular physiology (for example, gas vesicles in cells allow for buoyancy regulation), physiological response (use of nutrients and light), general morphology, cell structure, and cell size. Under the condition of particular water temperature and optimum light intensity, cyanobacteria predominate over other species [29].

The available light, weather conditions, appropriate water flow, and optimized temperature are physical factors leading to the proliferation of all phytoplankton



Fig. 5.3 Harmful algal bloom at Starvation Reservoir (USA) has been spotted, and the public is forbidden from swimming in this area (*Photo from Duchesne County Sheriff's Office*)

species (including cyanobacteria). Chemical factors include pH changes, the loading of nutrients (mostly in various nitrogen and phosphorus forms), and trace metals. An algal bloom has traditionally been closely associated with high nutrient levels in water bodies with low turbidity, permitting light transmission across the water bodies. However, the conditional factors include reservoir morphology, water circulation, viruses, grazing pressure from plant-eating fish, and microbial mechanisms [30].

More reports have been published on producing harmful species emerging in the areas not previously experiencing problems, such as the cyanobacteria in eutrophic lakes in Florida (Fig. 5.3) [31].

4 Reservoir Algal Bloom in Different Countries

In China, the algal bloom in the Three Gorges Reservoir is watched closely by the Chinese government and researchers. The algal blooms were observed in the Three Gorges Reservoir tributaries, such as Xiangxi, Shennong, and Daning, instead of the mainstream of the Yangtze River [32]. These algal blooms were formed by the proliferation of harmful algae, such as dinoflagellate *Prorocentrum donghaiense*, due to the enhanced phosphorus limitation and Biogenic silica (BSi) sedimentation, which has limited the primary production of diatoms [33]. A significant occurrence of harmful algal bloom in a reservoir in Inner Mongolia, China was also reported, which prevailed for 2 months with *Dinobryon* sp. (*chrysophyte*) species being the predominant algal species (abundance of 88,520 cells/mL) [34].

In the USA, Kislik et al. reported the seasonal and interannual heterogeneity of algal blooms in Copco and Iron Gate Reservoirs using the Sentinel-2 satellite imagery, and found algae appeared to be the highest in the spring and summer with the highest peaks observed in 2019 [35]. Two reservoirs in the Highland Lakes of central Texas were observed for the duration of an extended drought period from 2010 to 2015 [36], and the result showed that an increase in inorganic and organic nitrogen following a period of drought amplified the potential for harmful blooms of *Aphanizomenon*. Tábora-Sarmiento et al. analyzed the association of air, land, and water variables with the current distribution of toxic *Prymnesium parvum* bloom in reservoirs of the Brazos River and Colorado River, Texas, USA, and found that higher salinity and wetland deficiency facilitated the establishment of harmful algal bloom [37].

In other countries, cyanobacterial occurring in the Spanish reservoirs indicate that they are toxic and dominate a significant proportion (35-48%) during the study period [38]. The toxic cyanobacterial communities in Amphur Muang (Khon Kaen Province, Thailand) were investigated in four recreational reservoirs (Bueng Kaen Nakhon, Bueng Thung Sang, Bueng Nong Khot, and Bueng). *Microcystis* sp. and *Cylindrospermopsis* sp. were the dominant species to generate algal bloom in the water samples of Bueng Nong Khot and Bueng See Than, Thailand [39]. *Saxitoxins* were found in two samples (7%) obtained from two different Czech water reservoirs at concentrations from 0.03 to 0.04 µg/L [40]. Microcystis colonies in the Isahaya reservoir (Japan) were estimated to be 34.5 kg in the water and 8.4 kg in the surface sediment of the reservoir [41].

5 Methods Used to Control Harmful Algal Bloom in the Reservoirs

Sustainable management of algae aims at limiting nutrient inflow into the water body. Effective monitoring of critical parameters for water quality and algae measurements contributes to mitigating and lowering the risk of algae growth. Phytoplankton dynamics, such as chlorophyll A, phycocyanin, temperature, DO, pH, and turbidity, can be used to forecast harmful algae blooms. Bloom's evaluation and assessment help to select appropriate safety strategies. Algae control methods include aeration, chemical & biological additives, or ultrasound technology. The control of cyanobacteria's spread has become a major global challenge. All existing approaches have major advantages and disadvantages. Many control methods, for example, algaecides, are not environmentally safe. Moreover, some approaches, such as aeration, are very expensive.

5.1 Chemicals

Algae treated with a biocide is used to control the growth of algae. Chemical lanthanum, copper sulfate, hydrogen peroxide, copper chelate, and endothall are algaecides. Copper sulfate (CuSO₄) accumulates in reservoir sediments that may harm fish. Hydrogen peroxide can easily be applied; however, there are uncertain long-term effects on biota. In a laboratory study, chlorine effectively removed harmful algal blooms [42].

Advantages and Disadvantages of Chemical Control First, chemical control is efficient when handling the entire surface, and it can significantly reduce algae populations, potentially eliminate toxins, remove nutrients from the water column, and control the growth of harmful algal blooms. Algaecides are costly and have to be dosed regularly. They should be carefully used as they can break the algal cell and release toxins into the water. Blooms with high algal toxins may be harmful to fish and plants. Algaecides may have significant long-term effects on the environmental quality of the lake. Moreover, they are not suited for large water surfaces [42].

5.2 Aeration

Dissolved oxygen quality in the reservoir is essential. Oxygen tends to break down nutrients and rot plants in the water. Besides, microorganisms break up the silt at the bottom. Both aerobic and anaerobic bacteria contribute to decay [43]. Aerobic degradation needs a complete oxygen supply. When dissolved oxygen levels are at their maximum, aerobic bacteria break down, and anaerobic decomposition takes longer to complete. The final products are organic compounds that smell bad, such as alcohol and organic acids [43].

Air pumping or aeration is typically used to oxygenate the smaller reservoir areas. These may be involved in airlifts, hypolimnetic diffusers, downward-flow oxygen contractors, and onsite oxygen generators. The most efficient aeration is to oxygenate the area between the top and the bottom layers. If the region between the upper and lower layers remains oxygenated, enough iron will react to phosphorus available, and oxidized iron may form. However, the residue is not toxic to organisms. Aeration is an eco-friendly way to increase the amount of oxygen in the small reservoir. Aeration systems can help prevent the use of chemicals and develop a safe environment. However, aeration is better suited for smaller, concentrated treatment areas that achieve positive results [43]. Moreover, construction costs may be if underwater dam structures are needed to maintain cool water zones. However, underwater dam structures can incur annual maintenance and operating costs.

5.3 Mixing

Mixing circulating water in reservoirs can cause desertification. The method consists of mixing water to remove layers of draping. Epilimnion and metalimnion are typically used for algae control. It is intended to clean the surface water from iron, manganese, and anoxic odors, usually in the hypolimnion layer. This reduces conditions for the growth of algae in specific layers [44]. Mixing is one way to reduce epilimnion temperature (e.g., mix cool lower layer with warmer upper layer) and create fewer favorable conditions for the evolution and continuity of a harmful algal bloom. Hypolimnetic aeration increases water quality. Discharge of nutrient-rich water prevents the formation of harmful algal blooms. Mixing can increase turbidity and may influence beneficial phytoplankton to foster invertebrate and fish species [44].

Advantages and Disadvantages of Mixing Artificial mixing does not harm the atmosphere as organic ingredients. In deep reservoirs, it is usually more successful (mean depth >15 m). However, water circulation requires high wear and tear system maintenance. There is debate about the impact on overall cyanobacterium levels. Only surface layers are frequently affected by blending in the reservoir. Mixing sediments will potentially increase the nutrients that are available in large systems. This leads to further growth of algae in the short term. Nevertheless, the decrease in algal bloom can be accomplished over a long time [44].

5.4 Ultrasound

The destruction of many cyanobacteria and algae is proven by cavitation and ultrasonic disruption methods. Cavitation processes are combined with secondary treatment methods for toxins, such as superoxide radicals or ozone [45].

Advantages and Disadvantages of Ultrasonic Algae Control Ultrasound treatment of algae is a well-known technique that has been used for many years. It has been demonstrated to be beneficial for blue-green and green algae. Ultrasound is eco-friendly and safe for plants or fish. This can be used on small reservoirs. Ultrasound incorporated into real-time monitoring enables algal bloom prediction and algal bloom prevention. The whole surface of the reservoir may be covered. To achieve maximum efficiency, each space spot must be held for a minimum time [45].

6 Conclusion

The prominent literature findings have clearly shown that the ecosystem is resilient to harmful algal bloom with increasing anthropogenic activities and the continuously changing environment. There are many approaches to dealing with toxic blooms of algal in reservoirs. In larger reservoirs, the effectiveness of these methods decreases. Neither approach solves all problems individually. A combination of methods is probably needed in larger reservoirs. Comprehensive and focused work is required to track and manage harmful algal flora in the area of reservoir water.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and the Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- L. Stabili, M. Licciano, A. Giangrande, C. Caroppo, Filtration of the microalga Amphidinium carterae by the Polychaetes Sabella spallanzanii and Branchiomma luctuosum: A new tool for the control of harmful algal blooms? Microorganisms 10 (2022)
- C. Mora, T. McKenzie, I.M. Gaw, J.M. Dean, H. von Hammerstein, T.A. Knudson, R.O. Setter, C.Z. Smith, K.M. Webster, J.A. Patz, E.C. Franklin, Over half of known human pathogenic diseases can be aggravated by climate change. Nat. Clim. Chang. 12, 869–875 (2022)
- H.W. Paerl, V.J. Paul, Climate change: Links to global expansion of harmful cyanobacteria. Water Res. 46, 1349–1363 (2012)
- H.J. Kling, Bloom development and phytoplankton succession in Lake Winnipeg: A comparison of historical records with recent data. Aquat. Ecosyst. Health Manag. 14, 219–224 (2011)
- T. Zohary, A. Nishri, A. Sukenik, Present–absent: A chronicle of the dinoflagellate Peridinium gatunense from Lake Kinneret, in *Phytoplankton Responses to Human Impacts at Different Scales*, ed. by N. Salmaso, L. Naselli-Flores, L. Cerasino, G. Flaim, M. Tolotti, J. Padisák, (Springer, Dordrecht, 2012), pp. 161–174
- 6. J. Li, J. Zhang, W. Huang, F. Kong, Y. Li, M. Xi, Z. Zheng, Comparative bioavailability of ammonium, nitrate, nitrite and urea to typically harmful cyanobacterium Microcystis aeruginosa. Mar. Pollut. Bull. **110**, 93–98 (2016)
- J.M. O'Neil, T.W. Davis, M.A. Burford, C.J. Gobler, The rise of harmful cyanobacteria blooms: The potential roles of eutrophication and climate change. Harmful Algae 14, 313–334 (2012)
- B. Karlson, P. Andersen, L. Arneborg, A. Cembella, W. Eikrem, U. John, J.J. West, K. Klemm, J. Kobos, S. Lehtinen, N. Lundholm, H. Mazur-Marzec, L. Naustvoll, M. Poelman, P. Provoost, M. De Rijcke, S. Suikkanen, Harmful algal blooms and their effects in coastal seas of Northern Europe. Harmful Algae 102, 101989 (2021)
- 9. J.L. Briscoe, S.Y. Cho, I. Brener, Part-per-trillion level detection of microcystin-LR using a periodic nanostructure. IEEE Sens. J. **15**, 1366–1371 (2015)
- K.K. Schrader, A.M. Rimando, Off-flavors in aquaculture: An overview, in *Off-Flavors in Aquaculture*, ed. by A.M. Rimando, K.K. Schrader, (2003), pp. 1–12
- O.M. Skulberg, Microalgae as a source of bioactive molecules–experience from cyanophyte research. J. Appl. Phycol. 12, 341–348 (2000)

- 12. G. Speijers, G. Bakker, Toxins of cyanobacteria. Mol. Nutr. Food Res. 51, 7-60 (2007)
- Z. Svirčev, D. Lalić, G. Bojadžija Savić, N. Tokodi, D. Drobac Backović, L. Chen, J. Meriluoto, G.A. Codd, Global geographical and historical overview of cyanotoxin distribution and cyanobacterial poisonings. Arch. Toxicol. 93, 2429–2481 (2019)
- J. Wang, J. Wu, Occurrence and potential risks of harmful algal blooms in the East China Sea. Sci. Total Environ. 407, 4012–4021 (2009)
- E. Agathokleous, J. Peñuelas, R.A. Azevedo, M.C. Rillig, H. Sun, E.J. Calabrese, Low levels of contaminants stimulate harmful algal organisms and enrich their toxins. Environ. Sci. Technol. 56, 11991–12002 (2022)
- G.C. Pitcher, D.C. Louw, Harmful algal blooms of the Benguela eastern boundary upwelling system. Harmful Algae 102, 101898 (2021)
- S.Y. Wen, T.L. Sun, Z.Z. Wang, F. Li, D.M. Li, N. Liu, Spatial and temporal characteristics of harmful algal blooms in Zhejiang Province waters during 1933-2018. Mar. Biol. Res. 17, 545–553 (2021)
- E. Berdalet, L.E. Fleming, R. Gowen, K. Davidson, P. Hess, L.C. Backer, S.K. Moore, P. Hoagland, H. Enevoldsen, Marine harmful algal blooms, human health and wellbeing: Challenges and opportunities in the 21st century. J. Mar. Biol. Assoc. UK 96, 61–91 (2016)
- A.D. Persaud, A.M. Paterson, P.J. Dillon, J.G. Winter, M. Palmer, K.M. Somers, Forecasting cyanobacteria dominance in Canadian temperate lakes. J. Environ. Manag. 151, 343–352 (2015)
- P.R. Leavitt, C.S. Brock, C. Ebel, A. Patoine, Landscape-scale effects of urban nitrogen on a chain of freshwater lakes in Central North America. Limnol. Oceanogr. 51, 2262–2277 (2006)
- H.W. Paerl, Mitigating harmful cyanobacterial blooms in a human-and climatically-impacted world. Life 4, 988–1012 (2014)
- K. Fryirs, B. Chessman, I. Rutherfurd, Progress, problems and prospects in Australian river repair. Mar. Freshw. Res. 64, 642–654 (2013)
- T. Kutser, L. Metsamaa, A.G. Dekker, Influence of the vertical distribution of cyanobacteria in the water column on the remote sensing signal. Estuar. Coast. Shelf Sci. 78, 649–654 (2008)
- C.S. Reynolds, R.L. Oliver, A.E. Walsby, Cyanobacterial dominance: The role of buoyancy regulation in dynamic lake environments. N. Z. J. Mar. Freshw. Res. 21, 379–390 (1987)
- 25. H.W. Paerl, J. Huisman, Climate change: A catalyst for global expansion of harmful cyanobacterial blooms. Environ. Microbiol. Rep. 1, 27–37 (2009)
- N.-Y. Benayache, T. Nguyen-Quang, K. Hushchyna, K. McLellan, F.-Z. Afri-Mehennaoui, N. Bouaïcha, An overview of cyanobacteria harmful algal bloom (CyanoHAB) issues in freshwater ecosystems, in *Limnology—Some New Aspects of Inland Water Ecology*, (2019), pp. 1–25
- 27. J. Huang, Y. Zhang, G.B. Arhonditsis, J. Gao, Q. Chen, N. Wu, F. Dong, W. Shi, How successful are the restoration efforts of China's lakes and reservoirs? Environ. Int. **123**, 96–103 (2019)
- X.E. Yang, X. Wu, H.L. Hao, Z.L. He, Mechanisms and assessment of water eutrophication. J. Zhejiang Univ. Sci. B 9, 197–209 (2008)
- 29. H.F. Wang, Q.M. Shang, The combined effects of light intensity, temperature, and water potential on wall deposition in regulating hypocotyl elongation of Brassica rapa. PeerJ 8 (2020)
- C.J. Gobler, Climate change and harmful algal blooms: Insights and perspective. Harmful Algae 91, 101731 (2020)
- A.D. Chapman, C.L. Schelske, Recent appearance of Cylindrospermopsis (Cyanobacteria) in five hypereutrophic Florida Lakes1. J. Phycol. 33, 191–195 (1997)
- 32. Z. Yang, C. Wei, D. Liu, Q. Lin, Y. Huang, C. Wang, D. Ji, J. Ma, H. Yang, The influence of hydraulic characteristics on algal bloom in Three Gorges Reservoir, China: A combination of cultural experiments and field monitoring. Water Res. 211, 118030 (2022)
- 33. X. Ran, A.F. Bouwman, Z. Yu, J. Liu, Implications of eutrophication for biogeochemical processes in the Three Gorges Reservoir, China. Reg. Environ. Chang. **19**, 55–63 (2019)

- 34. Y. Zhao, J. Yu, M. Su, W. An, M. Yang, A fishy odor episode in a North China reservoir: Occurrence, origin, and possible odor causing compounds. J. Environ. Sci. 25, 2361–2366 (2013)
- 35. C. Kislik, I. Dronova, T.E. Grantham, M. Kelly, Mapping algal bloom dynamics in small reservoirs using Sentinel-2 imagery in Google Earth Engine. Ecol. Indic. 140, 109041 (2022)
- T.E. Gámez, L. Benton, S.R. Manning, Observations of two reservoirs during a drought in central Texas, USA: Strategies for detecting harmful algal blooms. Ecol. Indic. 104, 588–593 (2019)
- 37. S. Tábora-Sarmiento, R. Patiño, C. Portillo-Quintero, C. Coldren, Air, land, and water variables associated with the first appearance and current spatial distribution of toxic Prymnesium parvum blooms in reservoirs of the Southern Great Plains, USA. Sci. Total Environ. 836, 155567 (2022)
- A. Quesada, D. Sanchis, D.G.J.L. Carrasco, Cyanobacteria in Spanish reservoirs. How frequently are they toxic? Limnetica (2004)
- T. Somdee, T. Kaewsan, A. Somdee, Monitoring toxic cyanobacteria and cyanotoxins (microcystins and cylindrospermopsins) in four recreational reservoirs (Khon Kaen, Thailand). Environ. Monit. Assess. 185, 9521–9529 (2013)
- 40. D. Jančula, L. Straková, J. Sadílek, B. Maršálek, P. Babica, Survey of cyanobacterial toxins in Czech water reservoirs—the first observation of neurotoxic saxitoxins. Environ. Sci. Pollut. Res. 21, 8006–8015 (2014)
- 41. A. Umehara, T. Komorita, T. Takahashi, H. Tsutsumi, Estimation of production and sedimentation of cyanobacterial toxins (microcystin) based on nutrient budgets in the reservoir of Isahaya Bay, Japan. Ecotoxicol. Environ. Saf. 183, 109477 (2019)
- 42. J. Fan, L. Ho, P. Hobson, J. Brookes, Evaluating the effectiveness of copper sulphate, chlorine, potassium permanganate, hydrogen peroxide and ozone on cyanobacterial cell integrity. Water Res. 47, 5153–5164 (2013)
- S. Clark, In the shadow of the fourth circuit: Ohio Valley environmental coalition v. United States Army Corps of Engineers. Ecol. Law Q., 143–149 (2008)
- 44. D.L. Roelke, B.W. Brooks, J.P. Grover, D. Kalisek, B.L. Harris, Approaches to Golden Algae Control: In-Lake Mesocosm Experiments. 2012
- 45. Z. Wu, H. Shen, B. Ondruschka, Y. Zhang, W. Wang, D.H. Bremner, Removal of blue-green algae using the hybrid method of hydrodynamic cavitation and ozonation. J. Hazard. Mater. 235–236, 152–158 (2012)

Chapter 6 Microplastics Pollution in the Reservoir: Occurrence, Extraction, and Characterization



Marriya Sultan, Suman Thodhal Yoganandham, and De-Sheng Pei 💿

Abstract Modern life's dependence on plastic materials has led to increasing microplastics (MPs) contamination in the reservoir. The increasing abundance of MPs (usually of a size less than 5 mm) has instigated the scientific and management communities regarding the monitoring of MPs in freshwater bodies. Worldwide natural freshwater bodies, such as lakes, rivers, and reservoirs, are widely documented with MPs, which interact with the organism and affect their natural habitat. This chapter highlights the relevant aspects of MPs in the reservoir system, such as their sources and abundance in major reservoirs across the world. Besides, different methodologies for MPs sampling, pretreatment, and characterization are also discussed.

Keywords Microplastics · Reservoir · Sampling · Extraction · Characterization

1 Introduction

Plastic is regarded as the most useful synthetic material developed by humans with wide application and versatile properties, such as electrical insulation, ultralightweight, tensile strength, durability, and anti-corrosive nature [1]. As a result, plastic production has steadily increased since 1950, with global production of about 368 million tons in 2019 [2]. Its improper use or disposal has caused severe environmental problems [3], instigating significant environmental management challenges. Annually, more than 8 million tons of plastic waste goes into the fresh and marine waters around the world [4].

Microplastics (MPs) are usually defined as tiny plastic particles with a size <5 mm, either manufactured intentionally for industrial or domestic use, such as exfoliating facial scrubs, toothpaste, and resin used in the plastic industry (primary MPs) that enters the environment directly or can be formed by fragmentation of larger plastic products by weathering or mechanical abrasion (secondary MPs) [5–8]. MPs consist of synthetic polymers, manufactured by polymerization of different monomers, which include a variety of materials like polyvinyl alcohol (PA), polyamide (PA), polyethylene (PE), nylons, polyethylene terephthalate (PET), polypropylene (PP), polystyrene (PS), and polyvinyl chloride (PVC) [9, 10] with different

ratios of additives forming a variety of combinations. Due to their large surface area and hydrophobic nature, MPs adsorb persistent organic pollutants (such as OCPs, PAHs, PCBs, and PBDEs) and heavy metals, and carry them to other terrestrial and aquatic ecosystems [11]. Besides, because of their persistent nature and non-degradability, they pose threats to different aquatic and terrestrial organisms. Non-degradability of plastic granules affects various biological responses like cell apoptosis, necrosis, and genotoxicity. Further, it can also lead to several sub-lethal effects, such as growth inhibition, oxidative damage, and behavioral defects [12– 17].

Recently, worldwide cognizance about MPs pollution is attaining momentum with important declarations from prominent world economies, such as the approval of a bill banning plastic microbeads from the varied range of consumer products in the USA and the implementation of a ban on MPs in cosmetic items in France and the United Kingdom. Similarly, some other countries are currently preparing decisions for the same cause soon [18].

Reservoirs are regarded as significant resources of freshwater as they are the main source of drinking, food production (via irrigation of agricultural areas, aquaculture, and fisheries), energy provision (hydropower dams), and the regulation of drought and flood [19]. Regardless of all the services reservoirs provide, they have not acquired the appropriate attention for monitoring MPs in reservoirs. While many research studies have focused on the evaluation of MPs in freshwater systems, such as streams, lakes, ponds, and rivers. Studies on the occurrence and distribution of MPs in reservoirs have emerged recently during the last 6 years [20]. Therefore, more research about the fate and distribution of MPs in reservoirs is required in the coming years to understand the potential risks of MPs contamination.

2 Sources of MPs in Reservoirs

The sources of MPs are diverse and show discrepancies geographically, depending upon the land-use practices. MPs in freshwater bodies, such as rivers, lakes, and reservoirs, are released via runoffs from landfills, agricultural waste discharge, and wastewater solids or effluents [21, 22]. The plastic waste from industrial sources usually ends up being dumped in landfill or open waste dumping sites in solid form or as released in surface water as industrial effluent. In developed countries, plastic waste is mostly disposed of in landfills [23]. In Europe, about 1000–4000 MPs/kg dry mass of sludge was observed in landfill and agricultural areas [24]. In agricultural zones, plastic mulch films, plastic-coated fertilizers, municipal waste, biosolids, atmospheric deposition, and irrigation runoff are the main sources of MPs in freshwater bodies, such as reservoirs [25].

Urbanization also intensifies the spatiotemporal distribution of MPs in reservoirs [26], as it is observed that reservoirs located in urban areas contained a higher abundance of MPs, compared to the others. Consequently, in highly populated regions, the contamination of MPs is more prominent, compared to other regions

of the world [20]. However, in remote regions, the primary source of MPs is rainfall and atmospheric deposition. Besides, aquaculture and tourism are also regarded as prominent sources of MPs in rural and urban reservoirs [4].

3 Occurrence and Distribution of MPs in Reservoirs

MPs occurrence is distributed across wide geographic coverage with detections in different reservoirs from Asia, North America, and Europe [20]. However, the abundance of MPs in different reservoir varies widely across different regions due to differences in population statistics, economic development, and social status [27]. Table 6.1 presents the abundance of MPs in different reservoirs around the world. Ndlovu [28] reported elevated levels of MPs in Swedish Reservoirs Vombsjön and Bolmen with a mean abundance of 155.56 ± 91.6 particles/L and 80.56 ± 59.66 particles/L, respectively. Among the reservoirs in the USA, the abundance levels of MPs in Brownlee Reservoir, one of the biggest reservoirs in the western USA, were reported as 13.7 particles/m³ [29], whereas in Lake Mead Reservoir (USA), comparatively low levels of MPs were observed with concentration ranged between 0.44 and 9.70 particles/m³ [30]. Recent studies focusing on the spatial distribution of MPs in some main reservoirs of Asia also reported higher abundances of MPs in surface waters. For example, Turhan [31] reported the MPs abundance levels in the range between 106.6 and 200 particles/m³ in Sürgü Dam, Turkey. Similarly, in the Aras Dam Reservoir (Iran), the MPs levels in surface water from 19 locations ranged from 1 to 43 items/m³ (mean 12.8) with MPs of size 0.1-5 mm being more dominant [32]. Moreover, the MPs abundance in the Jatiluhur Reservoir (Indonesia) ranged between 0.71×10^4 and 4.59×10^5 particles/km² with Polyethylene (PE) being the common type of polymer found in surface water samples. Recently, Nousheen et al. [33] also assessed MPs occurrence in the Rawal Dam Reservoir (Pakistan) and observed the mean levels of 2.8 ± 1.44 particles/L (using the sieve method) and 0.025 ± 0.024 particles/L (using trawl method).

In China, the Danjiangkou Reservoir and Three Gorges Reservoir are the main reservoirs that reported the MPs distribution. MPs characteristics in these two reservoirs are comparable to the other freshwater bodies with fragments and fiber (size of most <1 mm) being more dominant, compared to the others [4]. Zhang et al. [34] assessed the MPs occurrence in the TGR region where the abundance was reported between 34.1×10^5 and 136.1×10^5 particles/km² (Yangtze River) and 0.55×10^5 – 342×10^5 particles km⁻² (Xiangxi Bay, the TGR), respectively. Two years later, Zhang et al. [35] also studied the Xiangxi River in the TGR region and reported the abundance of MPs in a range between 550 and 14,580 particles m⁻².

Di and Wang [36] assessed the occurrence and distribution of MPs alongside the Yangtze River of TGR and reported the overall abundance of MPs as 1597–12, 611 n/m³ in surface water with size 0.5–5 mm being more dominant. Tan et al. [39] reported the MPs in surface water of Feilaixia Reservoir with mean abundance levels

Reservoir	Abundance/levels in water	Size and polymer	References				
China							
Three Gorges Reservoir (TGR), Yangtze River	$34.1 \times 10^5 \text{ to } 136.1 \times 10^5$ particles/km ²		[34]				
Yangtze River (TGR)	$1597-12,611$ particles m^{-3}		[36]				
Xiangxi Bay, the TGR	$\begin{array}{c} 0.55 \times 10^{5} 342 \times 10^{5} \\ \text{particles } \text{km}^{-2} \end{array}$		[35]				
Xiangxi River (the TGR)			[37]				
Yangtze Estuary	500 n/m ³ to 10,200 n/m ³		[38]				
Feilaixia Reservoir	0.56 ± 0.45 items/m ³	PP (52.31%) PE (27.39%)	[39]				
Danjiangkou Reservoir	530–24,798 items/m ³ (7205 items/m ³)	PA(24.8%) PE (24%) PP (17.1%)	[40]				
Jiayan Reservoir	1.10×10^4 to 6.17×10^4 items/m ³	<300 µm, PE	[41]				
Other regions of the world							
Vombsjön Reservoir, Sweden	155.56 ± 91.6 particles/L		[28]				
Bolmen Reservoir, Sweden	80.56 ± 59.66 particles/L		[28]				
Brownlee Reservoir, USA	13.7 particles/m ³		[29]				
Lake Mead Reservoir, US	0.44–9.70 particles/m ³	355–1000 µm (73.1%) 1000–5600 µm (26.5%)	[30]				
Jatiluhur Reservoir, Indonesia	$\begin{array}{c} 0.7\times10^44.5\times10^5 \text{ par-}\\ \text{ticles/km}^2 \end{array}$	PE	[42]				
Sürgü Dam, Turkey	106.6–200 particles/m ³	Size:1–2 mm PP and PE	[31]				
Rawal Dam, Pakistan	$\begin{array}{l} 2.8 \pm 1.44 \text{ particles/L} \\ (\text{sieve method}) \\ 0.02 \pm 0.024 \text{ particles/L} \\ (\text{trawl method}) \end{array}$	Size: 0.1–0.9 mm PP (40–74%)	[33]				
Navua Irrigation Dam, Fiji	2.9 ± 0.4 particles/L	Size: 0.5–0.9 mm and 1.0–1.4 mm	[43]				
Aras River, Iran	1–43 items/m ³ (mean: 12.8 items/m ³)	Size: 0.1 to 5 mm PE (36.6%)	[32]				

Table 6.1 Abundance levels of MPs in different reservoirs around the world

of 0.56 \pm 0.45 items/m³. Recently, Lin et al. [40] assessed the MPs occurrence and distribution in the Danjiangkou Reservoir and reported very high levels with an abundance range of 530–24,798 items/m³ (7205 items/m³). Besides, Niu et al. [41] evaluated the MPs abundance in the Jiayan Reservoir and reported the levels within the range of 1.10 × 104 to 6.17 × 104 items/m³ with MPs particles of size<300 µm as more dominant among others.

4 MPs Sampling and Analysis Techniques

The collection of MPs from water or sediment and their extraction from organic and mineral material is quite challenging. The choice of sampling and preservation techniques generally relies on the research aim, economic affordability of the methods, sample characteristics, and the study matrix [44]. The quantification and identification techniques for MPs are not comprehensively standardized, which could result in a significant difference in output results from different studies carried out in the same place and time [45]. The detection of MPs in the reservoir includes three basic steps, such as sample collection, sample pretreatment, and characterization of MPs [46].

4.1 Sample Collection

Based on the type of compartment that is required to be examined (water, sediment, or biota), different sampling strategies could be adopted. Reduced volume and bulk sampling are the two main methods for collecting surface water [47]. Bulk sampling does not result in a reduction of water volume unlike the reduced volume sampling technique, in which on-site filtration by nets or sieving is carried out [48]. For a volume-reduced sampling of surface water, nets are mainly used that directly filter the debris from the large volume of samples [49]. Manta trawl and neuston plankton net are usually used for collecting samples from surface water. The size and type of MPs that are collected while sampling largely depends upon the mesh size of the collecting tool. Such smaller-sized mesh tools for the collection of MPs could result in the collection of large amounts of fibers, compared to other types of MPs, and could also result in over or underestimation of the abundance of MPs [46].

An alternative method of collecting the sample is through a pump or grab sampler. A pump sampler consists of a motor containing an inline filter, and water is pumped manually into the sampler. In grab sampling, a bucket is used to collect and sieve the water in the field [46, 50]. The manta trawling and pump sampling methods give different results in the abundance, shape, and size of MPs as observed in a study, where both of these methods were employed for the MPs sampling of Lake Tollense, Germany. It was recommended to use pump sampling instead of manta trawl, which does not retain the fibers and small-sized MPs sufficiently. However, pump samplers could filter large volumes of water generating reliable results [51]. Barrows et al. [52] observed that grab samplers collected MPs three times more in magnitude than the conventional neuston net sampling technique. However, for effective monitoring, a combination of both volume-reduced (net-based) and bulk sampling (pump and grab-based) approaches is reliable depending upon the objective of the study.

The sampling tools for the sediment matrices are selected for collecting places and objectives. For sediment matrices, manual grab samplers, such as stainless steel spoons and hand spades, are used for sampling from littoral zones of reservoirs. However, for deep sediments, different types of grab and core samplers, such as box corer [53] or Van Veen grabber [54] are used.

4.2 Sample Pre-treatment

After the collection of MPs samples from water or sediments, extraction and purification is carried out. MPs particles are separated from the matrix as sometimes the plastic is confused with biological material (algae fragments). Therefore, a simple digestion method is required to reduce the amount of organic matter without affecting the chemical and chemical integrity of the MPs [46]. Hydrogen peroxide is a vigorous oxidizing agent mainly used to digest organic matter without affecting the integrity of MPs [36]. Samples are generally preserved with methyl aldehyde (5%) [34], fixed in formalin (2.5%) [38], or could be submerged in 40% ethanol (EtOH) and kept at 4 °C before analysis [55].

4.3 Characterization of MPs

MPs characterization is of two types including physical characterization and chemical characterization. Physical characterization involves the classification of MPs according to their sizes, shapes, and color; whereas chemical characterization refers to the composition of MPs [56]. For the physical characterization, a stereomicroscope is mostly used to gauge the size and count the abundance of MPs. However, the visual identification of MPs relies on the operator as the stereomicroscope has low magnification, and it is difficult to distinguish different fibers. Therefore, a higher magnification stereomicroscope should be used to confirm the nature of MPs [57].

For chemical characterization of MPs, different technologies have been used, such as gas chromatography coupled to mass spectrometry (GC-MS) [58], liquid chromatography (LC) [59], Raman spectroscopy [60], energy-dispersive X-ray spectroscopy (EDS), scanning electron microscope (SEM) [61], attenuated total reflection-Fourier transform infrared (ATR-FTIR) spectroscopy [62]. Moreover, in many research studies, two to three multiple technologies have been used to identify MPs, such as the stereomicroscope, ATR-FTIR, SEM, μ -FTIR, and EDS. However, the most commonly used technologies for MPs analysis are μ -FTIR spectrophotometer, μ -Raman, and SEM [61].

4.3.1 FTIR Spectroscopy

FTIR spectroscopy provides spectral ranges for identifying plastics from other organic and inorganic materials. The MPs particle is exposed to infrared radiation, which forms a spectrum with distinguishing peaks corresponding to definite chemical bonds between atoms. To ascertain the sample composition, FTIR spectroscopy obtained polymer spectrum is identified by the well-established polymer spectrum library [63]. FTIR spectroscopy enables spectra to be collected in attenuated total reflectance (ATR) or transmission modes. ATR-FTIR and micro-FTIR are the common FTIR techniques used to characterize MPs. ATR is a unique sampling technique with infrared spectroscopy, which directly enhances the study of the samples in liquid and solid states without further preparation [56]. Micro-FTIR has become increasingly popular, because it characterizes the sample size up to 20 μ m. Micro-FT-IR analysis of plastics can be conducted in either transmission or reflectance mode. To obtain high-quality spectral data, infrared-transparent substrates are required in transmission mode [64].

4.3.2 Raman Spectroscopy

Raman spectroscopy is a vibrational spectroscopic technique, which produces a vibrational spectrum by the principle of inelastic stretching formed by the molecular vibrations of the MPs [37]. The Raman spectral data resembles a chemical structure with a fingerprint that allows the identification of components in the sample. The micro Raman combined with Raman spectroscopy is used to detect spatial resolution below 1 μ m. In this technique, the studied samples are irradiated with laser light between the frequency of visible, infrared, or ultraviolet ranges [36]. It displays enhanced spatial resolutions (up to 1 μ m) as compared to FTIR [65]. However, Raman spectroscopy is susceptible to fluorescence intrusion, caused by organic, inorganic, or microbiological entities in samples. Therefore, samples must be carefully purified before the analysis [59].

4.3.3 Scanning Electron Microscopy (SEM)

The SEM yields MPs images by scanning their surface with a concentrated beam of electrons, which characterizes the surface morphology of MPs [66]. Additionally, SEM-energy-dispersive X-ray spectroscopy (SEM-EDS) and environmental scanning electron microscopy-EDS (ESEM-EDS) are also used for identification of the elemental composition based on reflection and diffraction of radiations emanated from the surface of MPs surfaces alongside the characterization of surface morphology of MPs [22, 56].

5 Conclusion

With the increasing plastic utility, the discharge, distributions, and abundance of MPs in freshwater systems have increased, thereby increasing the health risks for the associated fauna. Owing to their health risks, MPs are widely documented in freshwater rivers and lakes but fewer studies have examined the MPs pollution in the reservoir system globally. Reservoirs, being an important source of water for various human activities, play a crucial role in maintaining the socio-economic wellbeing of the nations. Therefore, it is vital to assess the contamination of reservoirs in different countries. Moreover, the sampling and analytical techniques employed for MPs detection in reservoirs need to be standardized, because they vary dramatically among the published studies. Further, extensive research is required to develop an understanding of using appropriate sampling and analytical techniques for MPs detection in reservoirs.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- C. Jia, P. Das, I. Kim, Y.-J. Yoon, C.Y. Tay, J.-M. Lee, Applications, treatments, and reuse of plastics from electrical and electronic equipment. J. Ind. Eng. Chem. (2022)
- Y. Chen, B. Gao, D. Xu, K. Sun, Y. Li, Catchment-wide flooding significantly altered microplastics organization in the hydro-fluctuation belt of the reservoir. iScience, 104401 (2022)
- C.M. Rochman, M.A. Browne, B.S. Halpern, B.T. Hentschel, E. Hoh, H.K. Karapanagioti, L.M. Rios-Mendoza, H. Takada, S. Teh, R.C. Thompson, Classify plastic waste as hazardous. Nature 494, 169–171 (2013)
- 4. H. Zhao, Y. Zhou, Y. Han, Y. Sun, X. Ren, Z. Zhang, Q. Wang, Pollution status of microplastics in the freshwater environment of China: A mini review. Water Emerg. Contam. Nanoplastics 1, 5 (2022)
- 5. A.L. Andrady, Microplastics in the marine environment. Mar. Pollut. Bull. 62, 1596–1605 (2011)
- M. Wagner, C. Scherer, D. Alvarez-Muñoz, N. Brennholt, X. Bourrain, S. Buchinger, E. Fries, C. Grosbois, J. Klasmeier, T. Marti, Microplastics in freshwater ecosystems: What we know and what we need to know. Environ. Sci. Eur. 26, 12 (2014)
- H.S. Auta, C. Emenike, S. Fauziah, Distribution and importance of microplastics in the marine environment: A review of the sources, fate, effects, and potential solutions. Environ. Int. 102, 165–176 (2017)
- M. Cole, P. Lindeque, C. Halsband, T.S. Galloway, Microplastics as contaminants in the marine environment: A review. Mar. Pollut. Bull. 62, 2588–2597 (2011)

- C.G. Avio, S. Gorbi, M. Milan, M. Benedetti, D. Fattorini, G. d'Errico, M. Pauletto, L. Bargelloni, F. Regoli, Pollutants bioavailability and toxicological risk from microplastics to marine mussels. Environ. Pollut. **198**, 211–222 (2015)
- 10. S.A. Carr, J. Liu, A.G. Tesoro, Transport and fate of microplastic particles in wastewater treatment plants. Water Res. **91**, 174–182 (2016)
- A.W. Verla, C.E. Enyoh, E.N. Verla, K.O. Nwarnorh, Microplastic-toxic chemical interaction: A review study on quantified levels, mechanism and implication. SN Appl. Sci. 1, 1–30 (2019)
- L. Hermabessiere, A. Dehaut, I. Paul-Pont, C. Lacroix, R. Jezequel, P. Soudant, G. Duflos, Occurrence and effects of plastic additives on marine environments and organisms: A review. Chemosphere 182, 781–793 (2017)
- 13. T. Cedervall, L.-A. Hansson, M. Lard, B. Frohm, S. Linse, Food chain transport of nanoparticles affects behaviour and fat metabolism in fish. PLoS One 7, e32254 (2012)
- S. Kashiwada, Distribution of nanoparticles in the see-through medaka (Oryzias latipes). Environ. Health Perspect. 114, 1697 (2006)
- M.A. Browne, S.J. Niven, T.S. Galloway, S.J. Rowland, R.C. Thompson, Microplastic moves pollutants and additives to worms, reducing functions linked to health and biodiversity. Curr. Biol. 23, 2388–2392 (2013)
- V. Stock, L. Böhmert, E. Lisicki, R. Block, J. Cara-Carmona, L.K. Pack, R. Selb, D. Lichtenstein, L. Voss, C.J. Henderson, Uptake and effects of orally ingested polystyrene microplastic particles in vitro and in vivo. Arch. Toxicol. 93, 1817–1833 (2019)
- J.C. Prata, J.P. da Costa, I. Lopes, A.C. Duarte, T. Rocha-Santos, Environmental exposure to microplastics: An overview on possible human health effects. Sci. Total Environ. **702**, 134455 (2020)
- B. Jovanović, Ingestion of microplastics by fish and its potential consequences from a physical perspective. Integr. Environ. Assess. Manag. 13, 510–515 (2017)
- 19. L.T. Ho, P.L. Goethals, Opportunities and challenges for the sustainability of lakes and reservoirs in relation to the sustainable development goals (SDGs). Water **11**, 1462 (2019)
- Z. Guo, W.J. Boeing, Y. Xu, E. Borgomeo, S.A. Mason, Y.-G. Zhu, Global meta-analysis of microplastic contamination in reservoirs with a novel framework. Water Res. 207, 117828 (2021)
- Y. Chae, Y.-J. An, Current research trends on plastic pollution and ecological impacts on the soil ecosystem: A review. Environ. Pollut. 240, 387–395 (2018)
- M. Eriksen, S. Mason, S. Wilson, C. Box, A. Zellers, W. Edwards, H. Farley, S. Amato, Microplastic pollution in the surface waters of the Laurentian Great Lakes. Mar. Pollut. Bull. 77, 177–182 (2013)
- R.C. Hale, M.E. Seeley, M.J. La Guardia, L. Mai, E.Y. Zeng, A global perspective on microplastics. J. Geophys. Res. Oceans 125, e2018JC014719 (2020)
- S. Karbalaei, P. Hanachi, T.R. Walker, M. Cole, Occurrence, sources, human health impacts and mitigation of microplastic pollution. Environ. Sci. Pollut. Res. 25, 36046–36063 (2018)
- 25. V.S. Koutnik, J. Leonard, S. Alkidim, F. DePrima, S. Ravi, E. Hoek, S.K. Mohanty, Distribution of microplastics in soil and freshwater environments: Global analysis and framework for transport modeling. Environ. Pollut., 116552 (2021)
- 26. M.B. Alfonso, F. Scordo, C. Seitz, G.M.M. Manstretta, A.C. Ronda, A.H. Arias, J.P. Tomba, L.I. Silva, G.M.E. Perillo, M.C. Piccolo, First evidence of microplastics in nine lakes across Patagonia (South America). Sci. Total Environ. **733**, 139385 (2020)
- 27. H. Chen, Y. Qin, H. Huang, W. Xu, A regional difference analysis of microplastic pollution in global freshwater bodies based on a regression model. Water **12**, 1889 (2020)
- 28. O. Ndlovu, Microplastic Pollution in Swedish Freshwater Lakes: A Comparative Field Study of Microplastic Contamination in Two Drinking Water Reservoirs, Lake Vombsjön and Lake Bolmen 2018
- K.J. Kapp, E. Yeatman, Microplastic hotspots in the Snake and lower Columbia rivers: A journey from the Greater Yellowstone Ecosystem to the Pacific Ocean. Environ. Pollut. 241, 1082–1090 (2018)

- 30. A.K. Baldwin, A.R. Spanjer, M.R. Rosen, T. Thom, Microplastics in Lake Mead national recreation area, USA: Occurrence and biological uptake. PLoS One **15**, e0228896 (2020)
- D.Ö. Turhan, Evaluation of microplastics in the surface water, sediment and fish of Sürgü Dam Reservoir (Malatya) in Turkey. Turk. J. Fish. Aquat. Sci. 22 (2021)
- 32. A.H. Vayghan, M. Rasta, M. Zakeri, F.J. Kelly, Spatial distribution of microplastics pollution in sediments and surface waters of the Aras River and reservoir: An international river in Northwestern Iran. Sci. Total Environ. 843, 156894 (2022)
- 33. R. Nousheen, I. Hashmi, D. Rittschof, A. Capper, Comprehensive analysis of spatial distribution of microplastics in Rawal Lake, Pakistan using trawl net and sieve sampling methods. Chemosphere **308**, 136111 (2022)
- 34. K. Zhang, W. Gong, J. Lv, X. Xiong, C. Wu, Accumulation of floating microplastics behind the Three Gorges Dam. Environ. Pollut. 204, 117–123 (2015)
- 35. K. Zhang, X. Xiong, H. Hu, C. Wu, Y. Bi, Y. Wu, B. Zhou, P.K. Lam, J. Liu, Occurrence and characteristics of microplastic pollution in Xiangxi Bay of Three Gorges Reservoir, China. Environ. Sci. Technol. 51, 3794–3801 (2017)
- M. Di, J. Wang, Microplastics in surface waters and sediments of the Three Gorges Reservoir, China. Sci. Total Environ. 616, 1620–1627 (2018)
- K. Zhang, X. Chen, X. Xiong, Y. Ruan, H. Zhou, C. Wu, P.K. Lam, The hydro-fluctuation belt of the Three Gorges Reservoir: Source or sink of microplastics in the water? Environ. Pollut. 248, 279–285 (2019)
- S. Zhao, L. Zhu, T. Wang, D. Li, Suspended microplastics in the surface water of the Yangtze estuary system, China: First observations on occurrence, distribution. Mar. Pollut. Bull. 86, 562–568 (2014)
- 39. X. Tan, X. Yu, L. Cai, J. Wang, J. Peng, Microplastics and associated PAHs in surface water from the Feilaixia reservoir in the Beijiang River, China. Chemosphere 221, 834–840 (2019)
- 40. L. Lin, X. Pan, S. Zhang, D. Li, W. Zhai, Z. Wang, J. Tao, C. Mi, Q. Li, J.C. Crittenden, Distribution and source of microplastics in China's second largest reservoir-Danjiangkou reservoir. J. Environ. Sci. 102, 74–84 (2021)
- 41. J. Niu, B. Gao, W. Wu, W. Peng, D. Xu, Occurrence, stability and source identification of small size microplastics in the Jiayan Reservoir, China. Sci. Total Environ. **807**, 150832 (2022)
- A.H. Ramadan, E. Sembiring, Occurrence of microplastic in surface water of Jatiluhur reservoir, in E3S Web of Conferences, EDP Sciences, (2020), p. 07004
- J. Dehm, M. Volau, E. Ledua, C. Hewavitharane, Occurrence of microplastics within a freshwater aquaculture system in the Pacific Islands, Viti Levu, Fiji. Environ. Monit. Assess. 194, 1–12 (2022)
- 44. A. Lusher, I.L.N. Bråte, R. Hurley, K. Iversen, M. Olsen, Testing of Methodology for Measuring Microplastics in Blue Mussels (Mytilus spp.) and Sediments, and Recommendations for Future Monitoring of Microplastics (R & D-project), (2017)
- D. Elkhatib, V. Oyanedel-Craver, A critical review of extraction and identification methods of microplastics in wastewater and drinking water. Environ. Sci. Technol. 54, 7037–7049 (2020)
- 46. N. Razeghi, A.H. Hamidian, C. Wu, Y. Zhang, M. Yang, Microplastic sampling techniques in freshwaters and sediments: A review. Environ. Chem. Lett. **19**, 4225–4252 (2021)
- 47. L. Fok, T.W.L. Lam, H.-X. Li, X.-R. Xu, A meta-analysis of methodologies adopted by microplastic studies in China. Sci. Total Environ. 718, 135371 (2020)
- M. Tamminga, S.-C. Stoewer, E.K. Fischer, On the representativeness of pump water samples versus manta sampling in microplastic analysis. Environ. Pollut. 254, 112970 (2019)
- V. Hidalgo-Ruz, L. Gutow, R.C. Thompson, M. Thiel, Microplastics in the marine environment: A review of the methods used for identification and quantification. Environ. Sci. Technol. 46, 3060–3075 (2012)
- 50. R.A. Branch, Z. Yang, B. Maurer, L.M. Miller, L. Garavelli, Riverine Plastic Pollution: Sampling and Analysis Methods, (2022)

- M. Tamminga, E. Hengstmann, A.-K. Deuke, E.K. Fischer, Microplastic concentrations, characteristics, and fluxes in water bodies of the Tollense catchment, Germany, with regard to different sampling systems. Environ. Sci. Pollut. Res. 29, 11345–11358 (2022)
- A.P.W. Barrows, C.A. Neumann, M.L. Berger, S.D. Shaw, Grab vs. neuston tow net: A microplastic sampling performance comparison and possible advances in the field. Anal. Methods 9, 1446–1453 (2017)
- 53. A. Vianello, A. Boldrin, P. Guerriero, V. Moschino, R. Rella, A. Sturaro, L. Da Ros, Microplastic particles in sediments of Lagoon of Venice, Italy: First observations on occurrence, spatial patterns and identification. Estuar. Coast. Shelf Sci. 130, 54–61 (2013)
- 54. L. Mai, L.-J. Bao, L. Shi, C.S. Wong, E.Y. Zeng, A review of methods for measuring microplastics in aquatic environments. Environ. Sci. Pollut. Res. 25, 11319–11332 (2018)
- T. Mani, P. Burkhardt-Holm, Seasonal microplastics variation in nival and pluvial stretches of the Rhine River–from the Swiss catchment towards the North Sea. Sci. Total Environ. 707, 135579 (2020)
- 56. J. Sun, X. Dai, Q. Wang, M.C. van Loosdrecht, B.-J. Ni, Microplastics in wastewater treatment plants: Detection, occurrence and removal. Water Res. 152, 21–37 (2019)
- 57. K. Magnusson, F. Norén, Screening of Microplastic Particles in and Down-Stream a Wastewater Treatment Plant, 2014
- E. Dümichen, P. Eisentraut, M. Celina, U. Braun, Automated thermal extraction-desorption gas chromatography mass spectrometry: A multifunctional tool for comprehensive characterization of polymers and their degradation products. J. Chromatogr. A 1592, 133–142 (2019)
- 59. A.M. Elert, R. Becker, E. Duemichen, P. Eisentraut, J. Falkenhagen, H. Sturm, U. Braun, Comparison of different methods for MP detection: What can we learn from them, and why asking the right question before measurements matters? Environ. Pollut. 231, 1256–1264 (2017)
- C.F. Araujo, M.M. Nolasco, A.M. Ribeiro, P.J. Ribeiro-Claro, Identification of microplastics using Raman spectroscopy: Latest developments and future prospects. Water Res. 142, 426–440 (2018)
- P. Kumar, Y. Inamura, P.N. Bao, A. Abeynayaka, R. Dasgupta, H.D. Abeynayaka, Microplastics in freshwater environment in Asia: A systematic scientific review. Water 14, 1737 (2022)
- 62. A. Käppler, M. Fischer, B.M. Scholz-Böttcher, S. Oberbeckmann, M. Labrenz, D. Fischer, K.-J. Eichhorn, B. Voit, Comparison of μ-ATR-FTIR spectroscopy and py-GCMS as identification tools for microplastic particles and fibers isolated from river sediments. Anal. Bioanal. Chem. 410, 5313–5327 (2018)
- F. Murphy, C. Ewins, F. Carbonnier, B. Quinn, Wastewater treatment works (WwTW) as a source of microplastics in the aquatic environment. Environ. Sci. Technol. 50, 5800–5808 (2016)
- 64. M. Ojeda, P.F. Cossi, G.N. Rimondino, I.L. Chiesa, C.C. Boy, A.F. Pérez, Microplastics pollution in the intertidal limpet, Nacella magellanica, from Beagle Channel (Argentina). Sci. Total Environ. **795**, 148866 (2021)
- 65. P. Ribeiro-Claro, M.M. Nolasco, C. Araújo, Microplastics by Raman Spectroscopy, (2017)
- 66. A.M. Mahon, B. O'Connell, M.G. Healy, I. O'Connor, R. Officer, R. Nash, L. Morrison, Microplastics in sewage sludge: Effects of treatment. Environ. Sci. Technol. 51, 810–818 (2017)

Chapter 7 Distribution of Microorganisms in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract This chapter aims to determine the microorganism distribution in the Three Gorges Reservoir Area (TGRA). However, limited seasonal studies make it difficult to comprehend its real situation. Previous literature found that, before the dam construction, *alpha and gamma proteobacteria* were more prevalent in all surface water samples. While in 2009, *betaproteobacteria*, *gammaproteobacteria*, and *deltaproteobacteria* were ubiquitously found in the TGRA. In 2015, the general bacterial richness trend was found in the order of downstream > middle stream > upstream of the TGRA. Seasonally, in summer, most bacterial and archaeal communities were found due to the increased nutrient levels and higher temperatures. Ecological risk via the bacteria-based index of biotic integrity (Ba-IBI) showed that 25% were Excellent, but 50% and 25% were Good and Fair among all sampling sites. In conclusion, the risk score for overall bacterial ecology was appropriate. However, more studies should be conducted timely to evaluate ecological health and make early warnings when abnormal microorganisms appear in the reservoir.

Keywords Microorganisms \cdot Water quality \cdot Ecological risks \cdot Bacteria \cdot Early warnings

1 Introduction

Microorganisms, known as microbes, are microscopic unicellular or cell-cluster organisms and infectious agents that include bacteria, viruses, protozoa, and microscopic fungi & algae. Among all microorganisms, bacteria are considered the most abundant microorganisms in the aquatic ecosystem [1]. Microbes have a great influence on determining the health and stability of the ecosystem [2]. They serve as decomposers and play an important role in the cycle of carbon, phosphorous, and nitrogen [3]. Microorganisms may influence the function, structure, and services of aquatic environments [4]. Microbial activity initiates the decomposition of organic matter (OM), maintains the whole-reservoir respiration and carbon flow to the higher trophic levels, and thus upholds the entire food web structure of the aquatic ecosystem [5]. In the aquatic environment, bacterial composition and diversity vary with

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_7

the water quality. Many studies have been published previously focusing on the Three Gorges Reservoir Area (TGRA) regarding bacterial composition and diversity that shifts seasonally and annually with the pollution load. Similarly, nutrient pollution mainly from total nitrogen (N) aggravates the eutrophication process, which may change the microbial diversity in the TGRA [1].

Despite nutrient pollution, microorganisms play a mediated role in transforming mercury (Hg) to methyl mercury (MeHg) in sediments or water [6]. Recently, two genes *hcgA* and *hcgB* were discovered to play an important role in the methylation of mercury [7]. Moreover, *Deltaproteobacteria*, *Chloroflexi*, *Firmicutes*, and *Euryarchaeota* are involved in the methylation of mercury [6].

2 Factors Affecting the Microbial Diversity in the Aquatic Environment

Environmental factors, such as temperature, humidity, and precipitation, influence the microbial community pattern [1]. Similarly, seasonal variations and geographical distance alter the microbial distribution in the aquatic environment [8] (Fig. 7.1). The hydrological condition of water, such as flow velocity, water depth, water



Factors shaping microbial Communities

Fig. 7.1 Factors affecting the microbial diversity in the TGRA

temperature, pH, dissolved oxygen (DO), and nutrient concentrations may cause a spatiotemporal change in the microbial communities [9]. According to most of the previous studies, environmental factors were observed as more crucial to influence the aquatic microbial diversity, compared to spatial factors [1]. Notably, few other studies revealed that spatial factors significantly influence the distribution of microbes in water bodies [10–13].

Among hydrological parameters, pH plays a decisive role in nutrient metabolism. DO influences the efficacy of microbial degradation of substances and microbial growth [14]. Moreover, DO also regulates the redox reaction and affects loads of functional genes, involved in nitrogen and phosphorus cycling [15]. Different microorganisms have their physiological limitations for growth and reproduction within specific levels of DO, narrow ranges of pH, and nutrient availability, which influence the community structures. Moreover, the activities of microorganisms could alter the environmental properties, such as the concentrations of enzymes and nutrients, the form and amount of DO & carbon, and pH [16]. Seasonal and spatial variations also bring changes in concentrations of nutrients, which in return alter the microbial community composition [17]. Biotic processes regulating the seasonal variations of nutrients through the assimilation and mineralization processes are closely related to temperature variation, thus disturbing the propagation of bacterial communities in water bodies [18]. The spatial diversity of incoming water in the reservoir and climatic conditions cause variations in water temperature stratification patterns [19]. In the lower layer of the reservoir, the temperature is almost constant, compared to the upper surface layer, which also affects microbial diversity [14, 20].

3 Microbial Diversity and Phylogenetic Distribution Pattern in the TGRA

As the Three Gorges Reservoir (TGR) was established on the Yangtze River, it became a discontinuity for the riverine system due to the counter-season impoundment of the Three Gorges Dam (TGD). Thus, they affected river-borne nutrients transported by the river [21]. The upstream and downstream areas of the reservoir vary prominently in their physicochemical properties [22], which change the microbial communities. Sedimentation due to damming also shifts the proportion of freeliving and particle-attached bacteria. The previous studies described the longitudinal shifts in bacterioplankton community structure that are recognized to be caused by prolonged water residence time [23]. Since the construction of the TGR, bacterial diversity had changed gradually from the upstream to the downstream [24]. After the construction of the reservoir in June 2003, changes in bacterial community structure in the Yangtze River-East China Sea estuary were also observed. According to the report from Jiao et al. [25], the overall bacterial diversity was decreased and a lower abundance of *Betaproteobacteria* was found. However, the diversity of

Temporal Microbial diversity



Fig. 7.2 The temporal microbial diversity observed in the TGRA

Alphaproteobacteria and Cyanobacteria was improved because of the abrupt decline of river runoff and subsequent intrusion of ocean currents [26].

Recently, from environmental samples, the bacterial distribution can be determined by sequencing 16S RNA [27]. Before the construction of the TGRA, it was found that alpha and gamma proteobacteria were ubiquitously presented in all samples [6] (Fig. 7.2). Based on genotype and phenotype analysis, Alteromonas macleodii and Roseobacter spp. were dominant in most surface water samples. Contrarily, in 2009, Wang et al. [28] found that betaproteobacteria, gammaproteobacterial, and deltaproteobacteria were dominantly presented in the surface water samples of the TGRA. In 2015, bacterial abundance and water quality were closely observed along the upper, middle, and lower reaches of the TGRA by Niu et al. [29]. Results indicated that at the highest chemical oxygen demand (COD) level, the highest number/abundance of Firmicutes and Bacillus were found in the upper, middle, and down reaches [29]. Besides, the results of redundancy analysis revealed that COD and phosphates were the primary environmental parameters that significantly affected the bacterial community structure [29]. Furthermore, the general bacterial richness trend decreased in the order of downstream > middle stream > upstream [29]. Although Proteobacteria dominated the bacterial communities, Ochrophyta, Chlorophyta, and Ciliophora became the majority during specific stages in the TGR (Table 7.1) [30–32].

Microbial diversity of the TGR	Reference
Alpha-, Beta-, Gamma proteobacteria, Verrucomicrobia, and Planctomycetes Spi- rochaetes, Nitrospirae, Chloroflexi, and Acidobacteria	[28]
Acidobacteria, Spirochaetes, Chloroflexi, and Nitrospirae	[26]
Firmicutes, Proteobacteria, and Actinobacteria	[4]
Ochrophyta (Protist)	[30]
Ciliophora and Chlorophyta	[32]
Chironomidae, Heptageniidae, and Baetidae	[33]
Diatoma vulgare Melosira varians, Cocconeis placentula, Gyrosigma scalproides,	[34].
and Oscillatoria tenuis M. varians, Cymbella affinis, D. vulgare, Eucapsis alpina,	
and M. granulata, M. varians, C. affinis, and C. placentula,	

Table 7.1 The microbial community observed in the TGRA

4 Effect of Environmental Variations on Microbial Diversity in the TGR

Environmental factors play an important role in shaping the microbial community structure in the aquatic environment [35, 36]. Canonical correspondence analysis (CCA) exhibited relatively high changes in bacterial community composition, which are primarily correlated with the surface water quality parameters, such as temperature, COD, phosphates, nitrates, carbonates, and DO [37]. For the year 2009, the water quality parameters were found to be relatively stable [37], whereas in 2015, the water quality parameters particularly, COD and phosphate were relatively high [29]. From the water samples in 2017, redundancy analysis (RDA) revealed that the microbial population showed a significant correlation/association with environmental factors, such as NO_3 , NH_4 , total organic carbon (TOC), and total phosphates [8]. Due to the TGRA being geographically divided into three longitudinal zones, including riverine, transitional, and lacustrine, different bacterial diversities were found in each zone [8, 38, 39].

Chen et al. [37] evaluated the seasonal comparison of the bacterial diversity at different impoundment levels in the TGRA for the year 2009. It was found that, in the summer, more bacterial and archaeal communities were commonly present in the surface water of the TGRA, due to high nutrient levels and temperature. In terms of the topological distribution, more diverse bacterial populations have been found downstream than that in the middle and upper stream of the TGRA [37]. Moreover, the bacterial diversity via Simpson's index and Shannon index revealed that bacterial diversity dramatically changed between dry and wet seasons. Phylogenetic analysis showed that *Betaproteobacteria* and *Actinobacterium* were found to be the most dominant bacterial populations observed in wet and dry seasons [37]. Further, with the high or low water impoundment level, a variety of bacterial niches were found in the TGRA, which needs to be studied in detail. Also, interspecies interactions of the communities should be explored.

Water level fluctuations in the TGRA may affect the soil microbial compositions. It is worth mentioning that microorganisms contained a variety of aerobic, anaerobic, and facultative populations. However, in soil, anaerobes are barely found, which cannot be ubiquitously distributed in the surface water of the TGRA. It was found that, at low water impounding levels, the mercury methylation activity by microbes was also low [6]. The 16S RNA sequence implied that bacterial richness in soil was relatively higher in the dry season, compared to the wet season, and the bacterial abundance fluctuates seasonally [6]. At the phylogenetic level, *Deltaproteobacteria* and *Methanomicrobia* were found to be higher, which were involved in Hg methylation, and showed increased bacterial activity in the dry season [6]. Moreover, soil nutrients, such as organic matter, nitrates, and phosphates, are found to be higher in content in the dry season than wet season, which aggravate the soil microbe's growth [40].

5 The TGRA Ecological Health via Bacteria-Based Index

In the past years, there was a dire need to quantitatively evaluate the ecological health of the TGRA using the bacteria-based index of biotic integrity (Ba-IBI). Li et al. [3] found the bacterial ecological status in surface water and sediments samples at different water impoundment levels, such as a low water level (September 2016), a high impounding period (December 2016), and a sluicing period (April 2017) [3]. Moreover, certain ecological reference conditions including land use, urbanization, pollution, river connectivity, soil erosion, sediment load, riparian zone, and biodiversity determine the anthropogenic disturbances and the hydromorphology of the TGRA (Fig. 7.3). From the environmental matrices, *Acidobacteria, Geobacter*, and *Gemmatimonadetes* were found to be dominant and were selected to indicate the bacterial ecological risk in the TGRA. Among 12 sampling sites, 25% were reported to be Excellent. However, 50% and 25% were Good and Fair according to bacterial



Fig. 7.3 Reference conditions for determining the bacteria-based index of biotic integrity

ecological health [3]. The biotic integrity was greatly influenced by ecological and environmental changes along with nutrient fluctuations [3, 41, 42]. Besides, the ecological health in the low impoundment level and the sluicing period was good compared to the high impoundment level, implying the hydrodynamic changes due to the seasonal increase or decrease of the water level.

Similarly, another recent study highlighted that relative microbial population distributions are affected by ecological variables [8]. Their results showed that homogeneous selection, dispersal limitation, ecological drift, and variable selection accounted for 51.3, 25.5, 21.2, and 2.0% of the microbial metacommunity distributed in the TGRA [8]. Moreover, it was found that variable selection processes were relatively higher in the summer and also reached up to 100%, compared to the winter [8].

6 Conclusion

In summary, this chapter summarizes the distribution of microorganisms in the TGRA at different impoundment levels. It was found that before the construction of the dam, alpha and gamma proteobacteria were dominant in all surface water In 2009. betaproteobacteria, samples. gammaproteobacteria. and *deltaproteobacteria* were ubiquitously found in the TGRA. Notably, the bacterial richness trend decreased in the order of downstream > middle stream > upstream in 2015. The season-wise comparison revealed that more bacterial and archaeal communities were commonly found in the summer due to high nutrient levels and temperature. Ecological risk via the bacteria-based index of biotic integrity (Ba-IBI) can indicate water quality. In short, the overall bacterial ecological health of the TGRA was found to be stable. However, it may vary seasonally, therefore, more temporal and seasonal studies are required to determine the in-depth mechanistic details according to the bacterial diversity.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

 Q. He, Y. Qiu, H. Liu, X. Sun, L. Kang, L. Cao, H. Li, H. Ai, New insights into the impacts of suspended particulate matter on phytoplankton density in a tributary of the three gorges reservoir, China. Sci. Rep. 7, 13518–13518 (2017)

- D. Yin, Y. Wang, Y. Xiang, Q. Xu, Q. Xie, C. Zhang, J. Liu, D. Wang, Production and migration of methylmercury in water-level-fluctuating zone of the three gorges reservoir, China: Dual roles of flooding-tolerant perennial herb. J. Hazard. Mater. 381, 120962 (2020)
- Y. Li, N. Yang, B. Qian, Z. Yang, D. Liu, L. Niu, W. Zhang, Development of a bacteria-based index of biotic integrity (Ba-IBI) for assessing ecological health of the three gorges reservoir in different operation periods. Sci. Total Environ. 640, 255–263 (2018)
- 4. Z. Yu, J. Yang, S. Amalfitano, X. Yu, L. Liu, Effects of water stratification and mixing on microbial community structure in a subtropical deep reservoir. Sci. Rep. 4, 1–7 (2014)
- 5. L.H. Zeglin, Stream microbial diversity in response to environmental changes: Review and synthesis of existing research. Front. Microbiol. **6**, 454 (2015)
- Y. Xiang, Y. Wang, C. Zhang, H. Shen, D. Wang, Water level fluctuations influence microbial communities and mercury methylation in soils in the three gorges reservoir, China. J. Environ. Sci. 68, 206–217 (2018)
- J.M. Parks, A. Johs, M. Podar, R. Bridou, R.A. Hurt Jr., S.D. Smith, S.J. Tomanicek, Y. Qian, S.D. Brown, C.C. Brandt, A.V. Palumbo, J.C. Smith, J.D. Wall, D.A. Elias, L. Liang, The genetic basis for bacterial mercury methylation. Science **339**, 1332–1335 (2013)
- Y. Li, Y. Gao, W. Zhang, C. Wang, P. Wang, L. Niu, H. Wu, Homogeneous selection dominates the microbial community assembly in the sediment of the three gorges reservoir. Sci. Total Environ. 690, 50–60 (2019)
- H. Du, M. Ma, T. Sun, X. Dai, C. Yang, F. Luo, D. Wang, Y. Igarashi, Mercury-methylating genes dsrB and hgcA in soils/sediments of the three gorges reservoir. Environ. Sci. Pollut. Res. 24, 5001–5011 (2017)
- B.E. Beisner, P.R. Peres-Neto, E.S. Lindström, A. Barnett, M.L. Longhi, The role of environmental and spatial processes in structuring lake communities from bacteria to fish. Ecology 87, 2985–2991 (2006)
- S. Langenheder, H. Ragnarsson, The role of environmental and spatial factors for the composition of aquatic bacterial communities. Ecology 88, 2154–2161 (2007)
- E. Pagaling, H. Wang, M. Venables, A. Wallace, W.D. Grant, D.A. Cowan, B.E. Jones, Y. Ma, A. Ventosa, S. Heaphy, Microbial biogeography of six salt lakes in Inner Mongolia, China, and a salt lake in Argentina. Appl. Environ. Microbiol. **75**, 5750 (2009)
- M. Romina Schiaffino, F. Unrein, J.M. Gasol, R. Massana, V. Balague, I. Izaguirre, Bacterial community structure in a latitudinal gradient of lakes: The roles of spatial versus environmental factors. Freshw. Biol. 56, 1973–1991 (2011)
- 14. J. Wang, H. Fan, X. He, F. Zhang, J. Xiao, Z. Yan, J. Feng, R. Li, Response of bacterial communities to variation in water quality and physicochemical conditions in a river-reservoir system. Global Ecol. Conserv. 27, e01541 (2021)
- D. Güven, A. Dapena, B. Kartal, M.C. Schmid, B. Maas, K. van de Pas-Schoonen, S. Sozen, R. Mendez, H.J.O. den Camp, M.S. Jetten, Propionate oxidation by and methanol inhibition of anaerobic ammonium-oxidizing bacteria. Appl. Environ. Microbiol. **71**, 1066 (2005)
- S. Hongbo, C. Sixue, M. Brestic, Environment-Living Organism's Interactions from Physiology to Genomics (Hindawi, 2015)
- 17. J. Pinhassi, T. Berman, Differential growth response of colony-forming α-and γ-proteobacteria in dilution culture and nutrient addition experiments from Lake Kinneret (Israel), the eastern Mediterranean Sea, and the Gulf of Eilat. Appl. Environ. Microbiol. 69, 199 (2003)
- M.L.S. Fontes, P.C. Abreu, Spatiotemporal variation of bacterial assemblages in a shallow subtropical coastal lagoon in southern Brazil. Microb. Ecol. 58, 140–152 (2009)
- X. Casamitjana, T. Serra, J. Colomer, C. Baserba, J. Pérez-Losada, Effects of the water withdrawal in the stratification patterns of a reservoir. Hydrobiologia 504, 21–28 (2003)
- X.-F. Zhang, S. Ren, J.-Q. Lu, X.-H. Lu, Effect of thermal stratification on interflow travel time in stratified reservoir. J. Zhejiang Univ. Sci. A 16, 265–278 (2015)
- R.J. Batalla, D. Vericat, An appraisal of the contemporary sediment yield in the Ebro Basin. J. Soils Sediments 11, 1070 (2011)

- J. Pozo, E. Orive, H. Fraile, A. Basaguren, Effects of the Cernadilla–Valparaiso reservoir system on the River Tera. Regul. Rivers Res. Manage. 13, 57–73 (1997)
- C. Ruiz-González, L. Proia, I. Ferrera, J.M. Gasol, S. Sabater, Effects of large river dam regulation on bacterioplankton community structure. FEMS Microbiol. Ecol. 84, 316–331 (2013)
- 24. H. Sekiguchi, H. Koshikawa, M. Hiroki, S. Murakami, K. Xu, M. Watanabe, T. Nakahara, M. Zhu, H. Uchiyama, Bacterial distribution and phylogenetic diversity in the Changjiang estuary before the construction of the three gorges dam. Microb. Ecol. 43, 82–91 (2002)
- N. Jiao, Y. Zhang, Y. Zeng, W.D. Gardner, A.V. Mishonov, M.J. Richardson, N. Hong, D. Pan, X.-H. Yan, Y.-H. Jo, Ecological anomalies in the East China Sea: Impacts of the three gorges dam? Water Res. 41, 1287–1293 (2007)
- 26. S. Wang, R.M. Dong, C.Z. Dong, L. Huang, H. Jiang, Y. Wei, L. Feng, D. Liu, G. Yang, C. Zhang, Diversity of microbial plankton across the three gorges dam of the Yangtze River, China. Geosci. Front. 3, 335–349 (2012)
- H. Zeng, L. Song, Z. Yu, H. Chen, Distribution of phytoplankton in the three-gorge reservoir during rainy and dry seasons. Sci. Total Environ. 367, 999–1009 (2006)
- S. Wang, R.M. Dong, C.Z. Dong, L. Huang, H. Jiang, Y. Wei, L. Feng, D. Liu, G. Yang, C. Zhang, H. Dong, Diversity of microbial plankton across the three gorges dam of the Yangtze River, China. Geosci. Front. 3, 335–349 (2012)
- 29. A. Niu, L.-Y. Song, Y.-H. Xiong, C.-J. Lu, M. Junaid, D.-S. Pei, Impact of water quality on the microbial diversity in the surface water along the three gorge reservoir (TGR), China. Ecotoxicol. Environ. Saf. 181, 412–418 (2019)
- 30. G. Saxena, E.M. Marzinelli, N.N. Naing, Z. He, Y. Liang, L. Tom, S. Mitra, H. Ping, U.M. Joshi, S. Reuben, Ecogenomics reveals metals and land-use pressures on microbial communities in the waterways of a megacity. Environ. Sci. Technol. 49, 1462–1471 (2015)
- 31. Y. Xie, T. Floehr, X. Zhang, H. Xiao, J. Yang, P. Xia, G.A. Burton Jr., H. Hollert, In situ microbiota distinguished primary anthropogenic stressor in freshwater sediments. Environ. Pollut. 239, 189–197 (2018)
- 32. Y. Xie, J. Wang, Y. Wu, C. Ren, C. Song, J. Yang, H. Yu, J.P. Giesy, X. Zhang, Using in situ bacterial communities to monitor contaminants in river sediments. Environ. Pollut. 212, 348–357 (2016)
- 33. S. Chi, S. Li, S. Chen, M. Chen, J. Zheng, J. Hu, Temporal variations in macroinvertebrate communities from the tributaries in the three gorges reservoir catchment, China. Rev. Chil. Hist. Nat. 90 (2017)
- 34. J. Fu, L. Liu, X. He, H. Zhang, C. Dong, Y. Yang, J. Shi, Z. Wu, Dynamics of epilithic algae communities and their relationship with environmental factors during storage and non-storage periods in the three gorges reservoir. Huan Jing Ke Xue 40, 3099 (2019)
- 35. L. Yao, C. Chen, G. Liu, F. Li, W. Liu, Environmental factors, but not abundance and diversity of nitrifying microorganisms, explain sediment nitrification rates in Yangtze lakes. RSC Adv. 8, 1875–1883 (2018)
- 36. Z. Yu, J. Yang, S. Amalfitano, X. Yu, L. Liu, Effects of water stratification and mixing on microbial community structure in a subtropical deep reservoir. Sci. Rep. 4, 5821–5821 (2014)
- 37. Z. Chen, Z. Zhou, X. Peng, H. Xiang, S. Xiang, Z. Jiang, Effects of wet and dry seasons on the aquatic bacterial community structure of the three gorges reservoir. World J. Microbiol. Biotechnol. 29, 841–853 (2013)
- H. Sun, Y. Wu, H. Bing, J. Zhou, N. Li, Available forms of nutrients and heavy metals control the distribution of microbial phospholipid fatty acids in sediments of the three gorges reservoir, China. Environ. Sci. Pollut. Res. 25, 5740–5751 (2018)

- 39. X. Li, S. Guo, P. Liu, G. Chen, Dynamic control of flood limited water level for reservoir operation by considering inflow uncertainty. J. Hydrol. **391**, 124–132 (2010)
- 40. X. Zhang, C. Zhang, R.G. Sun, D.Y. Wang, Mercury dynamics of several plants collected from the water-level fluctuation zone of the three gorges reservoir area during flooding and its impact on water body. Huan Jing Ke Xue **35**, 4560–4566 (2014)
- B. Ma, K. Zhao, X. Lv, W. Su, Z. Dai, J.A. Gilbert, P.C. Brookes, K. Faust, J. Xu, Genetic correlation network prediction of forest soil microbial functional organization. ISME J. 12, 2492–2505 (2018)
- 42. N.-N. Ji, C. Gao, B. Sandel, Y. Zheng, L. Chen, B.-W. Wu, X.-C. Li, Y.-L. Wang, P.-P. Lü, X. Sun, L.-D. Guo, Late quaternary climate change explains soil fungal community composition rather than fungal richness in forest ecosystems. Ecol. Evol. 9, 6678–6692 (2019)

Part III Study Methods of Ecotoxicology in the Reservoir

Chapter 8 Ecotoxicity Test Methods of Primary Producers in the Reservoir



Marriya Sultan and De-Sheng Pei 💿

Abstract Primary producers occupy a bottom position in the food chain. For that reason, they play a vital role in the transfer of pollutants and ecotoxicity evaluations across the food chains. Ecotoxicity of freshwater bodies, such as reservoirs, is mainly evaluated using species of the primary producer, including *Vibrio fischeri* (bacterium), *Raphidocelis subcapitata* (microalgae), *Spirodela polyrhiza* (floating macrophytes), and *Myriophyllum* (submerged macrophytes). This chapter provides an overview of the use of primary producers in ecotoxicity assessment, standard ecotoxicity tests given by international organizations. Besides, some recent literature regarding the use of ecotoxicity tests with primary producers in water quality assessment of reservoirs is reviewed and issues related to primary producers-based ecotoxicity tests are also highlighted.

Keywords Primary producers · Microalgae · *Vibrio fischeri* · Reservoirs · Ecotoxicity

1 Introduction

Freshwater reservoirs are considered important water reserves for human and animal consumption, irrigation, and recreational activities. Over the past few decades, a growing concern is witnessed with the decrease in water quality due to enhanced domestic, agricultural, and industrial outputs [1]. To improve the water quality, several legislations have been implemented across the world, which advises the assessment of important freshwater bodies for pollutant concentrations and collection of acute and chronic toxicity data on aquatic organisms (primary producers, invertebrates, and vertebrates) for setting the water quality standards [2]. In response to these legislations (such as Directive 2000/60/E), various organizations have developed guidelines for testing water quality. Among them, the most widely accepted one is the OECD guideline, which includes the collection of internationally accepted ecotoxicity testing methods that can be used by independent laboratories, industries, and governments to determine the safety levels of different pollutants in various ecosystems (terrestrial & aquatic) [3].

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_8

Generally, chemical analysis is mainly carried out to monitor the pollutants, which does not give an accurate prediction of effects on organisms because it does not take into consideration of the direct interaction between organisms and the chemicals, such as dynamics of environmental heterogeneity, the bioavailability of the chemical, and organism life cycle. Conversely, bioassays provide more precise evidence of the effects produced after exposure to pollutants [2, 4]. In this perspective, ecotoxicological tests based on primary producers undertake an evident role in evaluating the quality status of freshwater bodies [5]. Being present at the bottom of the food chain, primary producers bioaccumulate the pollutants in water and transfer the pollutants to the herbivores and carnivores. Finally, humans will be contaminated along the food chain, resulting in the biomagnification process. Hence, ecotoxicological studies based on primary producers have the potential to identify the possible risks across the food chains [6]. Also, being the first interface amid abiotic and biotic constituents, plants react to water pollution earlier, compared to other organisms [7]. However, the use of primary producers, such as plants, in ecotoxicological tests is still limited [8, 9]. The most commonly used primary producers are bacterias (such as *Pseudomonas putida*) and unicellular algae [10]. Besides, aquatic micro- and macroinvertebrates (including Daphnia magna and Gammarus duebeni) and zebrafish (Danio rerio) are also frequently used organisms for ecotoxicity tests [7, 11].

2 Standardized Ecotoxicity Tests Using Primary Producers

Primary producer-based tests gained momentum during the last two decades [9, 12]. Microalgae were more commonly utilized as reference species, compared to other plant groups. Ecotoxicity tests using microalgae with various effluents and toxicants were the first to be designed and standardized since the 1960s by regulatory agencies, such as the ISO and the OECD [13]. The most common microalgal species used in ecotoxicology studies for a wide range of pollutants include *Pseudokirchneriella subcapitata (Selenastrum subcapitata)* and *Chlorella vulgaris* (Table 8.1) [7]. Afterward, since the 1970s, some aquatic pteridophytes (such as *Salvina molesta and Azolla pinnata*) and bryophytes (*Fontinalis antipyretica*) were also used for ecotoxicological tests with much lesser frequency, compared to other species. However, freshwater species, such as *Lemma* (floating macrophytes) and *Myriophyllum* (rooted flowering plants), have been used frequently for ecotoxicological studies since the 1950s.

Several standardized procedures for ecotoxicity testing have been proposed by globally recognized organizations, such as the US EPA (the United States Environmental Protection Agency), ISO (International Organization for Standardization), OECD (Organization for Economic Cooperation and Development), and ASTM (American Society for Testing and Materials) [28]. These organizations have given 13 standard tests using freshwater primary producers. Of which, three tests are selected for bacterial species, two tests for primary algal species (diatoms, green

Aquatic					
primary					
producer	Species	Compartment	Organization	Test no.	References
Bacteria	Pseudomonas putida	Water	ISO	10712	[14]
	Vibrio fischeri	Water	ISO	11348-1	[15]
	Salmonella typhimurium	Water	ISO	11350	[16]
Microalgae	Raphidocelis subcapitata	Water	OECD	201	[17]
	Pseudokirchneriella subcapitata	Water	ASTM	D3978- 04	[18]
	Anabaena flos- aquae	Water	USEPA	850.4550	[19]
Floating macrophytes	Lemna minor	Water	ISO	20079	[20]
	Spirodela polyrhiza	Water	ISO	20227	[21]
	Lemna gibba	Water	ASTM	E1415- 91	[22]
	Lemna sp.	Water	OECD	221	[23]
	Lemna spp.	Water	USEPA	850.4400	[24]
Submerged macrophytes	Myriophyllum aquaticum	Sediment	ISO	16191	[25]
	Myriophyllum spicatum	Water	OECD	238	[26]
	Myriophyllum spp./ Glyceria maxima	Water & Sediment	OECD	239	[27]

Table 8.1 Standardized ecotoxicity test using primary producers

algae, and cyanobacteria), five tests for floating macrophytes consisting of duckweed species, and three tests for submerged macrophytes (Table 8.1). The detailed procedure of some of the common standardized ecotoxicity tests using primary producers, such as bacteria (*Vibrio fischeri*), microalgae (*Raphidocelis subcapitata*), and floating macrophyte (*Spirodela polyrhiza*) is given as follows.

2.1 Luminescence Inhibition Test of Vibrio fischeri

This test is based on the hypothesis that the luminescence of bacteria is reduced when exposed to pollutants and the toxicity is expressed as the concentration of the toxicants (pollutants) [2]. Water samples are placed in glass cuvettes and *Vibrio fischeri* bacterium is transferred to cuvettes. After 15 min, the toxicity of water samples is evaluated by computing a 50% reduction in luminescence using a temperate luminometer. And results are reported as percentage inhibition [2, 15].

In the control group, the change in luminescence containing the bacteria suspension is determined during the incubation time t. From this data, a correction factor f is calculated as follows:
$$f = L_t/L_0$$

Where L_0 is the initial luminescence and L_t is the luminescence over time. The corrected initial luminescence (L_{0C}) is calculated as

$$L_{0C} = L_0 \times 100$$

In the treatment group, the change in luminescence after the required incubation time (L_t) is determined as $\Delta L (\Delta L = L_{0C} - L_t)$.

Whereas the percentage inhibition of the test chemical is calculated as:

$$\mathrm{INH}\% = \frac{\Delta L}{L_{\mathrm{0C}}} \times 100$$

2.2 Growth Inhibition Test of Raphidocelis subcapitata

This test is carried out in 24-well plates and a Woods Hole Marine Biological Laboratory (MBL) culture medium is used as a negative control. The natural water samples are assessed using three treatments and a blank (3 replicates each). In each treatment, the initial concentration of 5×10^4 cells/mL of microalgae (*Raphidocelis subcapitata*) is added. The microplates are then placed in a climatic chamber (Incubator) at 24 °C with a light intensity of about 7000 lux. The algal cultures are re-suspended every day to reduce sedimentation. After 72 h culture, the absorbance is measured in each well at $\lambda = 440$ nm using a UV-Spectrophotometer. In each treatment, the absorbance of the corresponding blank is removed to nullify the growth and existence of potential constituents that could affect the final results. Finally, the concentration of cells in each well is calculated using the following equation.

$$C = -17,107.5 + ABS \times 7,925,350$$

Where *C* is the concentration of algae in cells/mL and ABS is the absorbance dimension ($\lambda = 440$ nm). The final results are quantified as yield, which is the difference between initial and final biomass (cell densities- cells/mL) [17, 29].

2.3 Growth Inhibition Test of Spirodela polyrhiza

Spirodela polyrhiza is carried out in glass vials and the Steinberg medium is used as a negative control for growth inhibition assay. The natural water samples are assessed under three treatments with 3 replicates for each treatment with a final

volume of 100 mL. In each treatment and corresponding replicate, 9 fronds are added at the start of the test. The glass vials covered with perforated parafilm (for gaseous exchange) are retained in a climatic chamber (incubator water testing) for a week at 24 °C with a permanent light intensity of about 7000 lux. After 7 days culture, the number of fronds is counted and the total yield is calculated, which is the difference between the number of fronds at the end and the beginning of the test [21].

2.4 Vegetative Shoot Inhibition Test of Myriophyllum spicatum

Myriophyllum spicatum ecotoxicity test is to assess the pollutant-related effects on the vegetative growth of plants growing in freshwater water and sediment. The test is conducted by potting the shoot apices of healthy, non-flowering *Myriophyllum spicatum* plants in artificial sediment, supplemented with nutrients for the sufficient growth of plants. The potted plants are then retained in Smart and Barko medium. After an adequate period of root formation, plants are exposed to different test concentrations via water column or the sediment by spiking the sediment with the different concentrations of the test chemical. Later, plants are transplanted into the spiked sediment. After 14 days of culture, the effects on growth are assessed by numerically calculating the shoot length, fresh weight, and dry weight and by qualitative observations of growth deformities, or symptoms, such as chlorosis or necrosis [30].

For quantification of chemical-related effects, the growth of the plant in test solutions is compared with the control plants, and the concentration that causes growth inhibition is determined and expressed as the Effective Concentration (EC). Based on regulatory requirements, values of EC_{10} , EC_{20} , or EC_{50} are calculated.

The response variables, such as average specific growth rate and yield, are calculated. The average vegetative shoot growth rate is based on changes in the total shoot length, such as the total fresh weight of the shoot and the total dry weight of shoot. The mean weight and length of the three plants per test and the growth rate of each replicate are calculated using the following formulate:

$$\mu i - j = (\ln(Nj) - \ln(Ni))/t$$

Where $\mu i \cdot j$ is the mean specific growth rate from time *i* to *j*, N*i* is the measurement variable in the vessel at time *i*, N*j* is the measurement variable in the vessel at time *j*, and *t* is the time from *i* to *j*.

Percent inhibition of growth rate (Ir) is calculated for each test concentration according to the following formula:

$$\% Ir = \mu c - \frac{\mu t}{\mu c} \times 100$$

Where % *Ir* is the average growth inhibition in percentage, μc is the mean value for μ in the control, and μt denotes the mean value for μ in the treatment group.

Finally, the yield is calculated, based on variation in total shoot length, including total fresh weight and total dry weight of the shoot over time in both groups (controls and treatment). The mean percent inhibition in yield (% *Iy*) is calculated as:

$$\%Iy = bc - \frac{bt}{bc} \times 100$$

Where the % *Iy* is the percent reduction in yield, *bc* is the difference between final biomass from the initial biomass for the control group, and *bt* is the difference between final biomass from the initial biomass in the treatment group [26].

3 Ecotoxicity Assessment of Freshwater Reservoirs Using Primary Producers

The encouragement of using primary producers (mainly plants and algae) for ecotoxicity testing was given by the Toxic Substances Control Act (TSCA), which highlighted the sensitivity of freshwater algae to various pollutants [13]. Since then, various studies have used ecotoxicological tests with primary producers to determine the toxic effects of different pollutants, such as pesticides, herbicides, hydrocarbons, and trace and heavy metals. However, the studies regarding the use of primary producers for ecotoxicity testing of reservoirs are still scarce with few studies carried out during the last decades, featuring the freshwater reservoirs of Europe [1, 2, 31–33] and South America [34, 35] as shown in Table 8.2.

Pérez et al. [2] performed an ecotoxicity test using microalgae species, *Pseudokirchneriella subcapitata*, for evaluating the impacts of pesticides and heavy metals on primary producers of Alqueva Reservoir, Portugal. The results demonstrated that *P. subcapitata* was sensitive even toward the low concentrations of insecticides in the water. Baran et al. [32] evaluated the ecotoxicity of nutrients and trace elements in the sediments of Rożnów reservoir, Poland using *Vibrio fischeri* luminescence inhibition test and *S. alba* root and seed inhibition test, where the *S. alba* germination inhibition ranged between 22 and 53%, while root growth inhibition ranged within 23 to 63% with Germination Index (GI) values ranged between 30 and 152%. However, the *V. fischeri* luminescence inhibition was observed within the range of 28 to 57%. In another study, Szara et al. [33] verified the ecotoxicity level of sediment samples obtained from Rożnów reservoir, Southern Poland using *L. sativum S. saccharatum* and *S. alba*, and they found that the inhibition of root growth of these plants ranged between -70 to 37% for *S. saccharatum*, -47 to 45% for *S. alba* and -69 to 61% for *L. sativum*. L.

		10110001 1000	the sume prime for the	111111		
Primary						
producer	Test species	Matrix	Stressors	Endpoint measured	Location	References
Algae	Pseudokirchneriella subcapitata	Water	Pesticides & heavy metals	Growth inhibition	Alqueva reservoir Portugal	[2]
	Raphidocelis subcapitata	Water	Pesticides & heavy metals	Growth inhibition	Aguieira reservoir, Germany	[]]
Floating macrophyte	Spirodela polyrhiza	Water	Pesticides & heavy metals	Growth inhibition	Aguieira reservoir, Germany	Ξ
Bacteria	Vibrio fischeri	Water	Pesticides & heavy metals	Luminescence inhibition	Alqueva reservoir Portugal	[2]
	Vibrio fischeri.	Sediment	Trace metals	Luminescence inhibition	Rybnik reservoir, Poland	[31]
	Vibrio fischeri.	Sediment	Nutrients & trace elements	Luminescence inhibition	Rożnów reservoir, southern Poland	[32]
Plants	Sinapis alba	sediment	Nutrients & trace elements	Inhibition of seed germina- tion & root growth	Rożnów reservoir, southern Poland	[32]
	Sinapis alba	Sediment	Heavy metals	Inhibition of seed germina- tion and root growth	Ružín reservoir, Slo- vak Republic	[36]
	Sorghum saccharatum, Lepidium sativum & sinapis alba	Sediments	Trace elements	Inhibition of seed germina- tion and root growth	Rożnów reservoir, southern Poland	[33]
	Lepidium sativum & Sinapis alba	Sediments	Nutrients and trace elements	Germination index	Rio Grande reservoir, Brazil	[34]
	Sinapis alba	Water	Nutrients & estrogens	Germination index	Billings reservoir, Brazil	[35]

 Table 8.2
 Recent studies on the ecotoxicity of freshwater reservoirs using primary producers

Coelho et al. [35] evaluated nutrients and estrogens related to phytotoxicity using *Sinapis alba* in the Billings Reservoir (Brazil), and observed the GI values as 82.74% and 83.16% with no phytotoxic effect in site 1 and site 2 for wet and dry periods, respectively, whereas site 3 was classified as moderately phytotoxic with the GI values ranged between 60 and 80%. In another study, the growth inhibition ecotoxicity test with primary producers, such as *R. subcapitata* and *Spirodela polyrhiza* showed lower growth rates in the reservoir water during autumn and higher growth rates during the spring season. The growth inhibition of these two species during autumn was associated with the occurrence of herbicides, such as metolachlor, atrazine, and terbuthylazine, which are used mainly during summer in nearby agricultural areas of Aguieira reservoir, Germany [1].

4 Concerns in Standardized Primary Producers-Based Ecotoxicity Tests

In the previous study, several concerns have been reported regarding the use of these ecotoxicity tests, when these tests sometimes do not reciprocate the required results. The protocols of these ecotoxicity tests refer to their execution in controlled conditions of the laboratory that confine the ability of tests to deduce the real status of a natural system (such as other interactions and limiting factors that are ignored in lab-based experiments) [7]. Therefore, the obtained results do not fully reflect the real effects of tested pollutants on plants. For example, Ding et al. [37] studied the effects of silver nanoparticles (AgNPs) on Lemna minor, which are generally diminished in the humic acids (HA) as it reduces the absorption ability of AgNPs in Lemna. Besides, the exposure time given in standardized protocols does not always produce the anticipated response. For example, exposure extension (from 7 days to 14 days) could increase the intensity of the response providing an additional understanding of the pollutant toxicity [5]. Moreover, the use of standard growth undiluted growth medium (proposed by standard protocols) could also interfere with the tested chemical and could lead to various complexities in results interpretation. Gubbins et al. [5] found that silver nanoparticles (AgNPs) show different responses in the Lemna plant depending on the growth medium (either diluted or concentrated). As the interaction of AgNPs with the concentrated medium showed aggregation and sedimentation, a 100-fold diluted growth medium was used to reduce AgNP-medium aggregation. This assumes that the related problems might also occur with other standardized tests, therefore, preliminary tests are recommended to verify any interactions among pollutants and medium.

5 Conclusions

Because primary producers occupy the bottom place of the food chain, it is evident that they play role in pollutant transfer across the food chain and assist in ascertaining the possible toxic risks across the food chains. In this viewpoint, ecotoxicity tests based on primary producers are considered vital in evaluating the quality status of freshwater bodies. The guidelines for standardized ecotoxicity tests using primary producers are given by various organizations, such as the OECD, ISO, and ASTM, which have used sensitive species from different groups of primary producers for ecotoxicity tests. However, among the reported literature, the species of microalgae are more commonly utilized in ecotoxicity tests, compared to other plant groups. However, some of the common standardized ecotoxicity tests using primary producers include luminescence inhibition test with Vibrio fischeri, growth inhibition test with Raphidocelis subcapitata, Spirodela polyrhiza, and Myriophyllum spicatum. However, the standardized ecotoxicity test does not include the representative species from all major groups of primary producers and standard guidelines for ecotoxicological tests with some major groups, such as the missing mosses and lichens. Therefore, to increase ecological relevance, more test species representing all major groups are required to be incorporated in ecotoxicity testing, for which standard protocols are needed to be devised by authoritative organisms, such as the EPA, ISO, OECD, and ASTM. Besides, various concerns regarding ecosystem complexities, such as lack of understanding of synergistic effects, time scale, exposure time, and growth medium are required to be considered. Further, such tests should be designed to suitably represent the ecosystem complexity and achieve more consistent responses about processes occurring in the natural environment.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- I. Pinto, S. Rodrigues, O. Lage, S. Antunes, Assessment of water quality in Aguieira reservoir: Ecotoxicological tools in addition to the Water framework directive. Ecotoxicol. Environ. Saf. 208, 111583 (2021)
- J.R. Pérez, S. Loureiro, S. Menezes, P. Palma, R.M. Fernandes, I.R. Barbosa, A.M. Soares, Assessment of water quality in the Alqueva reservoir (Portugal) using bioassays. Environ. Sci. Pollut. Res. 17, 688–702 (2010)
- 3. A. Valavanidis, T. Vlachogianni, Ecotoxicity test methods and ecological risk assessment. Ecosystems 10, 13 (2015)

- C.A. Harris, A.P. Scott, A.C. Johnson, G.H. Panter, D. Sheahan, M. Roberts, J.P. Sumpter, Principles of sound ecotoxicology. Environ. Sci. Technol. 48, 3100–3111 (2014)
- E.J. Gubbins, L.C. Batty, J.R. Lead, Phytotoxicity of silver nanoparticles to Lemna minor L. Environ. Pollut. 159, 1551–1559 (2011)
- M.B. Costa, F.V. Tavares, C.B. Martinez, I.G. Colares, C.D.M.G. Martins, Accumulation and effects of copper on aquatic macrophytes Potamogeton pectinatus L.: Potential application to environmental monitoring and phytoremediation. Ecotoxicol. Environ. Saf. 155, 117–124 (2018)
- S. Ceschin, A. Bellini, M. Scalici, Aquatic plants and ecotoxicological assessment in freshwater ecosystems: A review. Environ. Sci. Pollut. Res. 28, 4975–4988 (2021)
- 8. I. Blinova, Use of freshwater algae and duckweeds for phytotoxicity testing. Environ Toxicol **19**, 425–428 (2004)
- G.D. Alkimin, D. Daniel, S. Frankenbach, J. Serôdio, A.M. Soares, C. Barata, B. Nunes, Evaluation of pharmaceutical toxic effects of non-standard endpoints on the macrophyte species Lemna minor and Lemna gibba. Sci. Total Environ. 657, 926–937 (2019)
- W. Wang, H. Gao, S. Jin, R. Li, G. Na, The ecotoxicological effects of microplastics on aquatic food web, from primary producer to human. Ecotoxicol. Environ. Saf. 173, 110–117 (2019)
- Y. Xue, W.J. Peijnenburg, J. Huang, D. Wang, Y. Jin, Trophic transfer of Cd from duckweed (Lemna minor L.) to tilapia (Oreochromis mossambicus). Environ. Toxicol. Chem. 37, 1367–1377 (2018)
- S. Kumar, N. Thakur, A.K. Singh, B.A. Gudade, D. Ghimire, S. Das, Aquatic macrophytes for environmental pollution control, in *Phytoremediation Technology for the Removal of Heavy Metals and Other Contaminants from Soil and Water*, (Elsevier, 2022), pp. 291–308
- M.A. Lewis, Use of freshwater plants for phytotoxicity testing: A review. Environ. Pollut. 87, 319–336 (1995)
- International Standard Organization (ISO) E. ISO, 10712 Ecotoxicity Test Using Pseudomonas putida, 1995
- 15. ISO, 11348-1, Water quality—determination of the inhibitory effect of water samples on the light emission of vibrio Fischeri (luminescent bacteria test), in *Part 1: Method Using Freshly Prepared Bacteria*, (International Standard Organization, Geneva, 2007), p. 23
- ISO 11350, Salmonella/microsome fluctuation test (Ames fluctuation test), in *Determination of* the Genotoxicity of Water and Waste Water, International Standard Organization, (Geneva, 2012)
- 17. OECD, Growth inhibition test (201), in *Guideline for Testing of Chemicals*, (Organization for Economic Cooperation and Development, Paris, 2011)
- 18. ASTM D3978-04, Standard Practice for Algal Growth Potential Testing with Pseudokirchneriella subcapitata (ASTM, 2012)
- 19. USEPA 850.4550, Cyanobacteria (Anabaena flos-aquae) toxicity, in USEPA Ecological Effects Test Guidelines, (United States Environmental Protection Agency, 2012)
- 20. ISO, 20079, Duckweed growth inhibition test, in *Water Quality—Determination of the Toxic Effect of Water Constituents and Waste Water on Duckweed (Lemna minor)*, (International Organization for Standardization Genève, 2005)
- 21. ISO, 20227, Method using a stock culture independent microbiotest, in Water Quality— Determination of the Growth Inhibition Effects of Waste Waters, Natural Waters and Chemicals on the Duckweed Spirodela polyrhiza, (2017)
- 22. ASTM, Standard Guide for Conducting Static Toxicity Tests with Lemna gibba E1415-91 (ASTM, 2012)
- 23. OECD, Lemna sp. Growth inhibition test, in *Guideline 221*, (Organisation for Economic Co-operation and Development, Paris, 2006)
- 24. USEPA, 850.4400, Aquatic plants field study, in *Ecological Effects Test Guidelines*, (2012)
- ISO, Determination of the toxic effect of sediment on the growth behaviour of Myriophyllum aquaticum, in Water Quality, 16191, (The International Organization for Standardization, 2013)

- 26. OECD, 238, Sediment-Free Myriophyllum spicatum Toxicity Test. OECD Guidelines for the Testing of Chemicals (Organization for Economic Cooperation, Development and Growth, 2014)
- 27. OECD, Water-Sediment Myriophyllum spicatum Toxicity Test. OECD Guidelines for the Testing of Chemicals, 239 (Organization for Economic Cooperation, Development and Growth (OECD), Paris, 2014)
- J.A. Vonk, M.H. Kraak, Herbicide exposure and toxicity to aquatic primary producers. Rev. Environ. Contam. Toxicol. 250, 119–171 (2020)
- U. Strotmann, D. Pastor Flores, O. Konrad, C. Gendig, Bacterial toxicity testing: Modification and evaluation of the luminescent bacteria test and the respiration inhibition test. Processes 8, 1349 (2020)
- A. Fusilero, J. De Rop, P. Spanoghe, G.H. Arts, K.A. De Schamphelaere, Mixture toxicity of herbicides with dissimilar modes of action to Myriophyllum spicatum. Environ. Toxicol. Chem. 41, 2209–2220 (2022)
- A. Baran, M. Tarnawski, T. Koniarz, Spatial distribution of trace elements and ecotoxicity of bottom sediments in Rybnik reservoir, Silesian-Poland. Environ. Sci. Pollut. Res. 23, 17255–17268 (2016)
- 32. A. Baran, M. Tarnawski, T. Koniarz, M. Szara, Content of nutrients, trace elements, and ecotoxicity of sediment cores from Rożnów reservoir (Southern Poland). Environ. Geochem. Health 41, 2929–2948 (2019)
- 33. M. Szara, A. Baran, A. Klimkowicz-Pawlas, M. Tarnawski, Ecotoxicological characteristics and ecological risk assessment of trace elements in the bottom sediments of the Rożnów reservoir (Poland). Ecotoxicology 29, 45–57 (2020)
- 34. T.D.A.G. Peduto, T.A. de Jesus, L.H.G. Coelho, W.S. Hanisch, M.R.D. Moretto, R.F. Benassi, Phytotoxicological assessment and its relationship with environmental variables of Rio Grande reservoir waters (São Paulo state, Brazil). Revista Eletrônica em Gestão, Educação e Tecnologia Ambiental 24, 15 (2020)
- 35. L.H.G. Coelho, T.A. de Jesus, M.Y. Kohatsu, G.T. Poccia, V. Chicarolli, K. Helwig, C. Hunter, J. Roberts, P. Teedon, O. Pahl, Estrogenic hormones in São Paulo waters (Brazil) and their relationship with environmental variables and Sinapis alba phytotoxicity. Water Air Soil Pollut. 231, 1–19 (2020)
- 36. O. Šestinová, L. Findoráková, J. Hančuľák, Toxicity testing of sediments. Nova Biotechnologica et Chimica 11, 111–116 (2012)
- Y. Ding, X. Bai, Z. Ye, D. Gong, J. Cao, Z. Hua, Humic acid regulation of the environmental behavior and phytotoxicity of silver nanoparticles to Lemna minor. Environ. Sci.: Nano 6, 3712–3722 (2019)

Chapter 9 Ecotoxicology Methods of Reservoir Water Using Invertebrates



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter explains different ecotoxicological methods utilizing invertebrates for the assessment of water quality of reservoirs and provides information on species of invertebrates that are involved mostly in ecotoxicological assessment assays. Invertebrates occupy a key position in the aquatic ecosystem and serve as intermediate consumers in the food chain, therefore, they are useful indicators of water toxicity. Various standard ecotoxicity tests approved by international organizations are being used in toxicity testing of water bodies, which includes acute and chronic toxicity tests using invertebrate species. The most acceptable organisms in ecotoxicity testing utilizing invertebrates are Daphids (*Daphnia magna, Daphina pulex*, and *Ceriodaphnia dubia*), which are also recognized as standard species. Previous studies also reported the use of crustaceans, such as amphipods, branchiopods, insect species, and rotifers.

Keywords Invertebrates · Ecotoxicology · Aquatic toxicology tests · Daphnia magna

1 Introduction

Inputs of various hazardous chemicals from agricultural activities, leaching of groundwater, sewage discharges, and runoff are affecting the overall quality of the freshwater reservoirs. Therefore, to reduce the risk instigated by hazardous compounds to the aquatic ecosystem, the assessment of water quality is essential for the execution of the monitoring programs. An integrated evaluation of harmful impacts of chemicals on aquatic ecosystems and populations is necessary, which must include characterization of chemicals complemented with bioassays as it allows the evaluation of effects from all components of water including effects from unknown substances [1]. For this purpose, ecotoxicological assessments are conducted to support the research on the harmful effects of pollutants on different organisms at various trophic levels [2]. Biological organisms mainly include vertebrates (such as fish and amphibians) and freshwater and marine invertebrate species. Different species are used in different toxicity tests. The difference is their proneness to chemicals due to their genetic factors and physiological factors, such as metabolic

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_9

rate, excretion rate, dietary factors, and health of the organism. The most commonly used standard aquatic test species include zebrafish, fathead minnow, sheep head minnow, rainbow trout, grass shrimps, midges, daphnids, oysters, mysids, scuds, and mussels [3, 4].

2 Invertebrates for Ecotoxicological Testing

Invertebrates have been used for ecotoxicological assessments for decades. They are considered useful to detect pollutants mainly due to their bioaccumulation and biological characteristics. Moreover, they could also be used as early warning indicators to predict toxicological effects at the population and community levels [5]. Invertebrates like crustaceans and mollusks are integral to ecotoxicological studies and are extensively used as test species in a laboratory for ecotoxicological assessments. Daphnia and intertidal copepods like Tigriopus japonicus are classic invertebrate examples [6]. Besides, various other freshwater invertebrate models like arthropods, annelids, and bivalves are also used for ecotoxicology evaluation [7]. Their small life period supports them to be utilized to study the multigenerational and reproductive effects of harmful chemicals with limited resources [5]. European methods of inter-calibration practices have shown that approaches using benthic macroinvertebrates were the most effective among other methods [8]. Invertebrates are considered major targets in ecotoxicology because of their vital roles in ecosystem function, wide biodiversity, their suitability for experimental methodologies in the field and laboratory, and the available genomic data of ecotoxicology model species [9].

3 Invertebrate Test Species

It is observed that organisms inhabiting different habitats are suitable for holistic study of aquatic pollutants, because their health assessment after exposure to these pollutants at trophic levels is desirable. Therefore, they are also ideal invertebrates for reservoir ecotoxicity assessment [7]. Among crustaceans, intertidal copepods are considered suitable test organisms. The *Tigriopus japonicus* is considered susceptible to stressors like copper, fluorene, phenanthrene, UV, and gamma radiation [10–12]. Recently, *Paracyclopina nana* is considered for its efficacy as a test organism in ecotoxicological studies [7]. They are often used in ecotoxicological studies of moderately soluble chemicals like drugs and pharmaceuticals [13–17]. For toxicological studies, planarians are commonly utilized [18, 19] and considered suitable for examining developmental toxicity, potential genotoxicity, teratogenicity, and tumorigenicity of chemicals [20]. Acute toxicity test using planarian species has been conducted in many studies [21, 22]. In a previous study [23], the acute toxicity of various pesticides and metals using the freshwater planarians model has been

reported, which demonstrates their susceptibility to cadmium, copper, mercury, tributyltin, and insecticides. The toxicities of ammonia and nitrite have also been evaluated using freshwater planarians [20].

Freshwater invertebrates, such as Daphnids (*Ceriodaphnia dubia*, *Daphnia pulex*, and *Daphnia magna*) were the recommended species for acute toxicity tests [24], because they could be cultured in the laboratory with ease and are sensitive to various classes of pollutants. Daphnids used for acute toxicity tests are usually less than 24 hours old. Other freshwater invertebrate species like mayflies, stoneflies, and amphipods should be in an early instar, whereas midges are in the second or third instar [25]. Besides, *Daphnia magna* is utilized in the majority of aquatic toxicity tests and cultured easily in the laboratory because of its small size, parthenogenetic reproduction, small reproductive cycle, and high fecundity [2]. The American Society for Testing and Materials (ASTM), American Public Health Association (APHA), Food and Agriculture Organization of the United Nations (FAO), and US Environmental Protection Agency (US EPA) are the known accepted regulatory units and give the list of some major species used as test organisms [3, 24, 25] (Table 9.1).

In benthic detrital and grazing food chains, the class of crustaceans and rotifers play a major role for primary consumers. Therefore, test methods with rotifers and crustaceans (copepods, amphipods, cladocerans, and insect larvae) have been developed, because they are considered representatives of major freshwater aquatic communities [3].

4 Ecotoxicity Tests

The toxicity tests provide qualitative and quantitative data for an ecological risk assessment, which aids in the evaluation of the adverse effects of environmental pollution on the survival of ecological receptors exposed to toxic compounds [2]. Aquatic ecotoxicology tests could be performed both in the field and laboratory. In the field experiment, multiple species are exposed to multiple toxicants, whereas in laboratory experiments exposure to single species is considered. To quantify the effects, a dose-response relationship is applied at certain criteria for adverse effects or selected end-points (lethal or sublethal effects) [4]. Toxicological assessments determine the potential bioavailability or biological damage caused by toxicants to tissues or organs. They are useful to describe the nature of a noxious effect [2]. Some generally include early life-stage tests for fish and invertebrates, whereas others are full life-cycle tests with invertebrates, such as shrimps and mysids [26]. The methods applied in reservoir water toxicity assessment could be classified as either instream or in situ observations on communities, instream bioassays, and laboratory tests (to assess acute or chronic toxicity to single species) [8] according to exposure time, test situation, and the effects to be measured [27].

A series of standard tests for ecotoxicity assessment have been published in scientific literature by widely accepted regulatory agencies, such as the APHA, US

Protozoans Ciliates Internal pyriformis APHA Vermes Annelids Namelids Namelids Branchiura sowerbyi APHA, FAO Immodrilus hoffmeisteri APHA, FAO Stylodrilus heringianus APHA APHA FAO Tubifex tubifex APHA FAO Playhelminthes Playhelminthes Dugesia tigrina ASTM ASTM Plays heterostropha ASTM Molluscs Gastropods ASTM Playsa heterostropha ASTM Physa heterostropha ASTM ASTM Playhelminthes Gammarus lacustris ASTM, USEPA, APHA, FAO Crustaceans Gammarus facciatus ASTM, USEPA, APHA, FAO Gammarus fasciatus ASTM, USEPA, APHA, VAO Physa heterostropha ASTM, USEPA, APHA, US EPA, APIA FAO Paotoporcia affinis APHA Valella azteca APHA, FAO Gammarus fasciatus ASTM, US EPA, OECD Daphnia magna APHA, FAO, ASTM, US EPA, OECD Daphnia nagna APHA, FAO, ASTM, US EPA, OECD Daphnia pulexa ASTM Daphnia spp. OECD Ceriodaplmia spp. <th>Phylum</th> <th>Order/Species</th> <th>Listed by organization</th>	Phylum	Order/Species	Listed by organization	
Tetrahymena pyriformisAPHAVermesAnnelidsBranchiura sowerbyiAPHA, FAOLinnodrilus hoffmeisteriAPHA, FAOStylodrilus heringianusAPHATubifex tubifexAPHAPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAnnicola limosaASTMPhysa integraASTMPhysa integraASTMPhysa integraASTMCrustaceansGamrarus lacustrisAstm, USEPA, APHA, FAOGarnrarus lacustrisASTM, USEPA, APHA, FAOGarnrarus speudolimnaellsASTM, USEPA, APHA, FAOGarnarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHAHyalella spp.US EPABranchiopodsDaphnia magnaASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.QECDCoronectes rusticusAPHA, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPA, FAO, ASTMProcambarus spp.APHA, US EPA, FAO, ASTMProcambarus spp.APHA, US EPA, FAO, ASTMProcambarus spp.APHA, US EPA, FAO, ASTMProcambarus spp.ASTMProcambarus spp.ASTMProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA	Protozoans	Ciliates		
VermesAnnelidsBranchiura sowerbyiAPHA, FAOLimnodrilus hoffmeisteriAPHA, FAOStylodrilus heringianusAPHATubifex tubifexAPHATubifex tubifexAPHA, FAOPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAnnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGammarus lacustrisASTM, USEPA, APHA, FAOGarnrarus lacustrisASTM, APHA, US EPA, FAOGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHA, FAOPontoporeia affinisAPHA, FAODaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.US EPABranchiopodsDaphnia spp.OECDCeriodaplinia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMDrocnectes rusticusAPHAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPacambarus spp.ASTMPacifastacus lenisculusASTMPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA		Tetrahymena pyriformis	АРНА	
Branchiura sowerbyiAPHA, FAOLimnodrilus hoffmeisteriAPHA, FAOStylodrilus heringianusAPHATubifex tubifexAPHATubifex tubifexAPHA, FAOPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAmnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGarrnarus lacustrisASTM, USEPA, APHA, FAOGarrnrvurs pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.US EPACambarus spp.APHA, US EPA, FAO, ASTMDecapodsCambarus spp.APHA, US EPA, FAO, ASTMDrocnectes rusticusAPHAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalalemonetes cummingiAPHA	Vermes	Annelids		
Limnodrilus hoffmeisteriAPHA, FAOStylodrilus heringianusAPHATubifex tubifexAPHA, FAOPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAmnicola limosaASTMPhysa integraASTMPhysa integraASTMCrustaceansAmphipodsGammarus lacustrisASTM, USEPA, APHA, FAOGarnravus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHAPhyalella aztecaAPHAHyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCrostectes rusticusAPHA, US EPA, FAO, ASTMOrconectes rusticusAPHA, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, ASTMProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPAPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPatifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHA		Branchiura sowerbyi	APHA, FAO	
Stylodrilus heringianusAPHATubifex tubifexAPHA, FAOPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAnnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansGamrarus lacustrisAstm, USEPA, APHA, FAOGarnrwus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHAPhyalella aztecaAPHAPhyalella aspp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.OECDCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes rusticusAPHAProcambarus spp.ASTM, US EPA, ASTMProcambarus spp.ASTMProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA		Limnodrilus hoffmeisteri	APHA, FAO	
Tubifex tubifexAPHA, FAOPlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAmnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGamrnarus lacustrisASTM, USEPA, APHA, FAOGarnrrwrus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHAHyalella aspp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTMProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPataASTMPataASTMCambarus spp.ASTMPataASTMPataASTMPalaemonetes cummingiAPHA		Stylodrilus heringianus	АРНА	
PlatyhelminthesDugesia tigrinaASTMMolluscsGastropodsAmnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGammarus lacustrisASTM, USEPA, APHA, FAOGarmrrwrus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHAHyalella astecaAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHA, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA		Tubifex tubifex	APHA, FAO	
Dugesia tigrinaASTMMolluscsGastropodsAmnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGammarus lacustrisASTM, USEPA, APHA, FAOGarnrwus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTMDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.OECDCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPA, FAO, ASTMProcambarus spp.APHA, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHA		Platyhelminthes		
Molluscs Gastropods Amnicola limosa ASTM Physa integra ASTM Physa integra ASTM Physa heterostropha ASTM Crustaceans Amphipods Gammarus lacustris ASTM, USEPA, APHA, FAO Garnrwus pseudolimnaells ASTM, APHA, US EPA Gammarus fasciatus ASTM, APHA, US EPA, FAO Hyalella azteca APHA, FAO Pontoporeia affinis APHA Hyalella spp. US EPA Branchiopods Daphnia magna Daphnia pulex ASTM, US EPA, OECD Daphnia pulex ASTM Daphnia spp. OECD Ceriodaplmia spp. OECD Ceriodaplmia spp. QECD Ceriodaplmia spp. APHA, US EPA, FAO, ASTM Orconectes rusticus APHA Orconectes spp. ASTM, US EPA Procambarus spp. ASTM Procambarus spp. ASTM Pacifastacus lenisculus ASTM, US EPA Procambarus spp. ASTM Pacifastacus lenisculus ASTM, US EPA Palaemonetes cummingi <t< td=""><td></td><td>Dugesia tigrina</td><td>ASTM</td></t<>		Dugesia tigrina	ASTM	
Amnicola limosaASTMPhysa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGamrarus lacustrisASTM, USEPA, APHA, FAOGarnrwus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAUS EPADecapodsPHAProcambarus spp.ASTM, US EPA, ASTM, US EPA, APHAOrconectes spp.ASTM, US EPA, FAO, ASTMOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA	Molluses	Gastropods		
Physa integraASTMPhysa heterostrophaASTMCrustaceansAmphipodsGammarus lacustrisASTM, USEPA, APHA, FAOGammarus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTM, US EPA, OECDDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCeriodaplmia spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHA, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMDachnia spp.OECDCeriodaplmia spp.US EPADecapodsProcambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHADAPHADacharus spp.ASTM, US EPA, FAO, ASTMDricenbarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPala		Amnicola limosa	ASTM	
Physa heterostrophaASTMCrustaceansAmphipodsGamrarus lacustrisASTM, USEPA, APHA, FAOGarnrwrus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTMDaphnia pulexASTMDaphnia spp.OECDCeriodaplmia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAProcambarus spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMProcambarus spp.ASTM, US EPA, FAO, ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPaluemonetes cummingiAPHA		Physa integra	ASTM	
CrustaceansAmphipodsGamrnarus lacustrisASTM, USEPA, APHA, FAOGamrnarus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTM, US EPA, OECDDaphnia pulexASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPaleamonetes cunmingiAPHAPalaemonetes cunmingiAPHA		Physa heterostropha	ASTM	
Gammarus lacustrisASTM, USEPA, APHA, FAOGammarus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPulaASTM, US EPAPalaemonetes cummingiAPHAPulaASTM, US EPAPalaemonetes cummingiAPHAPulaAPHAPulaASTMPulaAPHAPulaASTMPulaASTMPulaASTMPulaASTM, US EPAPulaASTM, US EPAPulaASTMPulaASTMPulaASTMPulaASTMPulaASTMPulaASTMPulaASTMPulaASTM <td>Crustaceans</td> <td>Amphipods</td> <td></td>	Crustaceans	Amphipods		
Garnrrwrus pseudolimnaellsASTM, APHA, US EPAGammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAP.APHAPalaemonetes cummingiAPHAP.ASTM, US EPA		Gamrnarus lacustris	ASTM, USEPA, APHA, FAO	
Gammarus fasciatusASTM, APHA, US EPA, FAOHyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphnia pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Orconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAP.ASTM, US EPA		Garnrrwrus pseudoIimnaells	ASTM, APHA, US EPA	
Hyalella aztecaAPHA, FAOPontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHADutAPHADetarASTM, US EPAPalaemonetes cummingiAPHA		Gammarus fasciatus	ASTM, APHA, US EPA, FAO	
Pontoporeia affinisAPHAHyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAD.DASTM, US EPAPalaemonetes cummingiAPHAD.ASTM, US EPA		Hyalella azteca	APHA, FAO	
Hyalella spp.US EPABranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHADetabarus spp.ASTM, US EPA		Pontoporeia affinis	АРНА	
BranchiopodsDaphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHADuticitational control of the second c		Hyalella spp.	US EPA	
Daphnia magnaAPHA, FAO, ASTM, US EPA, OECDDaphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAD.APHAD.ASTM, US EPA		Branchiopods		
Daphma pulexASTM, US EPA, OECDDaphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHA		Daphnia magna	APHA, FAO, ASTM, US EPA, OECD	
Daphnia pulicanaASTMDaphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAD.ASTM, US EPA		Daphma pulex	ASTM, US EPA, OECD	
Daphnia spp.OECDCeriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHADr.ASTM, US EPA		Daphnia pulicana	ASTM	
Ceriodaplmia spp.US EPADecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAD.ASTM, US EPA		Daphnia spp.	OECD	
DecapodsCambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAD.APHA		Ceriodaplmia spp.	US EPA	
Cambarus spp.APHA, US EPA, FAO, ASTMOrconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAP.ASTM		Decapods		
Orconectes rusticusAPHAOrconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA		Cambarus spp.	APHA, US EPA, FAO, ASTM	
Orconectes spp.ASTM, US EPAProcambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHA		Orconectes rusticus	АРНА	
Procambarus spp.ASTMPacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHAPalaemonetes cummingiAPHA		Orconectes spp.	ASTM, US EPA	
Pacifastacus lenisculusASTM, US EPAPalaemonetes cummingiAPHADalamonetes cummingiAPHA		Procambarus spp.	ASTM	
Palaemonetes cummingi APHA		Pacifastacus lenisculus	ASTM, US EPA	
		Palaemonetes cummingi	АРНА	
Palaemonetes APHA		Palaemonetes	АРНА	
Mysids		Mysids		
Mysis relicta APHA, FAO		Mysis relicta	APHA, FAO	
Insect larvae Plecopterans	Insect larvae	Plecopterans		
Pteronarcy dorsata APHA		Pteronarcy dorsata	АРНА	
Pteronarcy califonica APHA		Pteronarcy califonica	АРНА	
Pteronarcy spp. ASTM, FAO		Pteronarcy spp.	ASTM, FAO	
Hesperoperla lycorias APHA		Hesperoperla lycorias	АРНА	

 Table 9.1
 A list of major freshwater invertebrate species used for toxicity assessments

EPA, International Organization for Standardization (ISO), Environment and Climate Change Canada (ECCC), and Organization for Economic Co-operation and Development (OECD) [4]. The nature of the test design is subjected to available resources, time, laboratories, type of chemical pollutants, and physical and chemical properties of environmental matrices under investigation [4]. These tests are useful to characterize the spatial distribution of toxicity and temporal trends by monitoring at different times in different locations [2] (Table 9.2).

Organisms are exposed to the selected chemical pollutant based on four systems, including static assemblies (organisms are exposed to toxicants in standing water and the test solution is not altered [27]), the renewal test (a static test but the test solution is renewed after intervals), the recirculating system (organisms are retained in a chamber and the test solution is pumped inside through a filter maintaining oxygen and nutrient levels, which is usually used for chronic exposures), and a flow-through test (the test organisms are exposed to a test solution with flow through the compartments in which the test organisms are placed) [4].

4.1 Acute Toxicity Tests

Acute toxicity tests are short exposure period tests that generally range from a few hours to several days, and are used to assess the growth, immobility, and mortality of an organism [27]. In short-term exposure, organisms are exposed to a high dose of chemicals either in single or multiple events and produce instantaneous effects. Such tests are carried out using test species during a specific life cycle of the organism, usually called a partial life-cycle test. These tests are not considered valid if mortality is greater than 10% in the control [56].

Presently, *Daphnia* is the most commonly used for acute toxicity tests, which is recognized as the standard aquatic toxicity assay. These tests are comparatively easily performed when a good culture is established and are cost-effective as they are of shorter duration (48 vs. 96 hrs). Moreover, the maintenance of their culture needs less effort, equipment, and space [3]. Ecotoxicology protocols recommend *Daphnia magna* or *Daphnia pulex* as a test invertebrate species. *Daphnia magna* less than 24 hours of age are commonly used for this test [2]. Mortality and immobility are noted at 24 and 48 hours and matched with the values of control. The concentration of the test substance, dissolved oxygen, and pH are examined at the beginning and the completion of the test. The LC₅₀ or EC₅₀ is a major parameter to measure acute toxicity [2, 3].

Embryos of rotifer (*Brachionus calyciflorus*) used in this toxicity test are hatched from cysts by incubation in dilute water for 24 h at 25 °C in darkness. Commercially cultured strains of cysts are recommended because of their well-characterized sensitivity. In each culture plate, 10 newly hatched neonates are placed with 1 ml of test solution. At the end of the test, living and dead rotifers are counted and percent survival is noted with a dissecting microscope in each test chamber [57].

Species	Stressors tested	Endpoint measured	Type of test	Reference
P. nana	UV	Mortality, reproduc- tive parameters	life cycle	[28]
P. nana	Food (5 microalgae)	Fecundity, mortality, growth	diet	[29]
P. nana	Temperature, salinity,	Gene expression	mRNA expression	[30]
P. nana	Gamma radioisotope	Growth, fecundity	radiation	[28]
P. nana	Heavy metals, EDCs	Molecular, Vg expression	EDCs, HM	[31]
P. nana	UV	Clutch number, growth pattern, assimilation of diet, DNA repair	UV exposition	[32, 33]
P. nana	Light intensity	Survival, growth, productivity	light exposure	[34]
P. nana	Density, antioxidants	Naupliar production, gene expression	culture density	[35]
Acartia tonsa	Diflubenzuron	Reproductive rate, development Mor- phological abnormalities,	Life-cycle (part.)	[36]
Acartia tonsa	TBT	Development, acute toxicity, and mortality	Life-cycle (part.)	[37]
	Methyltestosterone, fenarimol		Exposition	
Acartia tonsa	Nickel	Mortality	Acute	[38]
Acartia tonsa	Zinc	Mortality	Acute	[38]
Acartia tonsa	Synthetic musks	Acute toxicity devel- opment, mortality	Partial life- cycle (larval development)	[39]
Acartia tonsa	Polybrominated diphenyl ether hydroxyecdysone, tetrabromobisphenol A	Acute toxicity devel- opment, mortality	Partial life- cycle(larval development)	[40]
Amphiascus tenuiremis	Atrazine	Reproduction, popu- lation growth rate, malformation,	Acute	[41]
Amphiascus tenuiremis	Fipronil	Mortality, reproduction	Population growth rate, acute	[42]
Amphiascus tenuiremis	Fipronil	Reproductive rate	Full life- cycle	[43]
Amphiascus tenuiremis	Fipronil, vitellin (VTN)		Full life- cycle	[44]
Amphiascus tenuiremis	Single-walled carbon nanotubes		Full life- cycle	[45]

Table 9.2 Studies reporting the assessment of various toxicity endpoints using invertebrates species

(continued)

Species	Stressors tested	Endpoint measured	Type of test	Reference
Eurytemora affinis	Cadmium, copper	Mortality	Acute	[46]
Daphnia magna	nonylphenol	Mortality, Reproduction	Acute	[47]
Daphnia magna	Juvenile hormone	Mortality, Immobility		[48]
Ceriodaphnia dubia	Styrene	Reproduction rate	Chronic	[49]
Daphina magna	Bisphenol A	Reproduction rate	chronic	[50]
Daphnia magna	Cadmium	Survival rate	Acute	[51]
Ceriodaphnia dubia and Daphnia carinata	Zinc, Copper, Lead	Reproduction and Mortality	Acute	[52]
Daphnia pulex	Organic selenium	Mortality	Acute	[53]
Daphnia magna	Fenvalerate (insecticide)	Mortality, survival	Acute	[54]
Daphnia magna	Propiconazole (pesticide)	Reproduction rate, offspring growth	Life cycle	[55]

Table 9.2 (continued)

4.2 Chronic Toxicity Tests

Long-term exposure tests that last from weeks to months or years relative to the life span (usually >10%) are known as chronic tests. The test organisms have to come in contact with low and frequent doses of chemical pollutants. These toxicity tests allow the long-term evaluation of concentrations of chemicals and produce sublethal effects [27]. Chronic toxicity tests are considered full life cycle assessments as they cover the full reproductive life cycle and are not regarded as valid if the percentage of mortality is more than 20% in the control sample. The results are stated in EC50s, the lowest observed effects level (LOECs), and no observed effects level (NOEL). There are sub-chronic exposures that are applied in early life stages and include the sensitive life stage of the organism (also called the critical life stage). Endpoints include reproduction survival and developmental changes.

4.2.1 The Survival and Reproduction Test of Ceriodaphnia dubia

The *Ceriodaphnia dubia* reproduction and survival test is a seven days chronic toxicity assay, which was proposed by Mount and Norberg [58]. Later, it was adopted by the US EPA and ASTM as a standard toxicity assay for chronic toxicity tests. *Ceriodaphnia dubia* is preferred because of its short exposure time, compared

to *Daphina magna* [58]. In this test, juvenile *Ceriodaphnia* is treated with different concentrations of pollutants in a static renewal system for 7 days. Each neonate is placed in 15 mL of test solution. During this period, the control organisms produce three broods. Each day organisms are fed and numbers of young produced and survived are noted. Results are centered on the reproduction capability and survival ratio. Developmental parameters, LOEC, and NOEC are determined [3]. Results are considered valid when 60% of survived females must produce three broods with each female producing 15 offspring and at least 80% of overall control organisms will survive [59].

4.2.2 The Reproduction Test of Daphina magna

In this chronic bioassay, the effect of a contaminant on the reproduction of *Daphnia* magna during a 21-day exposure is assessed. The test is set up by placing young female neonates (n = 60) in a test solution of five different concentrations (each neonate in 50–80 mL solution) in a static renewal system for 3 weeks. The numbers of surviving and produced organisms are determined after every two days. The experiment results are based upon the comparison of the offspring number per surviving female in the toxicant treatments with the reproductive output, compared to controls. The test is ended on the 21st day, and the LOEC and NOEC are calculated [3].

5 Bioaccumulation

Bioaccumulation tests are commonly utilized for chemicals that are hydrophobic as these chemicals could be stored in the fatty tissues of aquatic species. Due to they have low water solubilities, the species can accumulate such toxicants at a high level [60]. Their storage within the organism's body may cause cumulative toxicity. To predict the concentration of chemicals within the body of organisms, bioconcentration factors (BCF) are used. It is the ratio of average chemical concentration in the tissues of the organism, compared to the mean concentration in the water. There are different standard methods for reservoir bioaccumulation tests. Bioaccumulation tests could be performed in many ways like active bioaccumulation, passive biomonitoring laboratory setup, and simulating methods [4]. Four types of bioaccumulation monitoring are usually followed, such as stream active bioaccumulation monitoring, stream passive bioaccumulation monitoring, and laboratory simulation approaches. For carrying out active biomonitoring, invertebrate species are collected from unpolluted sites and then exposed to polluted sites in the field for a definite duration. The most commonly used organisms are freshwater mussels like Anodonta anatina and Dreissena polymorpha because of their extensive existence and potential of resisting the high concentration of toxic substances [61].

For regulatory purposes, bioconcentration factors are determined according to the OECD documents using fish flow-through tests, which are expensive, timeconsuming, and many organisms during the experiments. To overcome these issues, the freshwater amphipod species, Hyalella Azteca, is used as a substitute test species for such studies because of its high reproduction rate and fast growth, which meets the requirement of a large number of organisms for study [62]. Previous numerous laboratory tests have been conducted to assess the bioconcentrations of metals, organo-metals, insecticide DDT, and polycyclic aromatic hydrocarbons in Hyalella *azteca* [63-70]. Amphipods are raised in the laboratory, and 50 test organisms are in a 2 L flask. After 8-week development, they become mature enough to be used for bioconcentration studies. A 20 L of test solution is stocked with about 1200 test organisms and kept in 8/16 h dark. The uptake phase lasts from 2-12 days and organisms are exposed to a constant concentration of test chemicals [71]. For each chemical, the exposure period is adjusted based on the previous studies to make sure whether the equilibrium state has been achieved or not. After the completion of uptake, test organisms are moved into a new aquarium. Samples of test organisms are analyzed according to the standard procedure. BCF factors are calculated according to the quantity of the test chemical in the tissue represented as (Ch) and their comparable levels in water (Cw), which is given by the below equation [62].

BCFss = Ch/Cw

Besides, amphipods, calanoid copepods, *Pseudodiaptomus annandalei*, and *Eurytemora affinis* have also been used to study the bioaccumulation of heavy metals from surface water [72]. *Gammarus pulex* and *Notonecta glauca* were reported to study the uptake and bioaccumulation of pharmaceuticals [73].

6 Conclusion

Formerly, various chronic and acute toxicity tests were developed for reservoir ecotoxicity analysis, but only a few tests have been accepted as standard toxicity tests for routine toxicity assessments. The best recognized are chronic and acute toxicity tests with *Daphnids*. Invertebrates are ecologically diverse and there is great phylogenetic complexity among them. Consequently, they are sensitive to many ecotoxicological responses and exhibit variability after exposure to different stressors, which are considered suitable for ecotoxicity testing. Research efforts for developing new standard toxicity procedures using invertebrates are required.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of interest The author declares no conflict of interest.

References

- P. Palma, P. Alvarenga, V. Palma, C. Matos, R.M. Fernandes, A. Soares, I.R. Barbosa, Evaluation of surface water quality using an ecotoxicological approach: a case study of the Alqueva Reservoir (Portugal). Environ. Sci. Pollut. Res. 17, 703–716 (2010)
- A. Valavanidis, T. Vlachogianni, Ecotoxicity test methods and ecological risk assessment. Ecosystems 10, 13 (2015)
- 3. P.P. Calow, Handbook of ecotoxicology (Wiley, 2009)
- 4. G.M. Rand, S.R. Petrocelli, *Fundamentals of Aquatic Toxicology: Methods and Applications* (FMC Corp, Princeton, NJ, 1985)
- L. Lagadic, T. Caquet, Invertebrates in testing of environmental chemicals: are they alternatives? Environ. Health Perspect. 106, 593–611 (1998)
- 6. D. Papillon, M.J. Telford, Evolution of Hox3 and ftz in arthropods: insights from the crustacean Daphnia pulex. Dev. Genes Evol. **217**, 315–322 (2007)
- H.-U. Dahms, E.-J. Won, H.-S. Kim, J. Han, H.G. Park, S. Souissi, S. Raisuddin, J.-S. Lee, Potential of the small cyclopoid copepod Paracyclopina nana as an invertebrate model for ecotoxicity testing. Aquat. Toxicol. 180, 282–294 (2016)
- 8. R. Knoben, M. van Oirschot, C. Roos, *Biological Assessment Methods for Watercourses* (RIZA, 1995)
- A. Chaumot, B. Ferrari, O. Geffard, J. Garric, Ecotoxicology, aquatic invertebrates. Reference module in biochemical sciences, in *Encyclopedia of Toxicology*, (Elsevier Academic Press, 2014)
- J.-S. Rhee, I.T. Yu, B.-M. Kim, C.-B. Jeong, K.-W. Lee, M.-J. Kim, S.-J. Lee, G.S. Park, J.-S. Lee, Copper induces apoptotic cell death through reactive oxygen species-triggered oxidative stress in the intertidal copepod Tigriopus japonicus. Aquat. Toxicol. 132, 182–189 (2013)
- 11. J. Han, E.-J. Won, B.-Y. Lee, U.-K. Hwang, I.-C. Kim, J.H. Yim, K.M.Y. Leung, Y.S. Lee, J.-S. Lee, Gamma rays induce DNA damage and oxidative stress associated with impaired growth and reproduction in the copepod Tigriopus japonicus. Aquat. Toxicol. 152, 264–272 (2014)
- E.-J. Won, J.-S. Lee, Gamma radiation induces growth retardation, impaired egg production, and oxidative stress in the marine copepod Paracyclopina nana. Aquat. Toxicol. 150, 17–26 (2014)
- J.-P. Wu, H.-C. Chen, M.-H. Li, Bioaccumulation and toxicodynamics of cadmium to freshwater planarian and the protective effect of N-acetylcysteine. Arch. Environ. Contam. Toxicol. 63, 220–229 (2012)
- 14. J.-P. Wu, H.-L. Lee, M.-H. Li, Cadmium neurotoxicity to a freshwater planarian. Arch. Environ. Contam. Toxicol. 67, 639–650 (2014)
- J.-P. Wu, M.-H. Li, Inhibitory effects of pain relief drugs on neurological enzymes: implications on their potential neurotoxicity to aquatic animals. Environ. Toxicol. Pharmacol. **39**, 898–905 (2015)
- 16. J.-P. Wu, M.-H. Li, Low uptakes of Cd, Cu, and Zn in Dugesia japonica, a freshwater planarian with higher tolerance to metals. Chem. Ecol. **33**, 257–269 (2017)
- X. Zhang, Y. Mo, L. Zhou, Y. Wang, Z. Wang, B. Zhao, Induction of hsp70, hsp90, and catalase activity in planarian Dugesia japonica exposed to cadmium. Toxicol. Ind. Health 32, 1373–1380 (2016)
- T. Knakievicz, H.B. Ferreira, Evaluation of copper effects upon Girardia tigrina freshwater planarians based on a set of biomarkers. Chemosphere 71, 419–428 (2008)
- F. Calevro, S. Campani, C. Filippi, R. Batistoni, P. Deri, S. Bucci, M. Ragghianti, G. Mancino, Bioassays for testing effects of Al, Cr and Cd using development in the amphibian Pleurodeles

waltl and regeneration in the planarian Dugesia etrusca. Aquat. Ecosyst. Health Manage. 2, 281–288 (1999)

- 20. J.-P. Wu, M.-H. Li, The use of freshwater planarians in environmental toxicology studies: advantages and potential. Ecotoxicol. Environ. Saf. **161**, 45–56 (2018)
- M.-H. Li, Acute toxicity of benzophenone-type UV filters and paraben preservatives to freshwater planarian, Dugesia japonica. Toxicol. Environ. Chem. 94, 566–573 (2012)
- M.-H. Li, Acute toxicity of 30 pharmaceutically active compounds to freshwater planarians, Dugesia japonica. Toxicol. Environ. Chem. 95, 1157–1170 (2013)
- 23. T. Knakievicz, Planarians as invertebrate bioindicators in freshwater environmental quality: the biomarkers approach. Ecotoxicol. Environ. Contamin. 9, 1–12 (2014)
- 24. NERC, Methods for acute toxicity tests with fish, macroinvertebrates, and amphibians (National Environmental Research Center, 1975)
- 25. ASTM, Standard guide for conducting acute toxicity tests on test materials with fishes, macroinvertebrates, and amphibians, pp. 22 (2002)
- B. Anderson, B. Phillips, Saltwater Toxicity Tests, in *Marine Ecotoxicology*, (Elsevier, 2016), pp. 167–197
- 27. C. Palmer, R. Berold, W. Muller, Water Research Commission TT No 217/04, in, Water Research Commission, Pretoria, South Africa (2004)
- 28. E.-J. Won, Y. Lee, J. Han, U.-K. Hwang, K.-H. Shin, H.G. Park, J.-S. Lee, Effects of UV radiation on hatching, lipid peroxidation, and fatty acid composition in the copepod Paracyclopina nana. Comp. Biochem. Physiol. Part C. Toxicol. Pharmacol. 165, 60–66 (2014)
- K.W. Lee, H.G. Park, S.-M. Lee, H.-K. Kang, Effects of diets on the growth of the brackish water cyclopoid copepod Paracyclopina nana Smirnov. Aquaculture 256, 346–353 (2006)
- C.-B. Jeong, M.C. Lee, K.-W. Lee, J.S. Seo, H.G. Park, J.-S. Rhee, J.-S. Lee, Identification and molecular characterization of dorsal and dorsal-like genes in the cyclopoid copepod Paracyclopina nana. Mar. Genomics 24, 319–327 (2015)
- D.-S. Hwang, K.-W. Lee, J. Han, H.G. Park, J. Lee, Y.-M. Lee, J.-S. Lee, Molecular characterization and expression of vitellogenin (Vg) genes from the cyclopoid copepod, Paracyclopina nana exposed to heavy metals. Comp. Biochem. Physiol. Part C. Toxicol. Pharmacol. 151, 360–368 (2010)
- E.-J. Won, H.-U. Dahms, K.S. Kumar, K.-H. Shin, J.-S. Lee, An integrated view of gamma radiation effects on marine fauna: from molecules to ecosystems. Environ. Sci. Pollut. Res. 22, 17443–17452 (2015)
- 33. E.-J. Won, J. Han, Y. Lee, K.S. Kumar, K.-H. Shin, S.-J. Lee, H.G. Park, J.-S. Lee, In vivo effects of UV radiation on multiple endpoints and expression profiles of DNA repair and heat shock protein (Hsp) genes in the cycloid copepod Paracyclopina nana. Aquat. Toxicol. 165, 1–8 (2015)
- 34. K.-W. Lee, J.-H. Kang, H.-G. Park, Effect of light intensity on survival, growth and productivity of the cyclopoid copepod Paracyclopina nana: a laboratory study, Korean. J. Fish. Aquat. Sci. 44, 671–676 (2011)
- 35. K.-W. Lee, J.-S. Rhee, J. Han, H.G. Park, J.-S. Lee, Effect of culture density and antioxidants on naupliar production and gene expression of the cyclopoid copepod, Paracyclopina nana. Comp. Biochem. Physiol. Part A. Mol. Integr. Physiol. 161, 145–152 (2012)
- 36. P.A. Tester, J. Costlow, Effect of insect growth regulator Dimilin (TH 6040) on fecundity and egg viability of the marine copepod Acartia tonsa. Mar. Ecol. Prog. Ser. 5, 297–302 (1981)
- K.O. Kusk, S. Petersen, Acute and chronic toxicity of tributyltin and linear alkylbenzene sulfonate to the marine copepod Acartia tonsa. Environ. Toxicol. Chem. Int. J. 16, 1629–1633 (1997)
- S. Barka, J.-F. Pavillon, J.-C. Amiard, Influence of different essential and non-essential metals on MTLP levels in the copepod Tigriopus brevicornis. Comp. Biochem. Physiol. Part C. Toxicol. Pharmacol. 128, 479–493 (2001)

- L. Wollenberger, M. Breitholtz, K.O. Kusk, B.-E. Bengtsson, Inhibition of larval development of the marine copepod Acartia tonsa by four synthetic musk substances. Sci. Total Environ. 305, 53–64 (2003)
- L. Wollenberger, L. Dinan, M. Breitholtz, Brominated flame retardants: activities in a crustacean development test and in an ecdysteroid screening assay. Environ. Toxicol. Chem. Int. J. 24, 400–407 (2005)
- A.C. Bejarano, G.T. Chandler, Reproductive and developmental effects of atrazine on the estuarine meiobenthic copepod Amphiascus tenuiremis. Environ. Toxicol. Chem. Int. J. 22, 3009–3016 (2003)
- G.T. Chandler, T.L. Cary, D.C. Volz, S.S. Walse, J.L. Ferry, S.L. Klosterhaus, Fipronil effects on estuarine copepod (Amphiascus tenuiremis) development, fertility, and reproduction: a rapid life-cycle assay in 96-well microplate format. Environ. Toxicol. Chem. Int. J. 23, 117–124 (2004)
- T.L. Cary, G.T. Chandler, D.C. Volz, S.S. Walse, J.L. Ferry, Phenylpyrazole insecticide fipronil induces male infertility in the estuarine meiobenthic crustacean Amphiascus tenuiremis. Environ. Sci. Technol. 38, 522–528 (2004)
- 44. D.C. Volz, G.T. Chandler, An enzyme-linked immunosorbent assay for lipovitellin quantification in copepods: a screening tool for endocrine toxicity. Environ. Toxicol. Chem. Int. J. 23, 298–305 (2004)
- 45. R.C. Templeton, P.L. Ferguson, K.M. Washburn, W.A. Scrivens, G.T. Chandler, Life-cycle effects of single-walled carbon nanotubes (SWNTs) on an estuarine meiobenthic copepod. Environ. Sci. Technol. 40, 7387–7393 (2006)
- 46. B. Sullivan, E. Buskey, D. Miller, P. Ritacco, Effects of copper and cadmium on growth, swimming and predator avoidance in Eurytemora affinis (Copepoda). Mar. Biol. 77, 299–306 (1983)
- M. Comber, T. Williams, K. Stewart, The effects of nonylphenol on Daphnia magna. Water Res. 27, 273–276 (1993)
- N. Tatarazako, S. Oda, H. Watanabe, M. Morita, T. Iguchi, Juvenile hormone agonists affect the occurrence of male Daphnia. Chemosphere 53, 827–833 (2003)
- N. Tatarazako, Y. Takao, K. Kishi, N. Onikura, K. Arizono, T. Iguchi, Styrene dimers and trimers affect reproduction of daphnid (Ceriodaphnia dubia). Chemosphere 48, 597–601 (2002)
- S.W. Jeong, S.M. Lee, S.S. Yum, T. Iguchi, Y.R. Seo, Genomic expression responses toward bisphenol-A toxicity in Daphnia magna in terms of reproductive activity. Mol. Cell. Toxicol. 9, 149–158 (2013)
- R. Connon, H.L. Hooper, R.M. Sibly, F.-L. Lim, L.-H. Heckmann, D.J. Moore, H. Watanabe, A. Soetaert, K. Cook, S.J. Maund, Linking molecular and population stress responses in Daphnia magna exposed to cadmium. Environ. Sci. Technol. 42, 2181–2188 (2008)
- 52. N.L. Cooper, J.R. Bidwell, A. Kumar, Toxicity of copper, lead, and zinc mixtures to Ceriodaphnia dubia and Daphnia carinata. Ecotoxicol. Environ. Saf. **72**, 1523–1528 (2009)
- T.W. Schultz, S.R. Freeman, J.N. Dumont, Uptake, depuration, and distribution of selenium in Daphnia and its effects on survival and ultrastructure. Arch. Environ. Contam. Toxicol. 9, 23–40 (1980)
- 54. B.J. Pieters, M. Liess, Maternal nutritional state determines the sensitivity of Daphnia magna offspring to short-term Fenvalerate exposure. Aquat. Toxicol. **76**, 268–277 (2006)
- 55. A. Soetaert, L.N. Moens, K. Van der Ven, K. Van Leemput, B. Naudts, R. Blust, W.M. De Coen, Molecular impact of propiconazole on Daphnia magna using a reproduction-related cDNA array. Comp. Biochem. Physiol. Part C. Toxicol. Pharmacol. 142, 66–76 (2006)
- 56. G. Rand, P. Wells, L. McCarty, Introduction to Aquatic Toxicology, Fundamentals of Aquatic Toxicology Effects, Environmental Fate, and Risk Assessment (Taylor and Francis Publishers, North Palm Beach, FL, 1995), pp. 3–67
- 57. B. Anderson, P. Nicely, K. Gilbert, R. Kosaka, J. Hunt, B. Phillips, Overview of Freshwater and Marine Toxicity Tests: A Technical Tool for Ecological Risk Assessment. California

Environmental Protection Agency Office of Environmental Health Hazard Assessment Reproductive and Cancer Hazard Assessment Section Ecotoxicology Unit, in, 2004

- D. Mount, U.T. Norberg, A seven-day life cycle cladoceran toxicity test. Environ. Toxicol. Chem. Int. J. 3, 425–434 (1984)
- 59. USEPA, Short-term Methods for Estimating the Chronic Toxicity of Effluents and Receiving Waters to Freshwater Organisms, in, U.S. Environmental Protection Agency Office of Water (4303T) 1200 Pennsylvania Avenue, NW Washington, DC 20460, 2002, pp. 350
- 60. D. Alexander, Bioaccumulation, Bioconcentration, Biomagnification (Springer, 1999)
- J. Hemelraad, D. Holwerda, D. Zandee, Cadmium kinetics in freshwater clams. I. The pattern of cadmium accumulation in Anodonta cygnea. Arch. Environ. Contam. Toxicol. 15, 1–7 (1986)
- 62. C. Schlechtriem, S. Kampe, H.-J. Bruckert, I. Bischof, I. Ebersbach, V. Kosfeld, M. Kotthoff, C. Schäfers, J. L'Haridon, Bioconcentration studies with the freshwater amphipod Hyalella azteca: are the results predictive of bioconcentration in fish? Environ. Sci. Pollut. Res. 26, 1628–1641 (2019)
- S. Nuutinen, P.F. Landrum, L.J. Schuler, J.V. Kukkonen, M.J. Lydy, Toxicokinetics of organic contaminants in Hyalella azteca. Arch. Environ. Contam. Toxicol. 44, 0467–0475 (2003)
- 64. M. Shuhaimi-Othman, D. Pascoe, Bioconcentration and depuration of copper, cadmium, and zinc mixtures by the freshwater amphipod Hyalella azteca. Ecotoxicol. Environ. Saf. 66, 29–35 (2007)
- W. Norwood, U. Borgmann, D. Dixon, Chronic toxicity of arsenic, cobalt, chromium and manganese to Hyalella azteca in relation to exposure and bioaccumulation. Environ. Pollut. 147, 262–272 (2007)
- 66. L. Alves, U. Borgmann, D. Dixon, Kinetics of uranium uptake in soft water and the effect of body size, bioaccumulation and toxicity to Hyalella azteca. Environ. Pollut. 157, 2239–2247 (2009)
- 67. A.J. Bartlett, U. Borgmann, D.G. Dixon, S.P. Batchelor, R.J. Maguire, Accumulation of tributyltin in Hyalella azteca as an indicator of chronic toxicity: Survival, growth, and reproduction. Environ. Toxicol. Chem. Int. J. 23, 2878–2888 (2004)
- P.F. Landrum, J.A. Steevens, D.C. Gossiaux, M. McElroy, S. Robinson, L. Begnoche, S. Chernyak, J. Hickey, Time-dependent lethal body residues for the toxicity of pentachlorobenzene to Hyalella azteca. Environ. Toxicol. Chem. Int. J. 23, 1335–1343 (2004)
- 69. G.R. Lotufo, P.F. Landrum, M.L. Gedeon, E.A. Tigue, L.R. Herche, Comparative toxicity and toxicokinetics of DDT and its major metabolites in freshwater amphipods. Environ. Toxicol. Chem. Int. J. 19, 368–379 (2000)
- 70. J.-H. Lee, P.F. Landrum, C.-H. Koh, Toxicokinetics and time-dependent PAH toxicity in the amphipod Hyalella azteca. Environ. Sci. Technol. 36, 3124–3130 (2002)
- 71. O.f.E. Co-operation, Development, Test No. 305: Bioaccumulation in Fish: Aqueous and Dietary Exposure, OECD Publishing (2012)
- F.U. Kadiene, B. Ouddane, J.-S. Hwang, S. Souissi, Bioaccumulation of metals in calanoid copepods by oral intake. Sci. Rep. 9, 1–9 (2019)
- M. Meredith-Williams, L.J. Carter, R. Fussell, D. Raffaelli, R. Ashauer, A.B. Boxall, Uptake and depuration of pharmaceuticals in aquatic invertebrates. Environ. Pollut. 165, 250–258 (2012)

Chapter 10 Ecotoxicology Methods of Reservoir Water Using Fish



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter aims at elucidating the use of model fish species in ecotoxicological studies and the methodology of ecotoxicity testing based on fish. Chemicals are released into the reservoir and pose a harmful impact on the biota. Water quality and fish health are associated with each other. Therefore, fish is considered a suitable pollution indicator for monitoring reservoir pollution. To assess the effects of these chemicals, different standardized toxicity methods using fish have been developed by international organizations, which include acute and chronic toxicity tests. Species including zebrafish, rainbow trout, Japanese medaka, and fathead minnow are the most recommended fishes and are selected for a toxicity test according to the contaminant's type and the endpoints to be measured.

Keywords Fish \cdot Aquatic toxicology test \cdot Zebrafish \cdot Acute toxicity \cdot Chronic toxicity

1 Introduction

For decades, the assessment of reservoir ecotoxicological risks and the elucidation of molecular mechanisms of contaminant-induced dysfunctions have remained unambiguous [1]. The formal beginning of ecotoxicology as a separate science was procured in the late 1960s, and reservoir ecotoxicology is proposed in this book. Recently, many studies based on the risks of reservoir ecotoxicology are being conducted including the contamination characteristics, source apportionment, climate change, and potential ecotoxicological effects [2–4]. Environmental toxins profoundly affect the fish population and the health of wildlife and humans. Despite the presence of various species in the aquatic environment, fish is considered a more suitable indicator of water pollution. Fish health is associated with water quality when water contamination affects fish populations by disrupting lifespan, embryonic development, and reproductive health [5]. Toxicological testing using fish has been a common practice in the past, and fish death has been recognized as a general water pollution indicator. Current fish toxicology has moved towards mechanistic and multidisciplinary approaches [5].

2 Ecotoxicological Importance of Fish and Its Use in Ecotoxicity Testing

Fish is being utilized as sentinel organisms in ecotoxicological studies because of their capability of inhabiting all zones of the aquatic habitat where prevails suitable conditions for their survival, feeding habits, physiological diversity, reproductive strategies, and economic significance [6]. For anthropogenic pollutants, reservoir aquatic environments are acknowledged as the absolute sink, and fish death could be observed as a consequence of the toxic action of the pollutants. Thus, fish is commonly recognized as the substitute for evaluating the deleterious effects of contaminants on reservoir ecosystem health. The first ecosystem disturbances studied using fish were the impacts of mine-tailing effluents as reported by Carpenter [7]. In the middle of the twentieth century, techniques were standardized for acute fish toxicity testing. Toxicological tests were developed and validated to the internationally agreed testing procedures used by laboratories, government, and industry to illustrate prospective hazards of novel and existing contaminants. About 20% of the tests recognized by the OECD (Organization for Economic Co-operation and Development) for evaluating health effects on living systems are conducted using the fish model [8]. In the 1960s, concerns about long-term exposure of organisms to contaminants were raised and flow-through techniques were developed. Various biomarkers and endpoints were developed by studying the effects on early-life stages, reproductive cycles, and complete life cycles. New molecular techniques were developed and research was more concentrated on the detection and understanding of the toxicity mechanism of chemical substances [5]. The practice of using wild fish population indices for the assessment of ecological status in water bodies has also grown since the 2000s [9, 10]. Recently, the assessment of emerging contaminants, such as microplastics, pharmaceuticals, and pesticides, has been performed based on fish acute and chronic toxicity, and lethal and sublethal effects are evaluated using novel approaches of modern molecular biology. The acute and chronic toxicity of emerging chemicals using different fish species are presented in Table 10.1.

3 Fish Models

For ecotoxicity testing, fish species are selected based on different factors, such as size, ease of laboratory maintenance, suitability for testing, known sensitivity, available data, and the availability of test procedures and protocols that could be followed [8]. Zebrafish, rainbow trout (*Oncorhynchus mykiss*), Japanese medaka (*Oryzias latipes*), and fathead minnow (*Pimephales promelas*) are the most common species used for ecotoxicological studies. Zebrafish is considered the most popular testing model, because it shares common anatomy and development features, metabolism, as well as physiological and chemical-induced organ responses with

		Exposure		
Contaminant	Test Organism	duration	Toxicity	References
Diazinon	Anabas testudineus	96 h	Acute (LC 50) 6.55 ppm	[11]
Diazinon	Channa punctatus	96 h	Acute (LC 50) 3.09 ppm	[11]
Cadmium, cop- per, zinc	Rainbow trout (O. mykiss)	-	Acute and chronic (LC50 116 (µg/L)	[12]
Permethrin	Cyprinus carpio	24 h	Acute (LC 50) 35 µg/L	[13]
Dichlorovinyl Dimethyl phosphate	Zebrafish	24 h	Acute LC 50 39.75 mg/L,	[14]
Methyl parathion	Catla catla	96 h	Acute (LC 50) 4.8 ppm	[15]
Cypermethrin	Colisa fasciatus	96 h	Acute (LC 50) 0.02 mg/L	[16]
Malathion	Labeo rohita	96 h	Acute (LC 50) 15 mg/L	[17]
Endosulfan	Channa striatus	96 h	Acute (LC 50) 0.0035 ppm	[18]
Cypermethrin	Labeo rohita	96 h	Acute (LC 50) 4.0 µ/L	[19]
Endosulfan	Labeo rohita	96 h	Acute (LC 50) 2.15 µg/L	[20]
Dimethoate	Labeo rohita	96 h	Acute (LC 50) 24.55 µg/L	[21]
Poly Ethylene	Pomatoschistus microps	96 h	Acute: Reduced AChE activity	[22]
Poly Ethylene	Japanese medaka	2 months	Chronic: Histopathological alterations	[23]
Poly Ethylene	Japanese medaka	2 months	Chronic: Altered expression of a gene mediated by the estrogen receptor in the liver	[24]
Poly styrene	Zebrafish	7 days	Uptake and bioconcentration Accumulated in fish gills, liver, and gut	[25]
Chromium	Channa punctatus	60 d, 120 d	Chronic: 2.6 mg L^{-1} LDH activity inhibited in liver and kidney.	[26]
3-benzylidene camphor	Pimephales promelas	14 d, 21 d	Chronic: VTG LOEC 435, 74 μ g L ⁻¹	[27]
Oxybenzone UV filter	Oncorhynchus mykiss	14 d	Chronic: VTG LOEC 749 μ g L ⁻¹	[28]
Triclosan	Oncorhynchus mykiss	96 d	Chronic: Hatching, Survival No Effect, LOEC 71.3 μ g L ⁻¹	[29]
Triclosan	Oryzias latipes	14 d	Chronic: Hatching LOEC 213 μ g L ⁻¹	[30]
Triclosan	Oryzias latipes	21 d	Chronic Growth, Fecundity, HSI and GSId, VTGe LOEC 200 μ g L ⁻¹ , No Effect,	[30]
Fluoride	Salmon	-	Chronic: 0.5 mg/l Significant disruption of PGC migration	[31]

 Table 10.1
 Acute and chronic fish toxicity studies

(continued)

Contoninont	Test Oreanism	Exposure	Touisite	Dafamanaaa
Contaminant	Test Organism	duration	Toxicity	References
3-benzylidene	Pimephales	14 d	VTG, Reproduction, Gonad	[32]
camphor	promelas		Histology LOEC 434.6,	
			$74 \ \mu g \ L^{-1}$	
Benzylparaben	Pimephales	48 h	Acute LC ₅₀ 3.3 mg/L	[33]
	promelas			
Isobutylparaben	Pimephales	48 h	Acute LC ₅₀ 6.9 mg/L	[33]
	promelas			

Table 10.1 (continued)

humans. Moreover, other characteristics, such as small size, rapid development, the optical transparency of embryos, low cost, and easy maintenance, make it to be an ideal model species [34]. Zebrafish is also responsive to chemical and genetic screens and has a fully sequenced genome [35]. Further, zebrafish offers in vivo high-throughput assays, which are less costly than rodents. The huge population size of zebrafish provides a prompt assessment of multiple toxicity testing and facilitates the study of molecular mechanisms, and developmental and health effects related to exposure to contaminants across a population of organisms [36]. Thus, the OECD recommends the use of zebrafish as a model organism [34]. Japanese medaka has been used for toxicity testing for over 50 years, which is a small-sized (2–4 cm) freshwater fish and is well characterized as a model species because of being tolerant to wide salinity and temperature ranges. Japanese medaka is also being used as a model in the OECD test guidelines for developmental stages like early-life stages, juveniles, and adults [34].

The fathead minnow is also extensively used as a model species, especially in endocrine disruption studies [1, 37, 38]. It is native to North American and temperate waters and inhabits muddy pools of small rivers and streams. It is also among the three species validated by the OECD for ecotoxicity testing and has a huge toxicological database [39], which is mostly favored for embryotoxicity testing because of its well-known rapid development, transparent chorion, and sensitivity to toxic contaminants. The OECD has validated a test guideline for fathead minnow, zebrafish, and Japanese medaka [34]. Some of the fish species recommended by the European Center for Ecotoxicology and Toxicology of Chemicals (ECETOC) are given in Table 10.2 [40].

Fish species are selected and preferred according to the type of test and type of endpoint to measured, such as for toxicity test of early-life stage smaller species. For example, zebrafish, fathead minnow, and Japanese medaka are favored instead of rainbow trout because of their shorter test duration (30 days *versus* 90 days). Conversely, for longer exposure tests, rainbow trout are preferred to check endpoints [8].

Recommended test temperature range (°C)	Recommended the total length of test fish (cm)
21–25	2.0 ± 1.0
21–25	2.0 ± 1.0
20-24	3.0 ± 1.0
21–25	2.0 ± 1.0
21–25	2.0 ± 1.0
21–25	2.0 ± 1.0
13–17	5.0 ± 1.0
	Recommended test temperature range (°C) 21–25 21–25 20–24 21–25 21–25 21–25 21–25 21–25 21–25 21–25 21–25 21–25 21–25 21–25

Table 10.2 Recommended fish species for ecotoxicity testing

4 Fish Reservoir Ecotoxicity Tests

Reservoir toxicity tests aim at determining the level of biological response shown by adverse effects demonstrated by test species after being exposed to reservoir chemicals of concern. These tests are often conducted in controlled laboratory settings where the exposure concentration is of primary concern mainly regarding adverse biological effects associated with chemicals [41]. To enhance comparability, test methods are standardized. Toxicity tests are either conducted in situ or ex situ. Water samples may be collected from the contaminated area of the reservoir or prepared for simulated water after composition analysis [42]. Comprehensive assessments of the growth and reproduction of fish are more anticipated for population, multispecies, and community-level studies where reduction in the growth or the reproduction of particular species could be inferred regarding its ecological importance. For such kinds of reservoir toxicity assessments, standard toxicity tests including early-life stage fish tests are more suitable, compared to the other tests [43]. In reservoir environments, most of the exposures are chronic apart from those caused by accidental spill discharges. The results of acute and chronic tests are extrapolated to fluctuating aquatic environments and could be used to predict the effects of chemicals on the reservoir ecosystem [44].

Most of the time, the interaction between a fish and a contaminant under laboratory conditions is mostly more important. However, it is equally important to narrate the effect of chemicals on the fish population in reservoir ecosystems. The status of the fish population at the different concentration levels of contaminant is assessed in field surveys and results are then compared with data acquired from laboratory tests. Due to the fish being mostly exposed to multiple contaminants at a time, a combined toxicity assay in the laboratory is desirable. Besides, *in situ* experiments with caged fish in the reservoir are encouraged to conduct, which may provide more real data on reservoir toxicology [45].

Guideline designation	Organization	Title
203	OECD	Fish acute toxicity test
204	OECD	Fish prolonged toxicity test
210	OECD	Fish early-life stage toxicity test
212	OECD	Fish short-term toxicity test: embryo and sac-fry stage
215	OECD	Fish juvenile growth test
229	OECD	Fish, short-term reproduction assay
230	OECD	21-day fish assay
234	OECD	Fish, sexual development test
236	OECD	Fish, embryo acute toxicity test
850.1075	US EPA	Fish acute toxicity test, freshwater and marine
850.1085	US EPA	Fish acute toxicity mitigated by humic acid
850.14	US EPA	Fish early-stage toxicity test

Table 10.3 List of standard fish toxicity tests

Because of concerns about resource management and the release of chemicals from industries and factories into surface water bodies, aquatic toxicity tests were developed by world-renowned organizations on environmental protection like the US Environmental Protection Agency (US EPA) and OECD. Fish tests mostly relied on species, such as fathead minnow, zebrafish, and the cold-water rainbow trout [46]. A list of the standard fish toxicity test recommended by the OECD and US EPA [5, 46] is presented in Table 10.3.

4.1 Acute Toxicity Tests

When chemicals are released into the environment, they will find their way to enter lakes, rivers, and reservoirs. The EPA data require that fish acute toxicity tests should be typically conducted in three different fish species, including a cold-water freshwater species, a warmwater freshwater species, and a marine/estuarine species. Usually, an acute toxicity test is designed to check the safe concentration of pollutants, which gives a measure of acute lethality [45]. Despite being less ethically accepted as compared to tests with plants and invertebrates, fish acute toxicity commonly required ethics approval authorized by the official animal care and use committee [35, 47]. According to the EU Directive on animal protection utilized for scientific purposes, death as the endpoint should be avoided and substituted by some early endpoints [48]. The OECD guidelines for the testing of chemicals provide a helpful tool for assessing the potential effects of chemicals on human health and the environment. The OECD test guideline of fish acute toxicity tests (OECD TG 203) was published on July 17, 1992, in which fishes are exposed to the test substance for 96 hours under static or semi-static conditions [49]. Fish are exposed to five different concentrations of test chemicals for 96 h and deaths are documented at 24, 48, 72, and 96 h, respectively. Moreover, EC50 is determined to evaluate the concentration of the chemicals that gives a half-maximal response using a log-logistic model [48]. Recommended species are bluegill sunfish, common carp, zebrafish, fathead minnow, Japanese medaka, guppy, and rainbow trout.

4.1.1 Fish Larvae Test for Acute Toxicity

Rainbow trout acute toxicity test is 96 h static assays in which solutions are not renewed. Fish larvae are exposed to test solutions and are aerated at a rate of 6.5 ml/ min at 15 °C and a photoperiod of 16 h light: 8 h dark is maintained. In each test tank, ten fish larvae are retained. Fish are not fed during the experiment and even not 16 h before the test. After exposure for 96 h, numbers of survived fish larvae from each test concentration are counted and mortality is determined [50].

Fathead minnow toxicity test is conducted using a semi-static assay during which solutions are renewed after exposure for the first 48 h. Three replicates are prepared for each test concentration. Ten 4–6 days-old larvae are introduced into each tank with different concentrations. After 24 h of exposure to the test solution, fish larvae are transferred to a clean water container. Then, for the 72 h test, containers are placed in a controlled incubator at 25 °C with a 16 h light: 8 h dark photoperiod. To remove metabolic waste containers are renewed with fresh water. Finally, after exposure for 96 h, survived larvae are recorded from each concentration, and mortalities are observed [50].

4.1.2 Fish Embryo Test (FET) for Acute Toxicity

The fish embryo test (FET) is a potential animal alternative for the acute fish toxicity (AFT) test. The OECD test guideline for fish embryo acute toxicity test was published on July 26, 2013 (OECD 2013). In the beginning, this test was intended to ascertain the acute toxicity of chemicals at fish embryonic stages and was designed to replace the fish acute toxicity test later [35]. Because the early-life stages like embryogenesis are the sensitive period of the life cycle, the study of adverse impacts of chemicals on developmental processes is more discrete. Moreover, embryos are not anticipated under animal welfare regulations and could be utilized as screening tools [34].

Taking zebrafish as an example, the fertilized zebrafish eggs are exposed to the test chemical for 96 h with five different concentrations. After every 24 h observations, the indicators of acute toxicity are measured, which include the thickness of fertilized eggs, the absence of somite formation, the lack of heartbeat, and the non-detachment of the tail bud from the yolk sac. After the completion of the test, acute toxicity is determined based on the presence of any four of these observations, and LC50 is calculated. The test report should contain important information on physicochemical properties, such as pH, temperature, water hardness, the concentrations of the chemical being tested, and the conductivity [35, 51].

4.2 Chronic Toxicity Tests

Commonly, chronic toxicity tests are conducted to evaluate the adverse effects of contaminated medium, such as water, soil, or sediment under long-term exposure [46]. In 1956, a chronic exposure test was conducted by Olson and Foster to assess the toxicity of sodium dichromate to successive life stages (eggs, fry, and early juvenile) stages of salmonids [43]. During the exposure process, at least 10% of the test species remain alive after a complete life span. In chronic tests, survival is monitored and sublethal effects, such as reproductive success and growth, are observed. Statistical endpoints that are taken into account include no-observable (NOEC) and the lowest observable effect concentration (LOEC), which shows the maximum concentration of test substance that does not show any effect on the responses of test species under observation, and the minimum concentration of test chemical that shows the substantial effect on the response parameter compared to the control, respectively [52]. During life cycle tests, fish's younger developmental stages have constantly been shown as more sensitive than others. Short-duration tests using early developmental periods could also predict chronic toxicity. Chronic toxicity tests are considered more sensitive, compared to acute tests, because toxicity actions emphasize no adverse effects levels [46]. The conditions for chronic tests are different for different species. The selection of test species depends upon the consideration of whether the desired endpoints could easily be measured using the selected species or not. When the critical endpoints are the secondary sexual characters, the species nominated for testing should be fathead minnow or Japanese medaka instead of zebrafish. However, while using the endpoints such as fecundity, egg hatchability, and body size, zebrafish are preferred [8].

4.2.1 Full-Life Cycle Tests

Full-life cycle tests using fish were first carried out by Mount and Stephen [53]. Toxic effects were assessed for at least one generation under continuous exposure to chemicals [43]. Life cycle tests are carried out for a year or more to reveal more information, compared to the other tests, because hidden effects of contaminants could be exposed. They can provide evidence of not only fecundity and progeny, but also growth rate and disease resistance. The downside of long-term life cycle tests is that only a limited number of species could be assessed under limited exposure conditions [45]. They are carried out using rapidly growing and small-sized warmwater fish, such as zebrafish and fathead minnow [54]. For the life cycle toxicity test, freshwater fish like fathead minnow or zebrafish is cultured in the presence of the test chemicals from one stage of life to the other (whole life cycle) till the same stage of the next generation F_1 . The concentration of the test substance in the water is administered periodically during the experiment. By the end of the

experiment, the reproductive, behavioral, pathological, and physiological effects are assessed, and egg numbers, spawning ratio, fertility, and fecundity are recorded [55].

4.2.2 Partial-Life Cycle Tests

Before 1970, it was observed that when carrying out full-life cycle tests with numerous species after exposure to various chemicals, greater sensitivity has been shown by the early developmental stages, such as embryo, larvae, and juvenile stage, compared to the adult life stages [56, 57]. Consequently, in the mid-1970s, embryo-larval stages (30- to 60-day post-hatch) were proposed as a replacement for full-life cycle tests to lessen the time and cost [43]. embryo-larval stage tests intend to define the lethal and sublethal effects of a chemical on embryonic development, hatching, and larval growth are assessed [54]. According to the OECD guideline 210, these tests are recommended as a suitable and sensitive method for toxicity evaluation of chemicals [54]. In such tests, the embryo-larval stage of fish is exposed to three to five concentrations of the test chemicals under flow-through or semi-static conditions. Lethal and sublethal effects are evaluated and the lowest observed effect concentration (LOEC) is determined. The concentrations of the test chemicals are measured at regular intervals [58].

4.3 Bioconcentration and Bioaccumulation Tests

The uptake of pollutants from the external environment (usually water) is referred to as bioconcentration, and bioaccumulation is the absorption of a contaminant in biological tissues. In these tests, organisms are exposed to sublethal concentrations of the chemical, and their residues in the tissue of exposed organisms are evaluated until a steady state is achieved. Fish is usually used for such studies, because it is consumed by humans. Besides, soil invertebrates are also being assessed by fish for chemical uptake [52]. For fish, bioaccumulation and bioconcentration studies are carried out under flow-through and semi-static conditions. The test is divided into two phases: phase 1 is the uptake phase (exposure), which lasts normally 28 to a maximum of 60 days. During this phase, four fish of one species are exposed to at least two concentrations of the test chemical in separate groups. The second phase is the post-exposure or depuration phase, and fish are transferred into a medium devoid of the test chemical. Besides the two test concentrations, a control group without exposure to test chemicals is also performed in parallel. The concentration of the test chemical is monitored in fish in both phases of the test. Physicochemical parameters like pH, TOC, dissolved oxygen, salinity, total hardness, and temperature are also measured inside the test containers during the test. The lipid content is determined and the bioconcentration factor (BCF) at apparent steady state and the kinetic bioconcentration factor (BCFK) are calculated. Bioconcentration is expressed as a ratio of lipid content *versus* the whole bodyweight of fish [59].

5 Limitations of Fish Ecotoxicity Testing

Chronic fish toxicity tests are considered more sensitive than acute tests for the reason that the estimation of toxicity emphasizes endpoints other than survival, which can define better the no adverse effects levels. Moreover, chronic tests also provide a sound measure of responses for a population in the field. However, acute toxicity tests are regarded as fewer sensitive measures of toxic conditions, compared to chronic tests. Notably, chronic tests might not identify all sublethal effects [60]. Among chronic tests, the life cycle test is considered superior, but there is a limitation on time, space, and type of species that can be used. Other tests guarantee only a partial understanding of the impact of pollution on the fish's survival ability [8]. FET is considered a robust test and used as an alternative to the OECD 203 fish acute test [60].

6 Conclusions

Fish has been used as a sentinel organism for reservoir ecotoxicological testing. Various standardized tests have been designed according to contamination type and condition for evaluating the impacts of water-borne chemicals on fish. Standard toxicological tests are performed for acute lethality, fish embryo acute toxicity test, and chronic toxicity tests (full-life cycle toxicity tests). FET for acute toxicity. Fish bioaccumulation and bioconcentration tests are important because they reflect the reservoir ecotoxicology through the food chain. Reservoir toxicological studies also prefer to use small-size freshwater fish species like zebrafish, Japanese medaka, and fathead minnow. To enhance the predictive value and the extrapolation of acquired data at the ecosystem level, biochemical and molecular tools that can characterize the mode of action of chemicals should also be developed.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of Interest The author declares no conflict of interest.

References

- J.J. Stegeman, J.V. Goldstone, M.E. Hahn, Perspectives on zebrafish as a model in environmental toxicology, in *Fish Physiology*, (Elsevier, 2010), pp. 367–439
- S. Zhang, X. Li, D. He, D. Zhang, Z. Zhao, H. Si, F. Wang, Per- and poly-fluoroalkyl substances in sediments from the water-level-fluctuation zone of the Three Gorges Reservoir, China:

contamination characteristics, source apportionment, and mass inventory and loadings. Environ. Pollut. 299, 118895 (2022)

- S.A. Ali, S. Aadhar, H.L. Shah, V. Mishra, Projected increase in hydropower production in India under climate change. Sci. Rep. 8, 12450 (2018)
- 4. P.M. Mwanamoki, N. Devarajan, F. Thevenon, N. Birane, L.F. de Alencastro, D. Grandjean, P.T. Mpiana, K. Prabakar, J.I. Mubedi, C.G. Kabele, W. Wildi, J. Poté, Trace metals and persistent organic pollutants in sediments from river-reservoir systems in Democratic Republic of Congo (DRC): spatial distribution and potential ecotoxicological effects. Chemosphere 111, 485–492 (2014)
- 5. M. Christophe, A. Rachid, L. Mario, Fish as reference species in different water masses, in *Aquatic Ecotoxicology*, (Elsevier, 2015), pp. 309–331
- 6. P.P. Calow, Handbook of Ecotoxicology (Wiley, 2009)
- 7. K.E. Carpenter, On the biological factor involved in the destruction of river fisheries by pollution due to lead mining. Ann. Appl. Biol. **12**, 1–13 (1925)
- 8. OECD, Fish Toxicity Testing Framework (2014)
- 9. J. Belliard, R.B.d. Thomas, D. Monnier, Fish communities and river alteration in the Seine Basin and nearby coastal streams. Hydrobiologia **400**, 155–166 (1999)
- J. Breine, I. Simoens, P. Goethals, P. Quataert, D. Ercken, C. Van Liefferinghe, C. Belpaire, A fish-based index of biotic integrity for upstream brooks in Flanders (Belgium). Hydrobiologia 522, 133–148 (2004)
- M. Rahman, Z. Hossain, M. Mollah, G. Ahmed, Effect of Diazinon 60 EC on Anabas testudineus, Channa punctatus and Barbodes gonionotus. Naga, the ICLARM quarterly 25, 8–12 (2002)
- J.M. Besser, C.A. Mebane, D.R. Mount, C.D. Ivey, J.L. Kunz, I.E. Greer, T.W. May, C.G. Ingersoll, Sensitivity of mottled sculpins (Cottus bairdi) and rainbow trout (Onchorhynchus mykiss) to acute and chronic toxicity of cadmium, copper, and zinc. Environ. Toxicol. Chem. Int. J. 26, 1657–1665 (2007)
- I.M. Sial, M.A. Kazmi, Q.B. Kazmi, S.N.-u.H. Naqvi, Toxicity of biosal (phytopesticide) and permethrin (pyrethroid) against common carp, Cyprinus carpio, Pakistan. J. Zool. 41 (2009)
- T. Şişman, Dichlorvos-induced developmental toxicity in zebrafish. Toxicol. Ind. Health 26, 567–573 (2010)
- M. Ilavazhahan, R.T. Selvi, S. Jayaraj, Determination of LC of the bacterial pathogen. Pesticide and, Global Journal of Environmental Research (GJER) 4, 76–82 (2010)
- S.K. Singh, S.K. Singh, R.P. Yadav, Toxicological and biochemical alterations of cypermethrin (Synthetic Pyrethroids) against freshwater Teleost fish Colisa fasciatus at different season. World J. Zool. 5, 25–32 (2010)
- 17. C. Thenmozhi, V. Vignesh, R. Thirumurugan, S. Arun, Impacts of malathion on mortality and biochemical changes of freshwater fish Labeo rohita (2011)
- R. Ganeshwade, L. Dama, D. Deshmukh, A. Ghanbahadur, S. Sonawane, Toxicity of endosulfan on freshwater fish Channa striatus. Trends Fish. Res. 1, 29–31 (2012)
- 19. G.R. Marutirao, Histopathological changes in the gills of Puntius ticto (Ham) under Dimethoate toxicity. Bioscan 7, 423–426 (2012)
- R. Ilyas, M. Javed, Acute toxicity of endosulfan to the fish species Catla catla, Cirrhina mrigala and Labeo rohita. Int. J. Agric. Biol. 15 (2013)
- C. Dey, S. Saha, A comparative study on the acute toxicity bioassay of dimethoate and lambdacyhalothrin and effects on thyroid hormones of freshwater teleost fish Labeo rohita (Hamilton). Int. J. Environ. Res. 8, 1085–1092 (2014)
- M. Oliveira, A. Ribeiro, K. Hylland, L. Guilhermino, Single and combined effects of microplastics and pyrene on juveniles (0+ group) of the common goby Pomatoschistus microps (Teleostei, Gobiidae). Ecol. Indic. 34, 641–647 (2013)
- 23. C.M. Rochman, E. Hoh, T. Kurobe, S.J. Teh, Ingested plastic transfers hazardous chemicals to fish and induces hepatic stress. Sci. Rep. **3**, 3263 (2013)

- 24. C.M. Rochman, T. Kurobe, I. Flores, S.J. Teh, Early warning signs of endocrine disruption in adult fish from the ingestion of polyethylene with and without sorbed chemical pollutants from the marine environment. Sci. Total Environ. **493**, 656–661 (2014)
- 25. Y. Lu, Y. Zhang, Y. Deng, W. Jiang, Y. Zhao, J. Geng, L. Ding, H. Ren, Uptake and accumulation of polystyrene microplastics in zebrafish (Danio rerio) and toxic effects in liver. Environ. Sci. Technol. 50, 4054–4060 (2016)
- 26. M. Pal, S. Trivedi, Impact of chromium trioxide on haematological parameters of freshwater fish, Channa punctatus (Bloch). Eur. J. Exp. Biol. **6**, 40 (2016)
- P.Y. Kunz, T. Gries, K. Fent, The ultraviolet filter 3-benzylidene camphor adversely affects reproduction in fathead minnow (Pimephales promelas). Toxicol. Sci. 93, 311–321 (2006)
- M. Coronado, H. De Haro, X. Deng, M.A. Rempel, R. Lavado, D. Schlenk, Estrogenic activity and reproductive effects of the UV-filter oxybenzone (2-hydroxy-4-methoxyphenylmethanone) in fish. Aquat. Toxicol. 90, 182–187 (2008)
- D.R. Orvos, D.J. Versteeg, J. Inauen, M. Capdevielle, A. Rothenstein, V. Cunningham, Aquatic toxicity of triclosan. Environ. Toxicol. Chem. Int. J. 21, 1338–1349 (2002)
- H. Ishibashi, N. Matsumura, M. Hirano, M. Matsuoka, H. Shiratsuchi, Y. Ishibashi, Y. Takao, K. Arizono, Effects of triclosan on the early life stages and reproduction of medaka Oryzias latipes and induction of hepatic vitellogenin. Aquat. Toxicol. 67, 167–179 (2004)
- 31. S. Fleiss, Review of fluoride toxicity to aquatic organisms and its toxicity contribution in Volvo wastewater, in Gothenburg, Sweden: Department of Plant and Environmental Sciences ... (2011)
- K. Fent, P.Y. Kunz, E. Gomez, UV filters in the aquatic environment induce hormonal effects and affect fertility and reproduction in fish. CHIMIA Int. J. Chem. 62, 368–375 (2008)
- L.L. Dobbins, S. Usenko, R.A. Brain, B.W. Brooks, Probabilistic ecological hazard assessment of parabens using Daphnia magna and Pimephales promelas. Environ. Toxicol. Chem. 28, 2744–2753 (2009)
- 34. R. Capela, J. Garric, L.F.C. Castro, M.M. Santos, Embryo bioassays with aquatic animals for toxicity testing and hazard assessment of emerging pollutants: a review. Sci. Total Environ. 135740 (2019)
- 35. T. Braunbeck, B. Kais, E. Lammer, J. Otte, K. Schneider, D. Stengel, R. Strecker, The fish embryo test (FET): origin, applications, and future. Environ. Sci. Pollut. Res. 22, 16247–16261 (2015)
- K. Bambino, J. Chu, Zebrafish in toxicology and environmental health, in *Current Topics in Developmental Biology*, (Elsevier, 2017), pp. 331–367
- D.E. Hinton, S.W. Kullman, R.C. Hardman, D.C. Volz, P.-J. Chen, M. Carney, D.C. Bencic, Resolving mechanisms of toxicity while pursuing ecotoxicological relevance? Mar. Pollut. Bull. 51, 635–648 (2005)
- 38. G.T. Ankley, D.C. Bencic, M.S. Breen, T.W. Collette, R.B. Conolly, N.D. Denslow, S.W. Edwards, D.R. Ekman, N. Garcia-Reyero, K.M. Jensen, Endocrine disrupting chemicals in fish: developing exposure indicators and predictive models of effects based on mechanism of action. Aquat. Toxicol. 92, 168–178 (2009)
- T. Braunbeck, E. Lammer, Fish embryo toxicity assays. German Federal Environ. Agency 298 (2006)
- ECETOC, Alternative testing approaches in environmental safety assessment, Technical Report No. 97 (2005)
- 41. D.W. Sparling, G. Linder, C.A. Bishop, S. Krest, *Ecotoxicology of amphibians and reptiles* (CRC Press, 2010)
- 42. B. Anderson, P. Nicely, K. Gilbert, R. Kosaka, J. Hunt, B. Phillips, Overview of freshwater and marine toxicity tests: a technical tool for ecological risk assessment. California Environmental Protection Agency Office of Environmental Health Hazard Assessment Reproductive and Cancer Hazard Assessment Section Ecotoxicology Unit, in (2004)
- 43. D.M. Woltering, The growth response in fish chronic and early life stage toxicity tests: a critical review. Aquat. Toxicol. 5, 1–21 (1984)

- 44. M. Nikinmaa, An Introduction to Aquatic Toxicology (Elsevier, 2014)
- J. Alabaster, R. Lloyd, Fish toxicity testing procedures, in: water quality criteria for Fish Buthetworths scientific, pp. 315–343, (1982)
- 46. D.W. Sparling, Basics of Ecotoxicology (CRC Press, 2017)
- S. Scholz, E. Sela, L. Blaha, T. Braunbeck, M. Galay-Burgos, M. García-Franco, J. Guinea, N. Klüver, K. Schirmer, K. Tanneberger, A European perspective on alternatives to animal testing for environmental hazard identification and risk assessment. Regul. Toxicol. Pharmacol. 67, 506–530 (2013)
- 48. Z. Dang, L.T. van der Ven, A.S. Kienhuis, Fish embryo toxicity test, threshold approach, and moribund as approaches to implement 3R principles to the acute fish toxicity test. Chemosphere 186, 677–685 (2017)
- 49. OECD, OECD Guidelines for the Testing of Chemicals, Organization for Economic (1992)
- P.Y. Robidoux, B. Virginie, L. Judith, D. Marc, Assessment of acute and chronic toxicity of unweathered and weathered diluted bitumen to freshwater fish and invertebrates. Ecotoxicol. Environ. Saf. 164, 331–343 (2018)
- 51. A. Valavanidis, T. Vlachogianna, Ecotoxicity test methods and ecological risk assessment. Aquatic and terrestrial ecotoxicology tests under the guidelines of international organizations, Science Advances on Environmental Chemistry, Toxicology and Ecotoxicology Issues, 28 (2015)
- 52. J.R. Bidwell, In vivo ecotoxicology models, in *An Introduction to Interdisciplinary Toxicology*, (Elsevier, 2020), pp. 507–523
- 53. D.I. Mount, C.E. Stephan, A method for establishing acceptable toxicant limits for fishmalathion and the butoxyethanol ester of 2,4-D. Trans. Am. Fish. Soc. 96, 185–193 (1967)
- 54. R. Nagel, K. Isberner, Testing of chemicals with fish—a critical evaluation of tests with special regard to zebrafish, in *Fish Ecotoxicology*, (Springer, 1998), pp. 337–352
- 55. EPA, Ecological effect test guidelines OPPTS 850.1500 fish life-cycle toxicity (1996)
- J. McKim, J. Eaton, G.W. Holcombe, Metal toxicity to embryos and larvae of eight species of freshwater fish—II: copper. Bull. Environ. Contam. Toxicol. 19, 608–616 (1978)
- J. McKim, D. Benoit, K. Biesinger, W. Brungs, R. Siefert, Effects of pollution on freshwater fish. J. Water Pollut. Control Federation, 1711–1768 (1975)
- 58. OECD, Test No. 210: Fish, Early-Life Stage Toxicity Test (1992)
- OECD, 305, Bioconcentration, Flow-through Fish Test', in, OECD Guidelines for Testing of Chemicals (1996)
- 60. D.W. Sparling, *Ecotoxicology Essentials: Environmental Contaminants and Their Biological Effects on Animals and Plants* (Academic Press, 2016)

Chapter 11 Ecotoxicology Methodology of Sediment Toxicity in the Reservoir



Suman Thodhal Yoganandham and De-Sheng Pei 🝺

Abstract The reservoir sediment can act as a drain and contaminant source. Once contaminants are released into the water column, they can be toxic for biota. This chapter details various ecotoxicological studies using sediment derived from reservoirs. Aquatic organisms are critical in the aquatic environment as intermittent consumers in the food chain and valuable indicators of sediment toxicity. International organizations have developed numerous standardized toxicity methods for aquatic species, including acute and chronic lethal sublethal toxicity measures, to determine the effects of this sediment toxicity. Aquatic species, such as algae, amphipods, and bivalves, are the best recommended for a specific test depending on the type of test, the final measurement, and the easy use of ecotoxicity.

Keywords Sediment \cdot Toxicology \cdot Aquatic organism \cdot Acute toxicity \cdot Chronic toxicity

1 Introduction

Human health and growth rely on the efficiency and abundance of freshwater, mainly through food and water supply [1, 2]. Intensive urbanization and industrialization risk the quality of the water. Anthropogenic effluents discharged into aquatic systems known as industrial, farming, and domestic activities contain mixtures of organic and inorganic pollutants, which have negative impacts on ecosystems and organisms [3, 4].

In water irrigation, power generation, and flood control, reservoirs play essential roles. Dam construction alters the hydrological system, and the sediment of reservoirs contains multiple heavy metals (HMs) [5–8]. To monitor aquatic ecosystems, the bottom sediment is significant [9] for the assessment of pollution levels [10, 11] and the evaluation of environmental risks [12]. As large and small dams are built globally, water reservoirs are increasing [13, 14]. According to the report from Zarfl et al. [13], the number of dams is anticipated to double [13].

The sediment acts as a sink for pollutants [15]. But inside the water column, metals may be resuspended and a diffuse source of contamination, depending on environmental conditions [16]. In addition, the sediment can be essential to the

https://doi.org/10.1007/978-3-031-26344-6_11

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 127 D.-S. Pei et al., *Reservoir Ecotoxicology*,

ecosystem's trophic dynamic by geochemically interacting among the sediment and water column, including nutrient release [17]. Thus, surface sediment studies and water from the bottom and column may be used to elucidate pollutants' temporal and spatial dynamics and test reservoir management effectiveness.

Sediment toxicity testing under controlled exposure suggests toxicity assessment of the organism in the site [18–22]. Toxicity rate can be derived from chemical analyses, and pollutants can be quantified and compared to the sediment quality criteria, in which sediments are considered low or highly sensitive to adverse effects. However, chemical analyses can not evaluate the possible impacts of non-media pollutants that might lead to outcomes. Although chemical analyses can give details on potential pollutant bioavailability, sediment toxicity testing responses reflect bioavailable contaminant fractions, cumulative effects of pollutant mixtures, and the interaction of non-contaminant stressors as they influence test organisms. The sediment toxicity tests generally include further factual proof of pollutant effects than the benthic ecological, since community endpoints are difficult to interpret [23– 25].

2 Sediments Collection

Surface sediments are most frequently collected to determine risks associated with sediments. Generally, epifaunal and infaunal organisms occupied the top 10 cm of sediments. Epibenthic groups, including shrimps and amphipods, can only be subjected to surface sediment contaminants (0–1 cm). In contrast, others including bivalves and polychaetes may mainly be subjected to pollutants several centimeters deep. Epibenthic species can only contain surface sediment. Adequate information should be achieved to evaluate contaminant concentrations of sediments in both 0-2 cm and 2-10 cm depth for assessing significant pathways of exposure to contaminants in most species.

Many studies are available for different collection conditions [26]. Sediment processing systems need to limit the deterioration of sediment, otherwise, extensive erosion and other forms of sedimentary structural degradation may occur. The mixing can also demonstrate a possible disturbance with layers of varying particle dimensions and composition of preceding redox stratified chemical substrates that affect the bioavailability and potential for sediment toxicity [27].

The sediment quantities to be obtained are based on the tests and analyses to be carried out. Usually, 1 kg sediment should be adequate for most pollutants from each site in research. Both samples should be processed with the appropriate equipment and procedure for the study requested. Furthermore, bioaccumulation or toxicity tests require 2–3 kg, and these samples should be kept cold (not frozen).

Plastic devices and containers used for sample collection are soaked in 10% nitric acid and thoroughly washed [26]. Nitric acid is not ideal for the analysis of nitrogen forms. The sampling system must be cleanly rinsed with water from the sampling station before and after collecting samples. More cleaning of the sample may be
appropriate for evaluations, including (I) water and soap washing, (ii) rinsing with distilled water, (iii) rinsing acetone or ethanol, and (iv) on-site rinsing water. In the case of heavily polluted sediments, reference sites should be first sampled to avoid cross-contamination [18].

Before taking sediment for the subsequent biological, physical, toxicity, or chemical analyses, it is necessary to verify the reliability of the sample collected. Grab samples are acceptable when the surface layer appears unchanged, and the volume of the sediments is sufficient. For the desired material quantity, several replicate samples are needed for both grab and core sampling.

3 Manipulations of Sediments Before Testing

Until chemical or toxicity testing, sediments are frequently processed in the laboratory or field. Manipulation may include sieving to extract large particles and waste and homogenizing large samples so that multiple biological and chemical tests can be carried out [28]. Most sediment manipulations can influence the sediment's properties and pollutant bioavailability, and these effects will have to be tested. Therefore, the methods designed to process sediment samples for tests and analyses should eliminate perturbations.

It is desirable to examine how sample manipulation can affect the concentration, toxicity, and bioavailability of the pollutants in the sediments. Later, bioavailability and toxicity analysis will help to interpret the data of preliminary redox potentials, total organic carbon, pH, AVS, distribution of the iron and weakly extractable pollutants, and pore water contaminants in newly collected sediment samples that have been minimally manipulated. However, pH might be influenced if sediment samples are subjected to storage above 4 weeks longer [26].

4 Pathways for Pollutant Exposure

Sediment-dwelling organisms are vulnerable to pollutants by inadvertent and direct sediment intake, pore water, and excess water intakes [28, 29]. Contaminants can be released from sediments to water columns through absorption, disruption, and the distribution of dissolved chemicals into the water column [30]. In surface layers, normal sediment resuspension often preserves pollutants in oxidized types. For many benthic species, sediment-water exposure is the most critical measurement since feeding is conducted on organically rich particles in this area. Field studies have shown that sediment resuspension for transport and internal recycling in aquatic ecosystems is significant [31, 32]. Laboratory studies demonstrated a substantial rise in hydrophobic contaminant concentration in surrounding water in connection with simulated resuspension events [33, 34].

The benthic species play a significant role in sediment bioturbation. The deposition feeders consume organic/inorganic particles from sediment or within their sedimentation surface. Polychaete and oligochaete consume sub-surface lower sediments and transport them as fecal pellets to the sediment. A second relevant factor of bioturbation is the capacity, through feeding and continuous burrowing of diverse sub-surface feeders to increase the sediment/water area and the toxicants and oxygen fluxes. Several authors explain substantial rises in the flow of sediment-related pollutants from overlying water and sediment to pore water [35–38]. Moreover, the bio-mediated resuspension of particles can be considered under certain conditions and even surpass that of physical disruptions.

5 Test Organisms for Sediment Toxicity

The sensitivities of benthic organisms to sediment pollutants vary widely because of different organism properties, such as burrowing, life cycles, routes of pollutant exposure (for feeders to filter and deposit), and the contaminant's characteristics (such as partitioning and bioavailability). Therefore, in sediment quality assessments, it is critical that toxic exposure may use various organisms with diverse feeding approaches and behaviors. Bivalve (clams, mussels, and oysters), insect larvae (chironomid midges and mayflies), algae, nematodes, copepods, worms, and snails, may be used to examine the toxicity of sediments (Table 11.1). Despite the organism's close connection with sediment, large crustaceans, such as crabs, are reported for the toxicity test of sediments [46].

Organism	Test species	Duration/End point	Acute/ Chronic	References
Algae	Chlorella vulgaris	72 h/growth	Acute	[39]
Amphipod	Hyalella azteca	10- and 28-d survival and growth, 42-d survival, growth and reproduction	Chronic	[40]
	Corophium volutator	Survival	Chronic	[41]
Bivalve	Hyridella australis	28 d, growth, antioxidant.	Chronic	[42]
Worm	Limnodrilus hoffmeisteri	mortality and autotomy rates	Chronic	[43]
	Chironomus larvas	mortality and growth inhibition rates	Chronic	[43]
	Caenorhabditis elegans	Growth reproduction	Acute	[44]
Crustacean	Daphnia magna	48 h survival	Acute exposures	[45]

Table 11.1 Freshwater whole sediment toxicity tests

5.1 Algae

Algae are the common food source for many invertebrates, so transmitting harmful sediment pollutants to higher trophic levels is possible [47]. In sedimentary stimulation, toxic effects have no masked effects of ammonia stimulation [48, 49]. Regarding sediment research, algal inhibition appeared to be more suitable than whole algal growth as an endpoint. Flow cytometry is a speedy measurement of algal cells inside a moving fluid [50].

5.2 Amphipods

Amphipods are an important component of aquatic ecosystems (rivers and reservoirs). They are the main prey of fish, birds, and large invertebrates. Thus, they become crucial for moving pollutants from sediments to higher trophic levels. Amphipods have an environmental significance and vulnerability to polluted sediments, due to their large numbers, extensive distribution, ease of handling, and their suitability for sediment toxicity research. Many amphipods eat sediments directly and are treated with sediment-based pollutants combined with pollutants in overlying waters. There are many standardized or peer-reviewed methods for measuring survival, reproduction, development, and sediment avoidance in various fields for whole sediment toxicity studies [21, 51–53]. *Gammarus pulex* for freshwater sediment tests is widely used. The life cycle of amphipods was limited to ten days. Thus, bioassays of those species were commonly used to test sublethal reproductive effects [21]. As we know, amphipod species are ideal models for in situ sediment toxicity studies [54].

5.3 Bivalves

Bivalve is a significant and essential part of benthic estuarine, marine communities, and freshwater ecosystems [55]. Bivalves include clams, mussels, and oysters. Bivalves sometimes bury themselves in the top two to twenty centimeters of silt or sandy sediments. The bivalve species feed on organic-rich particles and algae from the surface layer with their siphons or filtering vast water column quantities. The bivalves are adversely affected by aqueous pollutants, such as pore water, burrow water, and overlying water. Their food contains bacteria, algae, plants, and inorganic sedimentary products unintentionally ingested [56–58]. Bivalves are vital preies of many fishes and invertebrates. Therefore, they are likely to transfer the contaminants from sediments and water to predators at the top of the food chain.

Freshwater bivalve has an extraordinary life history and physiological and anatomical qualities that make them beneficial organisms for detecting pollutants from contemporary and historical sources and determining the ecological value of pollution in reservoir ecosystems [59]. Therefore, Australian freshwater bivalves, including *Velesunio ambiguous*, *Velesunio angasi*, and *Hyridella depressa* were employed to examine metal accumulation from the sediment [60].

6 Toxicity Endpoints

Test endpoints are generally chronic or acute but can also be sublethal or lethal. Acute toxicity is usually a negative impact resulting from a short exposure period to a chemical, compared to the organism's lifespan. On the other hand, chronic toxicity is generally related to the negative effect on the organism of the lifetime or adverse outcome on a sensitive early life stage due to chemical exposure caused by a substantial concentration. A significant proportion of an organism's life will be over 10% [61]. Generally, a minimum of 48 h for a short life cycle and a 4 to 10 d test for a long life cycle. Thus, chronic sediment toxicity tests should be >10 days. Thus, 28–42 or 60 days are used for evaluating the longer-term effects of amphipod, bivalve, and worm species for survival, growth, and reproduction [20, 53, 62–66].

The organism's survival is the most common endpoint for sediment quality testing and is a general acute endpoint. The juvenile stage is more critical than the adult stage. These sublethal responses usually act as chronic endpoints and provide more detail about possible long-term consequences at the level of individual populations [19, 21, 66].

To have effective decisions on the mitigation alternatives for contaminated sediments, one needs to know about test endpoints' biological and ecological value. There are increasingly biochemical and physiological responses, and biomarker approaches give higher sensitivity and less variability than well-defined sublethal endpoints in whole sediment tests [21]. Direct associations between impaired reproduction and lysosomal instability are more susceptible to specific species [67]. The main goal of eco-genomics is to detect triggered genes so that molecular fingerprints are unique to that bioavailable chemical fraction [68, 69].

7 Conclusion

This chapter covers the basics of ecotoxicology techniques, focusing on sediment toxicity in the reservoir. Possible evaluations of sediment quality include acute and chronic lethal and sublethal toxicity measurements. Based on the kind of test, final measurement, and ease of using ecotoxicity, aquatic species including algae, amphipods, and bivalves are appropriate models for the sediment toxicity of reservoir sediment. How to design toxicity tests, what to evaluate, and how to interpret the toxicity results deserve us to systemically consider before the beginning of sediment toxicity in the reservoir.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- R.J. Naiman, S.E. Bunn, C. Nilsson, G.E. Petts, G. Pinay, L.C. Thompson, Legitimizing fluvial ecosystems as users of water: an overview. Environ. Manag. 30, 455–467 (2002)
- 2. K.E. Limburg, Aquatic Ecosystem Services Introductory Overviews Aquatic Ecosystem Services (Elsevier, Amsterdam, 2009)
- R.P. Schwarzenbach, T. Egli, T.B. Hofstetter, U. Von Gunten, B. Wehrli, Global water pollution and human health. Annu. Rev. Environ. Resour. 35, 109–136 (2010)
- J.G. Tundisi, Reservoirs: New challenges for ecosystem studies and environmental management. Water Security 4, 1–7 (2018)
- M. Sojka, J. Jaskuła, J. Wicher-Dysarz, T. Dysarz, The Impact of the Kowalskie Reservoir on the Hydrological Regime Alteration of the Główna River. J. Ecol. Eng. 17 (2016)
- 6. D. Vukovic, Z. Vukovic, S. Stankovic, The impact of the Danube Iron Gate Dam on heavy metal storage and sediment flux within the reservoir. Catena **113**, 18–23 (2014)
- J. Dąbrowska, H. Kaczmarek, J. Markowska, S. Tyszkowski, O. Kempa, M. Gałęza, E. Kucharczak-Moryl, A. Moryl, Shore zone in protection of water quality in agricultural landscape—the Mściwojów Reservoir, southwestern Poland. Environ. Monit. Assess. 188, 1–14 (2016)
- Y. Guo, B. Deng, Seasonal variation of heavy metals in suspended sediments downstream the Three Gorges Dam in the Yangtze River. Environ. Monit. Assess. 194, 1–16 (2022)
- N. Kolarova, P. Napiórkowski, Trace elements in aquatic environment. Origin, distribution, assessment and toxicity effect for the aquatic biota. Ecohydrol. Hydrobiol. 21, 655–668 (2021)
- P. Palma, L. Ledo, P. Alvarenga, Assessment of trace element pollution and its environmental risk to freshwater sediments influenced by anthropogenic contributions: the case study of Alqueva reservoir (Guadiana Basin). Catena 128, 174–184 (2015)
- Y. Wang, L. Yang, L. Kong, E. Liu, L. Wang, J. Zhu, Spatial distribution, ecological risk assessment and source identification for heavy metals in surface sediments from Dongping Lake, Shandong, East China. Catena 125, 200–205 (2015)
- G. Suresh, P. Sutharsan, V. Ramasamy, R. Venkatachalapathy, Assessment of spatial distribution and potential ecological risk of the heavy metals in relation to granulometric contents of Veeranam lake sediments, India. Ecotoxicol. Environ. Saf. 84, 117–124 (2012)
- C. Zarfl, A.E. Lumsdon, J. Berlekamp, L. Tydecks, K. Tockner, A global boom in hydropower dam construction. Aquat. Sci. 77, 161–170 (2015)
- T.B. Couto, J.D. Olden, Global proliferation of small hydropower plants-science and policy. Front. Ecol. Environ. 16, 91–100 (2018)
- H. Abdulla, D. Murgulet, Assessment of Organic Pollutants in Nueces Bay's Petroleum Brine Impacted Sediments (2021)
- A. Acquavita, F. Floreani, S. Covelli, Occurrence and speciation of arsenic and mercury in alluvial and coastal sediments. Curr. Opin. Environ. Sci. Health 22, 100272 (2021)
- T. Mayer, F. Rosa, R. Mayer, M. Charlton, Relationship between the sediment geochemistry and phosphorus fluxes in a Great Lakes coastal marsh, Cootes Paradise, ON, Canada, in *The Interactions Between Sediments and Water*, (Springer, Dordrecht, 2006), pp. 131–139

- ASTM (American Society for Testing and Materials), 2008a. Standard Guide for Conducting 10-day Static Sediment Toxicity Tests with Marine and Estuarine Amphipods (E1367-03 (2008)). Annual Book of ASTM Standards, Vol 11.06. West Conshohocken, PA, USA
- D. Greenstein, S. Bay, B. Anderson, G.T. Chandler, J.D. Farrar, C. Keppler, B. Phillips, A. Ringwood, D. Young, Comparison of methods for evaluating acute and chronic toxicity in marine sediments. Environ. Toxicol. Chem. 27, 933–944 (2008)
- A.J. Kennedy, J.A. Steevens, G.R. Lotufo, J.D. Farrar, M.R. Reiss, R.K. Kropp, J. Doi, T.S. Bridges, A comparison of acute and chronic toxicity methods for marine sediments. Mar. Environ. Res. 68, 118–127 (2009)
- S.L. Simpson, D.A. Spadaro, Performance and sensitivity of rapid sublethal sediment toxicity tests with the amphipod Melita plumulosa and copepod Nitocra spinipes. Environ. Toxicol. Chem. 30, 2326–2334 (2011)
- A. Rodríguez-Romero, A. Khosrovyan, T.A. Del Valls, R. Obispo, F. Serrano, M. Conradi, I. Riba, Several benthic species can be used interchangeably in integrated sediment quality assessment. Ecotoxicol. Environ. Saf. 92, 281–288 (2013)
- 23. E.L. Johnston, D.A. Roberts, Contaminants reduce the richness and evenness of marine communities: a review and meta-analysis. Environ. Pollut. **157**, 1745–1752 (2009)
- 24. G.A. Burton, E.L. Johnston, Assessing contaminated sediments in the context of multiple stressors. Environ. Toxicol. Chem. 29, 2625–2643 (2010)
- 25. K.A. Dafforn, S.L. Simpson, B.P. Kelaher, G.F. Clark, V. Komyakova, C.K. Wong, E.L. Johnston, The challenge of choosing environmental indicators of anthropogenic impacts in estuaries. Environ. Pollut. 163, 207–217 (2012)
- U.S. EPA (2007) Sediment toxicity identification evaluation (TIE) phases I, II and III: guidance document. EPA-600-R-07-080. Office of Research and Development, Washington, DC
- 27. S.L. Simpson, G.E. Batley, Predicting metal toxicity in sediments: a critique of current approaches. Integr. Environ. Assess. Manag. **3**, 18–31 (2007)
- O. Campana, S.L. Simpson, D.A. Spadaro, J. Blasco, Sub-lethal effects of copper to benthic invertebrates explained by sediment properties and dietary exposure. Environ. Sci. Technol. 46, 6835–6842 (2012)
- 29. I. Proulx, L. Hare, Differences in feeding behaviour among *C hironomus* species revealed by measurements of sulphur stable isotopes and cadmium in larvae. Freshw. Biol. **59**, 73–86 (2014)
- J.Y. Aller, S.A. Woodin, R.C. Aller, Organism-sediment Interactions. Columbia eSC SC: University of South Carolina Press (2001)
- J.S. Latimer, W.R. Davis, D.J. Keith, Mobilization of PAHs and PCBs from in-place contaminated marine sediments during simulated resuspension events. Estuar. Coast. Shelf Sci. 49, 577–595 (1999)
- 32. J.J. Bogdan, J.W. Budd, B.J. Eadie, K.C. Hornbuckle, The effect of a large resuspension event in southern Lake Michigan on the short-term cycling of organic contaminants. J. Great Lakes Res. 28, 338–351 (2002)
- 33. J. Feng, Z. Yang, J. Niu, Z. Shen, Remobilization of polycyclic aromatic hydrocarbons during the resuspension of Yangtze River sediments using a particle entrainment simulator. Environ. Pollut. 149, 193–200 (2007)
- 34. A.R. Schneider, E.T. Porter, J.E. Baker, Polychlorinated biphenyl release from resuspended Hudson River sediment. Environ. Sci. Technol. 41, 1097–1103 (2007)
- 35. A.S. Green, G.T. Chandler, Meiofaunal bioturbation effects on the partitioning of sedimentassociated cadmium. J. Exp. Mar. Biol. Ecol. **180**, 59–70 (1994)
- A.D. Briggs, N. Greenwood, A. Grant, Can turbidity caused by Corophium volutator (Pallas) activity be used to assess sediment toxicity rapidly. Mar. Environ. Res. 55, 181–192 (2003)
- J.E. Hedman, J.S. Tocca, J.S. Gunnarsson, Remobilization of polychlorinated biphenyl from Baltic Sea sediment: comparing the roles of bioturbation and physical resuspension. Environ. Toxicol. Chem. 28, 2241–2249 (2009)

- S. Josefsson, K. Leonardsson, J.S. Gunnarsson, K. Wiberg, Bioturbation-driven release of buried PCBs and PBDEs from different depths in contaminated sediments. Environ. Sci. Technol. 44, 7456–7464 (2010)
- 39. Z.T. Pei, R.R. Xu, H.Y. Liu, W.Q. Wang, M. Zhang, L.L. Zhang, J. Zhang, W.Q. Wang, R. Yu, L.W. Sun, Development and application of a novel whole sediment toxicity test using immobilized sediment and Chlorella vulgaris. Ecotoxicol. Environ. Saf. 189, 109979 (2020)
- 40. ASTM American Society of Testing and Materials, 2010. Standard test method for measuring the toxicity of sediment-associated contaminants with freshwater invertebrates. ASTM Standard E1706-05, Philadelphia, PA, USA
- 41. J. Stronkhorst, S. Ciarelli, C.A. Schipper, J.F. Postma, M. Dubbeldam, M. Vangheluwe, J.M. Brils, R. Hooftman, Inter-laboratory comparison of five marine bioassays for evaluating the toxicity of dredged material. Aquat. Ecosyst. Health Manage. 7, 147–159 (2004)
- 42. C.P.M. Wadige, A.M. Taylor, F. Krikowa, M. Lintermans, W.A. Maher, Exposure of the freshwater bivalve Hyridella australis to metal contaminated sediments in the field and laboratory microcosms: metal uptake and effects. Ecotoxicology 26, 415–434 (2017)
- 43. P. Zhang, X. Pan, Q. Wang, G. Ge, Y. Huang, Toxic effects of heavy metals on the freshwater benthic organisms in sediments and research on quality guidelines in Poyang Lake, China. J. Soils Sediments 20, 3779–3792 (2020)
- 44. S. Höss, W. Ahlf, M. Bergtold, E. Bluebaum-Gronau, M. Brinke, G. Donnevert, R. Menzel, C. Möhlenkamp, H.T. Ratte, W. Traunspurger, B.V. Danwitz, Interlaboratory comparison of a standardized toxicity test using the nematode Caenorhabditis elegans (ISO 10872). Environ. Toxicol. Chem. **31**, 1525–1535 (2012)
- 45. B.L. Stemmer, G.A. Burton Jr., U.S. Leibfritz-Frederick, Effect of sediment test variables on selenium toxicity to Daphnia magna. Environ. Toxicol. Chem. Int. J. 9, 381–389 (1990)
- 46. E.T. Rodrigues, M.Â. Pardal, The crab Carcinus maenas as a suitable experimental model in ecotoxicology. Environ. Int. **70**, 158–182 (2014)
- I. Moreno-Garrido, M. Hampel, L.M. Lubián, J. Blasco, Sediment toxicity tests using benthic marine microalgae Cylindrotheca closterium (Ehremberg) Lewin and Reimann (Bacillariophyceae). Ecotoxicol. Environ. Saf. 54, 290–295 (2003)
- M.U. Munawar, I.F. Munawar, Phytoplankton bioassays for evaluating toxicity of in situ sediment contaminants, in *Ecological Effects of In Situ Sediment Contaminants*, (Springer, Dordrecht, 1987), pp. 87–105
- C. Blaise, L. Menard, A micro-algal solid-phase test to assess the toxic potential of freshwater sediments. Water Qual. Res. J. 33, 133–152 (1998)
- 50. M.S. Adams, J.L. Stauber, Development of a whole-sediment toxicity test using a benthic marine microalga. Environ. Toxicol. Chem. 23, 1957–1968 (2004)
- H. Castro, F. Ramalheira, V. Quintino, A.M. Rodrigues, Amphipod acute and chronic sediment toxicity assessment in estuarine environmental monitoring: an example from Ria de Aveiro, NW Portugal. Mar. Pollut. Bull. 53, 91–99 (2006)
- 52. A. Scarlett, S.J. Rowland, M. Canty, E.L. Smith, T.S. Galloway, Method for assessing the chronic toxicity of marine and estuarine sediment-associated contaminants using the amphipod *Corophium volutator*. Mar. Environ. Res. 63, 457–470 (2007)
- 53. ASTM, Standard test method for measuring the toxicity of sediment-associated contaminants with estuarine and marine invertebrates, E1367–03. American Society for Testing and Materials, West Conshohocken, PA, USA (2014)
- 54. G.A. Burton, D.L. Denton, K. Ho, D.S. Ireland, Sediment toxicity testing issues and methods. Handb. Ecotoxicol. 5, 70–103 (1995)
- 55. Y. Rong, Y. Tang, L. Ren, W.D. Taylor, V. Razlutskij, L. Naselli-Flores, Z. Liu, X. Zhang, Effects of the filter-feeding benthic bivalve Corbicula fluminea on plankton community and water quality in aquatic ecosystems: a mesocosm study. Water 13, 1827 (2021)
- S.B. Griscom, N.S. Fisher, Bioavailability of sediment-bound metals to marine bivalve molluscs: an overview. Estuaries 27, 826–838 (2004)

- 57. C.K. King, M.C. Dowse, S.L. Simpson, Toxicity of metals to the bivalve Tellina deltoidalis and relationships between metal bioaccumulation and metal partitioning between seawater and marine sediments. Arch. Environ. Contam. Toxicol. 58, 657–665 (2010)
- O. Campana, J. Blasco, S.L. Simpson, Demonstrating the appropriateness of developing sediment quality guidelines based on sediment geochemical properties. Environ. Sci. Technol. 47, 7483–7489 (2013)
- 59. ASTM, Standard guide for conducting laboratory toxicity tests with freshwater mussels, E2455–06. American Society for Testing and Materials, Annual Book of ASTM standards, Vol 11.06, West Conshohocken, PA, USA (2013b)
- 60. R.A. Jeffree, S.J. Markich, P.L. Brown, Comparative accumulation of alkaline-earth metals by two freshwater mussel species from the Nepean River, Australia: consistences and a resolved paradox. Mar. Freshw. Res. 44, 609–634 (1993)
- 61. S.J. Markich, P.L. Brown, R.A. Jeffree, Divalent metal accumulation in freshwater bivalves: an inverse relationship with metal phosphate solubility. Sci. Total Environ. **275**, 27–41 (2001)
- 62. M.C. Newman, Fundamentals of ecotoxicology (CRC Press, 2009)
- 63. USEPA, Methods for measuring the toxicity and bioaccumulation of sediment associated contaminants with freshwater invertebrates. 2nd edn. EPA/600/R-99/064. US Environmental Protection Agency, Office of Water, Washington, DC, USA (2000)
- 64. OECD, Sediment–water chironomid toxicity using spiked water: Test no. 219. OECD Guidelines for the testing of chemicals, Section 2. Organisation for Economic Cooperation and Development, Paris, France (2004b)
- 65. ASTM, Standard test method for measuring the toxicity of sediment-associated contaminants with freshwater invertebrates, E1706–05. American Society for Testing and Materials, West Conshohocken PA, USA (2010)
- S.L. Simpson, G.E. Batley, I.L. Hamilton, D.A. Spadaro, Guidelines for copper in sediments with varying properties. Chemosphere 85, 1487–1495 (2011)
- K.J. Edge, K.A. Dafforn, S.L. Simpson, A.C. Roach, E.L. Johnston, A biomarker of contaminant exposure is effective in large scale assessment of ten estuaries. Chemosphere 100, 16–26 (2014)
- S.E. Hook, E.P. Gallagher, G.E. Batley, The role of biomarkers in the assessment of aquatic ecosystem health. Integr. Environ. Assess. Manag. 10, 327–341 (2014)
- A.D. Biales, M. Kostich, R.M. Burgess, K.T. Ho, D.C. Bencic, R.L. Flick, L.M. Portis, M.C. Pelletier, M.M. Perron, M. Reiss, Linkage of genomic biomarkers to whole organism end points in a toxicity identification evaluation (TIE). Environ. Sci. Technol. 47, 1306–1312 (2013)

Chapter 12 Mesocosm Study in the Reservoir Ecosystem



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter discusses the significance of mesocosm systems in studying the reservoir ecosystem. The use of various types of mesocosm to study the effects of contaminants on freshwater bodies has been in practice for the past three decades. They have been used widely to assess the toxic effects of water-borne contaminants on various biological levels of organizations and their impact on the overall ecosystem. Various studies have been conducted to evaluate the impact of contaminants like pesticides, microplastics, and persistent organic pollutants on macroinvertebrates and fish species, and endpoints like species richness, species diversity, and morphological and physicochemical parameters have been observed. These systems allow the control of various environmental factors and imitate the natural environment to create real exposure scenarios according to the study objectives, available cost, and time. However, if their design does not take account of the physical dimensions and their suitability with the use of suitable test organisms, the replicability and reproducibility of these systems get affected.

Keywords Mesocosms · Reservoirs · Freshwater ecosystem · Ecological realism

1 Introduction

Studying the effects of already present issues like eutrophication, removal of biota, and the addition of toxicants to aquatic ecosystems poses an additional challenge in assessing the overall effects [1]. Controlled laboratory toxicity tests using single species do not account for the complex biological, chemical, and physical interactions that occur in the natural environment. Such tests can only evaluate the real effects of a particular chemical or toxicant for a particular test organism used in the experiment. Thus, this approach is not useful to make comparisons with real-world data. On the contrary, field studies require adequate time, money, and replication to carry out such experiments. Moreover, the conditions are not controlled and a range of natural and anthropogenic stressors can affect the results. Therefore, for examining the aquatic toxicity, an improved test, which enhances the accuracy of ecotoxicological assessments and incorporates environmental realism, is required. For this purpose, some improvements in the design of toxicity testing have been made by the

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 137 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_12

development of aquatic mesocosms [2]. Aquatic mesocosms are the well-designed systems that assess the physical and functional parameters of aquatic ecosystems that could not be assessed in laboratory bioassays [3]. According to the description by Alexander et al. [4], mesocosm is a fusion of laboratory and field systems that permits the control of various factors of concern, such as species and habitats, and allows settings close to the natural conditions to create factual exposure setups than laboratory tests. Their originality is mainly based on ecological realism, which is achieved by incorporating basic constituents of natural ecosystems and maintaining biological, physicochemical, and toxicological parameters and controlled to some extent [5], among which biological characteristics include food chain and food web interactions and physical characteristics contain temperature-salinity flow, as well as chemical characteristics cover toxicant mixtures or fate and transport of chemicals [2]. They are designed in such a way, so that they may imitate the features and functions of shallow lake and reservoir ecosystems [6]. According to the International Union of Pure and Applied Chemistry (IUPAC), a mesocosm is an enclosed experimental system that mimics the natural setting and is used to evaluate the effects of contaminants on a greater scale, compared to a laboratory microcosm [7].

2 Use of Freshwater Mesocosms in Lakes and Reservoir Ecosystems

Mesocosms have been used by toxicologists, ecotoxicologists, and environmental scientists for about the past three decades to study aquatic and terrestrial ecosystems [2]. The term "mesocosm" was first used by Eugene P. Odum to describe moderatesized, simulated test setups, in which populations and ecosystems could be examined simultaneously [8]. Mesocosms have been used in experimental ecology for studying freshwater ecosystems since the twentieth century. Different approaches like freshwater in situ enclosure systems, small-bottle incubations, and multitrophic in situ enclosures containing water columns from surface to sediments have been utilized to assess acidification and eutrophication since the early 1970s [9]. In the USA, in the 1980s, large-scale mesocosm systems were employed for new pesticide registration. However, in the early 1990s, these test systems were restricted because their cost-effectiveness was questionable. Currently, in the USA, the assessment of the pesticides' adverse ecological effects is gathered using a tiered approach, and mesocosm studies are employed at the final tier for ecological risk assessment [10]. Various studies have been conducted using mesocosm to assess the effects of contaminants on freshwater bodies such as streams, rivers, lakes, ponds, and reservoirs. Studies using mesocosms mostly focus on understanding the effects on biological components of ecosystems, such as effects caused by chemical nutritional dilutions on trophic-level species richness, diversity, zooplankton, and phytoplankton community and the effect of chemical contaminants on physicochemical endpoints like turbidity, total suspended solids, total nitrogen phosphorus, etc. In most freshwater studies, fish species are used as indicator organisms. Moreover, macroinvertebrates, amphibians, and plankton communities are also being used. In several previous studies [11-15], the effects of pesticides, insecticides, and herbicides on phytoplanktonic, amphibians, and macroinvertebrates communities have been investigated. While other studies have focused on the impact of nutrient enrichment on various physicochemical parameters of freshwater using indicator species, such as fish [16, 17]. Recent studies have concentrated on the effects of microplastic resins on the diversity of aquatic organisms and reproductive toxicity at various trophic levels [18]. Studies carried out using the freshwater lake and reservoir mesocosms during the last two decades are presented in Table 12.1.

3 Significance of Mesocosms in Ecotoxicity Testing

The mesocosm tests are vital for understanding the impacts of activities, such as habitat fragmentation, species incursion, and habitat fragmentation (Stewart et al. [25]). They are considered useful when they replicate cause–effect relationships. An ideal mesocosm scheme fulfills the criteria of good spatial and temporal replication of the natural ecosystem, which establishes a representative community and covers appropriate predator–prey relationships in natural settings [4].

Mesocosm testing eliminates the need to assess biological and physicochemical parameters separately and provides hazard and exposure assessment in a single experiment [26]. The main objectives of applying the mesocosm approach in ecotoxicological research are to determine the fate and behavior of contaminants, and to observe the effect of toxicants on individual species, populations, food webs, and communities residing in a particular ecosystem [7]. In experimental ecology, they offer an association between field observations and controlled laboratory experimentations. They are applied to appraise the response of individual species or communities in their changing environment, such as the increases or decreases in pH, temperature, and CO₂. In mesocosms, various perturbations are facilitated together with species diversity, environmental variation, and the replication of experiments [9]. Moreover, mesocosms are also employed to refine NOECs and PECs approximations by exposing multiple species belonging to various taxa via fate studies. They could also be used to assess the recuperation of contaminated systems [10]. Thus, mesocosms work as efficient assays to investigate assumptions and generate new hypotheses [26].

4 Size and Types of Mesocosms

The size of mesocosms ranges from constructed ponds, ditches, and fabricated tanks (synthetic enclosure systems) to littoral enclosures, which are the quarantined parts of the natural habitat [27]. The OECD recommends that a size of 1 to 20 m³ is

Mesocosm	Toxicant	Duration	Organisms	Endpoint	Reference
Outdoor tanks, 5000 L	Atrazine, herbicide	25 d	Freshwater Phyto pK community	Chlorophyll fluorescence tolerance	[11]
Outdoor artificial ponds, 1480L	Carbaryl, insecticide Atrazine, herbicide	56-88 d	Amphibian community	Taxonomic composition Survival, body size, development, and time to metamorphosis	[12]
Pond enclosures (0.06 ha)	Trophic level	3 m	Common carp (Cyprinus carpio) Catfish (Ictalurus punctatus)	Turbidity, suspended solids, total phosphorus	[19]
Six macrophyte-dominated lakes: 11 mesocosms	Nutrient enrichment	5-6 w	Fish	pH, alkalinity, TSS, total phosphorus, ammonium, phytoplankton, and zooplankton.	[16]
Shallow eutrophic reservoir: 2 enclosed systems with 360 L water	Oligo-meso and eutro- phic treatments	31 d	Phytoplankton	Dissolved oxygen, species diversity, richness, and dominance, ammonium, free CO_2	[20]
Outdoor artificial ponds, 12,000 L	Ivermectin veterinary drug	265 d	Zoo Pk	Abundance Species richness	[21]
Outdoor tanks 11 m3	Chlorpyrifos OP insecticide	p 66	Zooplanktons	Population and community effects	[13]
Outdoor lentic microcosm (1.05 m; height: 0.9 m,780 L)	Gamma-cyhalothrin Pyrethroid	73 d	Lentic invertebrates	Population dynamics species composition Chlorophyll- <i>a</i> (phyto pK) macrophyte biomass leaf decomposition	[14]
Tropical reservoir in Brazil: outdoor 20 enclosures (9.8m ³)	Nutrient enrichment	5 w	Fish (<i>Nile tilapia</i>)	Total phosphorus and nitrogen, phyto- plankton, and zooplankton biomass	[17]
Subtropical mountain reser- voir 0.30 m depth	Solar UV-B	4 w	Phytoplankton	Chlorophyll-a soluble carbohydrate	[22]
Outdoor ponds 9 m ³		10 m	Benthic macroinvertebrate communities	Drift, the abundance of different taxa decomposition of leaf litter	[15]

Table 12.1 The use of lentic reservoir mesocosms to assess the effects of different environmental toxicants

	Different pesticides used in crop protection programs				
Outdoor: 6 enclosures of 400 L	Cyanobacteria, Microcystin		Bivalve (Limnoperna fortunei)	Total phosphorus and nitrogen, phyto- plankton periphyton, chlorophyll-a	[23]
Lentic mesocosm, 20 m long \times 1 m wide	Bisphenol A Plastics and resin constituent	165 d	Different trophic levels	Watercress volume macroinvertebrates community structure, abundance of fish and gonad morphology	[18]
50 L transparent polyethylene bags	Cyanotoxins	12 d	Zooplanktons and algal biomass	Increase and decrease in microcystin at different times	[24]

suitable for outdoor mesocosm studies. However, according to the literature, the average size of mesocosm in many studies was 1.7 m³ with 49 days of exposure duration [4]. For studies of shorter duration (3–6 months) using smaller organisms like planktonic species, smaller mesocosm with 1–5 m³ is more appropriate. However, for studies of longer periods (6 months or more), larger systems are recommended [28].

Mesocosms are either large enclosures placed in lakes and reservoirs or enclosed in artificial canals or ponds [29]. They can be small plastic enclosure systems, which are open-air [30], or large flexible/rigid polyethylene enclosures or ponds constructed by retaining water masses in dams or lagoons by placing bag systems, pond systems, or tanks. According to design and shape characteristics, various types of mesocosm systems are being used in different ecology studies [25].

The bag system comprises transparent, black-colored polyethylene or polyvinyl chloride bags, through which the separation of a considerable volume of water is attained. These bags are knotted to a floating wharf and have a narrowed bottom with a hose attached outward for water renewal. They are filled with filtered water and seeded with microalgae. Later, water is fertilized to promote the growth of algal blooms, and then planktons are added. After their development, the larvae of test organisms are released and exposed to environmentally relevant concentrations of contaminant [31]. Another variation of the system includes land-based dug-out ponds, which are easy and economical to construct and operate. After digging out the pond, it is sheltered with a plastic liner to avoid water leakage. Ponds are then exposed to sunlight for 3–4 days. Test organisms are cultured separately and then transferred to the ponds [32].

Cement tanks are also a common type of mesocosm, which are usually up to 50 m³ in size [33]. They generally vary in volume from 2000 to 20,000 L and represent a simple food web with ambient environmental conditions like light and temperature [34]. Biological communities are treated easily due to the presence of water and sediment from the same source. Cement tank mesocosms have low cost and cause less contamination of the natural surroundings. However, small-sized tanks cannot retain large predators and can cause wall effects, albeit large tanks may also induce wall effects in long-term mesocosm tests (>50 days) [4, 35]. The most promising tanks for mesocosm testing are super intensive mesocosm tanks known as the maximum tank system, which is intensified and controlled by steady readjustment using a computer-based subjective decision manipulation program. This system can control pH, nutrients, temperature, light, and biotic components (plankton, predators, and bacterial production) [33]. Circular tanks are more promising, compared to square designed tanks. They ensure high carrying capacity and deliver good flows with the advantage of self-cleaning ability because of continuous fish swimming. However, this design does not work well for small-sized less energetic organisms that could not overcome a higher water flow. Rectangular design tanks, also known as raceways, do not imitate usual environmental situations. They do not have good water flow, so higher velocities are not attained to self-clean the tank system [4]. Another mesocosm system, known as the Swedish pond, is considered to be the best for testing organisms, because it has more surface area with square tanks and rounded edges. Moreover, it also offers a self-cleaning process like a circular tank [36].

5 General Design of Aquatic Mesocosms

5.1 Preliminary Aspects

Conventional ecology comprising soil, water, rocks, air, and living biota is the prerequisite of the mesocosm system. However, for designing mesocosm, ecological engineering is also needed, because the system is developed artificially. The enclosed structure is constructed using plastic, metal, or cement. Besides, automated and electrical components are also required. The design and maintenance of ecological and engineering features of mesocosms and their interaction with each other are very crucial for the function of the mesocosm system [37].

Before carrying out any mesocosm test, it is important to define the objectives to determine a suitable experimental design according to the relevant endpoints. Questions, such as probable entry route of contaminants in the water body, frequency of entry, and physicochemical properties of contaminant, must be ascertained before the beginning of the experiment to identify components that are at risk as well as the sampling strategy and frequency. Preliminary studies and lab testing must be conducted before undertaking the mesocosm study when information related to a particular test design is not available [28]. The most important objective of mesocosm studies is to sustain realism, which could be achieved by adding basic components of the natural ecosystem into the mesocosm system. Although the reconstructed system may not mimic the exact natural conditions, it could be simplified by adding key features to ensure ecological representativeness [5].

While designing mesocosms, various scale-related parameters are kept in mind, such as size (radius, depth, and volume), the overall duration of the experiment, and sampling frequency. Moreover, the life cycle of test organisms utilized in the experiment, proper light intensity for temperature balance, primary productivity, construction material and cleaning frequency, temperature control for biogeochemical activities, selection of test organisms or communities, and horizontal and vertical mixing and flow of nutrients should be well considered [38].

5.2 Mesocosm Assembly

Mesocosms can be assembled using inert material either stainless steel, glass, or sealed concrete. To prevent water, exchange systems should be lined using PVC. Water added to the system should be originated from where organisms and sediments are collected. Besides, the physicochemical properties (pH, alkalinity, hardness, dissolved oxygen, and turbidity), chemical contaminants, and nutrients should

also be characterized. During the entire study, the water level should be retained at similar levels. Species communities should be developed by adding organisms from suitable external sources. Free-living species are not recommended, especially in cases where effects on plankton and macrovertebrates have to be studied [28].

For generating communities in mesocosm systems, organisms from laboratory cultures are used to standardize the system to assure the initial similarity of the replicates, although the communities will be unlike the natural communities of the surroundings. The other alternative is that the organisms are taken from natural surroundings and allowed to develop into communities for reaching equilibrium before the beginning of the experiment. Communities of fish, macroinvertebrates, and planktons are usually cultured in the systems [39]. Macrophytes can offer sanctuary for the mesocosm fauna and facilitate the diversity of invertebrates and algae [40]. Notably, they should be maintained according to the requirement of the study. For example, if the mesocosm system focuses on plankton, then macrophytes are recommended to limit their development to 20-30% of the bottom surface. However, if the study emphasizes macroinvertebrates, then, macrophytes should be enhanced to promote a large number of macroinvertebrates. According to study objectives, benthic and planktonic invertebrates are also required to be added to the mesocosms along with water and sediments. Invertebrates such as zooplanktons from phylum Rotifera (Branchiopoda) and benthic organisms from the phyla Annelida (Hirudinea Oligochaeta), Arthropoda (Insecta), Crustacea (Amphipoda), Mollusca (Bivalvia, Gastropoda), and Platyhelminthes (Turbellaria) are mostly investigated in the mesocosm studies. For mesocosm studies using fish, only endemic species are recommended to avoid the contamination of habitats of the native fish, and they should be added when the test system gets stabilized. Fish species used in most of the mesocosm studies include bluegill sunfish (Lepomis macrochirus), carp (Cyprinus carpio), fathead minnow (Pimephales promelas), golden orfe (Leuciscus idus), sticklebacks (Gasterosteus aculeatus), mosquito fish (Gambusia affinis), and rainbow trout (Oncorhynchus mykiss). Large mesocosms are recommended for fish studies and fish species are selected according to the size of the experimental system and the objective of the test. Before chemical dosing, mesocosms are adapted to ascertain the growth of a population community in terms of age and sex structure. The adaptation period varies with the origin of the water introduced in the system [28].

5.3 Protocol of Mesocosm Preparation

Mesocosm systems are first either dried for four days or treated with chemicals, such as HCl, to get rid of predators. Systems are then filled with nearby reservoir water, which is fertilized using poultry manure of about 40 g.m⁻³ along with chemical fertilizer (2.4 g superphosphate of lime, ammonium sulfate 1.6 g, and urea 1.08 g) for three days. Then, different planktonic community starts to develop in a process termed succession. The diatom may emerge first, followed by nano and

dinoflagellates. However, the rotifers and ciliates develop at the end. When an adequate population is established, cultured larvae of test organisms are added to the mesocosm system. Various biotic and abiotic factors are maintained and synchronized during the whole experiment, such as the maintenance of plankton growth rate and related environmental conditions (nutrients, temperature, and light intensity). Moreover, water quality parameters and other biotic factors are also regularly monitored. During the rearing period of test organisms, water analysis is carried out at regular intervals and food consumption is also observed [30].

6 Limitations

Mesocosms experiment contributed to understanding ecology community and ecosystem [41]. However, these tests are criticized for being unrealistic and less related to natural ecosystems. Other general limitation includes oversimplification and worse repeatability. Furthermore, the findings of mesocosm tests are difficult to extrapolate to bigger and natural ecosystems [42]. Mesocosms are generally considered replicable to the natural ecosystem. However, there is more variability among the biological parameters assessed at the individual level than those estimated at the community level [43]. Notably, large mesocosms can be replicable, if they are permitted to grow naturally and placed adjacent to the source system [44]. Achievement of an appropriate level of ecological realism is dependent on the project level and available funds. At a lower level of biological organization, there is high reproducibility but low realism. Larger systems, despite being ecologically realistic, still cannot offer better replication, compared to small systems [4]. The realism and replication aspects of mesocosm in the time and space domain are explained in Fig. 12.1, implying that there is a trade-off between these aspects and that none of the approaches is perfect [25]. Moreover, in the mesocosm test, the environmentally relevant levels of chemicals show less control and precision over the test progress. Certainly, concentrations up to sub-ppm to ppb can be detected, considering the existing instrumentation availability. However, the approaches used to determine the species distribution need concentrations away from the upper limit of the expected levels. Therefore, the physicochemical and biological parameters that could be assessed using standardized tests cannot be determined using mesocosm testing. Nevertheless, limitation this could be overcome by improving the instrumentation [26].

7 Conclusion

Mesocosm test is the fusion of laboratory and field experiments and has been used efficiently for the past few decades to assess the ecotoxicological impacts of contaminants at various biological scales. However, due to complex biological



Biological organization

Fig. 12.1 Conceptual diagram representing space and time domain for mesocosms

interactions, it is difficult to achieve reproducibility and ecological realism. Thus, their development and use at higher organizational levels should understand the knowledge of the environmental processes, interrelation of species and their habitat, and the engineering design of the system. Thus, before carrying out such experiments on bigger scales, pilot studies must be conducted to avoid the wastage of time and money. To implement scientific research, the community is required for bringing forth novel ideas to modify such systems on an engineering and ecological basis and improve the realism and replicability of these systems.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of Interest The author declares no conflict of interest.

References

- 1. G. Swartzman, K.A. Rose, Simulating the biological effects of toxicants in aquatic microcosm systems, in *Developments in Environmental Modelling*, (Elsevier, 1984), pp. 123–134
- 2. P.L. Pennington, The design, construction, operation, and maintenance of the replicated modular estuarine mesocosm (2007)

- L.W. Touart, Regulatory endpoints and the experimental design of aquatic mesocosm tests, in *Aquatic Mesocosm Studies in Ecological Risk Assessment*, (CRC Press Boca, Raton, FL, 1994), pp. 25–34
- 4. A. Alexander, E. Luiker, M. Finley, J. Culp, Mesocosm and field toxicity testing in the marine context, in *Marine Ecotoxicology*, (Elsevier, 2016), pp. 239–256
- 5. T. Caquet, L. Lagadic, S.R. Sheffield, Mesocosms in ecotoxicology (1): outdoor aquatic systems, in *Reviews of Environmental Contamination and Toxicology*, (Springer, 2000), pp. 1–38
- O. Sampedro, E. Schaum, J.R. Salgueiro, Inexpensive remote monitoring system for freshwater mesocosms. Measurement 108, 41–47 (2017)
- 7. C. Amiard-Triquet, How to improve toxicity assessment? From single-species tests to mesocosms and field studies, in *Aquatic Ecotoxicology*, (Elsevier, 2015), pp. 127–151
- T. Guy-Haim, H. Alexander, T.W. Bell, R.L. Bier, L.E. Bortolotti, C. Briseño-Avena, X. Dong, A.M. Flanagan, J. Grosse, L. Grossmann, What are the type, direction, and strength of species, community, and ecosystem responses to warming in aquatic mesocosm studies and their dependency on experimental characteristics? A systematic review protocol. Environ. Evid. 6, 6 (2017)
- 9. M.R. Olivier, The effects of climate change on an alpine lake system, a mesocosm experiment (2017)
- 10. T. Caquet, Use of aquatic mesocosms in ecotoxicology, state of the art and perspectives. Radioprotection-Colloques **37** (2002)
- F. Seguin, F. Le Bihan, C. Leboulanger, A. Bérard, A risk assessment of pollution: induction of atrazine tolerance in phytoplankton communities in freshwater outdoor mesocosms, using chlorophyll fluorescence as an endpoint. Water Res. 36, 3227–3236 (2002)
- 12. M.D. Boone, S.M. James, Interactions of an insecticide, herbicide, and natural stressors in amphibian community mesocosms. Ecol. Appl. 13, 829–841 (2003)
- P. Lopez-Mancisidor, G. Carbonell, A. Marina, C. Fernández, J.V. Tarazona, Zooplankton community responses to chlorpyrifos in mesocosms under Mediterranean conditions. Ecotoxicol. Environ. Saf. 71, 16–25 (2008)
- R.P. van Wijngaarden, I. Barber, T.C. Brock, Effects of the pyrethroid insecticide gammacyhalothrin on aquatic invertebrates in laboratory and outdoor microcosm tests. Ecotoxicology 18, 211–224 (2009)
- A. Auber, M. Roucaute, A. Togola, T. Caquet, Structural and functional effects of conventional and low pesticide input crop-protection programs on benthic macroinvertebrate communities in outdoor pond mesocosms. Ecotoxicology 20, 2042 (2011)
- B. Moss, D. Stephen, D. Balayla, E. Bécares, S. Collings, C. Fernández-Aláez, M. Fernández-Aláez, C. Ferriol, P. García, J. Gomá, Continental-scale patterns of nutrient and fish effects on shallow lakes: synthesis of a pan-European mesocosm experiment. Freshw. Biol. 49, 1633–1649 (2004)
- R.F. Menezes, J.L. Attayde, F.R. Vasconcelos, Effects of omnivorous filter-feeding fish and nutrient enrichment on the plankton community and water transparency of a tropical reservoir. Freshw. Biol. 55, 767–779 (2010)
- G. De Kermoysan, S. Joachim, P. Baudoin, M. Lonjaret, C. Tebby, F. Lesaulnier, F. Lestremau, C. Chatellier, Z. Akrour, E. Pheron, Effects of bisphenol A on different trophic levels in a lotic experimental ecosystem. Aquat. Toxicol. 144, 186–198 (2013)
- J.J. Parkos III, J. Santucci, J. Victor, D.H. Wahl, Effects of adult common carp (Cyprinus carpio) on multiple trophic levels in shallow mesocosms. Can. J. Fish. Aquat. Sci. 60, 182–192 (2003)
- L.O. Crossetti, C.E.D.M. Bicudo, Structural and functional phytoplankton responses to nutrient impoverishment in mesocosms placed in a shallow eutrophic reservoir (Garças Pond), São Paulo, Brazil. Hydrobiologia 541, 71–85 (2005)

- H. Sanderson, B. Laird, L. Pope, R. Brain, C. Wilson, D. Johnson, G. Bryning, A.S. Peregrine, A. Boxall, K. Solomon, Assessment of the environmental fate and effects of ivermectin in aquatic mesocosms. Aquat. Toxicol. 85, 229–240 (2007)
- C. Medina, B. Tracanna, M. Hilal, J. González, F. Prado, Effect of solar UV-B exclusion on the phytoplankton community in a sub-tropical mountain reservoir: a mesocosm study. Lakes Reserv. Res. Manag. 15, 237–253 (2010)
- 23. D. Cataldo, A. Vinocur, O. Inés, E. Paolucci, V. Leites, D. Boltovskoy, The introduced bivalve Limnoperna fortunei boosts Microcystis growth in Salto Grande reservoir (Argentina): evidence from mesocosm experiments. Hydrobiologia 680, 25–38 (2012)
- 24. J. dos Santos Severiano, V.L. dos Santos Almeida-Melo, M. do Carmo Bittencourt-Oliveira, M.A. Chia, A. do Nascimento Moura, Effects of increased zooplankton biomass on phytoplankton and cyanotoxins: a tropical mesocosm study. Harmful Algae 71, 10–18 (2018)
- R.I. Stewart, M. Dossena, D.A. Bohan, E. Jeppesen, R.L. Kordas, M.E. Ledger, M. Meerhoff, B. Moss, C. Mulder, J.B. Shurin, Mesocosm experiments as a tool for ecological climate-change research, in *Advances in Ecological Research*, (Elsevier, 2013), pp. 71–181
- 26. M. Auffan, A. Masion, C. Mouneyrac, C. De Garidel-Thoron, C.O. Hendren, A. Thiery, C. Santaella, L. Giamberini, J.-Y. Bottero, M.R. Wiesner, Contribution of mesocosm testing to a single-step and exposure-driven environmental risk assessment of engineered nanomaterials. NanoImpact 13, 66–69 (2019)
- T.P. Boyle, J.F. Fairchild, The role of mesocosm studies in ecological risk analysis. Ecol. Appl. 7, 1099–1102 (1997)
- OECD, Guidance Document on Simulated Freshwater Lentic Field Tests (Outdoor Microcosms and Mesocosms). ENV/JM/MONO 17 (2006)
- 29. M.L. Larramendy, *Ecotoxicology and Genotoxicology: Non-traditional Terrestrial Models* (Royal Society of Chemistry, 2017)
- 30. P. Lavens, P. Sorgeloos, FAO Fisheries technical paper 361 (University of Ghent, Ghent, 1996)
- 31. V. Øiestad, Growth and survival of herring larvae and fry (Clupea harengus L.) exposed to different feeding regimes in experimental ecosystems: outdoor basin and plastic bags, Inst. Marine Res. (1983)
- 32. L. Sturmer, Zooplankton composition and dynamics in fingerling red drum rearing ponds (1990)
- B. Urup, Methods for the production of turbot fry using copepods as food, European Aquaculture Society Special Publication (Belgium) (1994)
- 34. R.L. Graney, Aquatic Mesocosm Studies in Ecological Risk Assessment (CRC Press, 1993)
- 35. G. Grice, R. Harris, M. Reeved, J. Heinbokel, C. Davis, Large-scale enclosed water-column ecosystems an overview of Foodweb I, the Final CEPEX Experiment. J. Mar. Biol. Assoc. UK 60, 401–414 (1980)
- 36. R.G. Piper, *Fish hatchery management* (US Department of the Interior, Fish and Wildlife Service, 1982)
- 37. P. Kangas, W. Adey, Mesocosm Management (2008)
- J.E. Petersen, V.S. Kennedy, W.C. Dennison, W.M. Kemp, Enclosed experimental ecosystems and scale, in *Tools for Understanding and Managing Coastal Ecosystems*, (Springer, 2009)
- 39. M. Nikinmaa, An introduction to aquatic toxicology (Elsevier, 2014)
- 40. T. Brock, J. Lahr, P. Van den Brink, Ecological risks of pesticides in freshwater ecosystems; Part 1: herbicides, in, Alterra (2000)
- T.G. Benton, M. Solan, J.M. Travis, S.M. Sait, Microcosm experiments can inform global ecological problems. Trends Ecol. Evol. 22, 516–521 (2007)
- 42. A.C. Spivak, M.J. Vanni, E.M. Mette, Moving on up: can results from simple aquatic mesocosm experiments be applied across broad spatial scales? Freshw. Biol. 56, 279–291 (2011)
- T. Caquet, L. Lagadic, G. Monod, J.-C. Lacaze, A. Couté, Variability of physicochemical and biological parameters between replicated outdoor freshwater lentic mesocosms. Ecotoxicology 10, 51–66 (2001)

44. R.M. Harris, P.D. Armitage, A.M. Milner, M.E. Ledger, Replicability of physicochemistry and macroinvertebrate assemblages in stream mesocosms: implications for experimental research. Freshw. Biol. 52, 2434–2443 (2007)

Part IV Ecotoxicological Effects and Mechanism of Pollutants in the Reservoir

Chapter 13 Molecular Toxicity Mechanism of Heavy Metals in the Reservoir



Suman Thodhal Yoganandham and De-Sheng Pei 🝺

Abstract Protecting water quality related to metal exposure is a significant problem in a reservoir. Most heavy metals are carcinogenic substances. For example, arsenic, cadmium, chrome, and nickel are listed by the International Cancer Research Organization as carcinogens and are used extensively. Metal exposure derived from water, air, or food may result in acute or chronic toxicity. The bioaccumulation of heavy metals may cause toxic effects on different tissues and organs. Growth, proliferation, differentiation, and damage repair are all impeded after exposure to a high concentration of heavy metals. This chapter mainly discussed the molecular pathways of heavy metal toxicity in the reservoir, including reactive oxygen species (ROS) formation, antioxidant defense balance, enzyme inactivation, DNA damage, and cell death.

Keywords Heavy metals · Antioxidant · DNA damage · Cell death · Reservoir

1 Introduction

Heavy metal contamination is a major concern due to its serious health effects on the reservoir [1]. Because of the pervasiveness and persistence of chemicals, it is also regarded as a critical environmental problem [2]. Therefore, heavy metal pollution in the trophic chain must be quantified, characterized, and analyzed [3, 4]. Toxicity and bioaccumulation of heavy metals in the reservoir may have serious human health consequences, such as hepatitis, kidney damage, nervous system disorders, coronary artery disorders, hematological implications, reproductive outcomes, and cancer [4–9].

Environmental matrices in the reservoir, such as soil, air, and water also contain lead (Pb), cadmium (Cd), chromium (Cr), mercury (Hg), nickel (Ni), copper (Cu), zinc (Zn), iron (Fe), and manganese (Mg) [2]. However, heavy metals such as copper, iron, manganese, molybdenum, zinc, and nickel are essential micronutrients for the body's metabolism. Nevertheless, other heavy metals, such as mercury, cadmium, lead, and chromium, do not play a positive role even in minimal concentration and cause health risks [10–12].

Metals can be found in dissolved, suspended, colloid ions, and solid sediments in the reservoir [13, 14]. The concentration of these metal ions is heavily influenced by redox potential, biological processes, pH, ion strength, and the behavior of organic and inorganic chelators [15]. Over the last few decades, the contamination of water supplies by the indiscriminate dumping of heavy metals has caused global concern. Rapid population growth have contributed to significant water contamination in many developing countries, such as rivers, lakes, and reservoirs [16]. However, heavy metals can be accumulated in the sediments of reservoirs [17, 18]. Artificial reservoirs provide a significant role in functioning as sediment accumulation, which has attracted more attention recently [19].

2 Heavy Metal Pollution in the Reservoir

As shown in Table 13.1, in China, from 2014 to 2016, six heavy metals (zinc, chromium, cadmium, nickel, copper, and lead) were investigated in the Three Gorges Reservoir (TGR) to assess ecological risks. In the mainstream of the TGR sediments, concentrations of heavy metals and pollution levels were not significantly different between 2014 and 2016. Cadmium was high, while other metals were extremely low [20]. In Nigeria, the sediment composition of the Asejire Reservoir was examined to determine the heavy metal content. Twenty stations were chosen, sampled, and measured using standard methods. The findings revealed that sediment in all stations was mildly acidic, with low conductivity and organic matter. In Ghana, Anash et al. [22] investigated the concentration and distribution of heavy metals in the Weija Reservoir. Heavy metals in water, suspended particles, and sediments

Sample collection site	Sampling time	Elements	Determination method	References
Three Gorges Reser- voir, China	2016–2014	Cd, Cr, Cu, Ni, Pb, and Zn	ICP-MS	[20]
Asejire Reservoir, Nigeria		Fe, Pb, Cu, Zn, Mn, Al, Ba, Ni, and Cr		[21]
Weija Reservoir, Ghana		Cd, Cu, Ni, Pb, and Zn	AAS	[22]
Biliuhe, Tanghe, and Dahuofang Reservoirs	2015	(Fe, Mn, Cu, Cd, Pb, Zn, and Cr	AAS	[23]
Chah Nimeh water reservoir, Iran	2012	Cr, Cd, Cu, Mn, Fe, Pb, Zn, and Ni	AAS	[24]
Manwan Reservoir, China.	2011	As, Cd, Cr, Cu, Pb, and Zn	ICP-AES	[25]
Guanting Reservoir, China	2009	Cu, Zn, Cr, Ni, Cd, Pb, and As	ICP-MS	[26]
Hongfeng and Baihua Reservoirs, China	December 2010 and April 2012	Hg, Cd, Pb, Cr, Cu, and As	AAS and GFAAS	[27]

Table 13.1 Heavy metal pollution in the reservoirs

were measured using atomic absorption spectrophotometry. This study discovered that heavy metal accumulation concentrations in the sediments of Weija Reservoir are ranked in the following order: copper> manganese> iron> zinc> nickel> chromium> lead> arsenic> mercury> cadmium [22]. A high concentration of copper was found in the sediment [22]. Notably, a sampling campaign was carried out in February 2015 to investigate the heavy metal accumulation and potential toxicity of sediment cores from 5 reservoirs in the northeastern region of Liaoning and Jilin, China. The concentrations of most metals (manganese, iron, cadmium, copper, chromium, zinc, and lead) were detected. Interestingly, cadmium is accumulated significantly according to the findings [23]. In Iran, the concentrations of heavy metals (cadmium, chromium, manganese, copper, lead, iron, nickel, and zinc) were measured in the Chah Nimeh Reservoir of water and sediments [24]. Heavy metal concentrations in sediments were found to be higher in the reservoir water. The ranking concentrations of heavy metals in the sediments are: Iron > manganese > zinc > nickel > lead > chromium > cadmium > copper [24]. In 2011, sedimentsamples were collected from Manwan Reservoir (China) sites to detect the source of heavy metals (aluminum, arsenic, copper, iron, cadmium, chromium, manganese, lead, and zinc). The results showed that heavy metal sources were mainly categorized into natural and anthropogenic forms. Cadmium, arsenic, copper, chromium, zinc, and lead concentrations in some sediment areas exceed the standard's requirement [25]. In Guanting Reservoir (China), topsoil samples were assessed for the concentrations of zinc, nickel, chromium, lead, arsenic, and cadmium. Mean copper, zinc, chromium, nickel, cadmium, lead, and arsenic were quantified at 16.8, 59.4, 37.8, 18.3, 0.32, 20.1, and 8.67 mg/kg, respectively [26]. Further, mercury, cadmium, lead, chromium, and copper were detected in surface water and in the sediment of Hongfeng and Baihua Reservoirs (China) [27].

3 Bioaccumulation of Heavy Metals in the Reservoir's Organisms

Six commercial fish species were found in three key Cauvery Delta River Reservoirs, India, which were accumulated with heavy metals (iron, manganese, copper, chromium, lead, nickel, and zinc) [28]. The highest concentration of iron was contained in fish specimens, subsequently followed by lead, zinc, manganese, chromium, nickel, and copper (Table 13.2). The concentration of lead, chromium, and zinc in several samples was above the permitted limits of the Food and Agriculture Organization (FAO) of the United Nations [28]. In Rawal Lake Reservoir (Pakistan), four edible fish including *Cirrhinus mrigala*, *Tor putitora*, *Channa punctatus*, and *Labeo calbasu* were investigated. The results showed that the concentrations of Ni, Cr, and Pb in the muscle of four fish were higher than that requested by the World Health Organization (WHO) [29]. Heavy metal levels in the kidney and liver were relatively high. Metal concentrations were higher in

Sample collection site	Organism	Elements	Determination method	References
Three major reservoirs of	Fish	Fe, Mn, Cu,	AAS	29
the River Cauvery Delta		Cr, Pb, Zn,		
Region, India		and Ni		
Rawal	Fish	Zn, Cu, Cd,	AAS	30
Lake Reservoir, Pakistan		Pb, Co, Ni,		
		Mn, and Cr		
Keban Dam Reservoir,	1 mussel, 1 crayfish,	As, Cd, Cu,	AAS and	31
Turkey	6 wild fish, and	Pb, and Zn	GFAAS	
	1 farmed fish			
Shah Jamal Reservoir of	Fish	Cr, Ni, and Pb	AAS	32
India				
Wanzhou section, Three	Fish	Pb, Cr, Cd,	HG-AAS	33
Gorges Reservoir, China		As, and Hg		

Table 13.2 Heavy metal accumulation in the reservoir's organisms

post-monsoon fish organs (skins, muscles, and gills) than in pre-monsoon fish organs. Heavy metals concentrations were decreased in the following order: zinc > lead > iron > nickel > chromium > nickel > cobalt> copper > cadmium in the pre-monsoon season, while the post-monsoon season was followed by iron > lead > chromium > nickel > zinc > copper > manganese > cadmium. Similarly, the Cu levels were similar to that of *Cirrhinus mrigala*, but lower in *Tor putitora* and Channa punctatus [29]. In 2017, Varol and Sünbül examined the concentration of five heavy metals in aquatic organisms from the Keban Dam Reservoir (Turkey) on the Euphrates River [30]. The highest concentrations of cadmium, lead, and arsenic were found in mussels, while the highest concentrations of copper and zinc were observed in crayfish. The effects of heavy metal pollution on *Heteropneustes fossilis* and Channa striata were also investigated in the Shahjamal Reservoir (India) [31]. A considerably higher concentration of heavy metals such as nickel, lead, and chromium was measured in many fish tissues (muscle, liver, gill, and kidney) in this reservoir. The genotoxicity of the heavy metal for fish was confirmed using a micronucleus erythrocyte test and comet assay. The concentrations of heavy metals (chromium, nickel, and lead) have highly increased compared to the recommended values of the Federal Environmental Protection Agency (FEPA) [31]. Similarly, the muscles of eleven fish collected from the Yangtze River in China were tested for arsenic, chromium, lead, mercury, and cadmium. Hg, Cr, Pb, As, and Cd levels were found to be below the recommended limits [32].

4 Molecular Toxicity Mechanism of Heavy Metals

The detailed toxicological mechanism of heavy metals is discussed in Fig. 13.1. Initially, the reactive oxygen species (ROS) level increases and reduces the antioxidant level. The cells are protected from free radicals by antioxidants, such as



The attack of heavy metals on a cell and the balance between ROS production and the subsequent defense presented by antioxidants.

Fig. 13.1 The attack of heavy metals on cells and the balance between ROS production and the subsequent defense by antioxidants [33]

glutathione in its reduced form (GSH). The enzyme of glutathione peroxidase easily converts GSH into oxidized conditions states (GSSG). In normal conditions, GSH is 90% of the total glutathione amount, and the oxidized GSSG is 10%. However, the GSSG concentration under oxidative stress is higher than the GSH and catalyzed by a protein disulfide isomerase [34]. A further biomarker for oxidative stress is lipid molecules within the cell membrane, leading to lipid peroxidation [35, 36]. Moreover, ROS can cause significant damage to proteins, cells, nuclear acids, lipids, and membranes at high concentrations [37].

Toxic heavy metals, such as Cd and Hg, disrupt certain body functions in humans. Metallothionein (MT) can theoretically react with heavy metals and act as an antidote by removing heavy metals from the body [38]. SOD was determined to produce primarily H_2O_2 through superoxides, which can benefit species against free radicals. CAT can catalyze H_2O_2 into harmless H_2O and O_2 . Xu et al. found that the MT in *Crassostrea hongkongensis* could potentially respond to the presence of Cd and they established a practical approach to specifically monitor ChMT from the oyster tissues [39]. Chen et al. [40] found that Cd exposure altered the levels of transferrin in the livers of *Pseudosciaena crocea*. Feed ions in the *Pseudosciaena* serum expanded rapidly following the treatment of heavy metal Cd ions

[40]. Another study indicated that Cd and Zn were significantly absorbed in *Salmo platycephalus* gills after 15 days of exposure to Cd and Zn [41]. Moreover, oxidative stress caused by ROS is a well-known mechanism of heavy metal-induced damage, and apoptosis, caspase activation, and ultrastructural changes were observed after exposure to heavy metals [42].

5 Mechanism of Heavy Metal's Detoxification

Metal ions may bind to specific ligand molecules of living organisms under a phenomenon known as chelation [43]. Phytochelatins (PCs) are plant-derived protein-ligand molecules that chelate metal ions when plants are exposed to heavy metals [44–46]. Several studies have shown that PCs are synthesized and formed by a glutathione (GSH) PC syntheses enzyme [47, 48]. Metal ion PCs are effectively delivered and isolated from cellular proteins into vacuoles, reducing damage caused by heavy metal ions. ROS formation prevents DNA repair, and DNA crossconnection with proteins is a crucial factor in heavy metal carcinogenesis [49]. A homeostasis imbalance between the antioxidants and the pro-oxidants induced by the ROS, which contains radical hydroxyls (HO), radical superoxides (O_2^{-}) , and hydrogen peroxide (H₂O₂), which cause protein, DNA, and lipids to oxidative damage. Intracellular antioxidants inhibit this process by oxidation and reacting with free radicals [50, 51]. Intracellular antioxidants are different from complex systems, including heme oxygenase 1, GSH, NAD(P)H: quinone acceptor oxidoreductase 1 (NQO1), catalases, and superoxide dismutase (SOD) [51-54]. Moreover, nuclear factor (ERD 2)-like protein (NRF2) is a well-known antioxidant element regulator to respond to oxidative stress.

6 Conclusion

This chapter has addressed the critical sources of heavy metal exposure in a reservoir and its toxicological mechanistic pathways. Artificial reservoirs generally trap more sediments. Through the aquatic food chain, heavy metals in the water column eventually make their way into human bodies. Heavy metal exposure, whether direct or indirect, disrupts various intracellular processes. These mechanisms could be target markers of heavy metal-induced carcinogenesis. As, Cd, Cr, and Ni toxicity are frequently caused by oxidative pathways. Furthermore, research into heavy metal-induced cancers and diseases will comprehensively understand these complex mechanisms through pathway analysis.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- V.A. Calmuc, M. Calmuc, M. Arseni, C.M. Topa, M. Timofti, A. Burada, C. Iticescu, L.P. Georgescu, Assessment of heavy metal pollution levels in sediments and of ecological risk by quality indices, applying a case study: The Lower Danube River, Romania. Water 13, 1801 (2021)
- A.S. Worlanyo, L. Jiangfeng, Evaluating the environmental and economic impact of mining for post-mined land restoration and land-use: a review. J. Environ. Manag. 279, 111623 (2021)
- I. Papagiannis, I. Kagalou, J. Leonardos, D. Petridis, V. Kalfakakou, Copper and zinc in four freshwater fish species from Lake Pamvotis (Greece). Environ. Int. 30, 357–362 (2004)
- X. Tao, C.T.A. Chen, Heavy metal pollution status in surface sediments of the coastal Bohai Bay. Water Res. 46, 1901–1911 (2012)
- N. Saha, M.Z.I. Mollah, M.F. Alam, M.S. Rahman, Seasonal investigation of heavy metals in marine fishes captured from the Bay of Bengal and the implications for human health risk assessment. Food Control 70, 110–118 (2016)
- A.C. Bosch, B. O'Neill, G.O. Sigge, S.E. Kerwath, L.C. Hoffman, Heavy metals in marine fish meat and consumer health: a review. J. Sci. Food Agric. 96, 32–48 (2016)
- J. Djedjibegovic, T. Larssen, A. Skrbo, A. Marjanović, M. Sober, Contents of cadmium, copper, mercury and lead in fish from the Neretva River (Bosnia and Herzegovina) determined by inductively coupled plasma mass spectrometry (ICP-MS). Food Chem. 131, 469–476 (2012)
- M.K. Ahmed, N. Shaheen, M.S. Islam, M. Habibullah-al-Mamun, S. Islam, M. Mohiduzzaman, L. Bhattacharjee, Dietary intake of trace elements from highly consumed cultured fish (Labeo rohita, Pangasius pangasius and Oreochromis mossambicus) and human health risk implications in Bangladesh. Chemosphere **128**, 284–292 (2015)
- Y. Tao, Z. Yuan, H. Xiaona, M. Wei, Distribution and bioaccumulation of heavy metals in aquatic organisms of different trophic levels and potential health risk assessment from Taihu lake, China. Ecotoxicol. Environ. Saf. 81, 55–64 (2012)
- 10. G. Tyler, Heavy metals in soil biology and biochemistry, in *Soil Biochemistry*, (CRC Press, 2021), pp. 371–414
- T. Cohen, S.S.Q. Hee, R.F. Ambrose, Trace metals in fish and invertebrates of three California coastal wetlands. Mar. Pollut. Bull. 42, 224–232 (2001)
- M.R. Bruins, S. Kapil, F.W. Oehme, Microbial resistance to metals in the environment. Ecotoxicol. Environ. Saf. 45, 198–207 (2000)
- G.V. Iyengar, P.P. Nair, Global outlook on nutrition and the environment: meeting the challenges of the next millennium. Sci. Total Environ. 249, 331–346 (2000)
- F.A. Jan, M. Ishaq, S. Khan, I. Ihsanullah, I. Ahmad, M. Shakirullah, A comparative study of human health risks via consumption of food crops grown on wastewater irrigated soil (Peshawar) and relatively clean water irrigated soil (lower Dir). J. Hazard. Mater. 179, 612–621 (2010)
- 15. G. Yu, P. Li, G. Wang, J. Wang, Y. Zhang, S. Wang, K. Yang, C. Du, H. Chen, A review on the removal of heavy metals and metalloids by constructed wetlands: bibliometric, removal pathways, and key factors. World J. Microbiol. Biotechnol. 37, 1–12 (2021)
- C. Yang, Y. Wu, F. Zhang, L. Liu, R. Pan, Pollution characteristics and ecological risk assessment of heavy metals in the surface sediments from a source water reservoir. Chem. Speciat. Bioavailab. 28, 133–141 (2016)
- A.M. El Nemr, A. El Sikaily, A. Khaled, Total and leachable heavy metals in muddy and sandy sediments of Egyptian coast along Mediterranean Sea. Environ. Monit. Assess. 129, 151–168 (2007)

- Z. Borovec, Evaluation of the concentrations of trace elements in stream sediments by factor and cluster analysis and the sequential extraction procedure. Sci. Total Environ. 177, 237–250 (1996)
- E. García-Ordiales, J.M. Esbrí, S. Covelli, M.A. López-Berdonces, P.L. Higueras, J. Loredo, Heavy metal contamination in sediments of an artificial reservoir impacted by long-term mining activity in the Almadén mercury district (Spain). Environ. Sci. Pollut. Res. 23, 6024–6038 (2016)
- H. Bing, Y. Wu, J. Zhou, H. Sun, X. Wang, H. Zhu, Spatial variation of heavy metal contamination in the riparian sediments after two-year flow regulation in the Three Gorges Reservoir, China. Sci. Total Environ. 649, 1004–1016 (2019)
- A.I. Godwin, Assessment of heavy metals associated with bottom sediment of Asejire Reservoir, Southwest Nigeria. Int. Lett. Nat. Sci., 55 (2016)
- E. Ansah, D. Nukpezah, J.N. Hogarh, Levels and distribution of heavy metals in Weija Reservoir, Accra, Ghana. West Afr. J. Appl. Ecol. 26, 74–88 (2018)
- L. Zhu, J. Liu, S. Xu, Z. Xie, Deposition behavior, risk assessment and source identification of heavy metals in reservoir sediments of Northeast China. Ecotoxicol. Environ. Saf. 142, 454–463 (2017)
- 24. E. Bazrafshan, F.K. Mostafapour, M. Esmaelnejad, G.R. Ebrahimzadeh, A.H. Mahvi, Concentration of heavy metals in surface water and sediments of Chah Nimeh water reservoir in Sistan and Baluchestan province. Iran. Desalin Water Treat 57, 9332–9342 (2016)
- C. Wang, S. Liu, Q. Zhao, L. Deng, S. Dong, Spatial variation and contamination assessment of heavy metals in sediments in the Manwan Reservoir, Lancang River. Ecotoxicol. Environ. Saf. 82, 32–39 (2012)
- L. Xu, T. Wang, W. Luo, K. Ni, S. Liu, L. Wang, Q. Li, Y. Lu, Factors influencing the contents of metals and As in soils around the watershed of Guanting Reservoir, China. J. Environ. Sci. 25, 561–568 (2013)
- 27. B. Wu, G. Wang, J. Wu, Q. Fu, C. Liu, Sources of heavy metals in surface sediments and an ecological risk assessment from two adjacent plateau reservoirs. PLoS One **9**, 101 (2014)
- S. Dhanakumar, G. Solaraj, R. Mohanraj, Heavy metal partitioning in sediments and bioaccumulation in commercial fish species of three major reservoirs of river Cauvery delta region, India. Ecotoxicol. Environ. Saf. 113, 145–151 (2015)
- 29. R.N. Malik, M.Z. Hashmi, Y. Huma, Heavy metal accumulation in edible fish species from Rawal Lake Reservoir, Pakistan. Environ. Sci. Pollut. Res. **21**, 1188–1196 (2014)
- M. Varol, M.R. Sünbül, Organochlorine pesticide, antibiotic and heavy metal residues in mussel, crayfish and fish species from a reservoir on the Euphrates River, Turkey. Environ. Pollut. 230, 311–319 (2017)
- M. Fatima, N. Usmani, M.M. Hossain, M.F. Siddiqui, M.F. Zafeer, F. Firdaus, S. Ahmad, Assessment of genotoxic induction and deterioration of fish quality in commercial species due to heavy-metal exposure in an urban reservoir. Arch. Environ. Con. Tox. 67, 203–213 (2014)
- 32. J. Li, X. Xie, Heavy metal concentrations in fish species from Three Gorges Reservoir, China, after impoundment. Bull. Environ. Contam. Toxicol. **96**, 616–621 (2016)
- M. Jaishankar, T. Tseten, N. Anbalagan, B.B. Mathew, K.N. Beeregowda, Toxicity, mechanism and health effects of some heavy metals. Interdiscip. Toxicol. 7, 60–72 (2014)
- 34. T.Y. Suman, S.Y. Kim, D.H. Yeom, J. Jeon, Transformation products of emerging pollutants explored using non-target screening: perspective in the transformation pathway and toxicity mechanism—a review. Toxics 10, 54 (2022)
- 35. N. Wadhwa, B.B. Mathew, S. Jatawa, A. Tiwari, Lipid peroxidation: mechanism, models and significance. Int. J. Curr. Sci. **3**, 29–38 (2012)
- 36. S.J.S. Flora, M. Mittal, A. Mehta, Heavy metal induced oxidative stress & its possible reversal by chelation therapy. Indian J. Med. Res. 128, 501 (2008)
- B.B. Mathew, A. Tiwari, S.K. Jatawa, Free radicals and antioxidants: a review. J. Pharm. Res. 4, 4340–4343 (2011)

- D.F. Xiang, J.Q. Zhu, S. Jin, Y.J. Hu, F.Q. Tan, W.X. Yang, Expression and function analysis of metallothionein in the testis of *Portunus trituberculatus* exposed to cadmium. Aquat. Toxicol. 140, 1–10 (2013)
- D. Xu, B. Yu, Y. Zhang, M. Cui, Q. Zhang, Metallothionein protein expression of Crassostrea hongkongensis in response to cadmium stress. J. Shellfish Res. 34, 311–318 (2015)
- 40. J. Chen, Y.H. Shi, M.Y. Li, Changes in transferrin and hepcidin genes expression in the liver of the fish *Pseudosciaena crocea* following exposure to cadmium. Arch. Toxicol. 82, 525 (2008)
- B.H. Hansen, S. Rømma, L.I. Søfteland, P.A. Olsvik, R.A. Andersen, Induction and activity of oxidative stress-related proteins during waterborne Cu-exposure in brown trout (*Salmo trutta*). Chemosphere 65, 1707–1714 (2006)
- K. Renu, R. Chakraborty, H. Myakala, R. Koti, A.C. Famurewa, H. Madhyastha, B. Vellingiri, A. George, A. Valsala Gopalakrishnan, Molecular mechanism of heavy metals (Lead, Chromium, Arsenic, Mercury, Nickel and Cadmium)-induced hepatotoxicity—a review. Chemosphere 271, 129735 (2021)
- 43. S. Chu, X. Feng, C. Liu, H. Wu, X. Liu, Advances in chelating resins for adsorption of heavy metal ions. Ind. Eng. Chem. Res. 61(1), 11309–11328 (2022)
- 44. C.S. Cobbett, Phytochelatin biosynthesis and function in heavy-metal detoxification. Curr. Opin. Plant Biol. **3**, 211–216 (2000)
- 45. C.S. Cobbett, Phytochelatins and their roles in heavy metal detoxification. Plant Physiol. **123**, 825–832 (2000)
- 46. C. Cobbett, P. Goldsbrough, Phytochelatins and metallothioneins: roles in heavy metal detoxification and homeostasis. Annu. Rev. Plant Biol. **53**, 159–182 (2002)
- A.M. Kinnersley, The role of phytochelates in plant growth and productivity. Plant Growth Regul. 12, 207–218 (1993)
- W.E. Rauser, Structure and function of metal chelators produced by plants. Cell Biochem. Biophys. 31, 19–48 (1999)
- E.T. Snow, Metal carcinogenesis: mechanistic implications. Pharmacol. Therapeut. 53, 31–65 (1992)
- N. Ercal, H. Gurer-Orhan, N. Aykin-Burns, Toxic metals and oxidative stress part I: mechanisms involved in metal-induced oxidative damage. CTMCCL 1, 529–539 (2001)
- 51. S. Vertuani, A. Angusti, S. Manfredini, The antioxidants and pro-antioxidants network: an overview. Curr. Pharm. Des. 10, 1677–1694 (2004)
- H. Sies, Oxidative stress: oxidants and antioxidants. Exp. Physiol. Transl. Integr. 82, 291–295 (1997)
- 53. V. Lobo, A. Patil, A. Phatak, N. Chandra, Free radicals, antioxidants and functional foods: impact on human health. Phcog. Rev. 4, 118 (2010)
- 54. H.S. Kim, Y.J. Kim, Y.R. Seo, An overview of carcinogenic heavy metal: molecular toxicity mechanism and prevention. J. Cancer Prevent. **20**, 232 (2015)

Chapter 14 Molecular Toxicity Mechanism of Persistent Organic Pollutants (POPs) in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract Due to the ubiquitous environmental existence, persistent organic pollutants (POPs) and their harmful effects have drawn attention among scientific communities. This chapter particularly emphasizes the mechanistic toxicity pathways of POPs when interacting with aquatic species in the reservoir. Among previously published studies, these emerging chemicals mainly disrupt the aryl receptor (AhR) signaling pathway. Besides, some pesticides groups like polychlorinated biphenyls (PCBs), polycyclic aromatic hydrocarbons (PAHs), and organochlorinated pesticides (PCBs) induced neurodevelopmental disorders (NDD) and caused endocrine disruption, when exposed to mice and zebrafish at low environment-relevant concentrations (ERC). Furthermore, the increased oxidative stress and genetic toxicity with increased apoptosis levels and DNA damage were also observed even at low exposure doses. In summary, POPs possess deleterious effects when interacting with any biological organism. Therefore, regulative authorities should properly implement the rules and control the release of POPs in the environment.

Keywords Persistent organic pollutant \cdot AhR signaling pathway \cdot DNA damage \cdot Oxidative stress

1 Introduction

Over the past decades, persistent organic pollutants (POPs) are recognized as silent killers due to their bioaccumulative potential and their ubiquitous presence in the environment [1]. Generally, persistent organic chemicals belong to the industrial chemicals or the byproducts released from the combustion. They have been identified as significant components in the global ecosystem and cause health risks to human adipose tissue, blood, and fish [2, 3], which may be organochlorine pesticides, including DDT, lindane, chlordane, heptachlor, dieldrin, hexachlorobenzene, and halogenated hydrocarbons. The industrial compounds mainly comprise polycyhydrocarbons polychlorinated clic aromatic (PAHs), biphenyls (PCBs), polychlorinated naphthalene (PCNs), polybrominated biphenyls (PBBs), dibenzop-dioxins (PCDDs), polybrominated diphenyl ethers (PBDEs), and polychlorinated dibenzofurans (PCDFs) [1, 4, 5]. These different classes of chemicals cause various



Fig. 14.1 General classification of the POPs

diseases through cytotoxicity, mutagenicity, neurotoxicity, genetic toxicity, and estrogenicity [6], when interacting with any biological organism [3] (Fig. 14.1).

The integrated biomarker approach has been used widely for vertebrates and invertebrates for environmental monitoring [7]. However, the xenotoxicity mechanism of POPs is still elusive. Therefore, more in-depth studies are required to evaluate the mechanistic toxicity of the POPs. Moreover, the toxicology of POPs is highly complicated, and different chemicals can elicit diverse toxic responses in target organs, tissues, and species in a sex-dependent manner [8, 9]. Most xenobiotic chemicals can be mediated *via* the AhR signaling pathway and target different organs, including endocrine, neurological, and cardiovascular, leading to several well-characterized diseases [10–12].

Previously, various model animals including mouse, *Daphnia magna*, *C. elegans*, zebrafish, and marine medaka have been used to investigate the toxic potential of POPs, such as PFOS, PAHs, PCBs, and PAEs [6, 13–15]. Besides, zebrafish is considered a perfect animal model due to its 85% similarity to the human genome, high growth rate, and reproduction [6]. However, in toxicological studies, various exposure concentrations, exposure duration including acute or chronic, and multigenerational exposures are required to determine the complete toxicity scenarios [16–20]. Similarly, different chemical exposure durations may result in different toxic responses [21].

2 POPs Mechanistic Toxicity in the Three Gorges Reservoir (TGR)

The Three Gorges Reservoir (TGR), one of the largest hydropower reservoirs, extends 670 km and covers an area of 1084 km² from Chongqing to the Three Gorge Dam (TGD) in China. In the TGR, POPs toxicity has gained much intention owing to their increasing ecological risks. However, detailed mechanistic toxicological studies in this area are still elusive. In a recent study, PAHs pollution in the TGR and its toxicological risks were evaluated using the zebrafish model by Tang et al. [22] at the maximum impoundment level (175 m) [22]. Results revealed that transgenic fish Tg(cyp1a:gfp) induced by PAHs showed the highest fluorescent levels, particularly in the middle and lower reaches of the TGR. Moreover, the increased levels of oxidative stress further confirm the findings. Interestingly, in 2011 and 2012, sediment and benthic fish species were sampled and elucidated the organic pollutants levels and their mutagenic and genotoxic potentials [1]. It was found that PAHs, their derivatives, and non-target compounds are considered the main causative agents for the genotoxicity of the TGR's fish [1].

3 AhR Metabolism Mechanism

It is believed that POPs induced transcriptional changes in the detoxification of AhR pathway genes, resulting in the disturbance in AhR levels and induction of CYP1A [23, 24]. Zhou et al. conducted a study to determine the phylogenetic analysis of the AhR mechanism after exposure to the POPs in aquatic animals, and each species behaved differentially when exposed to different classes of POPs [24]. Generally, in the regulation of the AhR pathway, cytochrome P4501 is the major target molecule. When a xenobiotic compound, such as POPs, contacts any biological organism, CYP1A1, a xenobiotic metabolic enzyme, is activated and forms Ahr–ligand complexes. Therefore, AhR-dependent CYP1A1 has a high binding affinity against POPs metabolism [24]. When the AhR–ligand complexes translocate from cytoplasm to the nucleus, it induces the formation of AHR-ARNT and binds a precise xenobiotic response element (XRE) in the promotor region, resulting in disturbing the expression levels of the downstream genes and elevating the oxidative stress levels (Fig. 14.2).

4 POPs Toxicogenetic Endpoints In vitro and In vivo

Different model species have been performed in the past to determine the mechanistic toxicity mechanisms of the POPs. Here, a summary of numerous *in vivo* and *in vitro* studies on POPs' toxicity is presented in Table 14.1. The majority of the



Fig. 14.2 A detailed regulation description of the AhR pathway for the POPs. The figure was adapted from a previously published research [25]

previously published studies have focused on determining developmental toxicity *in vivo*, such as mortality, deformity, and oxidative stress [33, 34]. Likewise, perfluorooctane sulfonate (PFOS) exposure to *C. elegans* elevates the reactive oxygen species (ROS) and increased apoptosis levels with the significantly upregulated *HUS-1* expression [24]. When *C. elegans* was exposed to endosulfan, the expression levels of the germ apoptosis-related genes were increased [27] (Table 14.1).

Even at lower exposure concentrations of Pesticides, polychlorinated biphenyls (PCBs), and organochlorine phosphate (OCPs) perturb the central glucose metabolism signaling pathway, and caused reproductive toxicity and severe liver damage [7, 24]. Previously, a study published by He et al. [32] indicated the reproductive toxicity of the PCBs exposed to adult mice at low exposure concentrations [32]. DNA methylation was observed with the significant upregulation of the associated genes (*Dnmt1, 3a, 3b, 3 l, Uhrf1, Tet2*, and *Tet3*) in mice F₁ generation [32]. Therefore, it can be concluded that the PCBs' transgenerational effects were more related to epigenetic regulation. Furthermore, Lindane exposed to *C. elegans* activates the insulin-growth factor (IGF) pathway by decreasing the transcriptional gene expression (such as *daf-2, sgk-1, akt-1*, and *daf-16*) [30].

Interestingly, some pesticides of POPs induce neurotoxicity with acute exposure to aquatic species. For example, a recent study highlighted that PCBs disturbed the neurological networks and caused neurodevelopmental disorders (NDD) in mice even at low exposure concentrations [31]. The author further demonstrated that chronic exposure to low doses of many pesticides caused serious neurodegenerative diseases, particularly Parkinson's disease, resulting in silent neurotoxicity that may

Chemical	Species	Observation	Toxic endpoints	Reference
Perfluorooctane sul- fonate (PFOS)	C. elegans	Oxidative stress, cell metabolism	↑ROS; ↑Cell apopto- sis; ↑Distinct foci of HUS-1:GFP	[26]
Endosulfan	C. elegans	Cell apoptosis	↑Germ cell apoptosis in mev-1(kn-1) mutant; ↓Apoptosis cep-1(w40), egl-1 (n487), and hus-1 (op241); ↑HUS-1: GFP foci	[27]
Perfluorooctane sul- fonate (PFOS)	C. elegans	Insulin/IGF-1 signal- ing pathway	↓Average lifespan in daf-2(e1370) and daf-16b(KO) mutants	[28]
Polychlorinated biphenyls (PCBs), polybrominated diphenyl ethers (PBDE), polychlorinated dibenzo-p-dioxins (PCDDs), and polychlorinated dibenzo-p-dioxins/ furans (PCDD/Fs)	Aquatic birds	Developmental toxicity	Non-dioxin-like PCBs were ten times higher toxic for birds eggs than other ana- lyzed PCB congeners	[29]
Lindane	C. elegans	Insulin/IGF-1 signal- ing pathway	↓Level of insulin; ↓daf-2, sgk-1, akt-1, and daf-16 genes	[30]
DDT (1,1,1- trichloro-2,2-bis(<i>p</i> - chlorophenyl) ethane)	Mice	Liver damage, repro- ductive toxicity, glu- cose physiology, and central signaling pathway	200 hepatic genes affected by <i>p</i> , <i>p</i> '- DDE, perturb lipid metabolism, mito- chondrial dysfunc- tion, and alterations in glucose and central signaling pathways. Also impairs testos- terone metabolism in the liver revealing endocrine disruption properties.	[7]
Polychlorinated biphenyls (PCBs)	Mice	Neurodevelopmental disorders	Neurodevelopmental disorders (NDD), including autism spectrum disorder (ASD), a high preva- lence of gastrointesti- nal (GI) distress.	[31]

 Table 14.1
 Toxicogenetic endpoints of the POPs observed *in vitro* and *in vivo* assays at environmentally relevant concentrations (ERCs)

(continued)
Chemical	Species	Observation	Toxic endpoints	Reference
Polychlorinated biphenyls (PCBs)	Mice	DNA methylation, reproductive toxicity	DNA methylation patterns of the genes <i>H19, Snrpn, Peg3,</i> and <i>Igf2r</i> as well as the high expression levels of <i>Dnmt1, 3a,</i> <i>3b, 3 l, Uhrf1, Tet2,</i> and <i>Tet3</i> in fully grown germinal vesi- cle oocytes were found in mice off- spring after exposure to PCBs.	[32]
Tetrachlorodibenzo- p-dioxin (TCDD)	D. rerio	Aryl hydrocarbon receptor repressor (AHRR) xenobiotic metabolism pathway	Disturb cytochrome P450 1 (CYP1) metabolism detoxifi- cation pathway using environment-relevant concentrations (ERCs).	[24]

 Table 14.1 (continued)

disturb the nervous system entirely in old age [31]. In summary, POPs can induce different toxicity mechanisms even at lower exposure concentrations and significantly affect the integrity of the whole ecosystem.

5 Conclusion

In summary, POPs are ubiquitously present in the environment and exhibit intricate exposure scenarios for aquatic species. Although published data regarding the mechanistic toxicity of POPs in the TGR is limited, it can be found that some POPs, when interacting with any biological organisms, may target the AhR signaling pathway. For example, mice and zebrafish after pesticides exposure may induce neurodevelopmental disorders (NDD) and caused endocrine disruption properties, such as testicular toxicity even at low exposure concentrations. Exposure to PFOS in *C. elegans* elevates oxidative stress and cell apoptosis. However, there are significant challenges in determining the interactive effects of POPs mixtures. Thus, regulative authorities should focus their attention and ensure that it is adequately banned both in developed and developing countries to combat its serious health effects.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- T. Floehr, B. Scholz-Starke, H. Xiao, J. Koch, L. Wu, J. Hou, A. Wolf, A. Bergmann, K. Bluhm, X. Yuan, M. Roß-Nickoll, A. Schäffer, H. Hollert, Yangtze Three Gorges Reservoir, China: a holistic assessment of organic pollution, mutagenic effects of sediments and genotoxic impacts on fish. J. Environ. Sci. 38, 63–82 (2015)
- H. Chen, C. Wang, H. Li, R. Ma, Z. Yu, L. Li, M. Xiang, X. Chen, X. Hua, Y. Yu, A review of toxicity induced by persistent organic pollutants (POPs) and endocrine-disrupting chemicals (EDCs) in the nematode Caenorhabditis elegans. J. Environ. Manag. 237, 519–525 (2019)
- L.M. Jaacks, L.R. Staimez, Association of persistent organic pollutants and non-persistent pesticides with diabetes and diabetes-related health outcomes in Asia: a systematic review. Environ. Int. 76, 57–70 (2015)
- P. Bruce-Vanderpuije, D. Megson, E.J. Reiner, L. Bradley, S. Adu-Kumi, J.A. Gardella, The state of POPs in Ghana—a review on persistent organic pollutants: environmental and human exposure. Environ. Pollut. 245, 331–342 (2019)
- A.F. Leblanc, E.A. Attignon, E. Distel, S.P. Karakitsios, D.A. Sarigiannis, S. Bortoli, R. Barouki, X. Coumoul, M. Aggerbeck, E.B. Blanc, A dual mixture of persistent organic pollutants modifies carbohydrate metabolism in the human hepatic cell line HepaRG. Environ. Res. 178, 108628 (2019)
- O.M.L. Alharbi, A.A. Basheer, R.A. Khattab, I. Ali, Health and environmental effects of persistent organic pollutants. J. Mol. Liq. 263, 442–453 (2018)
- N. Morales-Prieto, J. Ruiz-Laguna, D. Sheehan, N. Abril, Transcriptome signatures of p, p'-DDE-induced liver damage in Mus spretus mice. Environ. Pollut. 238, 150–167 (2018)
- M.L. Bates, S.M. Bengtson Nash, D.W. Hawker, E.C. Shaw, R.A. Cropp, The distribution of persistent organic pollutants in a trophically complex Antarctic ecosystem model. J. Mar. Syst. 170, 103–114 (2017)
- T.Q. Doan, H.F. Berntsen, S. Verhaegen, E. Ropstad, L. Connolly, A. Igout, M. Muller, M.L. Scippo, A mixture of persistent organic pollutants relevant for human exposure inhibits the transactivation activity of the aryl hydrocarbon receptor in vitro. Environ. Pollut. 254, 113098 (2019)
- S.J. Varjani, E. Gnansounou, A. Pandey, Comprehensive review on toxicity of persistent organic pollutants from petroleum refinery waste and their degradation by microorganisms. Chemosphere 188, 280–291 (2017)
- A. Khezri, B. Lindeman, A.K. Krogenæs, H.F. Berntsen, K.E. Zimmer, E. Ropstad, Maternal exposure to a mixture of persistent organic pollutants (POPs) affects testis histology, epididymal sperm count and induces sperm DNA fragmentation in mice. Toxicol. Appl. Pharmacol. 329, 301–308 (2017)
- R. Islam, S. Kumar, J. Karmoker, M. Kamruzzaman, M.A. Rahman, N. Biswas, T.K.A. Tran, M.M. Rahman, Bioaccumulation and adverse effects of persistent organic pollutants (POPs) on ecosystems and human exposure: a review study on Bangladesh perspectives. Environ. Technol. Innov. 12, 115–131 (2018)
- C. Zhou, L. Gao, J.A. Flaws, Prenatal exposure to an environmentally relevant phthalate mixture disrupts reproduction in F1 female mice. Toxicol. Appl. Pharmacol. 318, 49–57 (2017)
- J. Zhou, Z.-H. Cai, K.-Z. Xing, Potential mechanisms of phthalate ester embryotoxicity in the abalone Haliotis diversicolor supertexta. Environ. Pollut. 159, 1114–1122 (2011)
- H.-T. Gao, R. Xu, W.-X. Cao, Q.-N. Di, R.-X. Li, L. Lu, Q. Xu, S.-Q. Yu, Combined effects of simultaneous exposure to six phthalates and emulsifier glycerol monosterate on male reproductive system in rats. Toxicol. Appl. Pharmacol. 341, 87–97 (2018)

- M. Saad, A. Matheeussen, S. Bijttebier, E. Verbueken, C. Pype, C. Casteleyn, G.C. Van, S. Apers, L. Maes, P. Cos, In vitro CYP-mediated drug metabolism in the zebrafish (embryo) using human reference compounds. Toxicol. In Vitro 42, 329 (2017)
- M. Saad, K. Cavanaugh, E. Verbueken, C. Pype, C. Casteleyn, C.V. Ginneken, S.V. Cruchten, Xenobiotic metabolism in the zebrafish: a review of the spatiotemporal distribution, modulation and activity of Cytochrome P450 families 1 to 3. J. Toxicol. Sci. 41, 1 (2016)
- K. Kurogi, T.A. Liu, Y. Sakakibara, M. Suiko, M.C. Liu, The use of zebrafish as a model system for investigating the role of the SULTs in the metabolism of endogenous compounds and xenobiotics. Drug Metab. Rev. 45, 431 (2013)
- Z.H. Li, D. Alex, S.O. Siu, I.K. Chu, J. Renn, C. Winkler, S. Lou, S.K. Tsui, H.Y. Zhao, W.R. Yan, Combined in vivo imaging and omics approaches reveal metabolism of icaritin and its glycosides in zebrafish larvae. Mol. BioSyst. 7, 2128 (2011)
- L.I. Ming, Y. Cao, Z. Zhao, S. Lin, A. Meng, Characterization and expression pattern of pouII1, a novel class II POU gene in zebrafish. Chin. Sci. Bull. 46, 1523–1527 (2001)
- 21. Y. Chang, L.K. Siddens, L.K. Heine, D.A. Sampson, Z. Yu, K.A. Fischer, C.V. Löhr, S.C. Tilton, Comparative mechanisms of PAH toxicity by benzo[a]pyrene and dibenzo[def,p] chrysene in primary human bronchial epithelial cells cultured at air-liquid interface. Toxicol. Appl. Pharmacol. **379**, 114644 (2019)
- 22. Y.-M. Tang, M. Junaid, A. Niu, S. Deng, D.-S. Pei, Diverse toxicological risks of PAHs in surface water with an impounding level of 175m in the Three Gorges Reservoir Area, China. Sci. Total Environ. 580, 1085–1096 (2017)
- N. Hamid, M. Junaid, D.-S. Pei, Individual and combined mechanistic toxicity of sulfonamides and their implications for ecological risk assessment in the Three Gorges Reservoir Area (TGRA), China. J. Hazardous Mater. 382, 121106 (2020)
- 24. H. Zhou, Y. Qu, H. Wu, C. Liao, J. Zheng, X. Diao, Q. Xue, Molecular phylogenies and evolutionary behavior of AhR (aryl hydrocarbon receptor) pathway genes in aquatic animals: Implications for the toxicology mechanism of some persistent organic pollutants (POPs). Chemosphere 78, 193–205 (2010)
- J. Mimura, M. Ema, K. Sogawa, Y. Fujii-Kuriyama, Identification of a novel mechanism of regulation of Ah (dioxin) receptor function. Genes Dev. 13, 20–25 (1999)
- 26. X. Guo, Q. Li, J. Shi, L. Shi, B. Li, A. Xu, G. Zhao, L. Wu, Perfluorooctane sulfonate exposure causes gonadal developmental toxicity in Caenorhabditis elegans through ROS-induced DNA damage. Chemosphere 155, 115–126 (2016)
- 27. H. Du, M. Wang, H. Dai, W. Hong, M. Wang, J. Wang, N. Weng, Y. Nie, A. Xu, Endosulfan isomers and sulfate metabolite induced reproductive toxicity in caenorhabditis elegans involves genotoxic response genes. Environ. Sci. Technol. 49, 2460–2468 (2015)
- 28. T. Xu, P. Li, S. Wu, D. Li, J. Wu, K.M. Raley-Susman, D. He, Chronic exposure to perfluorooctane sulfonate reduces lifespan of Caenorhabditis elegans through insulin/IGF-1 signaling. Bull. Environ. Contam. Toxicol. 97, 119–123 (2016)
- A.R. Reindl, L. Falkowska, A. Grajewska, Halogenated organic compounds in the eggs of aquatic birds from the Gulf of Gdansk and Wloclawek Dam (Poland). Chemosphere 237, 124463 (2019)
- 30. R. Chen, Z. Yu, D. Yin, Multi-generational effects of lindane on nematode lipid metabolism with disturbances on insulin-like signal pathway. Chemosphere **210**, 607–614 (2018)

- 31. K.M. Rude, M.M. Pusceddu, C.E. Keogh, J.A. Sladek, G. Rabasa, E.N. Miller, S. Sethi, K.P. Keil, I.N. Pessah, P.J. Lein, M.G. Gareau, Developmental exposure to polychlorinated biphenyls (PCBs) in the maternal diet causes host-microbe defects in weanling offspring mice. Environ. Pollut. 253, 708–721 (2019)
- 32. Q.L. He, X.Y. Wei, X.Y. Han, Q. Zhou, H.Q. Wang, N.Z. Ding, X.Q. Meng, H. Schatten, Q.Y. Sun, Effects of 2,3',4,4'5-pentachlorobiphenyl exposure during pregnancy on epigenetic imprinting and maturation of offspring's oocytes in mice 93, 2575–2592 (2019)
- 33. T. Lin, Y. Chen, W. Chen, Impact of toxicological properties of sulfonamides on the growth of zebrafish embryos in the water. Environ. Toxicol. Pharmacol. 36, 1068–1076 (2013)
- 34. M. Bilal, H.M.N. Iqbal, An insight into toxicity and human-health-related adverse consequences of cosmeceuticals—a review. Sci. Total Environ. **670**, 555–568 (2019)

Chapter 15 Molecular Toxicity Mechanism of Microplastics in the Reservoir



De-Sheng Pei 💿 and Suman Thodhal Yoganandham

Abstract Recently, the toxicity of microplastics (MPs) has attracted global public health concerns. In this chapter, we reviewed the published studies on MPs toxicity in the reservoir to identify knowledge gaps about the harm caused by MPs. Most toxicology studies of MPs have focused on ecotoxicity using apical endpoints, with only a small number of studies addressing molecular toxicity mechanisms. Moreover, MPs in the reservoir are easy to adsorb environmental pollutants, including heavy metals and organic pollutants, implying combined toxicity effects on organisms. Adverse Outcome Pathways (AOPs) framework may integrate published data and identify data gaps in the toxicity mechanisms of MPs. Generally, these findings point to the generation of reactive oxygen species (ROS) as the molecular initiating event (MIE) and then cause developmental and reproductive failure. However, there is not enough data about the association between key events (KEs). Thus, further research is needed to fill the gaps in the toxicity pathways of MPs.

Keywords Microplastics · Reservoir · Oxidative stress · Adverse outcome pathways

1 Introduction

Plastic materials have excellent properties, such as lightweight, robustness, flexibility, and low production costs [1]. However, the wide use and environmental effects of plastic waste have raised worldwide concerns. In 2016, approximately 322 million plastic products were produced globally [2]. Around 10% of plastic waste enters the aquatic environment due to widespread utilization, increased plastic product development, and poor management [3]. Plastic waste can reduce the aesthetic value of the water ecosystem, cause biodiversity loss, and influence public health [4]. Small plastic debris (<5 mm), commonly called microplastics (MPs) may cause a greater and broader ecological risk [5], due to their small size, difficult degradation [6], and easy adsorption of toxic organic and inorganic contaminants [7]. Generally, the polymers of MPs contain polyethylene terephthalate (PET), polystyrene (PS), polypropylene (PP), etc. Notably, in 2021, MPs were first reported in the human placenta [8]. Although the health risks of MPs have aroused great attention, the molecular toxicity mechanism of MPs in the water environment, especially in the reservoir, is still elusive [9]. Therefore, this chapter will clarify the MPs distribution in the reservoir, elucidate toxicity mechanisms caused by MPs, and reveal their combined toxicity and adverse outcome pathways (AOPs) relevance to MPs.

2 MPs Distribution in the Reservoir

MPs contamination in the reservoir is recently raising increasing concerns worldwide. Their distribution in the reservoir was reported in different countries, but lacks a holistic understanding of their occurrence. Guo et al. [10] constructed a global MPs dataset containing 440 collected samples from 43 reservoirs worldwide, which aimed to provide a comprehensive understanding of the drivers and mechanisms of MPs pollution in reservoirs from geographical distribution, driving forces, and ecological risks aspects. They found that small-sized MPs (<1 mm) accounted for more than 60% of the total MPs found in reservoirs worldwide; seasonal variation, geographic location, and land-use type were the main factors affecting MPs abundance.

Liu et al. [11] investigated the horizontal and vertical distribution of MPs in the Guanyingyan Reservoir (China), a dam reservoir, and found that MPs abundance in the horizontal distribution of the reservoir decreased significantly, and the vertical distribution of fibers MPs had less variation in the surface, intermediate, and deep layers, compared to other types of MPs. Moreover, MPs with a size <0.5 mm occupied the majority portion. Interestingly, Lin et al. [12] measured the distribution and source of MPs in the Danjiangkou Reservoir, China's second-largest reservoir. They found that MPs accumulated in the middle layer of the reservoir, and the MPs appearance, such as size and color, varied from the surface to the bottom, implying that surface water sampling cannot determine the MPs contamination for deep-water reservoirs. Shen et al. [13] monitored MPs pollution in the vicinity of ten dams in the Shaving River, a typical multigate dam-type river, by collecting water, sediment, and biological tissues. They found that dam construction altered the suspension, transportation, and deposition of MPs at different dams. Gao et al. [14] investigated MPs pollution at a sandy beach near the outlet of a major reservoir in north Mississippi (USA), and they found that the major form of MPs was fibers (64%), followed by fragments (23%), beads (7%), and films (6%). Further, more in-depth studies should be conducted to provide a comprehensive understanding of the occurrence, drivers, and potential risks of MPs in the reservoirs.

3 Toxicity Mechanisms of MPs

MPs exposure has been shown to inhibit different enzymes and metabolic pathways in invertebrates and vertebrates [15]. Exposure to MPs may disrupt detoxification systems and induce a high expression of genes involved in fundamental physiological processes, such as the arrest of cell cycle growth, oxidative stress, and apoptosis [16].

In our lab, Suman et al. [17] highlighted the molecular toxicity mechanism of MPs exposure using transcriptome analysis. Acute and chronic toxicity in brine shrimp were determined after exposure to polystyrene MPs with various concentrations, and the generation of reactive oxygen species (ROS) was observed. The histopathology analysis revealed the deformation of epithelial cells in the midgut region of brine shrimp. Moreover, the transcriptome analysis was performed after chronic exposure to polystyrene MPs, and the differential expression gene was further confirmed using qRT-PCR. Venn diagram of the transcriptome indicated that 3770 and 5448 genes were differentially expressed in the MPs exposure groups and the controls; 14,930 unigenes were co-expressed both in the treatment and controls (Fig. 15.1a). Compared to the control, 292 and 429 differentially expressed genes (DEGs) were significantly expressed after exposure to polystyrene MPs, indicating their vital role after exposure to PS-MPs (Fig. 15.1b). Enrichment analyses of KEGG and GO revealed the functional clusters and biochemical pathways of those DEGs. According to the biological process, the significantly DEGs were closely involved in the energy derivation by oxidation of organic compounds, polysaccharide biosynthetic process, cellular nitrogen compound metabolic process, glucan biosynthetic process, cellular polysaccharide biosynthetic process, RNA capping, and 7-methylguanosine RNA capping (Fig. 15.2a). Interestingly, 155 pathways were achieved through KEGG enrichment, including viral myocarditis, phagosome, arrhythmogenic right ventricular cardiomyopathy, fluid shear stress, hypertrophic cardiomyopathy, and atherosclerosis. Further, the regulations of actin



Fig. 15.1 The Venn diagram and volcano plot depicting the DEGs profile of brine shrimp treated with polystyrene MPs for 14 days. (a) Venn diagram shows the number of unigenes and changed expression after MPs exposure. (b) Volcano plot describes the DEGs profile after MPs exposure



Fig. 15.2 Gene Ontology (GO) and KEGG enrichment disclose the toxicity pathways after exposure to polystyrene MPs. (**a**) The hierarchy enrichment of GO terms highlights the annotations of biological processes with red boxes. (**b**) KEGG enrichment analysis of DEGs after exposure to polystyrene MPs

cytoskeleton were significantly affected by regulating the activities of ROS and apoptosis (Fig. 15.2b).

Moreover, MPs can disrupt cell surface structures or other extracellular matrix components at moderate levels and inhibit the cell signaling processes of extracellular receptor interactions between the ligand and the cell surface [18]. Besides, MPs may disrupt endocytic activity and activate the cellular innate immune system [19]. In the gastrointestinal tract, MPs were reported to be accumulated and caused inflammatory responses and oxidation [20, 21]. Moreover, MPs may produce reactive oxygen species (ROS), activate antioxidant-related enzymes, and boost glutathione S-transferase (GST) activity and MAPK signaling pathways.

4 Combined Toxicity of MPs with Other Environmental Pollutants

Because of their chemical and physical characteristics, MPs are easy to adsorb environmental pollutants, such as heavy metals [22] and organic pollutants [23], and show combined toxicity effects on organisms. The combined toxicity is related to the properties of adsorbates, particle size, and the composition of plastics [24]. When considering the interaction between particles and organisms, particle size is an important characteristic [25]. Because MPs are small and possess large hydrophobic surface areas, they are prone to serve as vehicles for microorganisms or chemicals. After exposure to these MPs mixed with other toxic substances, combined toxic effects were exhibited in organisms, including synergistic, additive, or antagonistic effects [26].

In our lab, Wang et al. [27] reported the combined toxicity of MPs and three concomitant heavy metals (Cd, Pb, and Zn) in the environment. We revealed the enrichment and distribution of heavy metals and MPs in marine medaka. Moreover, the individual and combined effects on intestinal toxicity and gonadal development were systematically investigated. As far as we know, this is the first report about the individual and combined effects of heavy metals (Cd, Pb, and Zn) and polystyrene MPs on intestinal toxicity and gonadal development of marine medaka.

MPs can adsorb polyfluoroalkyl substances (PFASs), polycyclic aromatic hydrocarbons (PAHs), organochlorine pesticides (OCPs), polychlorinated biphenyls (PCBs), antibiotics, and other organic pollutants [28]. The adsorption types mainly include surface adsorption, pore filling, and distribution [29]. Low-density polyethylene (LDPE) MPs in San Diego Bay were found to adsorb PAHs, PCBs, and polybrominated diphenyl ethers (PBDEs), and these compounds could be biomagnified in medaka (*Oryzias latipes*) and caused liver injuries, such as glycogen depletion, fat vacuolization, and cell necrosis [30]. Another study investigated the combined toxic effects of 5 μ m PS-MPs and F-53B. After their exposure in zebrafish, the bioavailability and bioaccumulation of F-53B were reduced due to the strong adsorption capacity of PS-MPs to F-53B. However, co-exposure to PS-MPs and F-53B induced severe oxidative stress and inflammation in zebrafish [31].

Environmental monitoring results showed that plastic chips can enrich metals in the ambient environment [32]. The concentrations of Al, Fe, Mn, Cu, Pb, and Zn reached 180 μ g·g⁻¹ in the plastic particles collected on the beach, and the concentrations of Cd, Cr, Co, and Ni reached 0.92 ng·g-1, which are close to or higher than those in the surrounding environment [33]. PE plastic microspheres (2–4 μ m) were reported to enrich Cd in seawater and caused serious reproductive toxicity to *Moina mongolica*, because the adsorbed Cd on PE plastic microspheres was released in the acidic digestive or gastrointestinal tract and subsequently transported to other tissues. Thus, stronger combined toxicity was generated than that of PE microspheres alone [34]. The surface area enhances as the particle size decreases, implying that micro-sized particles may achieve stronger adsorption of pollutants [24]. MPs in the environment have a coarser structure with a bigger surface area and possess stronger sorption abilities [25]. The combined toxicity of MPs is highly dependent on the plastic and absorbed compound types, the number of particles ingested, the release rate of contaminants, *etc*.

Except for the combined toxicity of plastics with other environmental pollutants, the toxicity of plasticizers cannot be ignored for their wide use to increase plasticity in the process of plastic synthesis [29, 35]. Different plasticizers are widely used globally and often exhibit ecological toxicity as environmental hormones [35, 36]. Organisms may be directly exposed to leached additives after MPs are ingested [29]. Such additives and monomers may interfere with the signaling pathways of estrogen and testosterone synthesis, resulting in endocrine disruption [37]. The most common plasticizers are phthalate esters (PAEs), which are primarily used in the manufacturing of PVC products, such as upholstery, shower curtains, flooring, and food containers [38]. PAEs as plasticizers are not covalently bonded to the polymer matrix and easily leach from plastics, especially at a high temperature and low pH [35]. Exposure to PAEs activates the CYP450 detoxification system and disturbs endocrine system, resulting in metabolic disorders, oxidative stress, endocrine disorders, and immunodepression [39].

5 AOPs Relevance to MPs Toxicity

AOPs may recognize the specific toxicity pathways and biological key events (KEs) occurring at different levels of the organism with adverse outcomes. Thus, the AOPs may indicate biological processes happening at several levels of the organization from the molecular to the cellular, organism, and tissue [40], which can be used to pinpoint potential threats and outline the chain of KEs that led to the risk effects. As a result, they explain how stressors might start the crucial sequence of KEs needed for the commencement of an adverse outcome. The connections of KEs describe the quantitative connection between two consecutive KEs to trigger the next KEs. The toxicological threshold, temporal dynamics, and dose–response connections are all described by KEs relationships [41, 42]. Individual KEs may be used to determine whether experimental endpoints and assays are the most important for measuring MPs toxicity.

AOPs relevance to MPs toxicity has not yet been fully known. Physical, structural (size and form), and chemical characteristics, such as the chemical composition of MPs and adsorbed substances on their surface, may contribute to MPs' toxicity. MPs can initiate biocorona and cell contact by interacting with lipids, proteins, and other small molecules in the cell membrane and/or biomolecules in the surrounding media. Chemical, physical, receptor-mediated, and mechanical interactions are all possible with the cellular microenvironment. When it comes to MPs, interactions between cells can be established by biocorona components to define the eventual AOPs [43]. Pattern recognition receptors play a crucial role in interactions by identifying agonists of the innate immune response. ROS risks can be amplified by photochemical weathering when the plastic is exposed to UV light [44]. Proinflammatory responses may be initiated by ROS as a result of an oxidative stress response.

Oxidative stress has been reported in *Artemia salina* after exposure to polystyrene MPs [17], however, there is no clear evidence to suggest that oxidative stress occurs in humans after exposure to plastic dust. Oxidative stress and inflammation are proposed to be the primary mechanisms of MPs-induced toxicity [19]. Although AOPs induced by MPs may not elucidate MPs' physical, chemical, and structural identity, some existing AOPs can certainly contribute to the prioritization of toxicity endpoints and assays for targeted investigation.

6 Conclusion

This chapter summarizes the reported effect of MPs and molecular toxicity mechanisms in the reservoir. According to the published literature, MPs are a threat to aquatic biota. The formation of ROS that has recently been identified as the molecular initiating event *via* AOPs, which is a more general phenomenon of the cellular stress response. However, few studies have been conducted on the toxicological mechanisms of MPs and the results are not sufficiently conclusive. To fill these gaps, more detailed toxicology research on MPs is required.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- 1. Z. Zhang, H. Peng, D. Yang, G. Zhang, J. Zhang, F. Ju, Polyvinyl chloride degradation by a bacterium isolated from the gut of insect larvae. Nat. Commun. **13**, 5360 (2022)
- O. Dogu, M. Pelucchi, R. Van de Vijver, P.H.M. Van Steenberge, D.R. D'Hooge, A. Cuoci, M. Mehl, A. Frassoldati, T. Faravelli, K.M. Van Geem, The chemistry of chemical recycling of solid plastic waste via pyrolysis and gasification: State-of-the-art, challenges, and future directions. Prog. Energy Combust. Sci. 84, 100901 (2021)
- D.K. Barnes, F. Galgani, R.C. Thompson, M. Barlaz, Accumulation and fragmentation of plastic debris in global environments. Philos. Trans. R. Soc. Lond. Ser. B. Biol. Sci. 364, 1985–1998 (2009)
- S.-A. Strungaru, R. Jijie, M. Nicoara, G. Plavan, C. Faggio, Micro-(nano) plastics in freshwater ecosystems: abundance, toxicological impact and quantification methodology. TrAC Trends Anal. Chem. 110, 116–128 (2019)
- D. Eerkes-Medrano, R.C. Thompson, D.C. Aldridge, Microplastics in freshwater systems: a review of the emerging threats, identification of knowledge gaps and prioritisation of research needs. Water Res. **75**, 63–82 (2015)
- 6. Z. Wang, Y. Zhang, S. Kang, L. Yang, X. Luo, P. Chen, J. Guo, Z. Hu, C. Yang, Z. Yang, T. Gao, Long-range transport of atmospheric microplastics deposited onto glacier in southeast Tibetan Plateau. Environ. Pollut. **306**, 119415 (2022)

- M. Arias-Andres, K. Rojas-Jimenez, H.-P. Grossart, Collateral effects of microplastic pollution on aquatic microorganisms: an ecological perspective. TrAC Trends Anal. Chem. 112, 234–240 (2019)
- A. Ragusa, A. Svelato, C. Santacroce, P. Catalano, V. Notarstefano, O. Carnevali, F. Papa, M.C.A. Rongioletti, F. Baiocco, S. Draghi, E. D'Amore, D. Rinaldo, M. Matta, E. Giorgini, Plasticenta: first evidence of microplastics in human placenta. Environ. Int. 146, 106274 (2021)
- Y.Y. Xu, F.K.S. Chan, J. He, M. Johnson, C. Gibbins, P. Kay, T. Stanton, Y.Y. Xu, G. Li, M.L. Feng, O. Paramor, X.B. Yu, Y.G. Zhu, A critical review of microplastic pollution in urban freshwater environments and legislative progress in China: recommendations and insights. Crit. Rev. Environ. Sci. Technol. 51, 2637–2680 (2021)
- Z. Guo, W.J. Boeing, Y. Xu, E. Borgomeo, S.A. Mason, Y.-G. Zhu, Global meta-analysis of microplastic contamination in reservoirs with a novel framework. Water Res. 207, 117828 (2021)
- Y. Liu, W. Cao, Y. Hu, J. Zhang, W. Shen, Horizontal and vertical distribution of microplastics in dam reservoir after impoundment. Sci. Total Environ. 832, 154962 (2022)
- L. Lin, X. Pan, S. Zhang, D. Li, W. Zhai, Z. Wang, J. Tao, C. Mi, Q. Li, J.C. Crittenden, Distribution and source of microplastics in China's second largest reservoir—Danjiangkou Reservoir. J. Environ. Sci. **102**, 74–84 (2021)
- 13. J. Shen, X. Gu, R. Liu, H. Feng, D. Li, Y. Liu, X. Jiang, G. Qin, S. An, N. Li, X. Leng, Damming has changed the migration process of microplastics and increased the pollution risk in the reservoirs in the Shaying River Basin. J. Hazard. Mater. **130067** (2022)
- 14. Z. Gao, K. Wontor, J.V. Cizdziel, H. Lu, Distribution and characteristics of microplastics in beach sand near the outlet of a major reservoir in north Mississippi, USA. Microplast. Nanoplast. **2**, 10 (2022)
- C.E. Enyoh, L. Shafea, A.W. Verla, E.N. Verla, W. Qingyue, T. Chowdhury, M. Paredes, Microplastics exposure routes and toxicity studies to ecosystems: an overview. Environ. Anal. Health Toxicol. 35, e2020004 (2020)
- M. Hu, D. Palić, Micro- and nano-plastics activation of oxidative and inflammatory adverse outcome pathways. Redox Biol. 37, 101620 (2020)
- T.Y. Suman, P.-P. Jia, W.-G. Li, M. Junaid, G.-Y. Xin, Y. Wang, D.-S. Pei, Acute and chronic effects of polystyrene microplastics on brine shrimp: first evidence highlighting the molecular mechanism through transcriptome analysis. J. Hazard. Mater. 400, 123220 (2020)
- J.B. Fleury, V.A. Baulin, Microplastics destabilize lipid membranes by mechanical stretching. Proc. Natl. Acad. Sci. USA 118 (2021)
- 19. C.Q.Y. Yong, S. Valiyaveettil, B.L. Tang, Toxicity of microplastics and nanoplastics in mammalian systems. Int. J. Environ. Res. Public Health 17 (2020)
- A. Tamargo, N. Molinero, J.J. Reinosa, V. Alcolea-Rodriguez, R. Portela, M.A. Bañares, J.F. Fernández, M.V. Moreno-Arribas, PET microplastics affect human gut microbiota communities during simulated gastrointestinal digestion, first evidence of plausible polymer biodegradation during human digestion. Sci. Rep. 12, 528 (2022)
- 21. G. Fackelmann, S. Sommer, Microplastics and the gut microbiome: how chronically exposed species may suffer from gut dysbiosis. Mar. Pollut. Bull. **143**, 193–203 (2019)
- B. Wen, S.R. Jin, Z.Z. Chen, J.Z. Gao, Y.N. Liu, J.H. Liu, X.S. Feng, Single and combined effects of microplastics and cadmium on the cadmium accumulation, antioxidant defence and innate immunity of the discus fish (Symphysodon aequifasciatus). Environ. Pollut. 243, 462–471 (2018)
- 23. Y. Ma, A. Huang, S. Cao, F. Sun, L. Wang, H. Guo, R. Ji, Effects of nanoplastics and microplastics on toxicity, bioaccumulation, and environmental fate of phenanthrene in fresh water. Environ. Pollut. 219, 166–173 (2016)
- S. Lambert, C. Scherer, M. Wagner, Ecotoxicity testing of microplastics: considering the heterogeneity of physicochemical properties. Integr. Environ. Assess. Manag. 13, 470–475 (2017)
- X. Sun, B. Chen, Q. Li, N. Liu, B. Xia, L. Zhu, K. Qu, Toxicities of polystyrene nano- and microplastics toward marine bacterium Halomonas alkaliphila. Sci. Total Environ. 642, 1378–1385 (2018)

- 26. M.O. Rodrigues, N. Abrantes, F.J.M. Gonçalves, H. Nogueira, J.C. Marques, A.M.M. Gonçalves, Impacts of plastic products used in daily life on the environment and human health: what is known? Environ. Toxicol. Pharmacol. 72, 103239 (2019)
- W. Yan, N. Hamid, S. Deng, P.-P. Jia, D.-S. Pei, Individual and combined toxicogenetic effects of microplastics and heavy metals (Cd, Pb, and Zn) perturb gut microbiota homeostasis and gonadal development in marine medaka (Oryzias melastigma). J. Hazard. Mater. **397**, 122795 (2020)
- A. Bakir, S.J. Rowland, R.C. Thompson, Enhanced desorption of persistent organic pollutants from microplastics under simulated physiological conditions. Environ. Pollut. 185, 16–23 (2014)
- P.S. Kumar, Chapter eight—Microplastics and its removal strategies from marine water, in Modern Treatment Strategies for Marine Pollution, ed. by P.S. Kumar, (Elsevier, 2021), pp. 125–144
- C.M. Rochman, E. Hoh, T. Kurobe, S.J. Teh, Ingested plastic transfers hazardous chemicals to fish and induces hepatic stress. Sci. Rep. 3, 3263 (2013)
- H. Yang, H. Lai, J. Huang, L. Sun, J.A. Mennigen, Q. Wang, Y. Liu, Y. Jin, W. Tu, Polystyrene microplastics decrease F-53B bioaccumulation but induce inflammatory stress in larval zebrafish. Chemosphere 255, 127040 (2020)
- 32. J. Wang, J. Peng, Z. Tan, Y. Gao, Z. Zhan, Q. Chen, L. Cai, Microplastics in the surface sediments from the Beijiang River littoral zone: composition, abundance, surface textures and interaction with heavy metals. Chemosphere 171, 248–258 (2017)
- 33. L.A. Holmes, A. Turner, R.C. Thompson, Adsorption of trace metals to plastic resin pellets in the marine environment. Environ. Pollut. **160**, 42–48 (2012)
- 34. Z. Wang, H. Dong, Y. Wang, R. Ren, X. Qin, S. Wang, Effects of microplastics and their adsorption of cadmium as vectors on the cladoceran Moina monogolica Daday: implications for plastic-ingesting organisms. J. Hazard. Mater. 400, 123239 (2020)
- A.M. Roark, Endocrine disruptors and marine systems, in *Encyclopedia of the World's Biomes*, ed. by M.I. Goldstein, D.A. DellaSala, (Elsevier, Oxford, 2020), pp. 188–194
- M. Biron, 10—Transition of plastics to renewable feedstock and raw materials: bioplastics and additives derived from natural resources, in *A Practical Guide to Plastics Sustainability*, ed. by M. Biron, (William Andrew Publishing, 2020), pp. 469–555
- M. Cole, P. Lindeque, C. Halsband, T.S. Galloway, Microplastics as contaminants in the marine environment: a review. Mar. Pollut. Bull. 62, 2588–2597 (2011)
- E. Białecka-Florjańczyk, Z. Florjańczyk, Chapter 22 Solubility of plasticizers, polymers and environmental pollution, in *Thermodynamics, Solubility and Environmental Issues*, ed. by T.M. Letcher, (Elsevier, Amsterdam, 2007), pp. 397–408
- 39. Y. Zhang, Y. Jiao, Z. Li, Y. Tao, Y. Yang, Hazards of phthalates (PAEs) exposure: a review of aquatic animal toxicology studies. Sci. Total Environ. **771**, 145418 (2021)
- 40. G.T. Ankley, R.S. Bennett, R.J. Erickson, D.J. Hoff, M.W. Hornung, R.D. Johnson, D.R. Mount, J.W. Nichols, C.L. Russom, P.K. Schmieder, J.A. Serrrano, J.E. Tietge, D.L. Villeneuve, Adverse outcome pathways: a conceptual framework to support ecotoxicology research and risk assessment. Environ. Toxicol. Chem. 29, 730–741 (2010)
- 41. D.L. Villeneuve, D. Crump, N. Garcia-Reyero, M. Hecker, T.H. Hutchinson, C.A. LaLone, B. Landesmann, T. Lettieri, S. Munn, M. Nepelska, M.A. Ottinger, L. Vergauwen, M. Whelan, Adverse outcome pathway (AOP) development I: strategies and principles. Toxicol. Sci. 142, 312–320 (2014)
- D.L. Villeneuve, D. Crump, N. Garcia-Reyero, M. Hecker, T.H. Hutchinson, C.A. LaLone, B. Landesmann, T. Lettieri, S. Munn, M. Nepelska, M.A. Ottinger, L. Vergauwen, M. Whelan, Adverse outcome pathway development II: best practices. Toxicol. Sci. 142, 321–330 (2014)
- 43. B. Fadeel, The right stuff: on the future of nanotoxicology. Front. Toxicol. 1, 1 (2019)
- 44. A. Rahman, A. Sarkar, O.P. Yadav, G. Achari, J. Slobodnik, Potential human health risks due to environmental exposure to nano- and microplastics and knowledge gaps: a scoping review. Sci. Total Environ. 757, 143872 (2021)

Chapter 16 Molecular Toxicity Mechanism of Plasticizers in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract Plasticizers, due to their robustness, flexibility, and low production costs, have broad industrial applications. However, they are released into the reservoir environment because of inappropriate disposal discharge and cause adverse health effects. This chapter mainly focuses on the molecular mechanism of the plasticizers, particularly phthalates (PAEs) and bisphenol A (BPA), when they are exposed to or interacted with organisms. Previously, various studies have reported that plasticizers affect growth development by altering the thyroid and estrogen axis, leading to infertility. Furthermore, these chemicals can disturb reproductive capacity and decrease egg production *via* reducing steroidogenesis, activating peroxisome proliferator receptors (PPAR), and enhancing oxidative stress levels. In a nutshell, plasticizer pollution in the reservoir should be deeply concerned, and more studies on the adverse health effects of plasticizers on aquatic species are critically needed.

Keywords Peroxisome proliferator-activated receptors (PPAR) pathway · Endocrine disruption · Reproductive toxicity · Phthalates

1 Introduction

Plasticizers are chemicals including flame retardants and surfactants that are combined with polymers to increase flexibility in industrial products. They are not permanently covalently bound to the plastics. Therefore, they can slowly diffuse into the reservoir environment and cause environmental pollution. Plasticizers are ubiquitous existence, and their metabolites were found in the air, soil, water, and biota [1–3]. Bisphenol A (BPA) and phthalates (PAEs) are the most essential plasticizers due to their inducing endocrine disruption abilities, particularly in mammals [4]. Presently, more than 20 types of PAEs have been found in the reservoir environment, of which six di-methyl phthalate (DMP), di-ethyl phthalate (DEP), di-butyl phthalate (DBP), benzyl-butyl phthalate (BBP), di-(2-ethylhexyl) phthalate (DEHP), and di-n-octyl phthalate (DNOP) are considered a priority chemical classified according to the United States Environmental Protection Agency (US EPA) [5]. Despite their restriction, their worldwide usage has rapidly escalated with a rate of 1.8 million tons to 8 million tons from the years 1975–2011 [6]. In Europe, about 1,150,000 metric tons of high-volume plastic monomer were produced in 2005–2006 [7]. Structurally, PAEs have a central ring with two esters, which vary with molecular weight. Longer chains have high molecular weight and possess a higher residence rate in the reservoir environment [4].

From 2010 to 2015, PAEs usage increased at an annual rate of 7.70% [8, 9]. Among all PAEs, only DEHP holds 80% of production in China [10]. Considering the high percentage prevalence in the reservoir environment, various species are exposed to and interacted with PAEs mixtures [10, 11]. In terms of toxicity, DEHP is the most widely studied and considered a toxic chemical due to its significant reproductive effects [12]. From the published studies, it was found that DBP, DEHP, and DNOP can disturb developmental growth, increase reactive oxygen species (ROS), and elevate oxidative stress that can cause oocyte apoptosis [12]. Moreover, it may also disrupt the peroxisome proliferator-activated receptors (PPARs) functions, perturb follicle production, and suppress vitellogenin (VTG) protein and estrogen receptor (ER) genes, which are considered the biomarkers of endocrine disruption [9, 13].

2 DEHP Mechanistic Toxicity in the Three Gorges Reservoir Area (TGRA)

Due to DEHP being known as the most toxic chemical congener of PAEs, a recently published study investigated the in-depth toxicity of DEHP exposed at environment-relevant concentrations (ERC) in the TGRA [14]. This study using cell lines and zebrafish as the model species elucidated the underlying mechanisms for DEHP toxicity and its associated ecological risks, which is the holistic approach for integrated toxicological assessment. In the TGRA, the levels of DEHP were considerably higher than that previously prescribed by the US EPA [14]. However, the general trend revealed a decrease in the DEHP levels from the upper, middle, and lower reaches of the TGRA. Further, *in vitro* toxicity of DEHP was determined by using a cell line. At the exposure levels of 100–800 μ g/L, a significant decrease in cell viability was observed [14]. Short-term exposure to 400 μ g/L DEHP in zebrafish embryos can activate the PI3K-AKT-mTOR pathway [14]. Moreover, a long-term (3 months) exposure to 10–33 μ g/L DEHP levels in the TGRA showed an enhanced potential to induce reproductive toxicity *in vivo* and DNA damage *in vitro* [14].

3 Plasticizers' Toxic Effect on Peroxisome Proliferator-Activated Receptors (PPARs)

PAEs can induce the transcriptional changes of PPAR and cause reproductive toxicity. PPARs are nuclear receptor proteins, which can bind to particular DNA sequences and affect the transcription levels of DNA [12]. They are categorized into three types: PPAR α participates in fatty acid degradation; PPAR β and PPAR γ control fatty acid metabolism and glucose levels, respectively [15]. It is confirmed that PAEs and BPA can disrupt the PPARs levels in mammals [16]. Similarly, a study was reported in mouse fibroblasts and observed the elevated expression levels of ppary2 after exposure to BPA (18 mg/L) [17]. Similarly, Deng et al. [18] found that $ppar\alpha$ and $ppar\beta$ transcription levels were also increased when rats were exposed to DEHP [18]. Furthermore, different structural phthalates have different metabolism abilities that may alter the PPARs and peroxisome proliferation [19]. In mice, an *in vitro* study investigated the toxic effects of BPA and found that two days of exposure elevated the PPAR γ 2 expression level [17]. Moreover, PPAR α and PPAR β transcript levels were also enhanced after DEHP exposure in rats [7]. PPAR protein levels were also altered in rats after exposure to monoester phthalates [20]. PAEs binding with PPARs can interact with retinoid X receptor (RXR), induce the change of hormones, and participate in the carbohydrate and lipid metabolism [17]. Notably, PAEs can alter the normal functioning of PPARs, and long-chain PAEs were more significantly bound than single-chain PAEs [19]. The detailed toxicity pathway of PAEs affected PPARs function and induce reproductive effects is presented in Fig. 16.1.

4 The Impairment of Thyroid and Estrogen Axes by Phthalates and BPA

If the hormonal axes are disturbed by plasticizers, the hypothalamic–pituitary– gonadal (HPG) axis and thyroid axis balance are also affected [21]. They are responsible to regulate metabolic, developmental, and reproductive functions [22]. Thyrotropin-releasing hormone (TRH) is released first from the hypothalamus and then converts into triiodothyronine (T₃) and thyroxine (T₄). Furthermore, these hormones are responsible for regulating the level of thyroid-stimulating hormone (TSH), which is released from the pituitary gland [23]. TSH mainly involves the synthesis of T₄ in the thyroid gland, converts them to T₃, and then further degrades it into T₂ [24]. The physiological effects of thyroid hormones mainly including T₃ and T₄ are mediated *via* binding to thyroid nuclear receptors α , β , and γ [16].



Fig. 16.1 The toxicity mechanism of PAEs (P) at different levels. 1—PAEs bind with thyroid hormone receptor (TR); 2—the formation of retinoid X receptor (RXR) complex; 3—PPARs gene and protein expression level is upregulated, and PAEs activate PPARs; 4—Fatty acid oxidation leads to the formation of ROS; 5—PAEs reduce the transport of fatty acids, decrease the testoster-one levels (6), and reduce the estradiol levels (7), which will have deleterious effects at the organ, individual, and population levels

4.1 Phthalates and BPA Effects on Thyroid Axis

In previous studies, the effects of phthalates and BPA on thyroid hormone (TH) have been reported in various aquatic species and mammals. Among the PAEs, DEHP is the potent inhibitor that specifically binds to the TR with the string ligand binding energy values [25]. A study conducted on African frogs (tadpoles) and treated with DBP at a low ERC of 2 mg/L showed an altered expression level of four TH-related genes [26]. Gayathri et al. [27] discussed that PAEs caused thyroid gland hyperactivity [27], whereas Wenzel et al. [28] suggested that PAEs amplified the iodide uptake and elevated the iodide symporter activity [28]. However, it is unknown whether PAEs can either modify or alter the TH levels in aquatic species. In amphibians, BPA alters the T₃ function resulting in shortened interocular distances in African tadpoles [29]. Furthermore, exposure to BPA perturbs the hatching ratio in zebrafish [30] because of the upregulated expression levels of the vitellogenin (Vtg) gene and the suppressed growth of somatic cells to vitellogenesis.

Studies have evidenced that PAEs and BPA can disturb the normal transcriptional levels of the genes involved in cholesterol transport and steroidogenesis [16, 31]. In steroidogenesis, the first step includes the conversion of cholesterol into pregnenolone with the help of an enzyme cytochrome P450 (CYP11A1) [11]. When DBP, DEHP, and DMP were exposed to rats, the gene expression level of CYP11A1 has been reduced significantly [32]. In addition, progestogens are changed into

androgens *via* CYP17A1, which are responsible for the mediation of both 17a-hydroxylase and 17, 20-lyase activities [29].

4.2 Phthalates and BPA Effects on Estrogen Axis

Many studies have been conducted and the results indicated that the estrogenic activity was altered after exposure to PAEs and BPA congeners in mammals, amphibians, fish, and human cell lines [23]. PAEs can easily bind with the estrogen receptor (ER) of humans, rats, rainbow trout, and zebrafish [7, 33]. When ERs bind to PAEs, they disturb the Vtg expression levels, particularly when exposed to DEHP in zebrafish, exposure to BBP in rainbow trout, and exposure to DBP in marine medaka [16, 34, 35]. PAEs exposure also affected the reproduction ability of female fish and rats. Previously, Xu et al. [31] reported that chronic exposure to DEHP disturbs the estrous cycle in rats [31]. Moreover, in aquatic species, egg ovulation has been delayed after exposure to DEHP. For example, 40 mg/L of DEHP exposure to zebrafish for three weeks impaired zebrafish reproduction, particularly the oogenesis. They further demonstrated that lower egg production might be due to the downregulated expression levels of ptgs2, which is considered an essential enzyme for zebrafish ovulation [36]. Similarly, like PAEs, BPA also possesses estrogenic effects, particularly on ER α and Er β . For example, exposure to a low concentration of BPA (0.23 mg/L) upregulated the ER α expression level in amphibians [37]. Besides, BPA may reduce the production of luteinizing hormone (LH), which could be the reason for disrupting the estrous cycle [25].

5 Combined Reproductive Effects of Plasticizers

Plasticizer mixtures also exhibited augmented high estrogenic activity, compared to individual chemical toxicity [38]. For the evaluation of plasticizer toxic mixture effect in animal models, previous studies on fish were enormously published, especially screening the combined toxicity of chemicals on various organs in zebrafish (*Danio rerio*). Other commonly used animal models encompassed mice (*Mus musculus*), *Caenorhabditis elegans*, *Carassius auratus*, *Daphnia magna*, and marine medaka (*Oryzias melastigma*) [39–41]. In contrast, *in vitro* models are commonly used to determine mixture toxicity, including *Vibro qinghaiensis sp-Q67*, *Vibrio fischeri*, *Photobacterium phosphoreum*, *Microtox*, *Folsomia fimetaria*, *Scenedesmus vacuolatus*, *Lemna minor*, *Escherichia coli*, *Bacillus subtilis*, *Photobacterium leiognathi*, *Anabaena CPB4337*, *Aliivibrio fischeri*, and H295R cells. Xu et al. [42] conducted a study to elucidate the joint toxic effects of the binary mixtures of DBP + EE2 on developmental growth, reproduction, and histological alterations using male zebrafish [42]. It was found that Vtg and AOX levels were significantly decreased in all binary mixtures, compared to individual

EE2 and DBP exposed groups [42]. It can be concluded that EE2 and DBP may act synergistically to enhance endocrine-disrupting effects in male zebrafish. Similar results were observed from the exposure to the binary combinations of antiinflammatory antibiotics (CBZ, PHO, SMX, and TMP), where male zebrafish showed high Vtg gene expression levels with an increase in hepatic mass [43]. Likewise, CBZ, DIC, EE2, and MET binary mixtures of antibiotics also induced reproductive abnormalities in progenies [44]. Dong et al. (2018) investigated the combined effects of DEHP+DEP on fetal Leydig cells in male rats [45]. Nevertheless, DBP + DEHP cocktails have shown synergistic and dose-additive effects to reduce sperm count, epididymis, and liver weight in male rats [46]. Besides, cystic ovaries disrupted the estrous cycle and reduced fertility when female rats were exposed to the quinary mixtures of phthalates (DEP, DEHP, DBP, DNOP, DIBP, and BBP) [47]. Furthermore, phthalates combinations with parabens also altered or modified the enzymatic activities for reproduction (estrogen metabolism) [48]. Antioxidant enzymatic actions, such as SOD, CAT, and GPx, were elevated in binary mixtures of phthalates (DBP + DEP) [46, 49]. The combined mixture effects of PAEs with Glycerin monostearate (GMS) resulted in a reduction in testosterone (T2) level; moreover, metabolomics and western blot analysis also confirmed the down-regulation of the expression of steroidogenic proteins in male rats (Table 16.1) [52]. Further, DEP + DEHP binary mixture effects were examined using *Caenorhabditis elegans*, and found the unregulated expression of genes linked with lipid metabolism and reproduction [41]. Some stress response genes were also upregulated, while the heat shock proteins were significantly downregulated [41]. In conclusion, alterations in the transcription levels or histological and antioxidant levels were more sensitive to mixture toxicities (Table 16.1).

6 Conclusion

When plasticizers are released into the reservoir, they remain persistent and interact with various aquatic organisms. Plasticizers mainly include phthalates (PAEs) and Bisphenol A (BPA). When exposed to aquatic species, plasticizers can disrupt the thyroid axis, estrogen axis, and various developmental and growth effects. Notably, plasticizers may reduce the cholesterol transported to mitochondria and decrease the cholesterol levels, which leads to disturbing the steroidogenesis and thyroid axis. The reproductive impact on both sexes (male and female) is predominant in the production of lower egg and sperm, which decreases the gene expression responsible for maintaining the estrogen or thyroid cycle. Taken together, the release of plasticizers in the reservoir should be strictly banned, so that these adverse effects can be avoided and ecosystem sustainability can be ensured.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

	Reference	[39]	[40]	[41]	[50]	[51]	[52]	(continued)
ially relevant concentrations (ERC)	Endpoint/mixture effects	Antagonistic effects using the inte- grated biomarker response index, alterations in antioxidant status in fish liver	Effect oxidative stress state in the testis, decrease the SOD and GSH-Px activities, increase CAT activity, testes damaged due to pro-duction of ROS	fasn-1, pod-2, fat-5, acs-6 and sbp-1, vtg. Upregulated. Stress response genes, ced-1 wah-1, daf-21, gst-4 upregulated; ctl-1, cdf-2, hsp-16.1, hsp-16.48, sip-1 were downregulated	Gestational (GD12–20) exposure of DEP + DEHP resulted in synergistic and/or dose-additive effects on the development of fetal Leydig cell	Fourteen lysoPCs, PC (18:4/18:1), lysoPE (18:2/0:0), phenylalanine, and tryptophan were identified as potential biomarkers for the DEHP +PCBs. RNA expressions indicate hepatic lipid accumulation	Steroidogenic proteins (StAR, P450scc, CYP17A1, 17β-HSD, and P450arom) indicated that MIXPs	
says at environmen	Exposure regime	7, 21 days	Time (9:30–11: 30 a.m.) for 4, 8, and 12 weeks	48 h	Postnatal day 1	12 days	15 weeks	
toxicity using in vivo as	Observation	Antioxidant response, SOD, CAT, MDA, GSH	SOD, GSH-Px, CAT, and MDA	Gene expression, lipid staining, fecun- dity assay	Combined effect on fetal Leydig cell	Metabolic responses	Reproductive toxicity	
zers on reproductive	Species exposure	Carassius auratus	Adult male rats	Caenorhabditis elegans	Male rats	Adult rats	Adult rats	
dixture effects of the plasticiz	Mixture	DBP, DEHP, Cu	DBP, BaP	DEHP, DEP	DEHP, DEP	DEHP, Aroclor 1254	DEHP, DEP, DMP, BBP, DEP, DNOP, glycerin monostearate (GMS)	
Table 16.1 N	Compound name	Phthalates, metal	Phthalates, PAH	Phthalates	Phthalates	Phthalates, PCBs	Phthalates, GMS	

16 Molecular Toxicity Mechanism of Plasticizers in the Reservoir

189

Table 16.1 (continued)					
Compound name	Mixture	Species exposure	Observation	Exposure regime	Endpoint/mixture effects	Reference
					exposure downregulated the expression of steroidogenic proteins and altered androgen metabolism.	
Phthalates, EE2	DBP, 17- α-ethynylestradiol (EE2)	Zebrafish (Danio rerio)	VTG induction, AOX protein level	21 days	Decreased VTG levels as compared to EE2-only groups; EE2 and DBP may act additively on VTG and antagonistically on AOX induction in males	[42]
Synthetic estrogen EE2, ZM	Ethynylestradiol (EE2) Antiestrogen ZM	Pimephales promelas gonads	Individual treatment- related expression increase/decrease	2 days, semi- static	Mixture exposure provides a dis- tinction between different modes of estrogenic action	[53]
Plasticizers	DEHP, DBP, ATBC	Zebrafish (Danio rerio)	Reproductive toxicity	3 months	Decrease in GSI, impairment of fecundity, disruption of oogenesis, spermatogenesis. Male zebrafish were more sensitive to the com- bined exposure	[6]
Phthalate, parabens	PP, DEHP, BP, triclosan, TBBPA	Adult rats	Influences of multiple EDCs, using ERCs	1 h	Results showed that EDCs interact <i>in vivo</i> , magnifying one another's effects, consistent with inhibition of enzymes that are critical for estro- gen metabolism	[48]
Antibiotics	CBZ, F, PHO, SMX, TMP,	Zebrafish (Danio rerio)	liver Cytohistological parameters, in zebrafish	21 days	Males zebrafish showed an increase of Vtg immunostaining, increased hepatic mass	[43]
Phthalates	BBP, DBP, DEHP, DIDP, DINP, DNOP	Oryzias melastigma, Danio rerio	Developmental toxic- ity and EDR estro- genic of phthalates	72 h	The mixture of the six phthalates exhibited enhanced estrogenic activity	[38]

190

[54]	[49]
Antagonism effects were found in the joint toxicity of Cu(II) combined with DBP or DEHP using the toxic unit method	Enhance the production of ROS and lipid peroxidation (LPO). The activity of antioxidant enzymes including SOD, CAT, and GPx was increased.
14 h	12 h
Combined effects of Cu and PAEs	Oxidative stress, innate immune response
Daphnia magna, Photobacterium phosphoreum	Zebrafish embryos (Danio rerio)
DBP, DEHP, Cu	DBP, DEP
Phthalates, metal	Phthalates

Conflicts of Interest The authors declare no conflict of interest.

References

- G. Yang, C. Chen, Y. Wang, Q. Peng, H. Zhao, D. Guo, Q. Wang, Y. Qian, Mixture toxicity of four commonly used pesticides at different effect levels to the epigeic earthworm, Eisenia fetida. Ecotoxicol. Environ. Saf. 142, 29–39 (2017)
- J. Zhou, Z.-H. Cai, K.-Z. Xing, Potential mechanisms of phthalate ester embryotoxicity in the abalone Haliotis diversicolor supertexta. Environ. Pollut. 159, 1114–1122 (2011)
- Y. Liu, Z. Chen, J. Shen, Occurrence and removal characteristics of phthalate esters from typical water sources in Northeast China. J. Anal. Methods Chem. 2013, 419349–419349 (2013)
- 4. J. Mathieu-Denoncourt, S.J. Wallace, S.R. de Solla, V.S. Langlois, Plasticizer endocrine disruption: highlighting developmental and reproductive effects in mammals and non-mammalian aquatic species. Gen. Comp. Endocrinol. 219, 74–88 (2015)
- 5. WHO, 3rd ed. Guidelines for Drinking-water Quality (Geneva), (World Health Organization), 1 (2004)
- S. Net, S. Rabodonirina, R.B. Sghaier, D. Dumoulin, C. Chbib, I. Tlili, B. Ouddane, Distribution of phthalates, pesticides and drug residues in the dissolved, particulate and sedimentary phases from transboundary rivers (France–Belgium). Sci. Total Environ. **521–522**, 152–159 (2015)
- A. Ohashi, H. Kotera, H. Hori, M. Hibiya, K. Watanabe, K. Murakami, M. Hasegawa, M. Tomita, Y. Hiki, S. Sugiyama, Evaluation of endocrine disrupting activity of plasticizers in polyvinyl chloride tubes by estrogen receptor alpha binding assay. J. Artif. Organs 8, 252 (2005)
- H. Liu, K. Cui, F. Zeng, L. Chen, Y. Cheng, H. Li, S. Li, X. Zhou, F. Zhu, G. Ouyang, T. Luan, Z. Zeng, Occurrence and distribution of phthalate esters in riverine sediments from the Pearl River Delta region, South China. Mar. Pollut. Bull. 83, 358–365 (2014)
- H.-T. Gao, R. Xu, W.-X. Cao, L.-L. Qian, M. Wang, L. Lu, Q. Xu, S.-Q. Yu, Effects of six priority controlled phthalate esters with long-term low-dose integrated exposure on male reproductive toxicity in rats. Food Chem. Toxicol. 101, 94–104 (2017)
- D.-W. Gao, Z.-D. Wen, Phthalate esters in the environment: a critical review of their occurrence, biodegradation, and removal during wastewater treatment processes. Sci. Total Environ. 541, 986–1001 (2016)
- N. Hamid, M. Junaid, R. Manzoor, P.-P. Jia, D.-S. Pei, Prioritizing phthalate esters (PAEs) using experimental in vitro/vivo toxicity assays and computational in silico approaches. J. Hazard. Mater. 398, 122851 (2020)
- Y.-B. Ma, P.-P. Jia, M. Junaid, L. Yang, C.-J. Lu, D.-S. Pei, Reproductive effects linked to DNA methylation in male zebrafish chronically exposed to environmentally relevant concentrations of di-(2-ethylhexyl) phthalate. Environ. Pollut. 237, 1050–1061 (2018)
- 13. G. Latini, Monitoring phthalate exposure in humans. Clin. Chim. Acta 361, 20–29 (2005)
- M. Junaid, P.-P. Jia, Y.-M. Tang, W.-X. Xiong, H.-Y. Huang, P.R. Strauss, W.-G. Li, D.-S. Pei, Mechanistic toxicity of DEHP at environmentally relevant concentrations (ERCs) and ecological risk assessment in the Three Gorges Reservoir Area, China. Environ. Pollut. 242, 1939–1949 (2018)
- J. Berger, D.E. Moller, The mechanisms of action of PPARs. Annu. Rev. Med. 53, 409–435 (2002)
- C.H. Hurst, D.J. Waxman, Environmental phthalate monoesters activate pregnane X receptormediated transcription. Toxicol. Appl. Pharmacol. 199, 266–274 (2004)

- P. Phrakonkham, S. Viengchareun, C. Belloir, M. Lombès, Y. Artur, M.-C. Canivenc-Lavier, Dietary xenoestrogens differentially impair 3T3-L1 preadipocyte differentiation and persistently affect leptin synthesis. J. Steroid Biochem. Mol. Biol. 110, 95–103 (2008)
- X. Deng, L. Pan, Y. Cai, Q. Jin, Transcriptomic changes in the ovaries of scallop Chlamys farreri exposed to benzo[a]pyrene. Genes Genom. 38, 509–518 (2016)
- G. Latini, E. Scoditti, A. Verrotti, C. De Felice, M. Massaro, Peroxisome proliferator-activated receptors as mediators of phthalate-induced effects in the male and female reproductive tract: epidemiological and experimental evidence. PPAR Res. 2008, 359267–359267 (2008)
- S.Y. Xu, H. Zhang, P.J. He, L.M. Shao, Leaching behaviour of bisphenol A from municipal solid waste under landfill environment. Environ. Technol. 32, 1269–1277 (2011)
- S.-K. Tong, H.-J. Hsu, B.-C. Chung, Zebrafish monosex population reveals female dominance in sex determination and earliest events of gonad differentiation. Dev. Biol. 344, 849–856 (2010)
- 22. E. Muth-Köhne, K. Westphal-Settele, J. Brückner, S. Konradi, V. Schiller, C. Schäfers, M. Teigeler, M. Fenske, Linking the response of endocrine regulated genes to adverse effects on sex differentiation improves comprehension of aromatase inhibition in a Fish Sexual Development Test. Aquat. Toxicol. **176**, 116–127 (2016)
- Y.-Q. Liang, G.-Y. Huang, S.-S. Liu, J.-L. Zhao, Y.-Y. Yang, X.-W. Chen, F. Tian, Y.-X. Jiang, G.-G. Ying, Long-term exposure to environmentally relevant concentrations of progesterone and norgestrel affects sex differentiation in zebrafish (Danio rerio). Aquat. Toxicol. 160, 172–179 (2015)
- 24. Y.-Q. Liang, G.-Y. Huang, G.-G. Ying, S.-S. Liu, Y.-X. Jiang, S. Liu, F.-J. Peng, The effects of progesterone on transcriptional expression profiles of genes associated with hypothalamic– pituitary–gonadal and hypothalamic–pituitary–adrenal axes during the early development of zebrafish (Danio rerio). Chemosphere **128**, 199–206 (2015)
- A. Ishihara, S. Sawatsubashi, K. Yamauchi, Endocrine disrupting chemicals: interference of thyroid hormone binding to transthyretins and to thyroid hormone receptors. Mol. Cell. Endocrinol. 199, 105–117 (2003)
- 26. O. Shen, W. Wu, G. Du, R. Liu, L. Yu, H. Sun, X. Han, Y. Jiang, W. Shi, W. Hu, L. Song, Y. Xia, S. Wang, X. Wang, Thyroid disruption by Di-n-butyl phthalate (DBP) and mono-n-butyl phthalate (MBP) in Xenopus laevis. PLoS One 6, e19159–e19159 (2011)
- N.S. Gayathri, C.R. Dhanya, A.R. Indu, P.A. Kurup, Changes in some hormones by low doses of di (2-ethyl hexyl) phthalate (DEHP), a commonly used plasticizer in PVC blood storage bags & medical tubing. Indian J. Med. Res. 119, 139–144 (2004)
- A. Wenzel, C. Franz, E. Breous, U. Loos, Modulation of iodide uptake by dialkyl phthalate plasticisers in FRTL-5 rat thyroid follicular cells. Mol. Cell. Endocrinol. 244, 63–71 (2005)
- 29. S. Imaoka, T. Mori, T. Kinoshita, Bisphenol A causes malformation of the head region in embryos of Xenopus laevis and decreases the expression of the ESR-1 gene mediated by Notch signaling. Biol. Pharm. Bull. **30**, 371–374 (2007)
- S. Santangeli, F. Maradonna, G. Gioacchini, G. Cobellis, C.C. Piccinetti, L. Dalla Valle, O. Carnevali, BPA-induced deregulation of epigenetic patterns: effects on female zebrafish reproduction. Sci. Rep. 6, 1–11 (2016)
- Y. Xu, S. Agrawal, T.J. Cook, G.T. Knipp, Maternal di-(2-ethylhexyl)-phthalate exposure influences essential fatty acid homeostasis in rat placenta. Placenta 29, 962–969 (2008)
- C.J. Thompson, S.M. Ross, K.W. Gaido, Di(n-butyl) phthalate impairs cholesterol transport and steroidogenesis in the fetal rat testis through a rapid and reversible mechanism. Endocrinology 145, 1227–1237 (2004)
- 33. T.R. Zacharewski, M.D. Meek, J.H. Clemons, Z.F. Wu, M.R. Fielden, J.B. Matthews, Examination of the in vitroandin vivoestrogenic activities of eight commercial phthalate esters. Toxicol. Sci. 46, 282–293 (1998)
- 34. Y. Nomura, N. Mitsui, U.K. Bhawal, M. Sawajiri, O. Tooi, T. Takahashi, M. Okazaki, Estrogenic activity of phthalate esters by in vitro VTG assay using primary-cultured Xenopus hepatocytes. Dent. Mater. J. 25, 533–537 (2006)

- 35. L.B. Christiansen, K.L. Pedersen, B. Korsgaard, P. Bjerregaard, Estrogenicity of xenobiotics in rainbow trout (Oncorhynchus mykiss) using in vivo synthesis of vitellogenin as a biomarker. Mar. Environ. Res. 46, 137–140 (1998)
- O. Carnevali, L. Tosti, C. Speciale, C. Peng, Y. Zhu, F. Maradonna, DEHP impairs zebrafish reproduction by affecting critical factors in oogenesis. PLoS One 5, e10201–e10201 (2010)
- G. Levy, I. Lutz, A. Krüger, W. Kloas, Bisphenol A induces feminization in Xenopus laevis tadpoles. Environ. Res. 94, 102–111 (2004)
- 38. X. Chen, S. Xu, T. Tan, S.T. Lee, S.H. Cheng, F.W. Lee, S.J. Xu, K.C. Ho, Toxicity and estrogenic endocrine disrupting activity of phthalates and their mixtures. Int. J. Environ. Res. Public Health 11, 3156–3168 (2014)
- 39. B. Huang, M. Feng, D. Li, Y. Yang, Antagonistic joint toxicity assessment of two current-use phthalates with waterborne copper in liver of Carassius auratus using biochemical biomarkers. Ecotoxicol. Environ. Saf. 116, 107–112 (2015)
- X. Chen, H. An, L. Ao, L. Sun, W. Liu, Z. Zhou, Y. Wang, J. Cao, The combined toxicity of dibutyl phthalate and benzo(a)pyrene on the reproductive system of male Sprague Dawley rats in vivo. J. Hazard. Mater. 186, 835–841 (2011)
- A. Pradhan, P.E. Olsson, J. Jass, Di(2-ethylhexyl) phthalate and diethyl phthalate disrupt lipid metabolism, reduce fecundity and shortens lifespan of Caenorhabditis elegans. Chemosphere 190, 375–382 (2018)
- 42. N. Xu, P. Chen, L. Liu, Y. Zeng, H. Zhou, S. Li, Effects of combined exposure to 17- α -ethynylestradiol and dibutyl phthalate on the growth and reproduction of adult male zebrafish (Danio rerio). Ecotoxicol. Environ. Saf. **107**, 61–70 (2014)
- 43. T.V. Madureira, M.J. Rocha, C. Cruzeiro, I. Rodrigues, R.A.F. Monteiro, E. Rocha, The toxicity potential of pharmaceuticals found in the Douro River estuary (Portugal): evaluation of impacts on fish liver, by histopathology, stereology, vitellogenin and CYP1A immunohisto-chemistry, after sub-acute exposures of the zebrafish model. Environ. Toxicol. Pharmacol. 34, 34–45 (2012)
- 44. S. Dietrich, F. Ploessl, F. Bracher, C. Laforsch, Single and combined toxicity of pharmaceuticals at environmentally relevant concentrations in Daphnia magna – a multigenerational study. Chemosphere **79**, 60–66 (2010)
- 45. X. Dong, X. Qiu, S. Meng, H. Xu, X. Wu, M. Yang, Proteomic profile and toxicity pathway analysis in zebrafish embryos exposed to bisphenol A and di-n-butyl phthalate at environmentally relevant levels. Chemosphere **193**, 313–320 (2018)
- 46. X.-M. Tian, Song Qi-ru, Combined toxic effects of di-n-butyl phthalate (DBP) and Di-2ethylhexyl Phthalate (DEHP) on reproductive ability in male rats, J. Environ. Health (2009)
- 47. C. Zhou, L. Gao, J.A. Flaws, Prenatal exposure to an environmentally relevant phthalate mixture disrupts reproduction in F1 female mice. Toxicol. Appl. Pharmacol. 318, 49–57 (2017)
- T. Pollock, R.E. Weaver, R. Ghasemi, D. deCatanzaro, A mixture of five endocrine-disrupting chemicals modulates concentrations of bisphenol A and estradiol in mice. Chemosphere 193, 321–328 (2018)
- 49. H. Xu, X. Shao, Z. Zhang, Y. Zou, X. Wu, L. Yang, Oxidative stress and immune related gene expression following exposure to di-n-butyl phthalate and diethyl phthalate in zebrafish embryos. Ecotoxicol. Environ. Saf. **93**, 39–44 (2013)
- 50. G. Hu, J. Li, Y. Shan, X. Li, Q. Zhu, H. Li, Y. Wang, X. Chen, Q. Lian, R.-S. Ge, In utero combined di-(2-ethylhexyl) phthalate and diethyl phthalate exposure cumulatively impairs rat fetal Leydig cell development. Toxicology **395**, 23–33 (2018)

- 51. J. Zhang, L. Yan, M. Tian, Q. Huang, S. Peng, S. Dong, H. Shen, The metabonomics of combined dietary exposure to phthalates and polychlorinated biphenyls in mice. J. Pharm. Biomed. Anal. 66, 287–297 (2012)
- 52. H.-T. Gao, R. Xu, W.-X. Cao, Q.-N. Di, R.-X. Li, L. Lu, Q. Xu, S.-Q. Yu, Combined effects of simultaneous exposure to six phthalates and emulsifier glycerol monosterate on male reproductive system in rats. Toxicol. Appl. Pharmacol. 341, 87–97 (2018)
- 53. N. Garcia-Reyero, K.J. Kroll, L. Liu, E.F. Orlando, K.H. Watanabe, M.S. Sepulveda, D.L. Villeneuve, E.J. Perkins, G.T. Ankley, N.D. Denslow, Gene expression responses in male fathead minnows exposed to binary mixtures of an estrogen and antiestrogen. BMC Genom. 10, 308 (2009)
- B. Huang, D. Li, Y. Yang, Joint toxicity of two phthalates with waterborne copper to Daphnia magna and Photobacterium phosphoreum. Bull. Environ. Contam. Toxicol. 97, 380–386 (2016)

Chapter 17 Molecular Toxicity Mechanism of PPCPs in the Reservoir



Naima Hamid and De-Sheng Pei 💿

Abstract Due to the ubiquitous environmental occurrence and associated health risks, pharmaceuticals and personal care products (PPCPs) as emerging contaminants widely exist in the reservoir. This chapter particularly emphasizes the mechanistic toxicity pathways of the PPCPs interacting with aquatic species. Previous studies revealed that these emerging chemicals mainly disrupted the reproductive system by decreasing the estrous cycle and reducing weight and sperm production. Besides, some pharmaceutical groups like sulfonamides (SAs) and tetracycline (TC) particularly affect the detoxification metabolism pathway. The cytotoxicity and genetic toxicity with increased apoptosis levels and DNA damage were also observed even at low exposure doses of PPCPs. In summary, the fate of PPCPs and their health risks to humans should be paid great attention to, and we must ensure their safe disposal to protect the ecosystem integrity in the reservoir.

Keywords Reproductive toxicity · PPCPs · Detoxification pathway · Zebrafish

1 Introduction

Over the past years, pharmaceuticals and personal care products (PPCPs) are recognized as emerging global pollutants owing to their increasing prevalence in the environment [1]. PPCPs possess unique bioaccumulative properties in the reservoir. However, even at lower exposure doses, PPCPs create alarming risks to aquatic species. The United States Environmental Protection Agency (US EPA) recognized numerous PPCPs predominance in the wastewaters. Therefore, it may threaten the receiving aquatic species from the surface water [2]. In 2000, the first list of 33 priority compounds was strictly banned. Later, in 2007, compounds such as diclofenac, carbamazepine, ibuprofen, triclosan, parabens, musk xylene, bisphenol, tetracycline, erythromycin, bezafibrate, indomethacin, and sulfonamides are also included in the prohibition list [3]. Previous studies reported that some of these compounds are endocrine disruptors [4, 5]. Another important reason of concern is the increase of antibiotic-resistant bacteria that have become ineffective for the treatment of numerous diseases caused by antibiotic-resistant bacteria [6].

Classification	Sub-groups	Chemical name	Adverse effects
Pharmaceuticals	Antibiotics, anti- inflammatory, antibacterial, antimalar- ial, estrogen, steroids, antiepileptic, lipid regu- lators, antifungal, anti- septics, antidepressants, analgesics, veterinary antibiotics, and growth hormones	Penicillin, amoxicillin, tetracycline, ibuprofen, paracetamol, naproxen, diclofenac, bezafibrate sulfonamides, indometha- cin, carbamazepine, erythromycin, 17β estra- diol, progesterone, testos- terone, zeranol	Endocrine disrup- tion, disrupt meta- bolic pathway, cytotoxicity
Personal care products (PCPs)	Disinfectants, fragrances, cosmetics, sunscreens, conservation agents	Musk xylol, parabens, musk xylene, tetracycline, triclosan, phthalates, benzophenone-3, octyl dimethyl-PABA, cam- phor, methylisothiazolinone, 1,4-dioxane formalde hyde, paraformaldehyde, benzalkonium chloride, methyldibromo glutaronitrile	Endocrine disrup- tion, disrupt meta- bolic pathway, apoptosis, and DNA damage

Table 17.1 General classification of the PPCPs as an emerging pollutant

Many PPCPs are toxic and are frequently used in personal care products, such as cosmetic formulations, sunscreens, paints, and both veterinary and human medicines [7–11]. The use of these chemicals can cause toxic severe implications including cytotoxicity, mutagenicity, neurotoxicity, genetic toxicity, and estrogenicity (Table 17.1). However, studies are elusive in terms of the mechanistic toxicity of these emerging contaminants. Next-generation toxicity testing in the twenty-first century (TOX₂₁) tools may elucidate the toxicology mechanism of prioritizing PPCPs.

2 PPCPs Mechanistic Toxicity in the Three Gorges Reservoir Area (TGRA)

In the TGRA, PPCPs are of great public concern due to their elevated ecological risks. In a recent study, selected PPCPs and non-steroidal anti-inflammatory drugs (NSAIDs) were quantified in the TGRA, and then their toxicities were determined using bioassays and confirmed *via* predicted TOX_{21} techniques [12]. Results showed that NSAIDs possess high ecological risks for *in vitro/in vivo* toxicity, which agrees with the predicted results. Furthermore, experimental results revealed that musk xylene from PPCPs and diclofenac from pharmaceuticals activate the PI3K-AKT-

mTOR pathway. Thus, musk xylene and diclofenac could be ranked as the priority pollutants that depressed the detoxification pathway.

Similarly, in three pharmaceuticals, the mechanistic toxicity was ranked as tetracycline hydrochloride (TH) > indomethacin, (IM) > bezafibrate (BF) both *in vitro* and *in vivo*. Contrarily, TOX₂₁ tools confirmed that IM is considered the most toxic pharmaceutical with the highest LC_{50} value, while TH appeared to be the highest LC_{50} value experimentally but the lowest LC_{50} estimated value in the PI3K-AKT-mTOR pathway [13]. IM significantly disrupted hepatogenesis and hematopoiesis with the downregulation of genetic biomarkers (*drl, mpx*, and *gata2a*) [12].

The sulfonamides (SAs) class also belongs to pharmaceuticals, particularly antibiotics, which pose higher ecological risks in the TGRA [14]. Among sulfonamide congeners, sulfamethoxazole (SMX) appeared as the individual chemical with elevated environmental impacts in the TGRA and increased *in vitro* or *in vivo* toxicity. Hamid et al. [14] applied various predicted models to confirm the toxicity potential among individual and combined sulfonamide mixtures in the TGRA. Besides, after SAs binary mixture treatment on zebrafish embryos, the average developmental growth was decreased with the suppression of the detoxification pathway [14]. The mathematical combination index (CI) and independent action (IA) model indicated maximum synergistic effects of the SAs mixtures with the population affected level fa = 0.9 [14]. In contrast, the IA model found the mixture effects ranked as additive > antagonistic > synergistic effects [14].

3 Mechanism of Detoxification Metabolism Pathway

It is believed that SAs induced transcriptional changes in the detoxification metabolism pathway resulting in hepatogenesis and hematopoiesis toxicity [14]. Generally, the detoxification mechanism of any chemicals involved various metabolic pathways, which are composed of three phases. Phase I consists of cytochrome p450 (CYP450) family genes. Phase II includes sulfotransferases (SULTs) and glucuronosyltransferases (UGTs). Lastly, phase III comprises numerous transmembrane proteins (TPs), which allow the pollutant to excrete out from the body [15]. Nevertheless, the comprehensive detoxification pathways studies are limited, particularly for aquatic species in the reservoir [7–11]. The detailed mechanism of SAs disrupting detoxifying metabolism pathway is presented in Fig. 17.1.

The single and joint chemical exposure of the SAs chemical group significantly disturbs the detoxification metabolism pathway in zebrafish (Fig. 17.1). Recently, a study reported by Hamid et al. [14] suggested that SMX as a single chemical treatment upregulated the phase I genes, implying its high toxicity potential. Moreover, the transcriptional results were consistent with developmental toxicity (elevated pericardial edema) [14]. After exposure to SDZ, SMR, ST, and SPY, the gene transcriptional levels of phase I were suppressed in a concentration-dependent manner, which is consistent with the previous report by Liu et al. [13].



Fig. 17.1 The detailed description of the detoxification pathway of the sulfonamides (individual and mixtures) was modified from the previous report [14]



Fig. 17.2 Transcriptomic expression levels of the detoxification pathway genes after SAs treatment (single and joint binary mixtures). This figure is modified from our previous study [14]

Regarding the detoxification pathway of Phase II, the single SAs behave as the suppressors of the *sult* family genes in all treated groups [14]. Generally, SULTs are used in sulfation by forming intermediate complexes of the drugs [16–18]. However, UGTs formulate glutathione intermediatory complexes that make the chemical water-soluble or increase its hydrophobicity [17]. Phase III comprises the transmembrane proteins that help the chemical to remove from the cell membrane. Notably, all the genes in phases I and II were reported to behave synergistically to disturb the detoxification pathway in zebrafish (Fig. 17.2) [14].

4 PPCPs Toxicogenetic Endpoints In vitro or In vivo

Toxicologists have taken up various technological approaches, such as transcriptomics (qRT-PCR, RNA-seq, and microarrays), proteomics (western blot, ELISA, 2D-PAGE, and mass spectrometry), and metabolomics (nuclear magnetic resonance spectroscopy-NMR, lipotoxicity test, Matador-luciferase cytotoxicity assay, and liquid chromatography-mass spectroscopic analysis-LC-MS) to reveal the mechanistic toxicological mechanisms of the PPCPs. Here, a summary of numerous *in vivo* studies on mixture toxicity is presented in Table 17.2. The majority of the published studies have focused on determining developmental toxicity in vivo, such as mortality, deformity, and oxidative stress [6, 19]. Likewise, non-steroidal anti-inflammatory drugs (NSAIDs) like naproxen severely decrease the body weight and length of the zebrafish at environmentally relevant concentrations (ERC) [35]. Specifically, most PPCPs possess endocrine disruption properties mainly by perturbing the hypothalamic–gonadal axis (HPG) pathway [33]. Previously, a study reported by Yu et al. [20] investigated the transgenerational reproductive effects of sulfamethoxazole antibiotics, in which the nematode lifespan and the HPG transcription levels were markedly affected [20]. Further, various antibiotics increased hepatic mass with an elevated vtg transcription level in adult male zebrafish [26]. Interestingly, some pharmaceuticals also disrupt the detoxification pathway by affecting the transcript levels even at lower ERCs. A recent study highlighted that sulfonamides, tetracycline, and indomethacin disturb the metabolism pathway even at low exposure concentrations [12, 14].

Except for the mixture effects at the protein level, phthalates and PPCPs mixtures downregulate the expression levels of the steroidogenic proteins in adult rats after being treated with the quinary mixtures of DMP, DEP, DBP, BBP, DEHP, DNOP, and GMS [36]. Similarly, atenolol, carbamazepine, diclofenac, and gemfibrozil joint exposure increased the cytotoxicity in *Dreissena polymorpha* [1]. Furthermore, binary mixtures of sulfonamides also disturb the metabolism detoxification pathway with the maximum synergistic effects [14]. Similarly, the ERC mixture of steroidal pharmaceuticals affects the transcript levels of the reproductive genes in rainbow trout [37]. Thus, PPCPs may disturb the HPG axis and participate in the detoxification pathway with prominent cytotoxic effects even at lower exposure doses.

5 Computational Approaches for Toxicological Screening

Numerous predictive modeling techniques have been used for assessing chemical toxicity [38]. Ligand descriptors are the individual chemical information of the compounds that are simulated by software programs [39]. Different modeling tools, such as molecular docking *via* quantitative structure–activity relationship and chemical index (CI) are the most widely used to determine the chemical

		Exposure		
Chemical	Species	time	Toxic endpoints	Reference
Sulfamethoxazole	D. rerio	8 hpf (hours per fertilization)– 96 hpf	8 hpf = yolk sac damaged, 24 hpf = eyespot delayed development 48 hpf = anomaly in the ovum and spinal cord flexure/bent spine 96 hpf = bending tail	[19]
Sulfadiazine	D. rerio	8hpf–96 hpf	8 hpf = yolk sack greatly convex, 24 hpf = mutation in zebrafish embryos, 48 hpf = spinal cord flex- ure/bent spine, 96 hpf = swim bladder deletion with temporary blood clot	[19]
Sulfamethoxazole	C. elegans	(F0–F6) generations	Lifespan was severely affected at embryo expo- sure (F1) and reproduction was affected following germline exposure (F2). In non-exposed F3–F6 gen- erations, nematode lifespan, and reproduction showed significant inhibitions.	[20]
Bezafibrate	Human sperm cells	15, 30, 45 min	DNA damage ↑ at 30 min Apoptotic cells ↑ at 45 min	[21]
Indomethacin	KB, Saos-2, 1321 N, U-87MG cell lines	72 h	Cell viability \downarrow 50.31% only for U-87MG cells.	[22]
Tetracycline	D. rerio	6–120 h	Hatching rate \downarrow (72-hpf) EC50 = 31.05 μ M malformation rate (EC50): 120 hpf = 36.92 μ M Mortality Rate (LC50): 120 hpf = 41.2 μ M Heartbeat \downarrow 20% at 337.5 μ M; acta1a, myl7, and gle1b significantly altered at 202.5 μ M	[23]
Bezafibrate	Human embryo cells	24 h	PPARα mRNA level ↑ at 300 and 1000 µM Inhibit the phosphorylation of Akt	[24]
Sulfadiazine	D. rerio	8hpf–96 hpf	Malformation, such as hemorrhage, blood	[25]

 Table 17.2
 Toxicogenetic endpoints of PPCPs in various bioassays exposed at the ERCs

(continued)

		Exposure		
Chemical	Species	time	Toxic endpoints	Reference
			coagulation, tail bending, pericardial edema, and swim bladder deletion	
Carbamazepine, Fenofibric acid, Propranolol, Sulfa- methoxazole, Trimethoprim	D. rerio	21 days	Male zebrafish showed an increase of Vtg, increased hepatic mass	[26]
Indomethacin	Gastric cells MGC-803	12, 24, 48 h	Cell viability ↓ with increasing dose and dura- tion. Gastric cancer cell apoptosis through Akt/GSK3β/NAG-1 pathway	[27]
Sulfamethoxazole, Sulfadiazine, Sulfamerazine, Sulfapyridine, Sulfameter	D. rerio	8–96 hpf	Disrupt the detoxification metabolism pathway	[14]
Atenolol, Carba- mazepine, Diclofenac, Gemfibrozil	Dreissena polymorpha	8–96 hpf	Elevated cytotoxicity of diclofenac and gemfibrozil.	[1]
Tetracycline	D. rerio	96 h	$\begin{split} EC50 &= 0.0071 \ \mu\text{M} \\ CYP1A \uparrow at 0.045 \ \mu\text{M} \text{ and} \\ 0.45 \ \mu\text{M} \\ LC_{10} &= 1241.52 \ \mu\text{M}, \\ LC_{50} &= 134.35 \ \mu\text{M} \end{split}$	[28]
Indomethacin	R. subcapitata	72 h	NOEC (No observed effect concentra- tion) = $2.35 \ \mu M$ EC10 = $12.86 \ \mu M$	[29]
Sulfadiazine	D. rerio	8hpf–96 hpf	8 hpf = yolk sack greatly convex, 24 hpf = mutation in zebrafish embryos, 48 hpf = spinal cord flex- ure/bent spine, 96 hpf = swim bladder deletion with temporary blood clot	[19]
Oral contraceptives (EE2, NOR)	Daphnia magna	25 days	Synergistic effects were observed with the 57% decrease in the offspring number, inhibited swimming	[30]
Parabens, phthalates	Adult rats	1 h	Results showed that EDCs interact <i>in vivo</i> ,	[31]

 Table 17.2 (continued)

(continued)

Chemical	Species	Exposure time	Toxic endpoints	Reference
			magnifying one another's effects, consistent with inhibition of enzymes that are critical for estrogen metabolism	
Carbamazepine, Diclofenac, EE2, Mefenamic acid	Daphnia magna	48 h	The drug mixture reduced the age at first reproduc- tion of daphnid in the F0 and F2 generation and increased the body length at first reproduction in the generations F0 and F3.	[32]
Diclofenac, Mefenamic acid, Naproxen, Acetylsalicylic acid, Ibuprofen	D. rerio	21 days	NSAIDs affect the gene transcription of the HPG axis pathway with the significant upregulation of the $fsh\beta$, $lh\beta$, $fshr$, and lhr genes.	[33]
Tetracycline	C. pyrenoidosa (CP) S. obliquus (SO)	96 h	$\begin{array}{l} \text{CP:EC}_{20-96h} = 5.96,\\ \text{EC}_{50-96h} = 15.44, \text{EC}_{80-96-}\\ \text{h:40.01 } \mu\text{M}\\ \text{SO: EC}_{20-96h} = 1.87,\\ \text{EC}_{50-96h} : 7.36\\ \text{EC}_{80-96h} : 28.96 \; \mu\text{M} \end{array}$	[34]

Table 17.2 (continued)

interaction with particular proteins [38]. These predictive models are extensively applied in bioinformatics studies for *in silico* toxicity prediction [39].

6 Conclusion

In recent years, PPCPs have emerged as emerging pollutants because of their harmful effects on the reservoir. Previously, published studies revealed that despite the lower concentrations in the reservoir, the endocrine disruption effects in the aquatic species become most prevalent. Furthermore, the body length and weight of most of the species have been reduced with elevated apoptosis levels and DNA damage. Recently, some studies also highlighted that sulfonamides and tetracycline might disturb the detoxification pathway. Moreover, prioritizing the PPCPs and pharmaceuticals toxicity, indomethacin, sulfamethoxazole, diclofenac, naproxen, and tetracycline have been categorized as the most toxic chemicals. Thus, more detailed studies are required to determine the exact mechanistic toxicity of these emerging contaminants.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- M. Parolini, B. Quinn, A. Binelli, A. Provini, Cytotoxicity assessment of four pharmaceutical compounds on the zebra mussel (Dreissena polymorpha) haemocytes, gill and digestive gland primary cell cultures. Chemosphere 84, 91–100 (2011)
- A.J. Ebele, M. Abou-Elwafa Abdallah, S. Harrad, Pharmaceuticals and personal care products (PPCPs) in the freshwater aquatic environment. Emerg. Contamin. 3, 1–16 (2017)
- M.K. Sarath Josh, S. Pradeep, K.S. Vijayalekshmy Amma, R. Sudha Devi, S. Balachandran, M.N. Sreejith, S. Benjamin, Human ketosteroid receptors interact with hazardous phthalate plasticizers and their metabolites: an in silico study. J. Appl. Toxicol. 36, 836–843 (2016)
- M.K. Sarath Josh, S. Pradeep, V.K. Adarsh, K.S. Vijayalekshmi Amma, R. Sudha Devi, S. Balachandran, M.N. Sreejith, U.C. Abdul Jaleel, S. Benjamin, In silico evidences for the binding of phthalates onto human estrogen receptor α, β subtypes and human estrogen-related receptor γ. Mol. Simul. 40, 408–417 (2014)
- 5. E.K. Shanle, W. Xu, Endocrine disrupting chemicals targeting estrogen receptor signaling: identification and mechanisms of action. Chem. Res. Toxicol. **24**, 6–19 (2011)
- M. Bilal, H.M.N. Iqbal, An insight into toxicity and human-health-related adverse consequences of cosmeceuticals—a review. Sci. Total Environ. 670, 555–568 (2019)
- M. Saad, A. Matheeussen, S. Bijttebier, E. Verbueken, C. Pype, C. Casteleyn, G.C. Van, S. Apers, L. Maes, P. Cos, In vitro CYP-mediated drug metabolism in the zebrafish (embryo) using human reference compounds. Toxicol. In Vitro 42, 329 (2017)
- M. Saad, K. Cavanaugh, E. Verbueken, C. Pype, C. Casteleyn, C.V. Ginneken, S.V. Cruchten, Xenobiotic metabolism in the zebrafish: a review of the spatiotemporal distribution, modulation and activity of Cytochrome P450 families 1 to 3. J. Toxicol. Sci. 41, 1 (2016)
- 9. K. Kurogi, T.A. Liu, Y. Sakakibara, M. Suiko, M.C. Liu, The use of zebrafish as a model system for investigating the role of the SULTs in the metabolism of endogenous compounds and xenobiotics. Drug Metab. Rev. **45**, 431 (2013)
- Z.H. Li, D. Alex, S.O. Siu, I.K. Chu, J. Renn, C. Winkler, S. Lou, S.K. Tsui, H.Y. Zhao, W.R. Yan, Combined in vivo imaging and omics approaches reveal metabolism of icaritin and its glycosides in zebrafish larvae. Mol. BioSyst. 7, 2128 (2011)
- L.I. Ming, Y. Cao, Z. Zhao, S. Lin, A. Meng, Characterization and expression pattern ofpouII1, a novel class II POU gene in zebrafish. Chin. Sci. Bull. 46, 1523–1527 (2001)
- M. Junaid, Y. Wang, N. Hamid, S. Deng, W.-G. Li, D.-S. Pei, Prioritizing selected PPCPs on the basis of environmental and toxicogenetic concerns: a toxicity estimation to confirmation approach. J. Hazard. Mater. 380, 120828 (2019)
- Y. Liu, M. Junaid, Y. Wang, Y.-M. Tang, W.-P. Bian, W.-X. Xiong, H.-Y. Huang, C.-D. Chen, D.-S. Pei, New toxicogenetic insights and ranking of the selected pharmaceuticals belong to the three different classes: a toxicity estimation to confirmation approach. Aquat. Toxicol. 201, 151–161 (2018)
- N. Hamid, M. Junaid, D.-S. Pei, Individual and combined mechanistic toxicity of sulfonamides and their implications for ecological risk assessment in the Three Gorges Reservoir Area (TGRA), China. J. Hazardous Mater. 382, 121106 (2020)
- T. Uno, M. Ishizuka, T. Itakura, Cytochrome P450 (CYP) in fish. Environ. Toxicol. Pharmacol. 34, 1–13 (2012)

- M.O. James, W. Li, D.P. Summerlot, L. Rowland-Faux, C.E. Wood, Triclosan is a potent inhibitor of estradiol and estrone sulfonation in sheep placenta. Environ. Int. 36, 942–949 (2010)
- Y.I. Mohammed, K. Kurogi, A.A. Shaban, Z. Xu, M.-Y. Liu, F.E. Williams, Y. Sakakibara, M. Suiko, S. Bhuiyan, M.-C. Liu, Identification and characterization of zebrafish SULT1 ST9, SULT3 ST4, and SULT3 ST5. Aquatic Toxicol. (Amsterdam, Netherlands) 112–113, 11–18 (2012)
- 18. K.M. Barakat, E.A. Beltagy, Bioactive phthalate from marine Streptomyces ruber EKH2 against virulent fish pathogens. Egypt. J. Aquat. Res. **41**, 49–56 (2015)
- 19. T. Lin, Y. Chen, W. Chen, Impact of toxicological properties of sulfonamides on the growth of zebrafish embryos in the water. Environ. Toxicol. Pharmacol. **36**, 1068–1076 (2013)
- Z. Yu, G. Sun, Y. Liu, D. Yin, J. Zhang, Trans-generational influences of sulfamethoxazole on lifespan, reproduction and population growth of Caenorhabditis elegans. Ecotoxicol. Environ. Saf. 135, 312–318 (2017)
- 21. L. Rocco, C. Peluso, F. Cesaroni, N. Morra, Genomic damage in human sperm cells exposed in vitro to environmental pollutants. J. Environ. Anal. Toxicol. **02** (2012)
- M.A. Hashemipour, H.M. Honarmand, F. Falsafi, M.T. Arashlo, S. Rajabalian, In vitro cytotoxic effects of celecoxib, mefenamic acid, aspirin and indometacin on several cells lines. J. Dent. 17, 219–225 (2016)
- Y. Zhang, X. Wang, X. Yin, M. Shi, R.A. Dahlgren, H. Wang, Toxicity assessment of combined fluoroquinolone and tetracycline exposure in zebrafish (Danio rerio). Environ. Toxicol. 31, 736–750 (2016)
- 24. Y. Zhao, M. Okuyama, H. Hashimoto, Y. Tagawa, T. Jomori, B. Yang, Bezafibrate induces myotoxicity in human rhabdomyosarcoma cells via peroxisome proliferator-activated receptor α signaling. Toxicol. In Vitro **24**, 154–159 (2010)
- 25. L.T. Lin, C. Yan-qiu, C. Wei, Toxic effect of sulfadiazine on the growth of zebrafish embryos in the water body. J. Saf. Environ. (2014)
- 26. T.V. Madureira, M.J. Rocha, C. Cruzeiro, I. Rodrigues, R.A.F. Monteiro, E. Rocha, The toxicity potential of pharmaceuticals found in the Douro River estuary (Portugal): evaluation of impacts on fish liver, by histopathology, stereology, vitellogenin and CYP1A immunohistochemistry, after sub-acute exposures of the zebrafish model. Environ. Toxicol. Pharmacol. 34, 34–45 (2012)
- R.P. Pang, H.U. Pin-Jin, Z.R. Zeng, W. Chen, Indomethacin induces gastric cancer cell apoptosis through Akt/GSK3β/NAG-1 pathway. Chin. J. Pathophysiol. 27, 1513–1518 (2011)
- Q. Zhang, J. Cheng, Q. Xin, Effects of tetracycline on developmental toxicity and molecular responses in zebrafish (Danio rerio) embryos. Ecotoxicology 24, 707 (2015)
- 29. I. Tamura, Y. Yasuda, K.I. Kagota, S. Yoneda, N. Nakada, V. Kumar, Y. Kameda, K. Kimura, N. Tatarazako, H. Yamamoto, Contribution of pharmaceuticals and personal care products (PPCPs) to whole toxicity of water samples collected in effluent-dominated urban streams. Ecotoxicol. Environ. Saf. 144, 338 (2017)
- 30. T. Goto, J. Hiromi, Toxicity of 17α -ethynylestradiol and norethindrone, constituents of an oral contraceptive pill to the swimming and reproduction of cladoceran Daphnia magna, with special reference to their synergetic effect. Mar. Pollut. Bull. **47**, 139–142 (2003)
- T. Pollock, R.E. Weaver, R. Ghasemi, D. deCatanzaro, A mixture of five endocrine-disrupting chemicals modulates concentrations of bisphenol A and estradiol in mice. Chemosphere 193, 321–328 (2018)
- 32. S. Dietrich, F. Ploessl, F. Bracher, C. Laforsch, Single and combined toxicity of pharmaceuticals at environmentally relevant concentrations in Daphnia magna—a multigenerational study. Chemosphere **79**, 60–66 (2010)
- 33. K. Ji, X. Liu, S. Lee, S. Kang, Y. Kho, J.P. Giesy, K. Choi, Effects of non-steroidal antiinflammatory drugs on hormones and genes of the hypothalamic-pituitary-gonad axis, and reproduction of zebrafish. J. Hazard. Mater. 254–255, 242–251 (2013)
- 34. D.M. Xu, Y.H. Wang, G.W. Rao, Cellular response of freshwater green algae to the toxicity of tetracycline antibiotics. Environ. Sci. **34**, 3386 (2013)
- 35. C. Xu, L. Niu, H. Guo, X. Sun, L. Chen, W. Tu, Q. Dai, J. Ye, W. Liu, J. Liu, Long-term exposure to the non-steroidal anti-inflammatory drug (NSAID) naproxen causes thyroid disruption in zebrafish at environmentally relevant concentrations. Sci. Total Environ. 676, 387–395 (2019)
- 36. H.-T. Gao, R. Xu, W.-X. Cao, Q.-N. Di, R.-X. Li, L. Lu, Q. Xu, S.-Q. Yu, Combined effects of simultaneous exposure to six phthalates and emulsifier glycerol monosterate on male reproductive system in rats. Toxicol. Appl. Pharmacol. 341, 87–97 (2018)
- 37. M. Schmitz, M. Beghin, S.N.M. Mandiki, K. Nott, M. Gillet, S. Ronkart, C. Robert, S. Baekelandt, P. Kestemont, Environmentally-relevant mixture of pharmaceutical drugs stimulates sex-steroid hormone production and modulates the expression of candidate genes in the ovary of juvenile female rainbow trout. Aquat. Toxicol. 205, 89–99 (2018)
- K. Khan, E. Benfenati, K. Roy, Consensus QSAR modeling of toxicity of pharmaceuticals to different aquatic organisms: ranking and prioritization of the DrugBank database compounds. Ecotoxicol. Environ. Saf. 168, 287–297 (2019)
- N. Klüver, K. Bittermann, B.I. Escher, QSAR for baseline toxicity and classification of specific modes of action of ionizable organic chemicals in the zebrafish embryo toxicity test. Aquat. Toxicol. 207, 110–119 (2019)

Chapter 18 Adverse Outcome Pathways (AOPs) of Pollutants in the Reservoir



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter explains the adverse outcome pathways (AOPs) for evaluating risks caused by pollutants found in the reservoir. Different kinds of pollutants exist in the reservoir at the same time, and these mixture contaminants pose a significant challenge in evaluating possible biological effects and risk assessment. Therefore, the AOP framework assists in deciphering the molecular or biochemical data from field-sampled organisms that are exposed to a combined mixture of contaminants. The resulting endpoints are later utilized to infer the possible risk. Despite being simple, the major shortcoming of traditional AOP is the poor replication of complex toxicological processes. A more advanced AOP is developed, called quantitative adverse outcome pathway (qAOP), which contains one or more biology-based computational models delineating key event (KE) correlations and combining a molecular initiating event (MIE) with an adverse outcome (AO).

Keywords Quantitative adverse outcome pathway $(qAOP) \cdot Molecular$ initiating event (MIE) \cdot Key events (KEs) \cdot Toxicodynamics \cdot Endpoints

1 Introduction

For reservoir water quality monitoring, a quantitative assessment related to the fate of pollutants in water bodies is required, which assists in the remediation of the contaminants. If only quantification in terms of concentrations is analyzed, then the analytically undetected contaminants may produce transformation products that are toxicologically substantial and elicit mixture effects. Thus, the approach of adverse outcome pathways (AOPs) is suggested, which may improve the environmental impact assessments [1]. The concurrent exposure of organisms to a variety of contaminants does not essentially mean that joint effects are aroused at measurable levels [2]. A single contaminant may act differently, and the association between the dose of contaminant and the concentrations found in the mixtures may not contribute to the detectable effect [3]. The AOPs framework assists in the interpretation, communication, and translation of pathway-specific systematic data into responses related to the assessment of risks of chemicals to the environment and human health [4].

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_18





Fig. 18.1 AOPs framework for reservoir water pollutants

The AOPs model was promoted by National Research Council in the early 1980s as an approach to anticipate diseases in humans developed from the concept of biomarkers [5]. AOPs model explains the toxicity mechanisms starting from the initiation of the molecular event to some other key events (KEs) that take place when a specific chemical interacts with the environment to the adverse outcome (AO) and is detected at the individual level [6]. AOPs have gained considerable approachability as a systematic framework for toxicological information. Associations like the Organization for Economic Cooperation and Development (OECD) with a workgroup of international experts have developed guidance for the assessment and technical review of AOPs and accepted data related to toxicology by various regulatory experts [4].

An AOP comprises molecular initiating events (MIEs), KEs, key event relationships (KERs), and AOs. The starting point of AOP is MIE, which involves the interaction of chemicals at the molecular level, such as the binding of ligandreceptor. The KEs are at the cell, tissue, or higher levels of biological organization. The MIE triggers the biological changes resulting in a typical endpoint of biologic dysfunction known as an AO that is further weighed up in the risk assessment [7].

Figure 18.1 explains the overall concept of AOPs across various scales of biological organization including the MIEs caused by most pollutants, such as receptor-ligand interaction, DNA binding, and protein oxidation at the molecular level. Similarly, at the cellular and tissue level, KE1 contains gene activation, protein production, and altered signaling, and other KEs include effects at the organ level (such as impaired development) and reproduction leading to population-level impacts (such as species extinction).

For impact assessment of various chemical contaminants in the reservoir, water bioanalytical tools are usually designed for apprehending KEs of biological reactions caused by exposure to contaminants and observed MIEs at the cell and organism, which may lead to AOs at the population level [3]. AOPs are characterized as linear events, however, parallel flows and crossing pathways could also be involved. They are deliberated as flexible configurations that ought to be incessantly refined by replenishing old and novel data. These repetitive refinement drills involve the explanation and calculation of the toxicodynamic interactions between contiguous KEs and the description of toxicokinetic settings leading to the activation of an AOP [8].

2 Development of AOPs for Assessment of Water Pollutants

To assess the AOs of pollutants in the reservoir ecosystem, the water pollution status is determined first by analyzing all groups of contaminants. The data of contaminants found in specific sites are then clustered hierarchically according to detected levels, such as low, moderate, and high concentration levels. Site-specific information could also be remarked to relate it to commercial, public, or agricultural activities, such as cultivation and the presence of meadows or factories, for predicting in advance what kind of pollutants could be detected in the reservoir and the river basin [9]. The complementary information on toxicity caused by pollutants detected in the reservoir water is also useful for examining the spatial and temporal distribution. Next, the dose-response relationship information of pollutants is added to component-based modeling approaches [2].

For the development of AOPs, selective aquatic vertebrates and invertebrate species are used as model organisms, because they are appropriate bioindicator species with a short lifespan that can be observed over a short duration. Moreover, they could be considered for in-depth studies at the molecular and population levels. However, the interspecific differences are an integral part of the natural environment between aquatic organisms. Thus, they need to be adapted to an AOP framework. Pollutant-mediated AOP enables an improved understanding of the environmental effects of pollutants in varied ecosystem communities [10]. Organisms model (such as zebrafish and other aquatic fish species) and toxicodynamic studies of single or combined pollutants are measured [3].

AOP cannot elucidate the toxicokinetics of pollutants. Consequently, no risk could be assessed directly, because AOP doesn't describe the specific chemical or contaminant. Therefore, absorption, distribution, metabolism, and elimination properties are not interpreted for specific chemicals, and the risk assessment couldn't be performed. However, AOP can explain the potential hazard caused by a specific pollutant at a concentration, which can trigger the MIE and further activate the AOP [11]. The overall schematic of AOP development for risk assessment of pollutants found in the reservoir is presented in Fig. 18.2. It requires the information from the exposure assessment to provide the basis for AOP development. If an MIE occurs, then an effect at an equal or higher level of organization may take place and can be defined as a KE. Examples might be the subsequent phosphorylation of a



Fig. 18.2 Schematic of AOP development for reservoir pollutants by integrating exposure pathway information as preliminary evidence

transcription factor activated as an MIE causing the upregulation of an enzyme that synthesizes a hormone. Upregulation of the enzyme would exist as one module that would be sequentially linked to another module of hormone increase, each of which would exist as KEs. In some cases, a linkage or relationship to a specific MIE may not be known [12]. The linkage between the KE modules is defined as a KE relationship (KER). In most cases, the KER may be inferred or determined between KEs or MIEs through hypothesis testing and literature reviews. The resulting information can be loaded into databases (i.e., AOP-wiki), where the KER can be reviewed, categorized, and confirmed through additional studies [5].

2.1 Bioassays Prioritization for AOPs

In common, a chemical contaminant could trigger different AOPs at different exposure levels. Like at low concentrations of phthalates, a direction may be reversed for some developmental effects [13]. Thus, for this purpose, a quantitative understanding of associations between the KEs and the AO is necessary to match the exposure of the target site with a relevant response of AOP. Such in vitro assays are required to match the KEs for explaining the link between the exposure concentration of chemicals and the predicted AOs [14]. While addressing complex pollutant

mixtures including emerging contaminants and natural toxicants, the chemical exposure scenarios of the contaminants and inherent innate biological characteristics (such as the reproduction and neurophysiology of organisms in the reservoir) are firstly taken into account. Based on the AOP concept, the bioassay prioritization approach is applied to predict a particular class of pollutants that is likely to be toxic for all organisms of the same species. Although the toxicity mechanism might not be clear for a given class of pollutant, the selection of a bioassay acute and chronic toxicity experiments could offer qualitative instruction, considering the mechanism of absorption, distribution, metabolism, excretion, and target sites [15]. Zebrafish, echinoderms, and molluscs are common organism models. Moreover, in silico studies and *in vitro* or *in vivo* bioassays are also being utilized to adapt the AOP approach [16]. According to the water directive framework, chemical indicators and ecosystem quality are not related to each other [17]. Regular environmental monitoring analysis of contaminants and biological examinations are regarded as distinct activities. Contaminant exposure and associated ecological impacts are studied for the single cause and effect relation of individual pollutants rather than taking into consideration the mixture toxicity of several contaminants together [9].

2.2 Design of Mechanistic Test Approaches

Systematic pre-tests act as filters for toxicological decisions and had more than 50 years of history in specific fields of toxicology [18]. In previous decades, numerous unnecessary in vivo studies related to carcinogenicity were waived. More commonly, when the results of animal and bacterial cell mutagenicity were positive, it was not considered necessary to execute carcinogenicity tests for a longer span in rats or mice because of the high likelihood of getting a positive result [19]. For the prediction of risks associated with exposure to pollutants, it is important to delineate a comprehensive set of KEs that are quantifiable *in vitro*, but under in vivo conditions, it displays adverse effects. In many AOPs, genotoxicity tests including chromosome breaks and DNA mutation are selected as KEs [11]. Mitochondrial toxicity is another example of pre-tests for the KE, as it is associated with several AOs [20]. Adverse effects appear when a chemical contaminant invades the target cells in tissue at concentrations that affect mitochondrial function, leading to cell or organ dysfunction [11]. Similarly, the predictive systems of developmental toxicity facilitated by contaminants impede neural crest cell migration, block the growth of neurite, and affect gene regulation during embryonic development [21]. Moreover, it can interfere with developmental signaling pathways leading to oxidative stress and endoplasmic reticulum stress [22, 23].

3 Endpoints in the AOP Framework for Risk Assessment

AOP is a vital part of the risk assessment conceptual model, because it provides the vital framework, through which ecotoxicological information is acquired, applied, and integrated at different levels of biological organization to efficiently evaluate outcomes. Information in AOPs consists of a chain of events starting from MIEs to AOs. They may also include such endpoints, whose mechanistic importance is ambiguous, but they are informative as a particular biomarker of the activation pathway [24].

One example is vitellogenin production in male fish, which indicates the activation of the ER signaling pathway related to reproductive dysfunction. With increased mechanistic details in AOPs, endpoints related to an adverse effect may contribute to identifying the responsible initiating event. The down-regulation of VTG expression in female fish collected from water could be deduced from the perspective of potential effects on reproductive ability, which could be utilized to identify particular chemicals responsible for this outcome [15]. For quantitative risk assessment of a single chemical contaminant, molecular and biochemical endpoints generally do not deliver adequate data. For such chemicals with little information, chemicals are placed within an established AOP and endorsed with an evaluation of marginal data. For example, the well-developed knowledge of potent ER agonists like EE2 could be utilized to understand the behavior of other potential estrogenic compounds in water [25].

Sometimes, the effects triggered by different chemical mixtures act *via* the same MIE. Thus, the AOP frameworks can elucidate such effects. Moreover, contaminants mixture may affect the pathways that later converge at shared intermediate steps, and could also be summed up for similar risks [24]. AOP framework also explicates the problem of interspecies extrapolation by identifying convergence or divergence spots within the pathways. The AOP is also an influential concept for the extrapolation of such species, which have conserved function mechanisms and were secured to endpoints pertinent to a particular risk problem. Like ecological receptors, the endpoints are fertility, reproduction, development, growth, and survival rate [23].

4 Use of the AOP for Risk Assessment of Pollutants

AOP has been applied in many studies for the assessment of chemical risk because of its promising characteristics [26]. In a previous study [27], an AOP was applied to a population model for white suckers in Great Lakes, and predicted the population based on the data on sex steroids in fish exposed to pulp and paper mill effluent. The chemicals consistently affected an early KE, such as enzyme inhibition, the activation of receptors, and the suppressive production of steroids, which resulted in decreased egg production [4]. In a previous study [26], the AOPs of microplastics based on their mechanism of ecotoxicity and human health toxicity were discussed.

The key MIE was a generation of ROS, and AOs were growth reduction, reproductive failure, and increased mortality. Similarly, a recent study was reported to monitor the San Francisco Bay-Delta watershed [5]. The AOPs were added to the preliminary monitoring list of contaminants using AOP-driven tools. Based on AOP principles, bioinformatics pathway software was developed, which served as a suitable tool to predict endpoints like behavioral changes and developmental lethality that influence wildlife populations. Recently, Corsi et al. performed an AOP analysis to evaluate pollutants from 57 great lake tributaries for identifying the contaminants of biological relevance [28].

San Francisco Bay-Delta has been studied widely over the years for effect-based analysis of the estrogenic activity of alkylphenols (Aps), alkylphenol ethoxylates (APEs), pyrethroid insecticide (bifenthrin), and persistent herbicides [29]. Estrogenic effects were not prominent when a single contaminant was assessed at environmentally relevant concentrations. However, exposure to mixture contaminants resulted in significant estrogenic activity, which is consistent with a recent study that estrogenic activity of Diuron is increased in the presence of APEs and Aps in male fathead minnows [30].

5 Outlook and Prospects

AOP has proved to be an important tool for evaluating the effects of chemical mixtures in the field [31]. AOP-based risk profiling and ranking based on relative toxicity are of great use for chemical prioritization. In the last few decades, AOP has been used to evaluate the negative impacts of contaminants found in water bodies on various biological scales. Currently, the use of quantitative adverse outcome pathways (qAOPs) has become a new approach in the AOP system. A qAOPs model including the KEs and their relationships within an AOP is developed based on scientific evidence that could be acquired from the AOP database, a resource that registers the AOPs and collects the essential mechanistic information [32]. Notably, the AOP knowledge base (https://aopkb.oecd.org/) is a crowd-based resource that catalogs AOPs and aggregates essential mechanistic information, such as auxiliary weight of evidence for adjoining KERs. An overall qAOP model elucidates the linkages between KEs and narrates the dose-response-based stimulation of an MIE to the response dynamics of KEs and AOs [33].

The overall AOPs description is well organized, which is followed by welldefined assistance. Although the AOPs applications are developing, regulatory gaps are required to be attained for replacing the currently used toxicological approaches with the AOP-based risk assessment [8]. However, it is still expected that the conceptual framework of AOP would undergo a mechanistic change in the risk assessment using the approach of weight evidence approach for the mechanistic testing of apical endpoints [34].

6 Conclusion

This chapter explains the development of a conceptual AOP model for reservoir pollutants. The pollutants trigger the initiating events at the molecular to the cellular level and later to the organ, organism, and population scale. The overall approach includes observational information, field experimental studies of contaminants, pollutants occurrence and concentrations in the environment, and their spatial distribution. This information is integrated with the AOP that assists in the development of mechanistic tests for assessing the adverse effects at different levels. This approach offers a platform to incorporate data of molecular and cellular biomarker that predicts the substantial endpoints.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of Interest The authors declare no conflict of interest.

References

- E. Malaj, C. Peter, M. Grote, R. Kühne, C.P. Mondy, P. Usseglio-Polatera, W. Brack, R.B. Schäfer, Organic chemicals jeopardize the health of freshwater ecosystems on the continental scale. Proc. Natl. Acad. Sci. 111, 9549–9554 (2014)
- R. Altenburger, H. Walter, M. Grote, What contributes to the combined effect of a complex mixture? Environ. Sci. Technol. 38, 6353–6362 (2004)
- R. Altenburger, S. Ait-Aissa, P. Antczak, T. Backhaus, D. Barceló, T.-B. Seiler, F. Brion, W. Busch, K. Chipman, M.L. de Alda, Future water quality monitoring—adapting tools to deal with mixtures of pollutants in water resource management. Sci. Total Environ. 512, 540–551 (2015)
- 4. G.T. Ankley, S.W. Edwards, The adverse outcome pathway: a multifaceted framework supporting 21st century toxicology. Curr. Opin. Toxicol. 9, 1–7 (2018)
- 5. D. Schlenk, Adverse outcome pathways in ecotoxicology, in *An Introduction to Interdisciplinary Toxicology*, (Elsevier, 2020), pp. 569–579
- 6. J. Jeong, J. Choi, Use of adverse outcome pathways in chemical toxicity testing: potential advantages and limitations. Environ. Health Toxicol. **33** (2018)
- D.S. Kang, J.H. Yang, H.S. Kim, B.K. Koo, C.M. Lee, Y.-S. Ahn, J.-H. Jung, Y.R. Seo, Application of the adverse outcome pathway framework to risk assessment for predicting carcinogenicity of chemicals. J. Cancer Prev. 23, 126 (2018)
- 8. M. Vinken, The adverse outcome pathway concept: a pragmatic tool in toxicology. Toxicology **312**, 158–165 (2013)
- R. Altenburger, M. Scholze, W. Busch, B.I. Escher, G. Jakobs, M. Krauss, J. Krüger, P.A. Neale, S. Ait-Aissa, A.C. Almeida, Mixture effects in samples of multiple contaminants– an inter-laboratory study with manifold bioassays. Environ. Int. 114, 95–106 (2018)
- J.W. Lee, E.-J. Won, S. Raisuddin, J.-S. Lee, Significance of adverse outcome pathways in biomarker-based environmental risk assessment in aquatic organisms. J. Environ. Sci. 35, 115–127 (2015)

- M. Leist, A. Ghallab, R. Graepel, R. Marchan, R. Hassan, S.H. Bennekou, A. Limonciel, M. Vinken, S. Schildknecht, T. Waldmann, Adverse outcome pathways: opportunities, limitations and open questions. Arch. Toxicol. **91**, 3477–3505 (2017)
- T.E. Allen, J.M. Goodman, S. Gutsell, P.J. Russell, Defining molecular initiating events in the adverse outcome pathway framework for risk assessment. Chem. Res. Toxicol. 27, 2100–2112 (2014)
- P.R. Gentry, H.J. Clewell III, R. Clewell, J. Campbell, C. Van Landingham, A.M. Shipp, Challenges in the application of quantitative approaches in risk assessment: a case study with di-(2-ethylhexyl) phthalate. Crit. Rev. Toxicol. 41, 1–72 (2011)
- 14. R.A. Clewell, J.A. Leonard, C.I. Nicolas, J. Campbell, M. Yoon, A.Y. Efremenko, P.D. McMullen, M.E. Andersen, H.J. Clewell III, K.A. Phillips, Application of a combined aggregate exposure pathway and adverse outcome pathway (AEP-AOP) approach to inform a cumulative risk assessment: a case study with phthalates. Toxicol. In Vitro **104855** (2020)
- T.H. Hutchinson, B.P. Lyons, J.E. Thain, R.J. Law, Evaluating legacy contaminants and emerging chemicals in marine environments using adverse outcome pathways and biological effects-directed analysis. Mar. Pollut. Bull. 74, 517–525 (2013)
- V.J. Kramer, M.A. Etterson, M. Hecker, C.A. Murphy, G. Roesijadi, D.J. Spade, J.A. Spromberg, M. Wang, G.T. Ankley, Adverse outcome pathways and ecological risk assessment: bridging to population-level effects. Environ. Toxicol. Chem. 30, 64–76 (2011)
- WFD, Directive 2000/60/EC, in: D.E.o.t.E.P.a.o.t.C.o.J.E. Communities (Ed.), Council of J. Eur. Communities 2000, pp. 1–73
- 18. S. Bhattacharya, A test for mutagenicity of methylcholanthrene. Nature 162, 573–573 (1948)
- D. Basketter, H. Clewell, I. Kimber, A. Rossi, B. Blaauboer, R. Burrier, M. Daneshian, C. Eskes, A. Goldberg, N. Hasiwa, A roadmap for the development of alternative (non-animal) methods for systemic toxicity testing (2012)
- 20. A. Limonciel, L. Aschauer, A. Wilmes, S. Prajczer, M.O. Leonard, W. Pfaller, P. Jennings, Lactate is an ideal non-invasive marker for evaluating temporal alterations in cell stress and toxicity in repeat dose testing regimes. Toxicol. In Vitro 25, 1855–1862 (2011)
- 21. E. Rempel, L. Hoelting, T. Waldmann, N.V. Balmer, S. Schildknecht, M. Grinberg, J.A.D. Gaspar, V. Shinde, R. Stöber, R. Marchan, A transcriptome-based classifier to identify developmental toxicants by stem cell testing: design, validation and optimization for histone deacetylase inhibitors. Arch. Toxicol. 89, 1599–1618 (2015)
- A.K. Krug, S. Gutbier, L. Zhao, D. Pöltl, C. Kullmann, V. Ivanova, S. Förster, S. Jagtap, J. Meiser, G. Leparc, Transcriptional and metabolic adaptation of human neurons to the mitochondrial toxicant MPP+. Cell Death Dis. 5, e1222–e1222 (2014)
- A. Bal-Price, P.J. Lein, K.P. Keil, S. Sethi, T. Shafer, M. Barenys, E. Fritsche, M. Sachana, M.B. Meek, Developing and applying the adverse outcome pathway concept for understanding and predicting neurotoxicity. Neurotoxicology 59, 240–255 (2017)
- 24. K.J. Groh, R.N. Carvalho, J.K. Chipman, N.D. Denslow, M. Halder, C.A. Murphy, D. Roelofs, A. Rolaki, K. Schirmer, K.H. Watanabe, Development and application of the adverse outcome pathway framework for understanding and predicting chronic toxicity: II. A focus on growth impairment in fish. Chemosphere **120**, 778–792 (2015)
- J.P. Laurenson, R.A. Bloom, S. Page, N. Sadrieh, Ethinyl estradiol and other human pharmaceutical estrogens in the aquatic environment: a review of recent risk assessment data. AAPS J. 16, 299–310 (2014)
- 26. J. Jeong, J. Choi, Adverse outcome pathways potentially related to hazard identification of microplastics based on toxicity mechanisms. Chemosphere 231, 249–255 (2019)
- D.H. Miller, J.E. Tietge, M.E. McMaster, K.R. Munkittrick, X. Xia, G. Ankley, D.A. Griesmer, Linking mechanistic toxicology to population models in forecasting recovery from chemical stress: a case study from Jackfish Bay, Ontario, Canada. Environ. Toxicol. Chem. 34, 1623–1633 (2015)
- S.R. Corsi, L.A. De Cicco, D.L. Villeneuve, B.R. Blackwell, K.A. Fay, G.T. Ankley, A.K. Baldwin, Prioritizing chemicals of ecological concern in Great Lakes tributaries using

high-throughput screening data and adverse outcome pathways. Sci. Total Environ. 686, 995–1009 (2019)

- D. Schlenk, R. Lavado, J.E. Loyo-Rosales, W. Jones, L. Maryoung, N. Riar, I. Werner, D. Sedlak, Reconstitution studies of pesticides and surfactants exploring the cause of estrogenic activity observed in surface waters of the San Francisco Bay Delta. Environ. Sci. Technol. 46, 9106–9111 (2012)
- 30. J. Crago, K. Tran, A. Budicin, B. Schreiber, R. Lavado, D. Schlenk, Exploring the impacts of two separate mixtures of pesticide and surfactants on estrogenic activity in male Fathead Minnows and Rainbow Trout. Arch. Environ. Contam. Toxicol. 68, 362–370 (2015)
- D.L. Villeneuve, M.M. Angrish, M.C. Fortin, I. Katsiadaki, M. Leonard, L. Margiotta-Casaluci, S. Munn, J.M. O'brien, N.L. Pollesch, L.C. Smith, Adverse outcome pathway networks II: network analytics. Environ. Toxicol. Chem. 37, 1734–1748 (2018)
- 32. OECD, Adverse Outcome Pathway Knowledge Base, in Paris, France (2019)
- 33. E.J. Perkins, R. Ashauer, L. Burgoon, R. Conolly, B. Landesmann, C. Mackay, C.A. Murphy, N. Pollesch, J.R. Wheeler, A. Zupanic, Building and applying quantitative adverse outcome pathway models for chemical hazard and risk assessment. Environ. Toxicol. Chem. 38, 1850–1865 (2019)
- 34. B. Buckley, A. Farraj, Conceptual model for assessing criteria air pollutants in a multipollutant context: a modified adverse outcome pathway approach. Toxicology **335**, 85–94 (2015)

Chapter 19 Invasive Alien Species Problem in the Reservoir



Marriya Sultan and De-Sheng Pei 💿

Abstract This chapter discusses the issue of invasive alien species in the Three Gorges Reservoir (TGR), highlights different natural and anthropogenic variables in the propagation of different exotic plants and fish species, and assesses their impacts on the reservoir ecosystem. After the construction of the TGR, many exotic plant and fish species made their way to the reservoir region; most of the invasive species were the result of factors, such as impoundments, international trade *via* water, road access, and recreational activities like aquaculture. Besides, these factors including natural climatic factors also influenced the proliferation of exotic species. The most common and problematic exotic plant and fish species in the reservoir were water hyacinth (*Eichhornia crassipes*) and red swamp crayfish, which could result in the deterioration of water quality, impede the growth of native species, influence the food chain, and cause geological hazards. Hence, we should properly monitor the reservoir threats caused by invasive species and the conservation of native species.

Keywords Three Gorges Reservoir (TGR) \cdot Invasive alien species \cdot Exotic plant species \cdot Reservoir ecosystem \cdot Species conservation

1 Introduction

On a national and international scale, the Three Gorges Reservoir (TGR) region is recognized as one of the most important biodiversity hotspots in China [1, 2]. The unique biodiversity is accredited to its distinct geographical locality, climate, and topography. The flora of the TGR consists of a range of species that evolved from temperate, tropical, and subtropical zones, whereas the fauna of this region is characterized by both aquatic and terrestrial animals [3]. The region is home to about 6400 plant species with 20% of the overall seed plants in China. Moreover, the vegetation of this area also includes about 36 endemic species and various rare and ancient species [1]. This area also contains 3400 species of insects and around 500 terrestrial vertebrate species, making up 8.5% and 22% of China's total insects and vertebrates, respectively [4].

The Three Gorges Dam (TGD) and the adjoining region of the reservoir with an overall area of around 58,000 km² are recognized as the TGR region [4]. Since the

completion of its construction in 2009, various environmental problems have been identified in the region [5]. The problem of alien species invasion became a threat to the native biodiversity of the TGR region [6]. Biological invasions are a threat to biodiversity and ecosystem functions that further inflict excessive economic and social damage [7]. The drawdown zone and riparian zone areas in the TGR are more susceptible to invasion, because they are exposed to the changes caused by human activities. The hydrological alterations assist the transport of non-native invaders, which later colonize the area and offer strong competition to the native species for their survival [1].

According to the report by Xiong et al. [6], about 42 non-native species (one alga, one crayfish, one mollusk, one frog, one freshwater turtle, 23 fishes, and 14 higher plants) were observed in the TGR region, among which 15 species originate from North American, 13 from Asia, 7 from Central and South America, 5 from Europe, and 2 from African origin. Further, the non-native aquatic species observed in the TGR region was 3.5 species year⁻¹ from the year 2003–2015 [8].

In this chapter, we focus on the spread of non-native (exotic) flora and fauna around the TGR region and elucidate their effect on the reservoir ecosystem and management strategies of alien species in the TGR.

2 Impact of Local and Landscape Variables on Exotic Species Invasion

Natural, landscape, and dam operational factors significantly influence the type of flora and fauna of the region surrounded or inundated by the reservoir. Local and landscape variables and dam operations-related factors result in the reduction of fish productivity, the discharge of toxic contaminants into the water, the initiation of sediment build-up, and the alteration of downstream habitats, such as loss of floodplains, river deltas, and riparian zones [9-11]. Human impact is also considered an imperative aspect, as the rivers flow through developed agricultural and populated areas surrounding the TGR region with around 40% cultivated land. The land characterized by patchy and disturbed natural parts is an increased occurrence of exotic species [5]. The aquatic species of the natural and regular flow patterns are disrupted by dam operations. The spawning grounds are also blocked, and aquatic populations are fragmented [3]. As a result, alien species successfully invade the vacant ecological niches. According to a previous study [12], 23 invasive fish species were found in the upper reaches of the Yangtze River and the main reservoir area. The population of some species is likely to increase, when some of them are in the population outbreak stage. The reason for their high reproduction is high nutrient inputs and increased primary productivity after the impoundment. The variables inducing the alien species invasion in the TGR region are presented in Fig. 19.1.

The TGD area was a closed region and had restricted the size of the waterway, therefore, ships had no such access to the water before the construction of the dam



Fig. 19.1 Variables influencing the exotic species invasion

[13]. Thus, non-native aquatic species were not observed in the TGD region. For the opening of ship locks for navigation purposes, many exotic species of various origins, such as North America, Central America, Europe, Asia (95% species), and Africa (5%), were reported in the reservoir region [8]. Among the local variables, international trade is an important vector that may escalate the number of non-exotic species. Due to an increasing number of vessels passing through the dam and carrying the imported goods from different continents to Chongqing, the number of non-native species may increase in the TGD region [6]. Several dam operation and management factors, such as impoundments, hydropower generation, and water flow regimes, also influence the introduction of exotic species in the reservoir, because they cause human disturbances, which may decrease the population of native species, thus providing an opportunity for non-native species to invade

[14]. Reservoir flooding and water flow regimes are also important sources of dispersion of exotic species, as water fluctuation is linked with the spread of exotic species. In the TGR region, when the water level is pushed up to 20 m or above from October to January, many exotic species from adjacent uplands are entrained and moved to drawdown areas. Similarly, during summer when the water recedes, nutrient content increases sustaining the decomposition process and reassuring the invasion of exotic species in the absence of competitors [5]. Besides, hydropower projects require road facilities, which act as an important vector for the spread of non-natives in the inland regions of the dam. Moreover, reservoirs linked to the roads had human access that likely increases recreational activities, such as sightseeing [14]. According to the earlier studies, regional and landscape factors are also considered better predictors of the community structure, compared to the local scale variables. In the vicinity of the TGR, agricultural activities were not allowed in the area between 175-275 m, and the former agricultural land was transformed into an artificial forest, in which the weeds colonized and thus exotic species invaded this area [5]. Habitat fragmentation because of flooding can lead to the formation of isolated islands, where the native species may decline, and exotic species make their way and establish their colonies [3, 15]. Natural conditions, such as cold-water environments, prevent the entrainment of non-native (exotic) fishes [16]. However, under the scenario of increasing temperature because of global warming, the establishment of non-native species could be enhanced in these regions that were formerly not suitable for their survival [14].

3 Exotic Plant Species in the TGR Region

Various exotic plant species were observed near the dam wall after the completion of the TGD project [17]. The most common was the South American native alien fleabane *Conyza sumatrensis* (*Asteraceae*) [18], an annual or biennial tall herbaceous plant, which reproduces by self-pollination. Because of high seed production, it is widely spread around the vicinity of the dam [15]. In the drawdown zone of TGD, plant community structure is also evolving, and the most troublesome alien plant species were *Eupatorium adenophorum* and *Alternanthera philoxeroides* [19]. Similarly, the exotic annual weed *Echinochloa crus-galli* had now become prevalent in many parts of the drawdown area [20]. According to a previous report [21], more or less 145 exotic plant species were found in the post-dam foliage in the Yangtze River basin between Chongqing and Yichang City, but only a few of them could form the dominant plant populations, such as *Alternanthera philoxeroides*, *Aster subulatus, Erigeron acer, Phytolacca Americana*, and *Xanthium sibiricum*.

In the Pengxi River area of the TGR region, about 186 plant species were detected, of which 21 were identified as invasive herbs and the most common were *Bidens pilosa*, *Conyza canadensis*, *Setaria viridis*, and *Xanthium sibericum*. Besides these *Chenopodium ambrosioides*, *Pistia stratiotes*, *Talinum paniculatum*, and *Canna indica* were found below 175 m or above 175 m. Some of the species

were found to spread in patches with less coverage like *Alternanthera philoxeroides*. *Conyza canadensis* was the most common species in the TGR area, whereas some of the species were rare, such as *Pistia stratiotes* with one or two occurrences in this area [5]. For restoration and reconstruction of vegetation, some aquatic plants were also introduced in the water level fluctuation zone of the TGD region. It also included non-native plants, such as *Chrysopogon zizanioides*. Moreover, *Canna indica, Cyperus alternifolius*, and *Elodea nuttallii* were also suggested to be introduced for ecological restoration [6]. In the Yangtze River of the TGD, more or less 24 non-native species were found with 5 species being observed in point bars and 13 and 4 in mid-channel bars of the downstream region. The non-native species occupied the areas that were disrupted by humans. Thus, the mid-channel bars had a high risk of species invasion, and the species richness of native plants was also less because of reachability *via* water flow and wind seed plants. Consequently, the number of short-lived herbs proliferated in this region [22]. A list of some common alien plant species around the vicinity of the TGR region is shown in Table 19.1.

4 Exotic Fish Species Along the TGR Region

According to a previous report [23], the biological invasion has put 27.7% of fish species at risk in the Yangtze River. Around eighteen exotic fish species belonging to 10 families were identified in the upper Yangtze River, of which *Ictalurus punctatus, Protosalanx hyalocranius, Megalobrama amblycephala*, and *tilapia* were wide-spread in the TGR. These exotic fish records were linked with the impoundment of the TGR, because no exotic fish species was reported in Zigui. The wide-spread species in the lacustrine zone were *Ameiurus melas, Cirrhinus molitorella, Hemisalanx brachyrostralis,* and *Polyodon spathula* [24]. According to the previous report [25], alien species, such *as Ameiurus melas, Tinca tinca,* and *Ictalurus punctatus,* were observed in the TGR region. Moreover, in the Wanzhou section of the Yangtze River, exotic piscivorous species, such as largemouth bass (*Micropterus salmoides*), were also observed.

Exotic fish including *Gambusia affinis* (western mosquitofish), *Megalobrama amblycephala* (bluntnose black bream), and *Pseudorasbora parva* (topmouth gudgeon) had invaded the Yangtze River. Among these fish species, the western mosquitofish (*Gambusia affinis*) population is more in the central Yangtze River because of its short life span, faster growth, and lower fecundity, which affects the presence of native species like medaka (*Oryzias latipes*) [26]. Non-native fish species are also introduced into the dams via aquaculture to increase the import of aquatic fish [27]. Non-native fish species like *Micropterus salmoides*, *Clarias batrachus*, and *Ictalurus punctatus* were introduced into the TGR to improve fisheries. Similarly, fish species, such as *Gambusia affinis* and *Oreochromis niloticus* in the TGR, may explain the reason for the decline of some endemic fish species [6]. As shown in Table 19.2, some of the exotic fish species were observed in the TGR region.

Location	Exotic Specie	Reference
Three Gorges Dam	Conyza sumatrensis (Asteraceae)	[15]
Drawdown zone of the TGR	Eupatorium adenophorum	[19]
	Alternanthera philoxeroides	
Yangtze River basin	Alternanthera philoxeroides	[21]
Chongqing and Yichang City	Aster subulatus	
	Erigeron acer	
	Phytolacca Americana	
	Xanthium sibiricum	
Drawdown zone of the TGR	Echinochloa crus-galli	[20]
Pengxi River	Bidens pilosa	[5]
	Conyza Canadensis	
	Setaria viridis	
	Xanthium sibiricum.	
	Chenopodium ambrosioides,	
	Pistia stratiotes	
	Talinum paniculatum	
	Oxalis corymbosa	
	Alternanthera philoxeroides	
	Crassocephalum crepidioides	
	Equisetum ramosissimum	
	Euphorbia maculate	
	Solanum aculeatissimum	
	Sonchus oleraceus	
	Veronica polita	
	Abutilon theophrasti	
	Senecio scandens	
	Amaranthus retroflexus	
	Euphorbia hirta	
	Canna indica	
The TGR region	E. crassipes,	[6]
	A. philoxeroides	
	Chrysopogon zizanioides	
	Elodea nuttallii,	
	Canna indica	
	Cyperus alternifolius	
Yangtze river	Salix variegata Franch.	[22]
	Myricaria laxiflora	
	Ranunculus sceleratus	

Table 19.1 Common non-native plant species found in the TGR region

5 Effect of Exotic Species on the Reservoir Ecosystem

The invasions of some exotic species have worsened the quality of water in the TGD region. Water hyacinth (*Eichhornia crassipes*), the foulest invasive aquatic weed has become widely dispersed in the TGD region forming dense coverings on the surface water [18]. It inhibits air diffusion into the water and reduces the dissolved oxygen content in water, resulting in hypolimnion conditions. As we know, the organic

Location	Exotic Specie	Reference
Yangtze River	Protosalanx hyalocranius	[24]
	Salangichthys tangkahkeii	
	Hemisalanx brachyrostralis	
	Neosalanx taihuensis	
	Acipenser schrenckii	
	Polyodon spathula	
	Huso dauricus,	
	Acipenser schrenckii	
	Tinca tinca	
	Megalobrama amblycephala	
	Cirrhinus molitorella	
	Ietalurus punetaus	
	Micropterus salmoides	
	tilapia	
	Lucioperca lucioperca	
	Colossoma brachypomus	
	Ameiurus melas	
	Gambusia affinis	
Yangtze River	Ameiurus melas,	[25]
	Tinca tinca,	
	Ictalurus punctatus	
The TGR region	Gambusia affinis	[6]
	Oreochromis niloticus	
	Micropterus salmoides	
	Claria batrachus	
	Ictalurus punctatus	
Yangtze River	Gambusia affinis (western mosquitofish)	[26]
-	Megalobrama amblycephala (bluntnose black bream)	
	Pseudorasbora parva (topmouth gudgeon)	

Table 19.2 Common non-native fish species in the TGR region

matter content is increased and water quality is deteriorated [28]. Exotic fishes also cause the displacement of native fishes, leading to interspecific competition and the spread of pathogens [26]. Besides, the introduction of red swamp crayfish (*Procambarus clarkii*) into the TGD region has resulted in its widespread in the watershed [29]. During the reproductive period of crayfish, intense burrowing is carried out to prevent desiccation, which could result in the collapse of the bank [6].

6 Management of Alien Species and Implications for the Reservoir Conservation

To cope with new invasions, ecological monitoring is required to improve [30]. In the TGR region, such efforts related to biodiversity conservation and tackling the issue of threats by alien species invasion have been not strongly fulfilled. To determine the dispersal patterns and spatial distribution of the alien species, a landscape-level monitoring system is needed [5]. The regional native species are required to be conserved and enriched by establishing nature reserves, national parks, and gene banks adjacent to the reservoir, so that the movement of invasive species is impeded. Target planting of native species should be carried out in the towns bordering the reservoir and stringent actions should be taken regarding the plantation of exotic species [1]. In narrow corridors, higher proportions of alien plants are observed. When the edge effect is at maximum with the corridor perimeter to area ratio, then the possibility of invasion is increased. Thus, the broad corridor can function as an effective "buffer," and this barrier must be maintained and established with native species [1]. For controlling the invasive alien species, mechanical, biological, and physical methods have been adopted, but the effects were not as much successful as predicted. Besides, mechanical and physical methods are also costly and chemical methods sometimes cause the problem of water pollution, but biological agents for controlling the alien species could threaten the non-target species [6].

7 Conclusion

The environmental changes in the TGR could directly or indirectly affect the reservoir ecosystem and the diversity of flora and fauna. Natural, local, landscape, and dam management-related variables influence the exotic species invasion in the TGR. In literature, very limited research related to the spread of exotic in the TGR region and different habitats of alien species has been documented. However, the available literature has addressed the negative impacts of invasive species on water quality, native species richness, and the spread of pathogens. Therefore, it is vital to pay attention to the spread of non-native alien species in the reservoir region with a special focus on monitoring and management programs for reservoir water quality and endemic species conservation.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014), Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714), and the CAS-TWAS Scholarship (No. 2019A8018537003 to M.S).

Conflicts of Interest The author declares no conflict of interest.

References

- 1. C. Chen, C.D. Meurk, S. Wu, The drawdown zone of the Three Gorges Reservoir: a high risk corridor for species invasion in China? Acta Ecol. Sin. **36**, 36–38 (2016)
- C. Huang, D. Zhao, L. Deng, Landscape pattern simulation for ecosystem service value regulation of Three Gorges Reservoir Area, China. Environ. Impact Assess. Rev. 95, 106798 (2022)

- J. Wu, J. Huang, X. Han, X. Gao, F. He, M. Jiang, Z. Jiang, R.B. Primack, Z. Shen, The three gorges dam: an ecological perspective. Front. Ecol. Environ. 2, 241–248 (2004)
- Q. Zhang, Z. Lou, The environmental changes and mitigation actions in the Three Gorges Reservoir region, China. Environ. Sci. Pol. 14, 1132–1138 (2011)
- C. Chen, S. Wu, C.D. Meurk, M. Ma, J. Zhao, X. Tong, Effects of local and landscape factors on exotic vegetation in the riparian zone of a regulated river: implications for reservoir conservation. Landsc. Urban Plan. 157, 45–55 (2017)
- W. Xiong, H. Wang, Q. Wang, J. Tang, P.A. Bowler, D. Xie, L. Pan, Z. Wang, Non-native species in the Three Gorges Dam Reservoir: status and risks. BioInvasions Record 7 (2018)
- D. Pimental, R. Zuniga, D. Morrison, Update on the environmental and economic costs associated with alien-invasive species in the United States. Ecol. Econ. 52 (2005)
- 8. H. Xu, K. Chen, Z. Ouyang, X. Pan, S. Zhu, Threats of invasive species for china caused by expanding international trade. Environ. Sci. Technol. **46**, 7063–7064 (2012)
- D.M. Rosenberg, P. McCully, C.M. Pringle, Global-scale environmental effects of hydrological alterations: introduction. Bioscience 50, 746–751 (2000)
- I. Sîrbu, A.-M. Benedek, B.L. Brown, M. Sîrbu, Disentangling structural and functional responses of native versus alien communities by canonical ordination analyses and variation partitioning with multiple matrices. Sci. Rep. 12, 12813 (2022)
- B.-Y. Wang, D.-C. Yan, A.-B. Wen, J.-C. Chen, Influencing factors of sediment deposition and their spatial variability in riparian zone of the Three Gorges Reservoir, China. J. Mt. Sci. 13, 1387–1396 (2016)
- 12. B. Jiawen, C. Daqing, Invasive fishes in Three Gorges Reservoir area and preliminary study on effects of fish invasion owing to impoundment [J]. J. Lake Sci. 2 (2012)
- W. Xiao, J. Li, C. Yu, J. Ma, R. Cheng, S. Liu, J. Wang, J. Ge, *Terrestrial animal and plant ecology of the Three Gorges of Yangtze River* (Southwest China Normal University Press, Chongqing, 2000)
- H. Sun, D. He, X. Sui, Y. Chen, Predicting impacts of future climate change and hydropower development towards habitats of native and non-native fishes. Sci. Total Environ. **707**, 135419 (2020)
- M.-X. Ren, X.-Q. Li, J.-Q. Ding, Genetic variation and spread pattern of invasive Conyza sumatrensis around China's Three Gorges Dam. Acta Oecol. 36, 599–603 (2010)
- F.J. Rahel, J.D. Olden, Assessing the effects of climate change on aquatic invasive species. Conserv. Biol. 22, 521–533 (2008)
- Y. Wang, Y. Liu, M. Ma, Z. Ding, S. Wu, W. Jia, Q. Chen, X. Yi, J. Zhang, X. Li, G. Luo, J. Huang, Dam-induced difference of invasive plant species distribution along the riparian habitats. Sci. Total Environ. 808, 152103 (2022)
- J. Ding, R.N. Mack, P. Lu, M. Ren, H. Huang, China's booming economy is sparking and accelerating biological invasions. Bioscience 58, 317–324 (2008)
- Z.-J. Lu, L.-F. Li, M.-X. Jiang, H.-D. Huang, D.-C. Bao, Can the soil seed bank contribute to revegetation of the drawdown zone in the Three Gorges Reservoir Region? Plant Ecol. 209, 153–165 (2010)
- J.M. Willison, R. Li, X. Yuan, Conservation and ecofriendly utilization of wetlands associated with the Three Gorges Reservoir. Environ. Sci. Pollut. Res. 20, 6907–6916 (2013)
- F. Yang, W.-W. Liu, J. Wang, L. Liao, Y. Wang, Riparian vegetation's responses to the new hydrological regimes from the Three Gorges Project: clues to revegetation in reservoir waterlevel-fluctuation zone. Acta Ecol. Sin. 32, 89–98 (2012)
- 22. X. Yi, Y. Huang, M. Ma, Z. Wen, J. Chen, C. Chen, S. Wu, Plant trait-based analysis reveals greater focus needed for mid-channel bar downstream from the Three Gorges Dam of the Yangtze River. Ecol. Indic. **111**, 105950 (2020)
- C. Ye, S. Li, Y. Zhang, Q. Zhang, Assessing soil heavy metal pollution in the water-levelfluctuation zone of the Three Gorges Reservoir, China. J. Hazard. Mater. 191, 366–372 (2011)
- 24. C. Peng, L. Zhang, H. Qin, D. Li, Revegetation in the water level fluctuation zone of a reservoir: an ideal measure to reduce the input of nutrients and sediment. Ecol. Eng. 71, 574–577 (2014)

- 25. H. Perera, L. Zhong-Jie, S. De Silva, Z. Tang-Lin, Y. Jing, Y. Shao-Wen, X. Yu-Guo, L. Jia-Shou, Effect of the distance from the dam on river fish community structure and compositional trends, with reference to the Three Gorges Dam, Yangtze River, China. Acta Hydrobiologica Sinica 38 (2014)
- 26. B. Kang, X. Huang, Y. Yan, Y. Yan, H. Lin, Continental-scale analysis of taxonomic and functional fish diversity in the Yangtze river. Global Ecol. Conserv. **15**, e00442 (2018)
- 27. S.S. De Silva, Aquaculture: a newly emergent food production sector—and perspectives of its impacts on biodiversity and conservation. Biodivers. Conserv. **21**, 3187–3220 (2012)
- 28. H. Wang, Q. Wang, P.A. Bowler, W. Xiong, Invasive aquatic plants in China. Aquat. Invasions 11 (2016)
- 29. T. New, Z. Xie, Impacts of large dams on riparian vegetation: applying global experience to the case of China's Three Gorges Dam. Biodivers. Conserv. **17**, 3149–3163 (2008)
- H.P. Rauch, M. von der Thannen, P. Raymond, E. Mira, A. Evette, Ecological challenges* for the use of soil and water bioengineering techniques in river and coastal engineering projects. Ecol. Eng. 176, 106539 (2022)

Part V Conclusions

Chapter 20 Final Thoughts and Concluding Remarks



231

De-Sheng Pei 💿 and Naima Hamid

Abstract Although reservoir ecotoxicology as a modern scientific discipline is proposed by us, there are many ecotoxicology issues in the reservoir that deserve thorough consideration. In several chapters of this book, we take the Three Gorges Reservoir as an example to describe the reservoir ecosystem (Chap. 2), distribution of pollutants (Chap. 3), migration of pollutants (Chap. 4), and microorganisms diversity (Chap. 7), etc., implying that different patterns may occur in other types of reservoir ecotoxicology. Taken together, in this book, we highlight the environmental characteristics in the reservoir system, discuss the study methods that are used to elucidate the ecotoxicology in the reservoir, and explain the detailed ecotoxicological effects and molecular mechanism of pollutants in the reservoir.

Keywords Reservoir ecotoxicology · Pollution load · Toxicity assessment · Toxicological effects · Mechanistic genetic toxicity

1 Summarized Viewpoint and Conclusion

Water resources are very precious and vital for human life. To protect and use the water efficiently, water reservoir plays an important role in both natural and artificial environment [1]. Over the past years, many reservoirs have been created primarily due to a high-water requirement for irrigation and energy demand [2]. Thus, reservoir ecotoxicology is an emerging academic discipline to protect reservoir native flora and fauna at the population, community, and ecosystem levels. We propose this discipline to appeal to more researchers to participate in this subject, because reservoir ecotoxicology is a multidisciplinary study, including ecology, ecotoxicology, environmental science, environmental hygiene, reservoir management, and water resources engineering. Although the Three Gorges reservoir in China is not the largest in the world, the Three Gorges Dam is the world's largest hydroelectric facility now [3]. Thus, the research results of the Three Gorges reservoir may be instructive for reservoir ecological health and environmental safety.

Among different types of reservoirs in the world, human anthropogenic disturbances affect the water quality of the reservoir so badly that it causes changes in the

[©] The Author(s), under exclusive license to Springer Nature Switzerland AG 2023 D.-S. Pei et al., *Reservoir Ecotoxicology*, https://doi.org/10.1007/978-3-031-26344-6_20

species composition of flora and fauna [4]. Different environmental stressors, including environmental pollutants, hydrologic alterations, sedimentation flux, nutrient loading, and erosion, also affected the ecological characteristics of the reservoir.

Various sources and distribution of environmental contaminants polluted the reservoir water. Heavy metals were mostly mainly reported among all the pollutants [5–7]. Similarly, polycyclic aromatic hydrocarbons (PAHs) distribution was attributed to the area's extensive pyrogenic and petrogenic combustion activities. The pollution of phthalate esters (PAEs) should be considered in most reservoirs because their levels exceeded the prescribed limits by World Health Organization (WHO) and the United States Environmental Protection Agency (US EPA) with high ecological risks [8, 9]. Pharmaceuticals and personal care products (PPCPs) and microplastic and nanoplastic pollution as emerging pollutants in the reservoirs attract more attention worldwide [10, 11]. Reservoir water-level fluctuation zone (WLFZ) increased the complexity of contaminant migration and conversion.

Ecological risk assessment using animal models is a vital approach to checking reservoir ecotoxicity levels. Many animal models, including zebrafish, mice, medaka, *Caenorhabditis elegans*, *Carassius auratus*, and *Daphnia magna*, can be used for elucidating the molecular mechanism of reservoir ecotoxicology. Moreover, bioconcentration and bioaccumulation toxicity tests are imperative, because bioconcentration can reveal the accumulation of a water-borne chemical by an aquatic organism [12]. However, bioaccumulation explains the uptake from all environmental sources including water, food, and sediment [13]. A simple aquatic food chain (green algae-*Daphnia magna*-fish) could provide insights into possible human health risks of chemicals at environmental relevant concentrations (ERCs) [14].

Detailed reservoir ecotoxicological effects of pollutants and their genetic mechanisms are fundamental for identifying their health risks. Elevated levels of chemicals in the ecosystem can result in adverse impacts on wildlife. Previous studies showed that heavy metals might induce oxidative stress in freshwater species [15, 16]. However, parts of persistent organic pollutants (POPs) could activate the aryl hydrocarbon receptor (AhR) signaling pathway [17], cause neurodevelopmental disorders [18], and possess endocrine disruption properties [19]. Besides, many PPCPs also exhibited endocrine disruption properties with elevated apoptosis levels and DNA damage [20]. For reservoir ecotoxicity study, the adverse outcome pathway (AOP) also is a vital approach for assessing the negative effects of reservoir pollutants at different levels [21].

With the development of reservoir ecotoxicity, the government will profoundly understand the toxicological mechanism of chemical mixture and formulate effective measures to decrease the pollution load in the reservoir.

Acknowledgments This work was supported by the funds from the High-level Talents Project of Chongqing Medical University (No. R4014) and Research Program of Chongqing Science and Technology Commission (No. cstc2019jcyj-zdxmX0035 and CSTCCXLJRC201714).

Conflicts of Interest The authors declare no conflict of interest.

References

- N.M. Rana, N. Ghahramani, S.G. Evans, A. Small, N. Skermer, S. McDougall, W.A. Take, Global magnitude-frequency statistics of the failures and impacts of large water-retention dams and mine tailings impoundments. Earth Sci. Rev. 232, 104144 (2022)
- M. Zhao, Y. Liu, Y. Wang, Y. Chen, W. Ding, *Effectiveness assessment of reservoir projects for flash flood control, water supply and irrigation in Wangmo Basin* (Science of The Total Environment, China, 2022), p. 157918
- X. Liu, Q. Xu, Hydropeaking impacts on riverine plants downstream from the world's largest hydropower dam, the Three Gorges Dam. Sci. Total Environ. 845, 157137 (2022)
- 4. R.M.B. Santos, L.F. Sanches Fernandes, R.M.V. Cortes, S.G.P. Varandas, J.J.B. Jesus, F.A.L. Pacheco, Integrative assessment of river damming impacts on aquatic fauna in a Portuguese reservoir. Sci. Total Environ. 601–602, 1108–1118 (2017)
- X. Li, J. Bing, J. Zhang, L. Guo, Z. Deng, D. Wang, L. Liu, Ecological risk assessment and sources identification of heavy metals in surface sediments of a river-reservoir system. Sci. Total Environ. 842, 156683 (2022)
- L.W. Ngatia, L.M. De Oliveira, O.C. Betiku, R. Fu, D.N. Moriasi, J.L. Steiner, J.A. Verser, R.W. Taylor, Relationship of arsenic and chromium availability with carbon functional groups, aluminum and iron in Little Washita River Experimental Watershed Reservoirs, Oklahoma, USA. Ecotoxicol. Environ. Saf. 207, 111468 (2021)
- S. Dhanakumar, G. Solaraj, R. Mohanraj, Heavy metal partitioning in sediments and bioaccumulation in commercial fish species of three major reservoirs of river Cauvery delta region, India. Ecotoxicol. Environ. Saf. 113, 145–151 (2015)
- N.N. Patil, E. Arun, K. Vimalkumar, B.R. Ramaswamy, Phthalate esters and their ecotoxicological risks from the rivers of India, in *Organic pollutants: toxicity and solutions*, ed. by M. Vasanthy, V. Sivasankar, T.G. Sunitha, (Springer, Cham, 2022), pp. 153–176
- L. Lin, L. Dong, X. Meng, Q. Li, Z. Huang, C. Li, R. Li, W. Yang, J. Crittenden, Distribution and sources of polycyclic aromatic hydrocarbons and phthalic acid esters in water and surface sediment from the Three Gorges Reservoir. J. Environ. Sci. 69, 271–280 (2018)
- X. Peng, W. Ou, C. Wang, Z. Wang, Q. Huang, J. Jin, J. Tan, Occurrence and ecological potential of pharmaceuticals and personal care products in groundwater and reservoirs in the vicinity of municipal landfills in China. Sci. Total Environ. 490, 889–898 (2014)
- M. Di, J. Wang, Microplastics in surface waters and sediments of the Three Gorges Reservoir, China. Sci. Total Environ. 616-617, 1620–1627 (2018)
- I.A. Duarte, J. Fick, H.N. Cabral, V.F. Fonseca, Bioconcentration of neuroactive pharmaceuticals in fish: Relation to lipophilicity, experimental design and toxicity in the aquatic environment. Sci. Total Environ. 812, 152543 (2022)
- F. Bernardo, A. Alves, V. Homem, A review of bioaccumulation of volatile methylsiloxanes in aquatic ecosystems. Sci. Total Environ. 824, 153821 (2022)
- Y. Chae, Y.-J. An, Toxicity and transfer of polyvinylpyrrolidone-coated silver nanowires in an aquatic food chain consisting of algae, water fleas, and zebrafish. Aquat. Toxicol. **173**, 94–104 (2016)
- K. Chen, X. Wu, Z. Zou, Y. Dong, S. Zhang, X. Li, M. Gouda, B. Chu, C.M. Li, X. Li, Y. He, Assess heavy metals-induced oxidative stress of microalgae by Electro-Raman combined technique. Anal. Chim. Acta 1208, 339791 (2022)
- 16. S.B. Eskander, H.M. Saleh, Heavy Metal-Induced Oxidative Stress and Related Cellular Process, in *Cellular and Molecular Phytotoxicity of Heavy Metals*, ed. by M. Faisal,

Q. Saquib, A.A. Alatar, A.A. Al-Khedhairy, (Springer International Publishing, Cham, 2020), pp. 99–123

- H. Zhou, Y. Qu, H. Wu, C. Liao, J. Zheng, X. Diao, Q. Xue, Molecular phylogenies and evolutionary behavior of AhR (aryl hydrocarbon receptor) pathway genes in aquatic animals: implications for the toxicology mechanism of some persistent organic pollutants (POPs). Chemosphere 78, 193–205 (2010)
- N. Davidsen, A.J. Lauvås, O. Myhre, E. Ropstad, D. Carpi, E.M.-d. Gyves, H.F. Berntsen, H. Dirven, R.E. Paulsen, A. Bal-Price, F. Pistollato, Exposure to human relevant mixtures of halogenated persistent organic pollutants (POPs) alters neurodevelopmental processes in human neural stem cells undergoing differentiation. Reprod. Toxicol. **100**, 17–34 (2021)
- K.E.M. Ahmed, H.G. Frøysa, O.A. Karlsen, N. Blaser, K.E. Zimmer, H.F. Berntsen, S. Verhaegen, E. Ropstad, R. Kellmann, A. Goksøyr, Effects of defined mixtures of POPs and endocrine disruptors on the steroid metabolome of the human H295R adrenocortical cell line. Chemosphere 218, 328–339 (2019)
- H. Wang, H. Xi, L. Xu, M. Jin, W. Zhao, H. Liu, Ecotoxicological effects, environmental fate and risks of pharmaceutical and personal care products in the water environment: A review. Sci. Total Environ. 788, 147819 (2021)
- 21. G.G. Del'haye, I. Nulmans, S.P. Bouteille, K. Sermon, B. Wellekens, M. Rombaut, T. Vanhaecke, Y. Vander Heyden, J. De Kock, Development of an adverse outcome pathway network for breast cancer: a comprehensive representation of the pathogenesis, complexity and diversity of the disease. Arch. Toxicol. (2022)